## SPECIFICATIONS for Saline Area Schools

# 2015 BOND PROGRAM 2017 PROJECTS PHASE 1

January 06, 2017

**VOLUME I OF I** 

<u>CONSTRUCTION MANAGER:</u> CLARK CONSTRUCTION COMPANY 29110 INKSTER ROAD SUITE 150 SOUTHFIELD, MI 48034

<u>ARCHITECT:</u> KINGSCOTT ASSOCIATES, INC. 229 EAST MICHIGAN AVE., SUITE 335 KALAMAZOO, MI 49007



Clark Project No. 15-2714

### **DIVISION 00 – PROCUREMENT AND CONTRACTING REQUIREMENTS**

## **INTRODUCTORY INFORMATION**

- 000101 PROJECT MANUAL COVER
- 000110 TABLE OF CONTENTS
- 000115 LIST OF DRAWING SHEETS

## **BIDDING REQUIREMENTS**

001100	ADVERTISEMENT/INVITATION FOR BIDS
001100	

- 002113 INSTRUCTIONS TO BIDDERS
- 002413 SCOPE OF WORK GENERAL NOTES
- 002416 SCOPE OF WORK BID CATEGORY SPECIFIC NOTES
- 003113 PROJECT MILESTONE SCHEDULE
- 004126 BID PROPOSAL FORM

## **CONTRACTING REQUIREMENTS**

- 005216 AGREEMENT FORM
- 006113 PERFORMANCE AND PAYMENT BOND
- 006200 CERTIFICATES AND OTHER FORMS
- 007216 GENERAL CONDITIONS
- 007316 INSURANCE REQUIREMENTS

## **DIVISION 01 – GENERAL REQUIREMENTS**

- 012100 ALLOWANCES
- 012200 UNIT PRICES
- 012300 ALTERNATES
- 012600 CONTRACT MODIFICATION PROCEDURES
- 012900 PAYMENT PROCEDURES
- 012973 SCHEDULE OF VALUES
- 013119 PROJECT MEETINGS
- 013323 SHOP DRAWINGS, PROJECT DATA AND SAMPLES
- 013529 HEALTH, SAFETY AND EMERGENCY RESPONSE PROCEDURES
- 013543 ENVIRONMENTAL PROCEDURES
- 014516 FIELD QUALITY CONTROL PROCEDURES
- 014517 NON-CONFORMANCE PROCEDURE
- 015000 TEMPORARY FACILITIES AND CONTROLS
- 017123 FIELD ENGINEERING
- 017329 CUTTING AND PATCHING
- 017700 CLOSEOUT PROCEDURES
- 017823 OPERATION AND MAINTENANCE DATA
- 017839 PROJECT RECORD DOCUMENTS

## **DIVISION 02 – EXISTING CONDITIONS**

024119 SELECTIVE DEMOLITION

**DIVISION 03 – CONCRETE** 

033000 CAST-IN-PLACE CONCRETE

### **DIVISION 04 – MASONRY**

042000 UNIT MASONRY

#### **DIVISION 05 – METALS**

051200	STRUCTURAL STEEL FRAMING
053100	STEEL DECKING
054000	COLD-FORMED METAL FRAMING
055000	METAL FABRICATIONS
055813	COLUMN COVERS
057300	DECORATIVE METAL RAILINGS

#### **DIVISION 06 – WOOD, PLASTICS, AND COMPOSITES**

061053	MISCELLANEOUS ROUGH CARPENTRY
061600	SHEATHING
064600	WOOD TRIM

#### **DIVISION 07 – THERMAL AND MOISTURE PROTECTION**

072100	THERMAL INSULATION
076100	METAL ROOFING SYSTEMS
076200	SHEET METAL FLASHING AND TRIM
077200	ROOF ACCESORIES
078100	APPLIED FIREPROOFING
078413	PENETRATION FIRESTOPPING
079200	JOINT SEALANTS

#### **DIVISION 08 – OPENINGS**

081113	HOLLOW METAL DOORS AND FRAMES
081416	FLUSH WOOD DOORS
081700	INTEGRATED DOOR OPENING ASSEMBLIES
081743	FRP FLUSH DOORS
083113	ACCESS DOORS AND FRAMES
083323	OVERHEAD COILING DOORS
084113	ALUMINUM FRAMED ENTRANCES
084413	GLAZED ALUMINUM CURTAIN WALLS
084523	FIBERGLASS-SANDWICH-PANEL ASSEMBLIES
087100	DOOR HARDWARE
087200	ACCESS CONTROL SYSTEM
088000	GLAZING
088010	GLASS WITH INTEGRAL BLINDS

**DIVISION 09 – FINISHES** 

- 092216 NON-STRUCTURAL METAL FRAMING
- 092900 GYPSUM BOARD
- 093013 CERAMIC TILING
- 095113 ACOUSTICAL PANEL CEILINGS
- **RESILIENT BASE AND ACCESSORIES** 096513
- 096516 **RESILIENT SHEET FLOORING**
- 096519 **RESILIENT TILE FLOORING**
- 096813 TILE CARPETING
- 097200 WALL COVERINGS
- SOUND ABSORBING WALL UNITS 098433
- 099123 **INTERIOR PAINTING**
- 099600 HIGH PERFORMANCE COATINGS

## **DIVISION 10 – SPECIALTIES**

101100	VISUAL DISPLAY UNITS
101110	

- DIMENSIONAL LETTER SIGNAGE 101419
- 101700 FRP TOILET PARTITION SYSTEM
- 102600 WALL AND DOOR PROTECTION
- 102800 TOILET, BATH AND LAUNDRY ACCESSORIES

## **DIVISION 11 – EOUIPMENT**

116623 GYMNASIUM EQUIPMENT (WALL PADDING)

## **DIVISION 12 – FURNISHINGS**

122200	GLASS FURNITURE WALLS
122413	ROLLER WINDOW SHADES
123216	MANUFACTURED PLASTIC-LAMINATE-FACED CASEWORK
123623.13	PLASTIC-LAMINATE-CLAD COUNTERTOPS
123661	SIMULATED STONE COUNTERTOPS
126100	EIVED SEATING

126100 FIXED SEATING

## **DIVISIONS 13 thru 21 (NOT USED)**

#### **DIVISION 22 – PLUMBING**

220517	SLEEVES AND SLEEVE SEALS FOR PLUMBING PIPING
220510	EGGUEGHEGNIG FOD DI LU (DINIG DIDINIG

- ESCUTCHEONS FOR PLUMBING PIPING 220518
- 220523.12 BALL VALVES FOR PLUMBING PIPING
- 220529 HANGERS AND SUPPORTS FOR PLUMBING PIPING AND **EOUIPMENT**
- 220553 IDENTIFICATION FOR PLUMBING PIPING AND EQUIPMENT
- 220719 PLUMBING PIPING INSULATION
- 221113 FACILITY WATER DISTRIBUTION PIPING
- 221116 DOMESTIC WATER PIPING
- 221119 DOMESTIC WATER PIPING SPECIALTIES

Saline Area Schools 2015 Bond Program – 2017 Projects Phase 1 Bid Release No.4

## **DIVISION 22 – PLUMBING (CONTINUED)**

- 221316 SANITARY WASTE AND VENT PIPING
- 221319 SANITARY WASTE PIPING SPECIALTIES
- 224213.13 COMMERCIAL WATER CLOSETS
- 224213.16 COMMERCIAL URINALS
- 224216.13 COMMERCIAL LAVATORIES
- 224216.16 COMMERCIAL SINKS
- 224716 PRESSURE WATER COOLERS

## **DIVISION 23 – HEATING, VENTILATING AND AIR CONDITIONING**

230513	COMMON MOTOR REQUIREMENTS FOR HVAC EQUIPMENT
230514	VARIABLE FREQUENCY MOTOR CONTROLLERS
230517	SLEEVES AND SLEEVE SEALS FOR HVAC PIPING
230518	ESCUTCHEONS FOR HVAC PIPING
230519	METERS AND GAGES FOR HVAC PIPING
230523.12	BALL VALVES FOR HVAC PIPING
230523.13	BUTTERFLY VALVES FOR HVAC PIPING
230523.14	CHECK VALVES FOR HVAC PIPING
230529	HANGERS AND SUPPORTS FOR HVAC PIPING AND EQUIPMENT
230553	IDENTIFICATION FOR HVAC PIPING AND EQUIPMENT
230593	TESTING, ADJUSTING AND BALANCING FOR HVAC
230713	DUCT INSULATION
230719	HVAC PIPING INSULATION
231123	FACILITY NATURAL GAS PIPING
232113	HYDRONIC PIPING
232116	HYDRONIC PIPING SPECIALTIES
232123	HYDRONIC PUMPS
232300	REFRIGERANT PIPING
233113	METAL DUCTS
233300	AIR DUCT ACCESSORIES
233423	HVAC POWER VENTILATORS
233713	DIFFUSERS, REGISTERS AND GRILLES
235123	GAS VENTS
235216	CONDENSING BOILERS
236313	AIR-COOLED REFRIGERANT CONDENSERS
237200	AIR-TO-AIR ENERGY RECOVERY EQUIPMENT
237433	DEDICATED OUTDOOR AIR UNITS
238213	VALANCE HEATING AND COOLING UNITS
238216.11	
238223	UNIT VENTILATORS
238239.13	CABINET UNIT HEATERS

### **DIVISION 26 – ELECTRICAL**

- 260500 BASIC ELECTRICAL REQUIREMENTS
- 260501 MINOR ELECTRICAL DEMOLITION
- 260519 LOW VOLTAGE ELECTRICAL POWER CONDUCTORS AND CABLES
- 2900-03 TABLE OF CONTENTS TOC-4
- 260526 GROUNDING AND BONDING FOR ELECTRICAL SYSTEMS
- 260529 HANGERS AND SUPPORTS FOR ELECTRICAL SYSTEMS
- 260534 CONDUIT
- 260537 BOXES
- 260553 IDENTIFICATION FOR ELECTRICAL SYSTEMS
- 260923 LIGHTING CONTROL DEVICES
- 262416 PANELBOARDS
- 262726 WIRING DEVICES
- 265100 INTERIOR LIGHTING
- 265600 EXTERIOR LIGHTING

#### **DIVISION 27 – COMMUNICATIONS**

270000	GENERAL TECHNOLOGY REQUIREMENTS
271000	GENERAL CABLING REQUIREMENTS

#### DIVISION 28 – ELECTRONIC SAFETY AND SECURITY

281300	ACCESS CONTROL
283101	FIRE ALARM SYSTEMS

## **DIVISION 31 – EARTHWORK**

311000	SITE CLEARING
312000	EARTH MOVING

#### **DIVISION 32 – EXTERIOR IMPROVEMENTS**

- 321216 ASPHALT PAVING
- 321313 CONCRETE PAVING
- 321373 CONCRETE PAVING JOINT SEALANTS
- 321723 PAVEMENT MARKINGS
- 321726 TACTILE WARNING SURFACING
- 321816.13 PLAYGROUND SURFACING
- 323113 CHAIN LINK FENCES AND GATES
- 323119 ALUMINUM ORNAMENTAL FENCE SYSTEM
- 329200 TURF AND GRASSES

#### **DIVISION 33 - UTILITIES**

334100 STORM UTILITY DRAINAGE PIPING334600 SUBDRAINAGE

SECTION 000110 PROJECT MANUAL TABLE OF CONTENTS 1/6/2017 END OF SECTION

## 1. LIST OF DRAWINGS PREPARED BY Kingscott Associate, Inc.

SHEET NO.	TITLE	DATE
T1	Cover Page	1/6/2017
T2	Data Page	1/6/2017
T3	Mounting Heights	1/6/2017
HS-C1.0	Topo Plan	1/6/2017
HS-C2.0	Demo Plan	1/6/2017
HS-C3.0	Site Plan	1/6/2017
HS-C3.1	Site Plan – Connector Sidewalk	1/6/2017
HS-C4.0	Detail Sheet	1/6/2017
HS-C5.0	Soil Erosion and Sediment Control Plan	1/6/2017
HS- A0.1	Code Compliance/ Composite Floor Plan	1/6/2017
HS-A1.1	First Floor Plan – Area E	1/6/2017
HS- A2.1	First Floor Reflective Ceiling Plan – Area E	1/6/2017
HS-A4.1	Wall Sections and Details	1/6/2017
HS-E0.0	Electrical General Information	1/6/2017
HS-E0.1	Electrical Partial 1 <sup>st</sup> /2 <sup>nd</sup> Floor Plans	1/6/2017
HS-M1.0	Partial Mechanical Plans	1/6/2017
HS-M2.0	Mechanical Details/Schedules	1/6/2017
HS-T4.0	Tech Plans – Composite 1 <sup>st</sup> Floor	1/6/2017
HS-T4.0A	Tech Units A,B,C,D,E,I 1st Floor	1/6/2017
HS-T4.0B	Tech Units F,G 1 <sup>st</sup> Floor	1/6/2017
HS-T4.1	Tech Plan – Composite 2 <sup>nd</sup> Floor	1/6/2017
HS-T4.1A	Tech Units A,B,C,D,E,I 2nd Floor	1/6/2017
HS-T4.1B	Tech Units F,G,H 2 <sup>nd</sup> Floor	1/6/2017
HS-T4.1C	Tech Units J,K,L 2 <sup>nd</sup> Floor	1/6/2017
HS-T4.2	Tech Plans – Composite 3 <sup>rd</sup> Floor	1/6/2017
HS-T4.2A	Tech Units A,B,C,D,E,I 3rd Floor	1/6/2017
HS-T4.2B	Tech Units J,K,L 3 <sup>rd</sup> Floor	1/6/2017
MS-C1.0	Topo Plan – East	1/6/2017
MS-C1.1	Topo Plan – West	1/6/2017
MS-C1.2	Topo Plan – Thibault Lane	1/6/2017
MS-C2.0	Demo Plan – East	1/6/2017
MS-C2.1	Demo Plan – Northwest	1/6/2017
MS-C2.2	Demo Plan – Southwest	1/6/2017
MS-C2.3	Demo Plan – Thibault Lane	1/6/2017
MS-C3.0	Overall Site Layout Plan	1/6/2017
MS-C3.1	Site Layout Plan – East	1/6/2017

MS-C3.2	Site Layout Plan – Northwest	1/6/2017
MS-C3.3	Site Layout Plan – Southwest	1/6/2017
MS-C3.4	Site Grading and Utility Plan – East	1/6/2017
MS-C3.5	Site Grading and Utility Plan – Northwest	1/6/2017
MS-C3.6	Site Grading and Utility Plan – Southwest	1/6/2017
MS-C3.7	Site Plan – Thibault Lane	1/6/2017
MS-C4.0	Detail Sheet	1/6/2017
MS-C4.0 MS-C4.1	Detail Sheet	1/6/2017
MS-C5.0	Soil Erosion and Sediment Control Plan	1/6/2017
MS-C5.1	Soil Erosion and Sediment Control Plan –	1/6/2017
NIS-C3.1	Thibault Lane	1/0/2017
MS-A0.1	Code Compliance and Composite Floor Plan	1/6/2017
MS-D1.1A	Demo Floor Plan – Area A	1/6/2017
MS-D1.1B	Demo Floor Plan – Area B	1/6/2017
MS-D1.1C	Demo Floor Plan – Area C	1/6/2017
MS-D1.1D	Demo Floor Plan – Area D,G	1/6/2017
MS-D1.1E	Demo Floor Plan – Area E	1/6/2017
MS-D1.1F	Demo Floor Plan – Area F	1/6/2017
MS-D1.1H	Demo Floor Plan – Area H	1/6/2017
MS-D1.1L	Demo Floor Plan – Area L	1/6/2017
MS-A1.1A	Floor Plan – Area A	1/6/2017
MS-A1.1B	Floor Plan – Area B	1/6/2017
MS-A1.1C	Floor Plan – Area C	1/6/2017
MS-A1.1D	Floor Plan – Area D,G	1/6/2017
MS-A1.1E	Floor Plan – Area E	1/6/2017
MS-A1.1F	Floor Plan – Area F	1/6/2017
MS- A1.1H	Floor Plan – Area H	1/6/2017
MS-A1.1L	Floor Plan – Area L	1/6/2017
MS-	Floor Plan – Gym	1/6/2017
A1.1GYM		
MS-A2.1C	Reflected Ceiling Plan – Area C	1/6/2017
MS-	<b>Reflected Ceiling Plan – Gym</b>	1/6/2017
A2.1GYM MS-A4.1	Wall Sections and Details	1/6/2017
MS-A4.1 MS-A6.1	Door Schedule and Door Details	1/6/2017
MS-A0.1 MS-A1.11C	Equipment Plan – Unit C	1/6/2017
MS-A1.11C MS-A1.11F		1/6/2017
	Equipment Plan – Unit F	
MS-A1.12F	Auditorium Seating Details	1/6/2017
MS-A1.13F	Auditorium Seating Details	1/6/2017
MS-A1.14F	Auditorium Seating Details	1/6/2017

MS-A2.1C	Reflected Ceiling Plan – Unit C	1/6/2017
MS-A2.1F	<b>Reflected Ceiling Plan – Unit F</b>	1/6/2017
MS-A2.1KL	Reflected Ceiling Plan – Gym	1/6/2017
MS-A2.1J	Reflected Ceiling Plan – Unit J	1/6/2017
MS-A8.1A	Color Layout Plan – Unit A	1/6/2017
MS-A8.1B	Color Layout Plan – Unit B	1/6/2017
MS-A8.1C	Color Layout Plan – Unit C	1/6/2017
MS-A8.1D	Color Layout Plan – Unit D,G	1/6/2017
MS-A8.1E	Color Layout Plan – Unit E	1/6/2017
MS-A8.1F	Color Layout Plan – Unit F	1/6/2017
MS-A8.1H	Color Layout Plan – Unit H	1/6/2017
MS-A8.1L	Color Layout Plan – Unit L	1/6/2017
MS-A8.1K	Color Layout Plan – Unit K	1/6/2017
MS-A9.1	Section, Interior Elevation, 3D Images	1/6/2017
MS-M1.1	Mechanical Plan – Unit A	1/6/2017
MS-M1.2	Mechanical Plan – Unit B	1/6/2017
MS-M1.3	Mechanical Plan – Unit C	1/6/2017
MS-M1.9	Mechanical Plan – Unit J	1/6/2017
MS-M1.10	Mechanical Plan – Unit K	1/6/2017
MS-M2.0	Mechanical/Electrical Schedules and Mechanical Details	1/6/2017
E0.0	Electrical General Information	1/6/2017
MS-ESD1.0	Electrical Demo Site Plan	1/6/2017
MS-ED1.1	Electrical Demo Plan – Area C	1/6/2017
MS-ED1.2	Electrical Demo Plan – Area F	1/6/2017
MS-ED1.3	Electrical Demo Plan – Gym	1/6/2017
MS-ES1.0	Electrical New Work Site Plan	1/6/2017
MS-EP1.0	Electrical Composite Plan	1/6/2017
MS-EP1.1	Power New Work – Area C	1/6/2017
MS-EL1.1	Lighting New work – Area C	1/6/2017
MS-EL1.2	Lighting New Work – Area F	1/6/2017
MS-EL1.3	Lighting New Work – Gym	1/6/2017
MS-E5.0	Electrical Details, One Line Diagram	1/6/2017
MS-E6.0	Electrical Panel Schedules	1/6/2017
MS-T4.1	Technology Composite Plan	1/6/2017
MS-T4.1C	Technology Plan – Unit C	1/6/2017
LS-C1.0	Topo Plan	1/6/2017
LS-C2.0	Demo Plan – West	1/6/2017
LS-C2.1	Demo Plan – South	1/6/2017
LS-C2.2	Demo Plan – East	1/6/2017

LS-C3.0	Overall Site Plan	1/6/2017
LS-C3.1	Site Layout Plan – West	1/6/2017
LS-C3.2	Site Layout Plan – South	1/6/2017
LS-C3.3	Site Layout Plan – East	1/6/2017
LS-C3.4	Site Grading and Utility Plan – West	1/6/2017
LS-C3.5	Site Grading and Utility Plan - South	1/6/2017
LS-C3.6	Site Grading and Utility Plan – East	1/6/2017
LS-C3.7	Site Plan – North	1/6/2017
LS-C4.0	Detail Sheet	1/6/2017
LS-C-4.1	Detail Sheet	1/6/2017
LS-A1.0	Composite Plan	1/6/2017
LS-D1.1A	Demo Plan – Unit A	1/6/2017
LS-D1.1B	Demo Plan – Unit B	1/6/2017
LS-D1.1C	Demo Plan – Unit C	1/6/2017
LS-D1.1D	Demo Plan – Unit D	1/6/2017
LS-A1.1A	Floor Plan – Unit A	1/6/2017
LS-A1.1B	Floor Plan – Unit B	1/6/2017
LS-A1.1C	Floor Plan – Unit C	1/6/2017
LS-A1.1D	Floor Plan – Unit D	1/6/2017
LS-A1.4	Enlarged Plans	1/6/2017
LS-A1.6	Composite Roof Plan	1/6/2017
LS-A1.7	Roof Details	1/6/2017
LS-A1.11A	Equipment Plan – Unit A	1/6/2017
LS-A1.11B	Equipment Plan – Unit B	1/6/2017
LS-A1.11C	Equipment Plan – Unit C	1/6/2017
LS-A1.11D	Equipment Plan – Unit D	1/6/2017
LS-A2.1	Reflected Ceiling Plan – Unit A	1/6/2017
LS-A2.2	Reflected Ceiling Plan – Unit B	1/6/2017
LS-A2.3	Reflected Ceiling Plan – Unit C	1/6/2017
LS-A2.4	Reflected Ceiling Plan – Unit D	1/6/2017
LS-A3.1	Exterior Elevations	1/6/2017
LS-A3.2	Exterior Elevations	1/6/2017
LS-A5.1	Building Details	1/6/2017
LS-A6.1	Door Schedule and Door Types	1/6/2017
LS-A6.2	Frame and Window Types	1/6/2017
LS-A6.3	Door and Window Details	1/6/2017
LS-A8.1A	Color Plan – Unit A	1/6/2017
LS-A8.1B	Color Plan – Unit B	1/6/2017
LS-A8.1C	Color Plan – Unit C	1/6/2017

LS-A8.1D	Color Plan – Unit D	1/6/2017
LS-MD1.1	Mechanical Demo Plan – Unit A	1/6/2017
LS-MD1.2	Mechanical Demo Plan – Unit B	1/6/2017
LS-MD1.3	Mechanical Demo Plan – Unit C	1/6/2017
LS-MD1.4	Mechanical Demo Plan – Unit D	1/6/2017
LS-M1.1	Mechanical Plan – Unit A	1/6/2017
LS-M1.2	Mechanical Plan – Unit B	1/6/2017
LS-M1.3	Mechanical Plan – Unit C	1/6/2017
LS-M1.4	Mechanical Plan – Unit D	1/6/2017
LS-M2.0	Enlarged Mechanical Plan	1/6/2017
LS-M2.1	Mech/Elec Schedules and Details	1/6/2017
LS-P0.1	Plumbing Foundation Plan	1/6/2017
LS-P1.0	Plumbing Plans – Units A,B,C,D	1/6/2017
LS-P2.0	Enlarged Plumbing Plan	1/6/2017
LS-E0.1	Electrical General Information	1/6/2017
LS-ESD1.0	Electrical Demo Plan	1/6/2017
LS-ED1.1	1 <sup>st</sup> Floor Electrical Demo – Area A	1/6/2017
LS-ED1.2	1 <sup>st</sup> Floor Electrical Demo – Area B	1/6/2017
LS-ED1.3	1 <sup>st</sup> Floor Electrical Demo – Area C	1/6/2017
LS-ED1.4	1 <sup>st</sup> Floor Electrical Demo – Area D	1/6/2017
LS-ES1.0	Electrical New Work Site Plan	1/6/2017
LS-EP1.1	1 <sup>st</sup> Floor Power Plan – Area A	1/6/2017
LS-EP1.2	1 <sup>st</sup> Floor Power Plan – Area B	1/6/2017
LS-EP1.3	1 <sup>st</sup> Floor Power Plan – Area C	1/6/2017
LS-EP1.4	1 <sup>st</sup> Floor Power Plan – Area D	1/6/2017
LS-EL1.1	1 <sup>st</sup> Floor Lighting Plan – Area A	1/6/2017
LS-EL1.2	1 <sup>st</sup> Floor Lighting Plan – Area B	1/6/2017
LS-EL1.3	1 <sup>st</sup> Floor Lighting Plan – Area C	1/6/2017
LS-EL1.4	1 <sup>st</sup> Floor Lighting Plan – Area D	1/6/2017
LS-E5.0	Electrical Details	1/6/2017
LS-T4.1	Technology Plan Composite	1/6/2017
LS-T4.1A	Technology Plan Area A,D	1/6/2017
HA-T4.1	Technology Plan Composite 1 <sup>st</sup> Floor	1/6/2017
НА-Т4.2	Technology Plan Composite 2 <sup>nd</sup> Floor	1/6/2017
HE-A0.1	Composite 1 <sup>st</sup> Floor Plan	1/6/2017
HE-A1.1	Enlarged Floor Plan	1/6/2017
HE-M1.0	Mechanical Roof Plan	1/6/2017
HE-M1.1	Mechanical Plan – Unit 100	1/6/2017
HE-M1.2	Mechanical Plan – Unit 200	1/6/2017

HE-M1.3	Mechanical Plan – Unit 300	1/6/2017
НЕ-Е0.0	Electrical General Information	1/6/2017
НЕ-Е0.2	Electrical Composite Roof Plan	1/6/2017
HE-T4.1	Technology Composite Plan	1/6/2017
PR-T4.1	Technology Composite Plan	1/6/2017
WM-A0.1	Composite 1 <sup>st</sup> Floor Plan	1/6/2017
WM-D1.1	Demo Plan – Unit 400	1/6/2017
WM-D1.2	Demo Plan – Unit 500	1/6/2017
WM-D1.3	Demo Plan – Unit 600	1/6/2017
WM-A1.1	Floor Plan – Unit 400	1/6/2017
WM-A1.2	Floor Plan – Unit 500	1/6/2017
WM-A1.3	Floor Plan – Unit 600	1/6/2017
WM-A8.1	Color Plan – Unit 400	1/6/2017
WM-A8.2	Color Plan – Unit 500	1/6/2017
WM-A8.3	Color Plan – Unit 600	1/6/2017
WM-M1.0	Mechanical Roof Plan	1/6/2017
WM-M2.0	Mechanical Plan – Unit 600	1/6/2017
WM-M2.1	Mechanical Plan – Unit 500	1/6/2017
WM-M2.3	Mechanical Plan – Unit 400	1/6/2017
WM-M3.0	Mechanical/ Electrical Schedules	1/6/2017
WM-E0.0	Electrical General Information	1/6/2017
WM-E0.2	Electrical Composite Roof Plan	1/6/2017
WM-T4.1	Technology Composite Plan	1/6/2017

## SECTION 001100 ADVERTISEMENT / INVITATION FOR BIDS

PROJECT:	Saline Area Schools 2015 Bond Program 2017 Projects Phase 1
CLARK PROJECT NO.:	2714
BID RELEASE NO.:	4
DESCRIPTION:	Renovation of 7 School Buildings
OWNER:	Saline Area Schools 7265 Ann Arbor Rd Saline, MI, 48176
ARCHITECT:	Kingscott Associates, Inc. 229 E Michigan Ave, Suite 335 Kalamazoo, MI 49007
CONSTRUCTION MANAGER:	Clark Construction Company
	3535 Moores River Drive/P. O. Box 40087 Lansing, MI 48901
PROJECT MANAGER: TELEPHONE:	Mark Paulus Mpaulus@clarkcc.com 517.282.9259
PROJECT ENGINEER:	Brad Guldan Bguldan@clarkcc.com 517.897.0463

#### 1. <u>PRE-BID CONFERENCE</u>

1.1. A pre-bid conference will be held at 3:15pm Local Time, Monday January 23, 2017 at Liberty School 7265 N. Ann Arbor St, Saline, MI 48176

#### 2. <u>BID PROPOSAL DUE DATE/LOCATION</u>

- 2.1. Lump sum sealed proposals will be received at the office of Liberty School 7265 N. Ann Arbor St, Saline, MI 48176 for the following Bid Categories.
- 2.2. Proposal must be delivered not later than 2:00 pm Local Time, Thursday February 2, 2017.
- 2.3. Fax proposals shall not be submitted.
- 2.4. Bids will be opened publicly 2:05 pm Local Time, Thursday February 2, 2017 at Liberty School 7265 N. Ann Arbor St, Saline, MI 48176

## SECTION 001100 ADVERTISEMENT / INVITATION FOR BIDS

## 3. <u>BID CATEGORIES</u>

04A - Demolition	04B - Concrete
04C- General Trades	04D - Roofing
04E – Glass and Glazing	04F – Access Control
04G – Floor covering and Ceramic Tile	04H – Painting and Wall Covering
04I – Auditorium Seating	04J – HVAC and Plumbing
04K - Electrical	04L - Communication
04M - Earthwork	04N - Asphalt
04O–Landscaping and Playground	04P - Fencing
Surfacing	

## 4. <u>BID DOCUMENT AVAILABILITY/DEPOSIT</u>

4.1. Bid Documents, drawings and specifications will be available at 8:00 am Local Time, Monday, January 16,2017 at the following location(s):

ARC Document Solution 2813 Boardwalk Ann Arbor, MI 48104

Or through the following

https://clarkcc.sharefile.com/i/ibbed5cd42be4ce89

- 4.2. Documents will also be made available for review at the following locations:
  - Dodge Reports Southfield
  - Builder's Exchange Lansing
  - Construction Association of Michigan (CAM)
- 4.3. Bidder is responsible for the entire cost of reproduction.

#### 5. <u>BID PROPOSAL REQUIREMENTS</u>

5.1. All questions during the bidding period shall be referred to Clark Construction Company via fax or email to:

CONTACT:	Brad Guldan, Project Engineer
TELEPHONE:	517.897.0463
EMAIL:	Bguldan@clarkcc.com

5.2. Bidders being considered for award of Contract shall submit the Clark Construction Company Trade Contractor Qualification Form completely filled out as well as all documents required with "Certificates and Other Forms".

#### END OF SECTION

#### 1. **DEFINITIONS:**

- 1.1. All definitions set forth in the General and Supplementary Conditions of the Contract for Construction are applicable to these Instructions to Bidders.
- 1.2. **Bid documents:** include the contract, the General Conditions, the Advertisement/ Invitation for Bids, the Project Manual (Specifications) and all Drawings issued for the purpose of preparing a bid proposal.
- 1.3. Bid Release: a set of Bid Documents.
- 1.4. **Bid Category:** areas of Work performed by a Trade Contractor and its Subcontractors.
- 1.5. Addenda: written or graphic instruments issued prior to the execution of the Contract which modify or interpret the Bidding Documents, including Drawings\_and Specifications, by additions, deletions, clarifications or corrections.
- 1.6. **Bid:** means a proposal prepared and submitted as required herein.
- 1.7. **Trade Contractor:** shall refer to the entity contracted to perform the Work of one or more bid categories whether referred to in the Contract Documents as Contractor, Subcontractor or Trade Contractor.

#### 2. <u>BIDDER'S REPRESENTATION</u>

2.1. Each Bidder, by submitting a bid, represents that the Bidder has read and understands the Bidding Documents, has visited the site and is familiar with the local conditions under which the Work is to be performed and has made his own review therefore of the facilities and difficulties attending the performance and completion of the Work.

#### 3. <u>BIDDING PROCEDURES</u>

- 3.1. Combined bids of two or more bid categories may be submitted.
- 3.2. All bids must be submitted on the unaltered Bid Proposal Forms provided as part of the Bid Documents and in accordance with these Instructions to Bidders.
- 3.3. Each Bidder shall ascertain prior to submitting a bid that he has received all Addenda issued and shall acknowledge receipt on the Bid Proposal Form.
- 3.4. All bids must be signed as follows:
  - A. The Company represents and warrants that the person signing the proposal has the authority to act on behalf of the Company and to bind the Company and all who may claim through it to the terms and conditions of this Agreement. Employee represents and warrants that he has the capacity to act on his own behalf and on behalf of all who might claim through him to bind them to the terms and conditions of this Agreement.
- 3.5. Bids shall be submitted in an opaque, sealed envelope. Facsimile bids will not be accepted unless otherwise noted in the Documents.

- 3.6. A bid shall be invalid if it is not delivered to the location indicated in the Advertisement/Invitation for Bids on or before the required bid due date and time.
- 3.7. No responsibility shall attach to the Construction Manager, the Owner, or representatives of either, for premature opening of any proposal, which is not properly addressed, delivered and identified.
- 3.8. Negligence in preparation, improper preparation, error in and/or omissions from the bid shall not relieve the Bidder from fulfillment of any and all applicable obligations and requirements of the contract documents.
- 3.9. The Owner, Architect/Engineer or Construction Manager, in making copies of the Bidding Documents available, do so only for the purpose of obtaining bids on the work and do not confer a license or grant for any other use.
- 3.10. All bids submitted shall be for the specified base scope of Work with no exceptions. Exceptions in the base bid may be considered as basis for rejection.
- 3.11. Voluntary alternates will be considered at the Architect and Owner's discretion.
- 3.12. Bidder shall not modify, withdraw or cancel a bid or any part thereof for ninety (90) days after the time designated for the receipt of bids.

#### 4. EXAMINATION OF BIDDING DOCUMENTS

- 4.1. Each Bidder shall examine the Bidding Documents carefully and, not later than seven (7) days prior to the date for receipt of bids, shall make written request to Clark Construction Company for interpretation or correction of any ambiguity, inconsistency or error therein discovered. Only written interpretation or correction by Addendum shall be binding.
- 4.2. Each Bidder shall submit a bid proposal based on the entire set of Bid Documents. Bidders shall review all Drawings and Specifications to identify work related to its respective bid category. Complete sets of Drawings are available for review at the locations indicated in the Advertisement/Invitation for Bids or may be obtained from Clark Construction Company.
- 4.3. Failure to review the complete set of documents and to identify items reasonably interpreted to be in the scope of the Bid Category shall not relieve the Trade Contractor of its responsibility to perform the Work.
- 4.4. Any incidental item of material, labor or detail, required for proper execution and completion of the Work, omitted from the Contract Documents, but required by governing codes, local regulations, trade practices, operational functions, and quality workmanship, shall be provided as part of the contract Work at no additional cost, even though not specifically detailed or noted.
- 4.5. Trade Contractor shall not scale Drawings to obtain dimensions.

#### 5. <u>BIDDER'S EXAMINATION OF PREMISES</u>

5.1. The Bidder represents that it has carefully inspected the Project site and examined the Drawings and Specifications and other Contract Documents and is familiar with and has

satisfied itself as to the nature, location and amount of the Work, the Bidder's access thereto and ability to perform the Work, local code requirements applicable to the Work and requirements of permits and inspections, safety and barricade requirements, the terms and conditions of any applicable project labor and collective bargaining agreements, as well as the quality, quantity and availability of labor, materials, equipment and facilities and other items required for the performance of the Work and the possible limiting physical and other conditions which may be encountered in the performance of the Work and assumes all risks therefrom. The Bidder has determined, by its own investigation and research, all the conditions affecting the Work to be performed and materials to be furnished and does not rely upon any representation by the Owner or Construction Manager in connection therewith. In performing the Work, Bidder\_accepts the condition of the Project site as-is and assumes the risks with regard to existing conditions at the Project site.

- 5.2. The Bidder shall take its own measurements and be responsible for the correctness of same.
- 5.3. The Bidder shall be held to have made such examinations of the premises and no allowances will be made on its behalf by reason of error or omission on its part.
- 5.4. Plans, diagrams and other descriptive information which depict existing conditions are provided for scope identification and scheduling purposes only. Quantities, elevations, measurements and locations shown may have been approximated and/or gathered from dated, incomplete original construction documents. <u>This data should not be used for bidding purposes without field verification by the Bidder.</u>
- 5.5. The Bidder's own Contract Document review and site inspection review of the work areas shall be relied upon to provide the Bidder information he may require to properly execute and complete the work. Questions must be submitted in writing to the Construction Manager at least seven (7) days before the date for receipt of bids to allow time for written Addenda to be issued.
- 5.6. Each Bidder shall examine the premises carefully and, not later than seven (7) days prior to the date for receipt of bids, shall make written request to Clark Construction Company for any additional information required. Only written interpretation or correction by Addendum shall be binding.

#### 6. <u>SUBSTITUTIONS</u>

- 6.1. Each bid shall be based upon materials and equipment described in the Bidding Documents.
- 6.2. Material and equipment substitutions will not be considered unless written request has been submitted for approval at least fourteen (14) days prior to bid due date. Only approvals in a written Addendum shall be binding. Each substitution request shall include the following:
  - A. Complete description of the proposed substitution.
  - B. Information regarding specified material or equipment for which substitution is being submitted.

- C. Drawings.
- D. Manufacturer's literature.
- E. Performance and test data.
- F. Effect on performance characteristics.
- G. Impact on the Work of other trades.
- H. Any other data or information necessary for a complete evaluation.

#### 7. <u>OWNER/CONSTRUCTION MANAGER RIGHTS</u>

- 7.1. Bidder acknowledges the right of the Owner or Construction Manager to reject any or all bids and to waive any informality or irregularity in any bid received. In addition, the Bidder recognizes the right of the Owner or Construction Manager to reject a bid for no reason or any reason including but not limited to the following:
  - A. Bidder fails to furnish or submit data required in the Bidding Documents;
  - B. Bid is in any way incomplete or irregular;
  - C. Bidder's performance as a Trade Contractor was unsatisfactory under a prior contract for the construction, repair, modification, or demolition of a facility with the Owner, Construction Manager or Architect;
  - D. Known poor performance on prior contracts with parties other than the Construction Manager, Owner or Architect; or
  - E. Unsatisfactory financial condition.
- 7.2. The Owner may accept alternates which serve its own best interest. The Owner shall have the right to accept alternates in any order or combination and to determine the low Bidder on the basis of the sum of the base bid and bid alternates accepted.
- 7.3. Bidders to whom award of a contract is under consideration shall submit to the Construction Manager upon request, a properly executed Subcontractor Qualification Form (Refer to "Certificates and other Forms" section).

#### 8. EQUAL OPPORTUNITY EMPLOYMENT POLICIES

8.1. The Bidder agrees that it will not discriminate against any employee or applicant for employment because of race, religion, color, sex, national origin or age. The Bidder shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to race, religion, color, sex or national origin.

#### 9. <u>BID SECURITY</u>

9.1. Each bid shall be accompanied by a bid security in an amount of not less than 5% of the total base bid sum, pledging that the Bidder will enter into a contract with the Construction

Manager under the terms stated in the Bidding Documents. (Refer to "Performance and Payment Bond" section).

- 9.2. The Construction Manager shall have the right to retain the bid security of Bidders under consideration until either:
  - A. the Contract has been executed and bonds have been furnished, or
  - B. the specified time has elapsed so that bids may be withdrawn, or
  - C. All bids have been rejected.
- 9.3. The amount of the bid security shall be forfeited to the Construction Manager upon failure of the successful Bidder to enter into a Contract within fifteen (15) days after acceptance of the proposal.

#### 10. PERFORMANCE AND LABOR AND MATERIAL PAYMENT BONDS

10.1. Performance and Labor and Material Payment Bonds <u>may be</u> requested as a condition of the Contract. **Cost of said bonds shall not be included in the base bid.** 

#### 11. PREVAILING WAGE

- 11.1. All Trade Contractors shall comply with the Michigan Prevailing Wage Act.
- 11.2. Trade Contractors shall be responsible to comply with Prevailing Wage Rates as they may be revised by the State of Michigan during the course of the Work.
- 11.3. Trade Contractors shall provide evidence to the Construction Manager if requested of payment of Prevailing Wage in the form of a certified payroll or other form as may be required.
- 11.4. The Owner and Construction Manager shall at its option order an audit of the Trade Contractors' record to confirm payment of Prevailing Wage. The Trade Contractor shall be responsible for payment of such audit in the event of any failure discovered to make full payment to any individual in accordance with Prevailing Wage requirements.

#### END OF SECTION

## SECTION 002413 SCOPE OF WORK GENERAL NOTES

## 1. <u>GENERAL</u>

- 1.1. All Trade Contractors shall be bound to all requirements and conditions applicable to the Work of their respective area of Work of Division 00 Procurement and Contracting Requirements and Division 01 General Requirements, Drawings, Specifications and Addenda.
- 1.2. Scope of Work General Notes shall be <u>applicable to all</u> Trade Contractors. The Scope of Work of each bid category shall include the following items required for, caused by or resulting from its Work.

#### 2. <u>GENERAL NOTES</u>

- 2.1. Coordinate and cooperate with all entities associated with the project including the Owner, Architect, Independent Testing and Inspection agencies, local and other governmental authorities and other Trade Contractors. Cost incurred as a result of lack of coordination of Work, deliveries, access required by others, testing/inspection, schedule, penetrations, etc., shall be the responsibility of the Trade Contractor failing to notify others of action to be taken or other requirements.
- 2.2. Provide written Daily Reports to include the following minimum information:
  - A. Description of Work performed
  - B. Trade personnel classification and hours
  - C. Equipment used
  - D. Significant events/issues
  - E. Weather
  - F. Other items as requested
- 2.3. Commencement of Work subsequent to associated Work by other Trade Contractor shall indicate acceptance of the condition of the Work by other Trade Contractor and/or existing conditions.
- 2.4. Delivery and storage of materials and equipment shall be the responsibility of each Trade Contractor (Refer to "Temporary Facilities and Controls" section).
- 2.5. All means and methods, labor, material, tools, construction equipment and machinery necessary, including cranes, hoisting equipment, scaffolding, shoring and bracing, to complete the Work.
- 2.6. Perform all Work related to the Work of the Bid Category included in "Temporary Facilities and Controls" section.
- 2.7. The Trade Contractor shall obtain and pay for all permits and fees associated with its Work and provide evidence of same to Construction Manager.

## SECTION 002413 SCOPE OF WORK GENERAL NOTES

- 2.8. Minimal benchmarks and control lines will be provided. Trade Contractor shall be responsible for all other engineering and layout required for the performance of its Work (Refer to "Field Engineering" section).
- 2.9. Barricades, traffic maintenance and control as required.
  - A. Wherever fencing is required adjacent to a public walking area and where fence base supports are required which would otherwise project into the walking area creating a potential trip hazard the following barrier/fence system shall be utilized.
    - 1. TraFix Water-Wall Fence by TraFix Devices (or equivalent).
      - a. Product Website:
        - i. <u>https://www.traffixdevices.com/products/barriers/water-wall-fence</u>
      - b. Rental Supplier Information:
        - i. Spartan Barricade
          - 1. Contact: Ken McNeilly
            - a. Phone: (517) 694-1500
            - b. Email: spartanbarr@aol.com
- 2.10. Confine operations at the site to areas permitted by the Construction Manager.
- 2.11. Protection and safekeeping of Trade Contractor's own materials and equipment stored on the premises.
- 2.12. Take care and protect as required to avoid damaging Work in place by other Trade Contractors. In the event of damage to Work in place, the Trade Contractor causing the damage will be responsible for cost of repair. In the event the responsible Trade Contractor cannot be identified, the cost of repair will be prorated on a man hour basis to those Trade Contractors reasonably considered to be in the area of the damage.
- 2.13. Protect all existing structures, equipment, trees, landscaping, etc., to remain.
- 2.14. Dust control to prevent nuisance and hazard.
- 2.15. Cleaning of street of mud and debris (Refer to "Temporary Facilities and Controls" section).
- 2.16. Daily clean-up shall include broom cleaning, dumpster cost and removal of identifiable debris and rubbish from site (Refer to "Temporary Facilities and Controls" section for specific construction cleaning and cleaning of Work in place).
- 2.17. Pumping water required to perform the Work.
- 2.18. Temporary utilities as needed to perform the Work (Refer to "Temporary Facilities and Controls" section for specific temporary utility information).

## SECTION 002413 SCOPE OF WORK GENERAL NOTES

- 2.19. Cutting, coring, patching and penetrations of any structural, architectural, mechanical and electrical materials required to complete the Work unless noted otherwise (Refer to "Cutting and Patching" specification section for specific cutting and patching requirements).
- 2.20. Use of tobacco products will not be allowed on the project site.

END OF SECTION

## 1. <u>GENERAL</u>

- 1.1. Bid Category Specific Notes are <u>applicable to a specific Bid Category</u>. Notes within each Bid Category are provided to clarify the work to be performed under the respective Bid Category. Specific notes shall not be interpreted as a complete itemization of the work to be performed under the Bid Category. The Trade Contractor shall be responsible to perform all Work reasonably interpreted to be included in its scope of Work in accordance with the plans and specifications in addition to the Bid Category notes of clarification.
- 1.2. **Base Specifications -** Each Bid Category Scope of Work shall include <u>all</u> Work specified in or reasonably inferred from the <u>Base Specifications</u> sections listed within the Bid Category.
- 1.3. **Reference Specifications** Each Bid Category Scope of Work shall include <u>portions</u> of the Work specified in or reasonably inferred from the <u>Reference Specification</u> sections listed within the Bid Category.

## END OF SECTION

## **BID CATEGORY 04A - Building Demolition**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

<u>BASE SPECIFICATION</u> – (Include <u>all</u> Work specified or reasonably inferred) <u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred) 024119 - SELECTIVE DEMOLITION

Include (Furnish and Install u.n.o):

- 1. Verify location of existing utilities prior to beginning the work. Hand dig as required to avoid contact with underground utilities.
- 2. Dust control for the duration of the work of this bid category.
- 3. Maintain Streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.
- 4. Remove debris from the site in a timely manner.
- 5. Legal disposal of material off site.
- 6. Pumping water to accomplish the work of this bid category.
- 7. Remove above ground, overhead and below grade materials (e.g. concrete foundations, pavement and base, utilities, etc.) indicated in the Documents.
- 8. Excavation and backfill associated with the work of this bid category.
- 9. Patching of auditorium floor after seating removal.
- 10. Salvaging all items as noted in the documents or as requested by the Owner.
- 11. Exterior window demolition to coordinate with the glass contractor to minimize temporary protection.

#### Exclude:

- 1. Mechanical demolition.
- 2. Electrical demolition.
- 3. Site demolition.
- 4. Concrete demolition for electrical work.
- 5. Roofing demolition.

### BID CATEGORY 04 B - Concrete

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include <u>all</u> Work specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

033000 – CAST-IN-PLACE CONCRETE

321313 – CONCRETE PAVING

321373 - CONCRETE PAVING JOINT SEALANTS

321726 - TACTILE WARMING SURFACING

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below)

312000 – EARTH MOVING

## Include (Furnish and Install u.n.o):

- 1. Concrete flatwork including "thickened" slabs and elevated concrete flatwork, etc.
- 2. Concrete housekeeping/equipment pads. Coordinate required location and dimensions with other Trades.
- 3. Verify existing grade prior to performing the work. Report discrepancies to Clark Construction Company in writing.
- 4. Pumping and/or hoisting materials and concrete as required.
- 5. Temporary shoring and bracing as required.
- 6. Maintain streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.
- 7. Confirm curing compounds and other materials applied to surfaces are compatible with finish materials and adhesives.
- 8. Coordinate special slab requirements with other Trades (e.g. depressed slab at hard tile, floor mats, etc.).
- 9. Confirm and coordinate slope to floor drains.
- 10. Coordinate location and confirm elevation of materials within concrete by other Trades prior to performing the Work (e.g. cleanouts, electrical duct banks, boxes, etc.).
- 11. Proof rolling and fine grading of subgrade as required prior to placement of granular base.
- 12. Pumping water to accomplish the work of this bid category.
- 13. Reinforcing steel and wire mesh associated with the work of this bid category.

- 14. Clean reinforcing of loose or flaking rust, dried concrete, oil, and other foreign material that may reduce the bond between the reinforcing and concrete.
- 15. Openings, pockets, box outs, etc. Coordinate required location and dimensions with other Trades.
- 16. Infill of box outs at later date as directed by Clark Construction Company.
- 17. Install sleeves furnished by others. Coordinate required location with other Trades.
- 18. Locate on site and remove from site all concrete wash out as directed by Clark Construction Company.
- 19. Protection of surrounding surfaces. Clean and restore materials affected by concrete spatter.
- 20. Restore grade to condition that existed prior to performing the work.
- 21. Concrete joint sealants.
- 22. Coordinate and cooperate with the project Independent Testing and Inspection Company for inspection and/or testing of the Work.
- 23. Coordinate dowel location requirements with other Trades.
- 24. Protect anchor bolt threads with grease or other approved means.
- 25. Restore grade to conditions that existed prior to performing the Work.
- 26. Exterior (site) concrete (e.g. flatwork, curb and gutter, etc.).
- 27. Weather protection to meet the project schedule.
- 28. Proof rolling of sub-grade as required prior to placement of granular base.
- 29. Coordinate adjustment of all storm catch basins, manhole covers, and valve boxes, etc. with other Trades prior to performing the Work of this bid category.
- 30. Furnish, grade and compact granular base material.
- 31. Finished concrete flatwork shall have positive drainage over the entire surface as required.
- 32. Tactile warning surfaces.
- 33. Fill/seal existing track with grout to match existing floor.
- 34. Vapor barrier.
- 35. Backfill of all exterior concrete Work performed by this bid category.
- 36. Exterior concrete equipment pads including foundations. Coordinate required location and dimensions with other Trades.

## Exclude:

- 1. Excavation and backfill for concrete foundations.
- 2. Light pole bases.
- 3. Fence posts concrete.

## **BID CATEGORY 04C - General Trades**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include <u>all</u> Work specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or

reasonably inferred)

042000 – UNIT MASONRY

051200 – STRUCTURAL STEEL FRAMING

053100 – STEEL DECKING

054000 – COLD FORMED METAL FRAMING

055000 - METAL FABRICATIONS

055813 – COLUMN COVERS

057300 – DECORATIVE METAL RAILINGS

061053 - MISCELLANEOUS ROUGH CARPENTRY

061600 – SHEATHING

064600 – WOOD TRIM

072100 - THERMAL INSULATION

078100 – APPLIED FIREPROOFING

078413 - PENETRATION FIRESTOPPING

081113 – HOLLOW METAL DOORS AND FRAMES

081416 – FLUSH WOOD DOORS

083113 - ACCESS DOORS AND FRAMES

083323 – OVERHEAD COILING DOORS

087100 – DOOR HARDWARE

087200 - ACCESS CONTROL SYSTEM

092216 – NON STRUCTURAL METAL FRAMING

092900 - GYPSUM BOARD

095113 – ACOUSTICAL PANEL CEILINGS

098433 – SOUND ABSORBING WALL UNIT

101100 – VISUAL DISPLAY UNITS

101200 - DISPLAY CASES

101419 – DIMENSIONAL LETTER SINGAGE

101700 – FRP TOILET PARTITIONS

102600 – WALL AND DOOR PROTECTION

102800 - TOILET, BATH AND LAUNDRY ACCESSORIES

116623 – GYMNASIUM EQUIPMENT (WALL PADDING)

122413 – ROLLER WINDOW SHADES

123216 - MANUFACTURED PLASTIC LAMINATE FACED CASEWORK

123623.13 – PLASTIC LAMINATE CLAD COUNTERTOPS

123661 – SIMULATED STONE COUNTERTOPS

 Saline Area Schools 2015 Bond Program – 2017 Projects Phase 1
 002416-1

 Bid Release No. 4
 Project No. 2714

 Section 002416-Scope of Work-Bid Category Specific Notes, Bid Category 6X-Rough and Finish Carpentry and General Trades
 Version 2

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below)

079200 – JOINT SEALANTS

## Include (Furnish and Install u.n.o):

- 1. Wood blocking indicated in the documents, including blocking for specialties, roofing, and plywood sheathing as required. Coordinate blocking requirements with equipment suppliers and other Trades.
- 2. Architectural woodwork.
- 3. Casework and countertops including required blocking.
- 4. Coordinate location of door frame openings with metal stud and other Trade Contractors as required for high quality installation.
- 5. Security of materials furnished by others for installation by this bid category.
- 6. Millwork items included in the documents, including standing and running trim.
- 7. Joint sealants associated with the Work of this bid category.
- 8. Fire extinguishers for use during construction in accordance with specification section "Temporary Facilities and Controls".
- 9. Wood doors.
- 10. Allowance in the amount of \$10,000.00 to provide labor and material to construct temporary protection, doors, covers, etc. as directed by Clark Construction Company.
- 11. Coordinate dowel location requirements with other Trades.
- 12. Reinforcing and other accessories (e.g. ties, anchors, drainage weeps, etc.).
- 13. Protect existing floors from mortar and other damage. Clean floors upon completion of the Work or more frequently as required.
- 14. Control joints in accordance with the Documents and as required to meet quality masonry practice. Review Control joint layout with Clark Construction Company and the Architect prior to performing the Work.
- 15. Review special conditions (e.g. joining of dissimilar masonry and other materials, etc.) with Clark Construction Company and the Architect prior to beginning the Work.
- 16. Reinforcing steel and grouting associated with the Work of this bid category.
- 17. Smoke and fire stopping as required.
- 18. Construct box outs as required. Coordinate required location and dimensions with other Trades.
- 19. Install sleeves furnished by others. Coordinate required location with other Trades.
- 20. Grout hollow metal frames as required.
- 21. Install access doors required and furnished by others.
- 22. Expansion joints as required and as indicated in the documents.
- 23. Glazed block.
- 24. Field touch-up of shop primers, paint, galvanizing, etc.
- 25. Remove excess materials from the site in a timely manner as directed by Clark Construction Company.

 Saline Area Schools 2015 Bond Program – 2017 Projects Phase 1
 002416-2

 Bid Release No. 4
 Project No. 2714

 Section 002416-Scope of Work-Bid Category Specific Notes, Bid Category 6X-Rough and Finish Carpentry and General Trades
 Version 2

- 26. Handrails.
- 27. Toilet Partition support framing.
- 28. Toilet room countertop support framing.
- 29. Loose masonry lintels.
- 30. Furnish required shop drawings in a timely manner to allow for adequate review time by Clark Construction Company and the Architect to meet requirements of the project schedule.
- 31. Coordinate with the Mechanical Contractor and other Trades requiring openings and support framing in the structural steel for the HVAC units and other material and equipment.
- 32. All welders shall be certified for the welding Work they will be performing. Each welder's certificate shall be submitted prior to performing the Work.
- 33. Hoisting equipment support mats required to complete the Work of this bid category.
- 34. Provide generator, electrical power as needed to perform the Work of this bid category.
- 35. Remove temporary shoring and bracing associated with the Work of this bid category. Schedule removal to assure safety.
- 36. Removal of rust, scale and slag from steel in areas to be exposed or to receive paint, etc.
- 37. Prime painting except in locations where it will inhibit the quality of Work by other Trades (e.g. steel embedded in concrete, steel to receive fireproofing, galvanizing, etc.).
- 38. Clean dirt and mud from all materials following erection.
- 39. Trade Contractor representatives involved in the Work of this bid category including project manager, detailer, fabricator, erector, safety representative, etc. shall attend a "Structural Steel Start-Up Meeting".
- 40. All hollow metal frames (including anchoring devices), hollow metal doors, and finish hardware.
- 41. Reinforcing of framing and doors as required for hardware installation.
- 42. Deliveries shall be phased to meet requirements of the project schedule.
- 43. Construction cores and final keying of all hardware furnished by this bid category.
- 44. Number all door frames on the center hinge knockout.
- 45. Lock cylinders and cores for aluminum doors.
- 46. Final adjustments to hardware as directed by Clark Construction Company.
- 47. Coordinate undercutting of doors with floor finishes.
- 48. Coordinate opening and other dimensions as required with other Trades.
- 49. Electric motors and operators.
- 50. Verify door travel and adjust as required upon completion of the project.
- 51. Blocking required for a complete installation whether or not indicated in the Documents.
- 52. Coordinate ceiling layout with other Trades.
- 53. Cutting of material for penetrations by other Trades.

 Saline Area Schools 2015 Bond Program – 2017 Projects Phase 1
 002416-3

 Bid Release No. 4
 Project No. 2714

 Section 002416-Scope of Work-Bid Category Specific Notes, Bid Category 6X-Rough and Finish Carpentry and General Trades
 Version 2

- 54. Provide additional lighting as needed to provide high quality gypboard finish.
- 55. Fire stopping associated with the work of this bid category.
- 56. Framing for built in equipment (i.e., recessed overhead projector screens, recessed projector lifts, built in televisions, etc.).
- 57. Temporary disconnect projector and reinstall after new ceiling is in place.
- 58. Remove mortar and provide joint sealants in masonry.
- 59. Coordinate location and dimensions of wall and ceiling openings required by other Trades. Provide opening framing as required.
- 60. Reinstall tackboards, markerboards and operable partitions.
- 61. Remove and re-instal Owner's dedication placques.
- 62. Rework existing ceiling around new work items.
- 63. Sound baffles.
- 64. Prefabricated column covers.
- 65. Hoisting and scaffolding as required to complete the Work of this bid category.
- 66. Masonry mock ups for Owner/Architect approval.
- 67. Coordinate window, door and other opening dimensions with other Trades. Construct opening dimensions within tolerances provided by others to allow for ordering and manufacturing of window systems, doors and other material prior to masonry installation.
- 68. Sealants associated with the Work of this bid category.
- 69. Special brick shapes (i.e. lip brick, circular, corner etc.) as required.
- 70. Cleaning of masonry.
- 71. Installation of sound attenuating material at voids at top of masonry and other openings as required.
- 72. Masonry "toothing" required at new openings in existing walls. Reuse old brick when possible.
- 73. Maintain streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.
- 74. Install roof drain sump pans furnished by others.
- 75. Structural framing shall be designed by a registered engineer. Design documents shall include stamp of the registered engineer acceptable to regulatory agencies.
- 76. Gypboard joint compound application requirements beneath painted finish shall meet or exceed recommendations for the type of paint (i.e., flat, semi gloss, gloss) and highest quality finished appearance. Building lighting and natural lighting shall be taken into consideration in application of joint compound to assure highest quality finished appearance.
- 77. Minor spotting of gypboard prior to final coat of paint.
- 78. Acoustical wall panels.
- 79. Building insulation.
- 80. Sound insulation.

#### Health and Safety Special Requirements:

- 1. Provide a minimum of 20 pound U.L. rated multi-purpose (Type A, B, C) fire extinguishers on stands spaced throughout the building at all times spaced such that the travel distance to a fire extinguisher shall not exceed 100 feet. Fire extinguisher stand shall be "Fire-Safety Mate" of equal as manufactured by Goria Corporation, (telephone 888.464.6742).
- 2. Steel Erector's Qualified Person shall develop a detailed erection plan in accordance with OSHA 1926, Subpart R, Appendix A. The erection plan shall be reviewed with Clark Construction Company Project Superintendent.
  - a. Changes in the erection plan shall be approved by the Erector's Qualified Person
  - b. The steel erection plan shall be maintained at the project site and a copy provided to Clark Construction Company's Project Superintendent.
- 3. Steel erection shall not start prior to written conformation from Clark Construction Company that concrete associated with steel erection has attained 75% of the minimum intended compressive strength or sufficient strength to support the loads imposed during steel erection. Refer to MIOSHA Part 26 requirements.
- 4. Submit structural steel bracing plan to Clark Construction Company prior to start of erection of structural steel.

## Exclude:

- 1. Aluminum door system hardware.
- 2. Electrical connection of access control system.

## **BID CATEGORY 04 D - Roofing**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include <u>all</u> Work specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

072100 - THERMAL INSULATION

076100 - METAL ROOFING SYSTEM

076200 - SHEET METAL FLASHING AND TRIM

077200 – ROOF ACCESSORIES

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below)

- 024119 SELECTIVE DEMOLITION
- 061053 MISCELLANEOUS ROUGH CARPENTRY
- 061600 SHEATHING

078100 – APPLID FIREPROOFING

078413 - PENETRATION FIRESTOPPING

079200 – JOINT SEALANTS

Include (Furnish and Install u.n.o):

- 1. Roof accessories as required for a complete system whether or not indicated in the documents.
- 2. Flashing associated with the Work of this bid category whether or not indicated in the documents.
- 3. Wood blocking as required but not indicated in the documents.
- 4. Expansion joint assemblies associated with the Work of this bid category.
- 5. Flashing of all piping, conduit and other penetrations required by other Trades.
- 6. Coordinate installation of roof drains installed by others.
- 7. Sealants associated with the Work of this bid category as required.
- 8. Sheet metal, copings, etc. associated with the Work of this bid category.
- 9. Girts, furring and other supports systems as required to accomplish the Work of this bid category whether or not indicated in the documents.
- 10. Verify quality of substrate installed by other Trades to result in quality results of this bid category. Report discrepancies to Clark Construction Company in writing.
- 11. Field measure to assure proper fit as necessary to meet the project schedule.

- 12. Coordinate location and dimensions of penetrations through the system as required by other Trades. Provide weather tight seal and/or accessories as required.
- 13. Sealants including joints abutting dissimilar materials and the Work of other Trades.
- 14. Miscellaneous closures and other metal fabrications as required for a complete system whether or not indicated in the documents.
- 15. Roofing demolition.
- 16. Single ply roofing.
- 17. Coordinate location of reglets with other Trades as required.
- 18. Insulation associated with the Work of this bid category including vapor barriers as required.

#### Health and Safety Special Requirements:

1. Utilize warning lines, guard rails, safety nets, fall arrest systems and safety monitoring in accordance with regulatory Safety Authorities.

## Exclude:

1. Wood blocking indicated in the Documents.

## **BID CATEGORY 04E - Glass and Glazing**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include all Work specified or reasonably inferred)

002413 - SCOPE OF WORK GENERAL NOTES - (Include all Work specified or reasonably inferred)

081743 - FRP FLUSH DOORS

084113 – ALUMINUM FRAMED ENTERANCES

084413 – GLAZED ALUMINUM CURTAIN WALLS

084523 – FIBERGLASS SANDWICH PANEL ASSEMBLIES

087100 – DOOR HARDWARE

088000 - GLAZING

088010 - GLASS WITH INTEGRAL BLINDS

122200 – GLASS FURNITURE WALLS

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below)

024119 - SELECTIVE DEMOLITION

- 061053 MISCELLANEOUS ROUGH CARPENTRY
- 079200 JOINT SEALANTS

087200 – ACCESS CONTROL SYSTEM

281300 – ACCESS CONTROL

Include (Furnish and Install u.n.o):

- 1. Interior and exterior glazing including frames, doors, etc., furnished by others.
- 2. Coordinate rough opening requirements with other Trades.
- 3. Coordinate routing of line and low voltage wiring, pneumatic lines, etc. with other Trades to accomplish all such Work through the framing system. Review routing with Clark Construction Company and the Architect prior to performing the Work.
- 4. Field verify opening dimensions and condition of surfaces in advance of arriving on site to perform the work of this Bid Category to allow adequate time for corrections to be made as not to delay the project schedule. Immediately report discrepancies to Clark Construction Company in writing.
- 5. Wood blocking as required but not indicated in the documents.
- 6. Complete shimming and anchorage.
- 7. Finish hardware associated with the Work of this bid category.
- 8. Reinforcing of framing and doors as required for hardware installation.
- 9. Closure metal for a complete installation.

Saline Area Schools 2015 Bond Program – 2017 Projects Phase 1 Bid Release No. 4

002416-1

- 10. Isolate all dissimilar metals.
- 11. Interior and exterior joint sealant associated with the Work of this bid category.
- 12. System water leakage testing.
- 13. Test windows and doors hardware for ease of operation upon completion of installation.
- 14. Final cleaning of framing system and glass.
- 15. Security film.
- 16. Complete keyed reflected ceiling plan note #7 include demolition, temporary protection and proper installation.

Exclude:

1. Card access.
### **BID CATEGORY 04F-Access Control**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include <u>all</u> Work as specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

270000 - GENERAL TECHNOLOGY REQUIRMENTS

271000 – GENERAL CABLING REQUIRMENTS

281300 – ACCESS CONTROL

- 1. MEP Contractors are required to submit MEP Cost Breakdown as a start-up submittal for approval.
- 2. Coordinate design of system material and equipment routing with other Trade Contractors as required to avoid conflicts.
  - a. This Trade Contractor shall develop shop drawings equal to scale as that to be used by other Trade Contractors in which conflicts have potential to occur. Drawing scale shall be coordinated prior to start of shop drawings.
  - b. Shop drawings shall be extensively and thoroughly coordinated with each Trade Contractor with which conflicts have potential to occur to the extent required to eliminate conflicts prior to fabrication and installation.
  - c. Conflicts encountered during fabrication and installation which could have been foreseen through a more extensive coordination effort shall be corrected by this bid category.
  - d. All costs for modifications which are a result of conflicts with items noted in the Contract Documents shall be born by this bid category.
  - e. In the event of conflict with items indicated in shop drawings of other Trade Contractors and not in the Contract Documents, the cost shall be shared equally by the Trade Contractors involved.
- 3. Coordinate openings required to accommodate installation of equipment prior to enclosure of spaces by other Trades.
- 4. Hangers, supports, unistrut or misc. steel required for a complete installation.
- 5. Equipment support structures as required.
- 6. Furnish only, for installation by others, access panels as required for this Work but not indicated in the documents. Confirm type of panel required with Clark Construction Company prior to ordering.

- 7. Fire stopping of penetrations associated with the Work of this bid category.
- 8. Furnish sleeves as needed and not indicated in the Documents for installation by others. Provide layout and assistance with placement.
- 9. Coring as required to perform the Work of this bid category.
- 10. This Contractor to post revised drawings for the area under review on a regular schedule as defined by Clark Construction Company. Failure to do so results in a Notice to Comply.
- 11. All drawings must incorporate all portions of the system to show the actual size of the system (i.e., insulation on systems, different types of hanging systems and reduction in system size as it progresses through the building).
- 12. Obtain permits required to complete the Work of this bid category. Post at the jobsite prior to performing the Work.
- 13. Warranty and Guarantee start date shall be the project Substantial Completion date.
- 14. Train Owner's representative(s) on proper start-up, operation and maintenance of all mechanical systems. Provide video recording of training for Owner.

#### Exclude:

1. Door hardware listed in specification section 087200 – Access Control System as furnished by Door Hardware Division 8 and Installed by Others. Final connection is the responsibility of this bid category.

# **BID CATEGORY 04G – Floor Covering and Ceramic Tile**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION: (Include all Work specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

093013 - CERAMIC TILING

094000 – PORTLAND CEMENT TERRAZZO FLOORING (PATCH AND REPAIR)

096513 – RESILIENT BASE AND ACCESSORIES

096516 - RESILIENT SHEET FLOORING

096519 – RESILIENT TILE FLOORING

096813 - TILE CARPETING

096816 - SHEET CARPETING

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below) 079200 – JOINT SEALANTS

- 1. Inspect substrate for quality of surface and compatibility with materials of this bid category prior to performing the work. Report discrepancies to Clark Construction Company in writing.
- 2. Substrate surface preparation including minor filling, grinding and sweeping as required prior to performing the work. Report other than minor preparation work required prior to beginning the work to Clark Construction Company in writing.
- 3. Inspect surfaces to receive base material prior to installation and report any deficiencies to Clark Construction Company.
- 4. Sweeping and cleaning of substrate prior to installation of flooring.
- 5. Wall base associated with the work of this bid category.
- 6. Transitions material as required to accomplish acceptable transition between dissimilar materials.
- 7. Sealants associated with the work of this bid category.
- 8. Cleanup of adhesives, loose carpet fibers, etc. upon completion of the Work.
- 9. Wall and other surface scuffing shall be avoided. Cost incurred as a result of such scuffing shall be born by this bid category.
- 10. Deliver, unload and place in storage specified maintenance materials when and where as directed by Clark Construction Company.

- 11. Expansion joint assemblies required within this Work.
- 12. Vacuum carpet when directed by Clark Construction Company.
- 13. Include an allowance of 80 hours for substrate surface patching. Work tickets signed by Clark Construction Company shall be provided to document hours used. A deduct change order will be issued for any unused portion of this allowance.
- 14. Final cleaning and waxing of resilient flooring. Schedule of this work shall be coordinated with Clark Construction Company.
- 15. Preparation of existing tile surface for proper installation of new floor covering material.
- 16. Provide additional lighting as required to provide high quality workmanship.
- 17. Protection of adjacent surfaces.
- 18. Subsurface transition as required.
- 19. Substantial barricading of area of Work to prevent foot traffic until completion of the Work and adequate curing for use. Include removal of barricading.
- 20. Cleaning of flooring upon completion.
- 21. Stone thresholds.
- 22. Terrazzo.

### **BID CATEGORY 04H -Painting and Wall Covering**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include <u>all</u> Work specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

097200 – WALL COVERINGS

099123 - INTERIOR PAINTING

099600 – HIGH PERFORMANCE COATINGS

099666 - DRY ERASE CLEAR COAT

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below)

033000 – CAST-IN PLACE CONCRETE

042000 – UNIT MASONRY

079200 – JOINT SEALANTS

081113 - HOLLOW METAL DOORS AND FRAMES

092216 – GYPSUM BOARD

- 1. Painting of exposed interior and exterior surfaces.
- 2. Inspect substrate for quality of surface and compatibility with materials of this bid category prior to performing the Work. Report discrepancies to Clark Construction Company in writing.
- 3. Cleaning and preparation of surfaces as required prior to application of paint and/or wall covering.
- 4. Provide supplemental lighting as required to provide high quality workmanship.
- 5. Notify Clark Construction Company of completion of each coat to verify Work completed.
- 6. Sealants between dissimilar materials including door frames, window frames, etc. to provide a complete finished product.
- 7. Wood filler and sanding as required.
- 8. Remove and replace door silencers, door hardware and other items as required.
- 9. Protect fire rating labels from paint.
- 10. Protect adjacent surfaces not to be painted including exposed concrete floors.
- 11. Protection tape and other materials shall be removed in a timely manner to avoid damage to material and equipment surfaces.
- 12. Deliver, unload and place specified maintenance materials in storage when and where as directed by Clark Construction Company.

- 13. Sand and refinish face of existing stage.
- 14. De-gloss factory applied primers as required to provide proper surface for painting.
- 15. Report potentially high VOC levels to Clark Construction Company in writing prior to painting.
- 16. Paint all backboards with fire retardant paint.
- 17. Minor patch and paint touch-up prior to owner occupancy. "Minor" as used here shall mean not more than:
  - a. Three nicks or mars up to <sup>1</sup>/4" deep per 320 square feet in heavy traffic areas.
  - b. Two nicks or mars up to <sup>1</sup>/<sub>4</sub>" deep per 320 square feet in moderate traffic areas.
  - c. Three nicks or mars up to  $\frac{1}{4}$ " deep per 500 square feet in rooms on average.
  - d. Two nicks or mars in hollow metal doors and/or frames in heavy traffic areas.
  - e. One nick or mar in hollow metal doors and/or frames in medium and light traffic areas.

#### **BID CATEGORY 04I – Auditorium Seating**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

<u>BASE SPECIFICATION</u> – (Include <u>all</u> Work specified or reasonably inferred) <u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred) 126100 – FIXED AUDIENCE SEATING

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below) 033000 – CAST-IN-PLACE CONCRETE 260500 – BASIC ELECTRICAL REQUIREMENTS 260534 – CONDUIT 260537 – BOXES

Include (Furnish and Install u.n.o):

- 1. Fixed audience seating.
- 2. Conduit from the audience seating to the junction box.
- 3. All audience seating anchors.

Exclude:

- 1. Demolition and proper disposal of seating including anchors.
- 2. Electrical connections to the junction box.

# **BID CATEGORY 04J – HVAC and Plumbing**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION - (Include all Work specified or reasonably inferred) 002413 - SCOPE OF WORK GENERAL NOTES - (Include all Work specified or reasonably inferred) 220517 - SLEEVES AND SLEEVE SEALS FOR PLUMBING PIPING 220518 - ESCUTCHEONS FOR PLUMBING PIPING 220523.12 – BALL VALVES FOR PLUMBING PIPING 220529 – HANGERS AND SUPPORTS FOR PLUBING PIPING AND EQUIPMENT 220553 – IDENTIFICATION FOR PLUMBING PIPING AND EQUIPMENT 220719 – PLUMBING PIPING INSULATION 221113 - FACILITY WATER DISTRIBUTION PIPING 221116 – DOMESTIC WATER PIPING 221119 – DOMESTIC WATER PIPING SPECIALTIES 221316 - SANITARY WASTE AND VENT PIPING 221319 - SANITARY WASTE PIPING SPECIALTIES 224213.13 – COMMERCIAL WATER CLOSETS 224213.16 - COMMERCIAL URINALS 224216.13 – COMMERCIAL LAVATORIES 224216.16 - COMMERCIAL SINKS 224716 – PRESSURE WATER COOLERS 230513 - COMMON MOTOR REQUIRMENTS FOR HVAC EQUIPMENT 230514 - VARIABLE FREQUENCY MOTOR CONTROLLERS 230517 - SLEEVES AND SLEEVE SEALS FOR HVAC PIPING 230518 - ESCUTCHEONS FOR HVAC PIPING 230519 - METERS AND GAGES FOR HVAC PIPING 230523.12 – BALL VALVES FOR HVAC PIPING 230523.13 - BUTTERFLY VALVES FOR HVAC PIPING 230523.14 – CHECK VALVES FOR HVAC PIPING 230529 - HANGERS AND SUPPORTS FOR HVAC PIPING AND EQUIPMENT 230553 - IDENTIFICATION FOR HVAC PIPING AND EQUIPMENT 230593 - TESTING, ADJUSTING AND BALANCING FOR HVAC 230713 - DUCT INSULATION 230719 - HVAC PIPING INSULATION 230993 – SEQUENCE OF OPERATIONS 231123 - FACILITY NATURAL GAS PIPING 232113 – HYDRONIC PIPING 232116 - HYDRONIC PIPING SPECIALTIES

232123 – HYDRONIC PUMPS

- 232300 REFRIGERANT PIPING
- 233113 METAL DUCTS
- 233300 AIR DUCT ACCESSORIES
- 233423 HVAC POWER VENTILATORS
- 233713 DIFFUSERS, REGISTERS AND GRILLES
- 235123 GAS VENTS
- 235216 CONDENSING BOILERS
- 236313 AIR COOLED REFRIGERANT CONDENSERS
- 237200 AIR TO AIR ENERGY RECOVERY EQUIPMENT
- 237413 PACKAGED, OUTDOOR, CENTRAL-STATION AIR-HANDLING UNITS
- 237433 DEDICATED OUTDOOR AIR UNITS
- 238213 VALANCE HEATING AND COOLING UNITS
- 238216.11 HYDRONIC AIR COILS
- 238223 UNIT VENTILATORS
- 238239.13 CABINET UNIT HEATERS

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below)

- 024119 SELECTIVE DEMOLITION
- 033000 CAST-IN-PLACE CONCRETE
- 051200 STRUCTURAL STEEL FRAMING
- 055000 METAL FABRICATIONS
- 078100 APPLIED FIREPROOFING
- 078413 PENETRATION FIRESTOPPING
- 079200 JOINT SEALANTS
- 031200 EARTH MOVING

- 1. MEP Contractors are required to submit MEP Cost Breakdown as a start-up submittal for approval.
- 2. Coordinate design of system material and equipment routing with other Trades as required to avoid conflicts.
  - a. This Trade Contractor shall develop shop drawings equal to scale as that to be used by other Trade Contractors in which conflicts have potential to occur. Drawing scale shall be coordinated prior to start of shop drawings.
  - b. Shop drawings shall be extensively and thoroughly coordinated with each Trade Contractor with which conflicts have potential to occur to the extent required to eliminate conflicts prior to fabrication and installation.
  - c. Conflicts encountered during fabrication and installation which could have been foreseen through a more extensive coordination effort shall be corrected by this bid category.
  - d. All costs for modifications which are a result of conflicts with items noted in the Contract Documents shall be born by this bid category.
  - e. In the event of conflict with items indicated in shop drawings of other Trade Contractors and not in the Contract Documents, the cost shall be shared equally by the Trade Contractors involved.

# SECTION 002416

# SCOPE OF WORK

# BID CATEGORY SPECIFIC NOTES

- 3. Coordinate openings required to accommodate installation of equipment prior to enclosure of spaces by other Trades.
- 4. Hangers, supports, unistrut or miscellaneous steel required for a complete installation.
- 5. Verify existing grade prior to performing the Work. Report discrepancies to Clark Construction Company in writing.
- 6. Excavation, backfill and compaction associated with this Work.
- 7. Restore disturbed grade to conditions that existed prior to performing the Work.
- 8. Final hook-up of plumbing equipment and fixtures furnished by Owner and others.
- 9. Final connection of kitchen equipment.
- 10. Coring as required to perform the Work of this bid category.
- 11. Insulation associated with the Work of this bid category.
- 12. Furnish sleeves as needed and not indicated in the Documents for installation by others. Provide layout and assistance with placement.
- 13. Coordinate floor drain elevations with Clark Construction Company and other Trades.
- 14. Paint touch-up of factory furnished equipment.
- 15. Identification, labeling and tagging system equipment, piping, etc.
- 16. Remove spoils from the site and backfill in accordance with the plans and specifications requirements.
- 17. Furnish roof drain sump pans for installation by others. Provide layout and assistance with placement.
- 18. Warranty and Guarantee start date shall be the project Substantial Completion date.
- 19. Train Owner's representative(s) on proper start-up, operation and maintenance of all mechanical systems. Provide video recording of training for Owner.
- 20. Material bends and offsets in addition to those indicated in the documents associated with the Work of this bid category and required to avoid installation conflicts. Contract Documents shall not be interpreted to include all bends and offsets required to avoid conflicts.
- 21. Coordinate roof and other opening dimensions and locations required with Structural and other Contractors.
- 22. Remove temporary covers over mechanical openings installed by others prior to performing the Work (including cutting metal deck or burning of welds). Maintain openings in safe manner.
- 23. Furnish, for installation by others, access panels as required for this Work but not indicated in the documents. Confirm type of panel required with Clark Construction Company prior to ordering.
- 24. Access panels as required in duct to access in-duct coils or other equipment.
- 25. HVAC Work as described in specification section "Temporary Facilities and Controls".
- 26. Mechanical insulation.
- 27. Roof curbs associated with the Work of this bid category.
- 28. Concrete housekeeping pads for mechanical equipment required and not indicated in the documents.
- 29. Support steel required for mechanical equipment and not indicated in the documents.
- 30. Fire stopping at penetrations through fire walls and smoke barriers.
- 31. Testing and balancing of system. Submit final test and balance reports in a timely manner as not to delay other project requirements.

- 32. Complete check, test and start up of mechanical systems. Provide written plan and documentation to Clark Construction Company of check, test and start of each component of the system.
- 33. Identification, labeling and tagging system equipment, piping, duct, etc.
- 34. Framed identification charts in mechanical rooms providing valve and major equipment information.
- 35. Obtain permits required to complete the Work of this bid category. Post at the jobsite prior to performing the Work.
- 36. Temperature controls including raceways, wiring, complete commissioning and user training.
- 37. Participate in fire alarm testing process as it relates to the Work of this bid category.
- 38. Mechanical and plumbing demolition indicated in the documents. Include cutting and capping of items to remain. Dispose of materials off site.
- 39. Disconnects and motor starters as required for mechanical equipment.
- 40. Pumps and associated variable frequency drives.
- 41. Breeching, flues, stacks and chimneys for all fuel fired equipment.

### **BID CATEGORY 04K - Electrical**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include <u>all</u> Work as specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

- 260500 BASIC ELECTRICAL REQUIRMENTS
- 260501 MINOR ELECTRICAL DEMOLITION
- 260519 LOW VOLTAGE ELECTRICAL POWER CONDUCTORS AND CABLES
- 260526 GROUNDING AND BONDING FOR ELECRICAL SYSTEMS
- 260529 HANGERS AND SUPPORTS FOR ELECTRICAL SYSTEMS
- 260534 CONDUIT
- 260537 BOXES
- 260553 IDENTIFICATION FOR ELECTRICAL SYSTEMS
- 260923 LIGHTING CONTROL DEVICES
- 262416 PANELBOARDS
- 262726 WIRING DEVICES
- 265100 INTERIOR LIGHTING
- 265600 EXTERIOR LIGHTING
- 283101 FIRE ALARM SYSTEMS

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below)

- 024119 SELECTIVE DEMOLITION
- 033000 CAST-IN-PLACE CONCRETE
- 078100 APPLIED FIREPROOFING
- 078413 PENETRATION FIRESTOPPING
- 079200 JOINT SEALANTS
- 031200 EARTH MOVING

- 1. MEP Contractors are required to submit MEP Cost Breakdown as a start-up submittal for approval.
- 2. Coordinate openings required to accommodate installation of equipment prior to enclosure of spaces by other Trades.
- 3. Hangers, supports, unistrut or misc. steel required for a complete installation.
- 4. Equipment support structures as required.
- 5. Fire stopping of penetrations associated with the Work of this bid category.

- 6. Coordinate design of system material and equipment routing with other Trade Contractors as required to avoid conflicts.
  - a. This Trade Contractor shall develop shop drawings equal to scale as that to be used by other Trade Contractors in which conflicts have potential to occur. Drawing scale shall be coordinated prior to start of shop drawings.
  - b. Shop drawings shall be extensively and thoroughly coordinated with each Trade Contractor with which conflicts have potential to occur to the extent required to eliminate conflicts prior to fabrication and installation.
  - c. Conflicts encountered during fabrication and installation which could have been foreseen through a more extensive coordination effort shall be corrected by this bid category.
  - d. All costs for modifications which are a result of conflicts with items noted in the Contract Documents shall be born by this bid category.
  - e. In the event of conflict with items indicated in shop drawings of other Trade Contractors and not in the Contract Documents, the cost shall be shared equally by the Trade Contractors involved.
- 7. Furnish only, for installation by others, access panels as required for this Work but not indicated in the documents. Confirm type of panel required with Clark Construction Company prior to ordering.
- 8. Concrete housekeeping pads for electrical equipment required and not indicated in the documents.
- 9. Concrete removal, concrete, excavation, sawcutting, backfill and compaction associated with the Work of this bid category.
- 10. Warning tape in trenches above conduit and wire.
- 11. Restore grade to condition that existed prior to start of the Work.
- 12. Furnish sleeves as needed and not indicated in the Documents for installation by others. Provide layout and assistance with placement.
- 13. Coring as required to perform the Work of this bid category.
- 14. Perform final hook up of equipment furnished by Owner and others. Coordinate rough in location with equipment supplier and Clark Construction Company.
- 15. Electrical Work associated with fire protection system equipment (e.g. tamper and flow switches, etc.) installed by Fire Protection Contractor.
- 16. Electrical identification and labeling.
- 17. Obtain permits required to complete the Work of this bid category. Post at the jobsite prior to performing the Work.
- 18. Site electrical Work.
- 19. Site lighting including light pole bases.
- 20. Fire alarm system.
- 21. Verify with electrical utility company that high voltage rating indicated on the documents to be actual voltage prior to submission of service equipment material submittals.
- 22. Starters and disconnects for equipment.

- 23. Install variable speed drives furnished by others including conduit and wiring.
- 24. Warranty and Guarantee start date shall be the project Substantial Completion date.
- 25. Train Owner's representative(s) on proper start-up, operation and maintenance of all mechanical systems. Provide video recording of training for Owner.
- 26. Electrical demolition indicated in the documents. Include cutting and capping of items to remain. Dispose of materials off site.
- 27. Connection and wiring for the auditorium seating.
- 28. Relocate copier.

#### Exclude:

1. Electrical conduit from auditorium seating to junction box.

### **BID CATEGORY 04L - Communications**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

<u>BASE SPECIFICATION</u> – (Include <u>all</u> Work as specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

270000 - GENERAL TECHNOLOGY REQUIRMENTS

271000 - GENERAL CABLING REQUIRMENTS

271500 – DATA AND VOICE HORIZONTAL CABLING

274000 – GENERAL A/V SYSTEMS REQUIREMENT

282300 - VIDEO SURVEILLANCE

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below)

078100 – APPLIED FIREPROOFING

078413 – PENETRATION FIRESTOPPING

079200 – JOINT SEALANTS

- 1. MEP Contractors are required to submit MEP Cost Breakdown as a start-up submittal for approval.
- 2. Coordinate design of system material and equipment routing with other Trade Contractors as required to avoid conflicts.
  - a. This Trade Contractor shall develop shop drawings equal to scale as that to be used by other Trade Contractors in which conflicts have potential to occur. Drawing scale shall be coordinated prior to start of shop drawings.
  - b. Shop drawings shall be extensively and thoroughly coordinated with each Trade Contractor with which conflicts have potential to occur to the extent required to eliminate conflicts prior to fabrication and installation.
  - c. Conflicts encountered during fabrication and installation which could have been foreseen through a more extensive coordination effort shall be corrected by this bid category.
  - d. All costs for modifications which are a result of conflicts with items noted in the Contract Documents shall be born by this bid category.
  - e. In the event of conflict with items indicated in shop drawings of other Trade Contractors and not in the Contract Documents, the cost shall be shared equally by the Trade Contractors involved.

- 3. Coordinate openings required to accommodate installation of equipment prior to enclosure of spaces by other Trades.
- 4. Hangers, supports, unistrut or misc. steel required for a complete installation.
- 5. Equipment support structures as required.
- 6. Furnish only, for installation by others, access panels as required for this Work but not indicated in the documents. Confirm type of panel required with Clark Construction Company prior to ordering.
- 7. Fire stopping of penetrations associated with the Work of this bid category.
- 8. Furnish sleeves as needed and not indicated in the Documents for installation by others. Provide layout and assistance with placement.
- 9. Coring as required to perform the Work of this bid category.
- 10. This Contractor to post revised drawings for the area under review on a regular schedule as defined by Clark Construction Company. Failure to do so results in a Notice to Comply.
- 11. All drawings must incorporate all portions of the system to show the actual size of the system (i.e., insulation on systems, different types of hanging systems and reduction in system size as it progresses through the building).
- 12. Obtain permits required to complete the Work of this bid category. Post at the jobsite prior to performing the Work.
- 13. Warranty and Guarantee start date shall be the project Substantial Completion date.
- 14. Train Owner's representative(s) on proper start-up, operation and maintenance of all mechanical systems. Provide video recording of training for Owner.

### **BID CATEGORY 04M – Earthwork and Site Utilities**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include <u>all</u> Work specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

311000 - SITE CLEARING

312000 – EARTH MOVING

334100 – STORM UTILITY DRAINAGE PIPING

334600 – SUBDRAINAGE

- 1. Dust control for the duration of this work.
- 2. Maintain Streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.
- 3. Remove debris from the site in a timely manner.
- 4. Legal disposal of materials off site.
- 5. Pumping water to accomplish the work of this bid category.
- 6. Excavation and backfill associated with the work.
- 7. Coordinate, accomplish, verify and provide evidence of utility shut downs and/or disconnects.
- 8. Obtain permits and licenses required to complete the work of this bid category from governing authorities.
- 9. All site demolition included but not limited to tree removal, clearing, grubbing, exterior concrete, asphalt and signage.
- 10. Remove above ground, overhead and below grade materials (e.g. pavement and base material, utilities, etc.) indicated in the documents.
- 11. Import and export of soils required to accomplish the work of this bid category.
- 12. Protect existing structures and/or trees and landscaping to remain.
- 13. Grade, seed and mulch areas disturbed outside of the building area as a result of the work of this bid category.
- 14. Overhead and underground utility cutting, capping, valves, etc., as required to accomplish the work and as required by authorities having jurisdiction.
- 15. Coordinate turnover of material to be salvaged with Clark Construction Company.
- 16. Site grading to plus or minus 1/10 of a foot of design.
- 17. Verify on-site and surrounding grades are in accordance with the Documents prior to start of the work. Report discrepancies to Clark Construction Company in writing.

- 18. Proof rolling of sub-grade as required prior to placement of stone base, asphalt, concrete or other materials.
- 19. Strip, stockpile and re-spread topsoil at a later date as directed by Clark Construction Company. Include remobilization.
- 20. Submit soil erosion control plan to governing authorities and secure all permits.
- 21. Furnish, install, maintain and inspect as required all soil erosion control including all existing storm structure and storm system sedimentation guards as required in accordance with the Documents and requirements of authorities having jurisdiction.
- 22. Remove soil erosion control after permanent vegetation is established and approved by governing authorities and as directed by Clark Construction Company.
- 23. Provide soil erosion system inspections by a certified storm water operator as required by authorities having jurisdiction.
- 24. Verify location of existing utilities prior to performing the Work. Hand dig as required to avoid contact with underground utilities. Report location, invert elevation discrepancies or interferences to Clark Construction in writing.
- 25. Remove spoils and debris from the site in a timely manner.
- 26. Perform water, sanitary and storm tie-in to public utility system in accordance with requirements of authorities having jurisdiction.
- 27. Install drainage structures such that Subgrade will drain completely by gravity. Final adjustments of the drainage structures shall be performed by this bid category at a later date just prior to placement of finished surfaces and as directed by Clark Construction Company. Final adjustments shall include required excavation, backfill and compaction.
- 28. Coordinate and perform final adjustments of valve boxes and other utilities prior to placement of finished surfaces by others.
- 29. Subdrainage with proper bedding.
- 30. Saw cutting and removal of existing curbs may be required for ingress and egress.
- 31. Grade, seed and mulch areas disturbed outside of the building area as a result of the Work of this bid category.
- 32. Clean dirt and debris out of storm water system upon completion of the project when directed by Clark Construction Company.

#### Exclude:

- 1. Site electrical demolition.
- 2. Fence posts.

#### **BID CATEGORY 04N - Asphalt Paving**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION-(Include <u>all</u> Work specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

321216 – ASPHALT PAVING

321723 – PAVEMENT MARKINGS

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below) 312000 – EARTH MOVING

- 1. Verify Contract Document asphalt design will provide positive drainage in all areas prior to performing the work. Report potential problem areas to Clark Construction Company in writing.
- 2. Verify existing grade elevations prior to performing the work. Report discrepancies to Clark Construction Company in writing.
- 3. Maintain streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.
- 4. Coordinate adjustment of all storm catch basins, manhole covers, valve boxes, etc. with other Trades prior to beginning the work.
- 5. Proof roll sub-grade prior to placement of stone base.
- 6. Fine grading of subbase material.
- 7. Furnish, place and grade stone base material.
- 8. Finished asphalt shall have positive drainage over the entire surface.
- 9. Pavement markings.
- 10. Site signage.
- 11. Maintain streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.
- 12. Milling of existing asphalt.
- 13. Bond coat for asphalt.

### **BID CATEGORY 40 - Landscaping and Playground Surfacing**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

<u>BASE SPECIFICATION</u> – (Include <u>all</u> Work specified or reasonably inferred) <u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred) 321816.13 – PLAYGROUND SURFACING 329200 – TURF AND GRASSES

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below) 312000 – EARTH MOVING

-----

- 1. Verify location of existing utilities prior to performing the work. Hand dig as required to avoid contact with underground utilities.
- 2. Furnish sleeves to other Trades to accommodate the work of this bid category. Provide layout and assistance with placement.
- 3. Verify existing grades at planting and lawn areas prior to performing the work. Report discrepancies to Clark Construction Company.
- 4. Topsoil raking and removal of rocks, stones and debris.
- 5. Clean-up on a daily basis.
- 6. Legally dispose of debris off site.
- 7. Broom cleaning of hard surfaces.
- 8. Initial staking/support of plantings.
- 9. Weed control and fertilizer per specifications.
- 10. Temporary watering as required until irrigation system is fully functioning.
- 11. Furnish, place and grade topsoil in accordance with the documents.
- 12. Rock hounding of subgrade prior to placement of topsoil and/or sod.
- 13. Topsoil.
- 14. Playground surfacing.
- 15. Maintain streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.

#### **BID CATEGORY 04P – Fencing**

GENERAL – The following shall not be interpreted as a complete itemization of the work to be performed under this Bid Category. This Bid Category Trade Contractor shall be responsible to perform all work reasonably interpreted to be included in its scope of work in accordance with the drawings and specifications in addition to these Bid Category notes of clarification.

BASE SPECIFICATION – (Include <u>all</u> Work specified or reasonably inferred)

<u>002413 – SCOPE OF WORK GENERAL NOTES</u> – (Include <u>all</u> Work specified or reasonably inferred)

323113 – CHAIN LINK FENCES AND GATES

323119 – ALUMINUM ORNAMENTAL FENCE SYSTEM

<u>REFERENCE SPECIFICATION</u> – (Include <u>portions</u> of the Work specified as noted below) 033000 – CAST-IN-PLACE CONCRETE

312000 – EARTH MOVING

- 1. Dust control for the duration of this work.
- 2. Maintain Streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.
- 3. Remove debris from the site in a timely manner.
- 4. Legal disposal of materials off site.
- 5. Pumping water to accomplish the work of this bid category.
- 6. Excavation, backfill and concrete associated with the work.
- 7. Chain link fences and gates with required fabric and hardware.
- 8. Aluminum ornamental fence system.
- 9. Import and export of soils required to accomplish the work of this bid category.
- 10. Protect existing structures and/or trees and landscaping to remain.
- 11. Grade, seed and mulch areas disturbed outside of the building area as a result of the work of this bid category.
- 12. Maintain streets and public areas free of dirt, mud and debris. Daily or more frequent road sweeping as required when debris is tracked onto roads.
- 13. Site grading to plus or minus 1/10 of a foot of design.

# SECTION 003113 PROJECT MILESTONE SCHEDULE

# 1. <u>GENERAL</u>

1.1. Milestone dates identified herein shall become binding under the Contract. The Bidder represents that it will provide sufficient labor, material, supervision, equipment and other necessary resources to achieve the milestone dates.

### 2. WORK HOURS

- 2.1. Project site standard work hours shall be 7:00 am to 3:30 pm.
- 2.2. Trade Contractor shall notify and obtain authorization from the Construction Manager in the event work is required to be performed outside of standard work hours.
- 2.3. For work required outside of standard work hours which is the result of the Trade contractors failure to perform the work as scheduled during standard hours, the Trade Contractor shall reimburse the Construction Manager for the Construction Managers staff cost to be on site while such work is performed.

#### 3. <u>SCHEDULE REQUIREMENTS</u>

- 3.1. Bidders shall review the milestone schedule for Work included in the construction documents and advise, at bid time, of any severe discrepancies identified.
- 3.2. Bid Proposals shall be based on overall time duration provided in the milestone schedule for all construction activities.
- 3.3. Trade Contractors will be required to submit monthly progress reports and updated schedules as directed by the Construction Manager.
- 3.4. The Construction Manager will endeavor to incorporate Trade Contractor input into the final project schedule. Under no circumstances will data be incorporated that causes delay to any milestone or to the overall project completion date.
- 3.5. Trade Contractor representatives shall attend Pull Planning Meetings as required and as described in Section 013119 Project Meetings.
- 3.6. The Trade Contractor shall review, check, approve and submit, in such sequence as to cause no delay in the Work or in the Work of the Owner or any separate Trade Contractor, all shop drawings, product data and samples required by the Contract Documents.
- 3.7. Trade Contractors failing to meet the project schedule shall be required to submit a detailed recovery schedule within two (2) days of notification by the Construction Manager. Such recovery schedule shall contain sufficient detail to satisfy requirements of the Construction Manager and shall include overtime as necessary to accomplish recovery in the shortest possible duration.

# SECTION 003113 PROJECT MILESTONE SCHEDULE

- 3.8. The Trade Contractor shall be responsible for all additional cost associated with a project delay caused by the Trade Contractor. Project delay costs include but are not limited to: 1) Construction Manager personnel cost associated with additional scheduling and/or coordination time including phone calls, emails, meetings etc.; 2) overtime cost expended by other Trade Contractors and that of the failing Trade Contractor necessary to recover lost schedule time.
- 3.9. All schedule "float time" shall belong exclusively to the Construction Manager. Trade Contractor shall endeavor to improve upon the project schedule dates.

#### 4. <u>MILESTONE SCHEDULE</u>

Start Date: Monday June 19, 2017

Finish Date: Friday August 25, 2017.

PROJECT:	Saline Area Schools 2015 Bond Program 2017 Projects Phase
CLARK PROJECT NO.:	2714
BID RELEASE NO.:	4
OWNER:	Saline Area Schools
ARCHITECT:	Kingscott Associates, Inc.
CONSTRUCTION MANAGER:	Clark Construction Company 3535 Moores River Drive Lansing, MI 48911
PROJECT MANAGER:	
ATTENTION:	Mark Paulus
TELEPHONE: FAX:	517.372.0940 517.372.0668

### 1. <u>PROPOSAL</u>

BIDDER'S NAME:

1.1. This offer has been prepared after our examination of the complete drawings and specifications, together with their related documents, and our examination of the conditions surrounding the construction of the proposed work including the availability of materials, equipment and labor. The undersigned submits the following offer to enter into a Contract with Clark Construction Company and agrees to furnish all labor, material, equipment and service to complete the Work in accordance with the Contract Documents for:

А.	Bid	Category No.: Description:		
	1.	For the Lump Sum Base Bid of:	(\$	)
				Dollars
B.	Bid	Category No.: Description:		
	1.	For the Lump Sum Base Bid of:	(\$	)
				Dollars

1.2. All appropriate sales taxes are included in the above Lump Sum Base Bid.

#### 2. <u>COMBINED BIDS</u>

2.1. Combined bids of two (2) or more Bid Categories may be submitted. Separate bids for each Bid Category included in a combined bid are required.

A.	Bid	Category Numbers:		
	Bid	Category Descriptions:		
	1.	For the Lump Sum Base Bid of:	(\$	)
				Dollars

#### 3. <u>ADDENDA</u>

3.1. The undersigned acknowledges receipt of the following Addenda and has included the cost thereof in the Lump Sum Base Bid:

No. 1, dated	No. 4, dated
No. 2, dated	No. 5, dated
No. 3, dated	No. 6, dated

#### 4. TRADE HOURS

- 4.1. Total estimated trade hours required to perform the Work \_\_\_\_\_\_ Trade Hours
- 4.2. The undersigned acknowledges that the estimated trade hours provided above are for Clark Construction Company scheduling purposes only and shall not be deemed a limit to trade hours required to perform the work and shall not be considered as a basis for claim.

#### 5. <u>BID SECURITY</u>

5.1. Bid security in the amount of 5% of the proposal, shall accompany this proposal.

#### 6. PERFORMANCE AND LABOR AND MATERIAL PAYMENT BOND

- 6.1. The undersigned confirms that the cost of required Bonds is included in the base bid amount. The undersigned agrees to furnish a Performance and Labor and Material Payment Bond, if required, by the Owner or Construction Manager.
  - A. The cost of such bonds, to be added to the base bid amount is: (\$\_\_\_\_\_)

Dollars

#### 7. <u>REJECTION OF BID</u>

7.1. The undersigned acknowledges the right of the Construction Manager and Saline Area Schools to reject any or all bids and to waive any informality or irregularity in the bid.

#### 8. **PROJECT SCHEDULE**

8.1. The undersigned acknowledges that it shall meet requirements of the Project Schedule (Section 003113).

#### 9. EXTRA WORK

- 9.1. The undersigned agrees that:
  - A. A maximum of 15% overhead and profit will be allowed for Changes in the Work performed by the Trade Contractor.
  - B. A maximum of 5% overhead and profit will be allowed for Changes in the Work for any tier Subcontractor.
  - C. For changes involving both additional costs and credits to the Contract, the mark-up will be allowed on the net add only after all credits have been deducted from the additional work.

#### 10. <u>ALTERNATES</u>

- 10.1. General
  - A. Each Bidder <u>must</u> furnish alternate pricing for the Work of its respective Bid Category for the following alternates.
  - B. Alternates shall not be included in the Lump Sum Base Bid.
  - C. Alternate price shall include all cost for labor, material, equipment, service, overhead and profit including any bonds and taxes as required in the Bid Documents to complete the Work of the Bid Category.

#### 10.2. List of Alternates

Alternate #1 –

1.

a. Add/Deduct (Circle one) \$\_\_\_\_\_
2. Alternate #2 
a. Add/Deduct (Circle one)
3. Alternate #3 
a. Add/Deduct (Circle one)
\$\_\_\_\_\_

#### 11. <u>ALLOWANCES</u>

- 11.1. General
  - A. Each Bidder for Bid Categories listed below, <u>must</u> include allowance amount in its base bid (Refer to "Allowances" section).

B.	Allowances for specific materials (i.e., carpet, wall covering, etc.) shall cover the
	material cost only. All other costs including labor, overhead and profit and
	incidentals shall be included in the base bid.

C. Refer to bid Category and Allowance Specification Sections for allowance amount to include in base bid.

BIDDER'S NAME:	
LEGAL ADDRESS:	
	ZIP CODE:
CONTACT NAME:	
TELEPHONE NO.:	
FAX NO.:	
EMAIL ADDRESS:	
LWAIL ADDRESS.	

The Bidder declares the following legal status in submitting this Proposal: (Check one)

A Corporation organized and existing under the laws of the State of Michigan

\_\_\_\_A Partnership

\_\_\_\_Other

#### CONTRACT ACKNOWLEDGEMENT

Trade Contractor hereby acknowledges acceptance of the terms of the Contract Agreement and will enter into the Agreement with no modifications to the terms of the Agreement.

#### **SIGNATURE**

Respectfully submitted:

 SIGNATURE

 BY:

 TITLE:

 DATE:

 WITNESSED BY:

 Federal Employer Identification No.:

004216-4 Project No.2714

State License No.:

#### BIDDER'S NAME:

#### FAMILIAL RELATIONSHIP

#### 1. <u>GENERAL</u>

1.1. Each Bidder shall complete, execute and submit with its Bid Proposal the following Familial Relationship Sworn Statement.

#### 1.2. SWORN STATEMENT

Date:

School District: Saline Area Schools		
Project: Saline Area Schools 2015 Bond 2017 Summer	Work Phase	1

I/We disclose below any familial relationship that exists between the Owner or any employee of Bidder and any member of the Board of Education, Board of Directors or the Superintendent of (Insert School District Name)

Familial Relationships:   None   Listed Below				
Bidder Employee/Position	Relationship		School District Associate/Position	
		-		
(Company Name)		Subsci	ribed and sworn to be	
By:(Authorized signer)		this	_day of	, year
(Print or type Name and Title of Signer)	(Signed	by No	tary Public)	
Address:		Му со	mmission expires:	
Telephone:				

#### **IRAN ECONOMIC SANTION ACT 517 OF 2012**

- 1. On December 28, 2012, Governor Snyder signed Public Act 517 of 2012, commonly known as the "Iran Economic Sanctions Act" (the "Act"). The Act provides that beginning April 1, 2013 an "Iran Linked Business" is not eligible to submit a bid on a request for proposal with a "public entity." Under the Act, a "public entity" includes school districts and intermediate school districts. The Act also requires that a person that submits a bid in response to a public entity's request for proposal must certify to the public entity that it is not an Iran Linked Business. This requirement applies to <u>all</u> requests for proposals issued by a public entity, and not just to construction projects.
- 2. The Act defines an Iran Linked Business as: 1) a person engaging in investment activities in the energy sector of Iran, including a person that provides oil or liquefied natural gas tankers or products used to construct or maintain pipelines used to transport oil or liquefied natural gas for the energy sector of Iran; or 2) a financial institution that extends credit to another person if that person will use the credit to engage in investment activities in the energy sector of Iran.
- 3. If the public entity determines, using credible information available to the public, that a person or entity has submitted a false certification, the public entity must provide written notice to the person or entity of its determination and of its intent not to enter into or renew the contract. The notice must include information on how to contest the determination. The notice must also specify that the individual or entity may become eligible for future contracts with the public entity if the activities that caused it to be an Iran Linked Business are ceased.
- 4. The Attorney General may bring a civil action against any individual or entity reported to have submitted a false certification. If the civil action results in a finding that certification was false, the person or entity will be responsible for a civil penalty of not more than \$250,000.00 or two times the amount of the contract for which the false certification was made, whichever is greater. In addition to the fine the individual or entity will be responsible for the cost and reasonable attorney fees incurred by the public entity. An individual or entity who has submitted a false certification was determined to be false.

### IRAN ECONOMIC SANCTION ACT 517 OF 2012

#### **BIDDER CERTIFICATION FORM**

- 1. Beginning April 1, 2013, an Iran linked business is not eligible to submit a bid on a request for proposal with a public entity.
- 2. Beginning April 1, 2013, a public entity shall require a person that submits a bid on a request for proposal with the public entity to certify that it is not an Iran linked business.
- 3. Pursuant to Michigan law, (the Iran Economic Sanctions Act, 2012 PA 517, MCL 129.311 et seq.), before accepting any bid or proposal, or entering into any contract for goods or services with an prospective Contractor, the contractor must first certify that it is not an "IRAN LINKED BUSINESS", as defined by law.
- 4. Each Contractor submitting a bid on this project shall include a letter with their bid certifying that they have full knowledge of the requirements and possible penalties under the law MCL 129.311 et seq. that the Contractor is NOT an "IRAN LINKED BUSINESS", as required by MCL 129.311 et seq., and as such that Contractor is legally eligible to submit a bid and be considered for a possible contract to supply goods and/or services to Saline Area Schools.

I certify that I am a duly authorized representative of	and
(Name of Company)	
confirm that neither I nor the company is an "Iran Linked Business"	

Company Representative Name \_\_\_\_\_

Signature \_\_\_\_\_

Date

### SECTION 005216 AGREEMENT FORM

### 1. <u>GENERAL</u>

- 1.1. The Contract form to be used for this project shall be Clark Construction Company Subcontract (Reference Trade Contractor Contract Sample (406FR002).
  - A. The Trade Contractor shall execute the Contract with no modifications to the terms of the Agreement.

# SECTION 006113 PERFORMANCE AND PAYMENT BOND

### 1. <u>GENERAL</u>

- 1.1. Whether the Performance and Labor Material Payment Bond is single or dual obligee, trade contractor must provide a "Performance Bond and Labor and Material Payment Bond", AIA Document A312, December 1984, Third Printing 1987, of the American Institute of Architects, which are hereby made a part of the Contract Documents, the same as if bound herein. The Performance and Payment Bond shall be part of the Contract and shall apply to all Trade Contractors Contract Documents.
- 1.2. All bonding companies must be listed by the U.S. Treasury.
- 1.3. All bonding companies must be rated "A" or better by AM Best.
- 1.4. All bonding companies must be licensed to do business in Michigan, or the State in which the Project is located.
- 1.5. Copies of the Performance Bond and Payment Bond may be obtained from the American Institute of Architects, national office, 1735 New York Avenue, NW, Washington, DC 20006.

# SECTION 006200 CERTIFICATES AND OTHER FORMS

# 1. <u>GENERAL</u>

- 1.1. Sample documents, included in this section, shall be used through the life of the project.
- 1.2. Sample forms:
  - A. Trade Contractor Qualification Form
  - B. Trade Contractor Contract Sample
  - C. Trade Contractor Change Order Sample
  - D. Certificate of Insurance Sample
  - E. MEP Cost Breakdown (required start-up submittal for MEP Contractors)
  - F. Sworn Statement
  - G. Partial Unconditional Waiver
  - H. Full and Final Unconditional Waiver
  - I. Materials Stored Payment Form
  - J. Labor Rate Calculations
  - K. Form of Guarantee
  - L. State of Michigan Prevailing Wage Rate Schedule
  - M. General Conditions

# CLARK CONSTRUCTION COMPANY TRADE CONTRACTOR QUALIFICATION FORM

PROJECT NAME:		DATE		
GENERAL				
Legal name of Business:				
Principal Address:Street				
P. O. Box City			State	Zip
Contact Person: E-mail Address: Telephone No.:		Duns No.:		
CorporationPartnership	Individual	Sole Proprietorsh	ip Joint Venture	e <u>Other</u>
If Incorporated, State of Incorporation:			Year Incorpo	orated
Average number of office employees:			_Field:	
Is your company qualified as (Enclose certificate) SMALL	MBE BUSINESS SDB SDVOSB	HUB ZO WOSB	ONE 8A	E SB
Does your company have a written EEC	D policy?		Yes	No
Does your company have a current Cert	tificate of Awardab	ility?	Yes	No
List company officers:				
Name	1	ìtle	Years with Orga	nization
State the type(s) of work in which you s	specialize and regul	arly perform with you	ur own personnel:	
Labor Relations: Uni	on	Op	en Shop	
<b>FINANCIAL</b>				
Please submit current interim financial and contract schedule. This information			ement, including work	in progress
Bank Reference:(Bank Name)			(Telephone)	
			(reteptione)	
Contact Person:(Contact Name)				
Amount of Line of Credit:	Sec	ured? YesNo	_ Amount available	

P:\2714 - Saline Area Schools 2015 Bond Program\5 Bidding.Procurement\4 Bid Pack # - Description\2 Front End\BP #1\Forms & Certificates\A - Trade Contractor Qualification Form.doc -Version 7 Page 1 of 3

#### CLARK CONSTRUCTION COMPANY TRADE CONTRACTOR QUALIFICATION FORM

Is your company currently in default on any loan agreement or financial agreement with any bank, financial institution or other entity? (If yes, attach details, circumstances and prospects for resolution) YES No **BONDING INFORMATION** Furnish a signed statement from the surety certifying the following: If bonded - statement not required Current Bonding Capacity of Company: Aggregate\_\_\_\_\_\_Single Project\_\_\_\_\_ Amount of Work Currently Bonded\_\_\_\_\_Bond Rate \_\_\_\_\_ Name of Bonding Company\_\_\_\_\_ Co. Rating \_\_\_\_\_ Length of Time with Bonding Company Have Performance or Payment Bond claims ever been made to a surety? Yes No Yes\_\_\_No\_\_\_\_ Has any surety company refused to bond the firm or any affiliate companies on any project? Current capacity exists to cover the amount of this contract? Yes No

#### **INSURANCE**

Furnish a Certificate from your insurance carrier verifying all limits & project requirements to main office.

Experience Modification Rating (EMR) for the last three years:

# SAFETY

Please attach copies of OSHA No. 300A Logs for the last three years along with your most current log to-date.

Does your company have a written Safety Program? If yes, provide an electronic copy or verify that a current copy is on file at Clark's Main Office.	Yes	_No
Does your company have a Substance Abuse Program?	Yes	_No
Does your company have a safety officer? If YES, provide name:	Yes	_No
Does your company hold weekly craft "tool box" talk safety meetings?	Yes	_No
Does your company provide foreman safety training? If YES, how often?	Yes	_No
Does your company conduct safety inspections? If YES, how often?	Yes	_No
Does your company give orientation/safety instruction to new hires?	Yes	_No
In the past three years, has your company been cited by state or federal OSHA for any willful violations?	Yes	_No

P:\2714 - Saline Area Schools 2015 Bond Program\5 Bidding.Procurement\4 Bid Pack # - Description\2 Front End\BP #1\Forms & Certificates\A - Trade Contractor Qualification Form.doc -Version 7
#### CLARK CONSTRUCTION COMPANY TRADE CONTRACTOR QUALIFICATION FORM

Please attach list on a separate sheet, the summary details for all violations including date, type, description and amount for the last three years

 During the past three years, has your company experienced any employee fatalities?
 Yes\_\_\_No\_\_\_

 If YES, please attach details of the accident on a separate sheet
 Yes\_\_\_No\_\_\_

#### **EXPERIENCE**

Attach list of recent major projects completed and work in progress, including average manpower required and anticipated duration of contract. Please include any projects completed with Clark Construction Company.

#### LIST THREE (3) SUPPLIER REFERENCES

Supplier	Project	Contact	Phone	Amount
Have you been	deemed to be in default of	on any contract?		Yes No

Have you failed to complete any work awarded to you?	Yes	_No
Have you ever been adjudged bankrupt or filed a petition in bankruptcy?	Yes	_No
Have you filed any lawsuits or requested arbitration?	Yes	_No

(If you answered YES to any of the above, please attach a brief explanation)

Please attach the following documents:

• • • •	MBE/WBE/DBE Certificate (if applicable) Current interim Financial Statement Last Certified Financial Statement Statement of Surety or Bid Bond Insurance carriers EMR verification MIOSHA 300A Logs for the last 3 years and the most surrant log to dote	• • •	Electronic Safety Manual OSHA violations for the last 3 years Completed Project List Work in Progress List Trade Contractor Qualification Form Schedule of Values – to be submitted at Bro. Award Macting
	most current log to-date		Pre-Award Meeting

#### As an Authorized Representative for

I hereby certify that the answers to the foregoing questions, and all documents contained herein, are true and correct. I understand that submission of this information is in no way a guarantee of contract award by the Owner or Clark Construction Company and that the information is provided for review and evaluation purposes only.

(Signature)

(Date)

(Above Name typed or Printed)

(Title of Authorized Representative)

### TRADE CONTRACT



VENDOR No. DO NOT USE

CONTRACT No.

9999-001

THIS NUMBER MUST APPEAR ON ALL INVOICES

**TO:** ABC Excavating - Sample Company

1234 North Ave. Smallville, MI 99999 Attn:Joe Smith Phone:1-234-567-8910 Fax:1-234-567-8910

DATE:9/20/13

#### JOB: PROLOG 2008 SAMPLE PROJECT

Perform all work per Specifications dated 1 and Plans referenced therein including Addendum 1, 2, 3, 4, Post Bid Addendum 1, 2, and Exhibit "A" - Trade Contractor Pre-Award Meeting form dated 9/20/13.

Scope of Work:	This is the area whre you would add any additional exhibits (ie: Exhibit B, C, etc.) and any		
	other contract specific information. This field is unlimited and populates in the body of the		
	contract.		

Budget Codes	Description	Scheduled Cost
000101-01-000010	This field will default to the budget code line description	\$200,000.00
000101-01-000010	Site Utilities (this ist he default for the budget code)	\$200,000.00
000101-02-000010	Construction manager's FEe	\$10,000.00
000102-03-000010	CM Fe4e For Building A-e	\$15,000.00

#### TOTAL AMOUNT OF THIS CONTRACT:

Sample

**NOTICE:** Substitute forms must not be used. Signed copy certifies your acceptance of the terms and conditions appearing above and in the attached Trade Contract Conditions and/or exhibits to this Trade Contract.

Clark Construction Company	ABC Excavating - Sample Company
CONTRACTOR	TRADE CONTRACTOR/SUPPLIER
By Racheal Cline	Ву
Signature	Signature
Date	Date

\$425,000.00

The Trade Contractor agrees:

- a. That it is bound to Clark Construction Company (Contractor herein) by the terms of all Contract Documents of Clark Construction Company's Contract with the Owner (hereinafter referred to as Prime Contract) (including, but not limited to drawings, specifications, general conditions, special and supplemental conditions and addenda), and to assume toward Clark Construction Company all obligations and responsibilities that Clark Construction Company by said documents, assumes toward the Owner.
  - b. The Trade Contractor binds itself to Clark Construction Company for the performance of Trade Contractor's Work in the same manner as Clark Construction Company is bound to the Owner for such performance under the Prime Contract. The Prime Contract or the pertinent parts of such Contract will be made available upon Trade Contractor's request.
  - c. The Contract Documents consist of the Trade Contract, these Trade Contract Conditions, the attached Trade Contract Rider (if any), the Prime Contract, the drawings, specifications, general conditions, special and supplemental conditions, change orders and all addenda.
  - d. The Work is the completed construction required of the Trade Contractor by the Contract Documents and includes all labor, material, and equipment necessary to perform the Work
  - e. The Contract price includes, and Trade Contractor shall obtain, all permits and licenses, and pay all permit and license fees, sales tax, use taxes, freight and unloading of material at jobsite necessary for the completion of the Work. Trade Contractor shall pay all taxes applicable to this Contract.
- 2) Trade Contractor shall timely furnish all items of the Work described in this Contract, including, but not limited to all labor, materials, equipment, services, layout, taxes, and any other expressly or impliedly required or necessary by inference to complete all portions of the Work in accordance with the Contract Documents.
- 3) a. Should any question arise with respect to the interpretation of the drawings and specifications, such questions shall be submitted to the Architect/Engineer as required and in like manner as the Prime Contract and such decision shall be final and binding upon the Trade Contractor. If there is no Architect/Engineer for this project, Clark Construction Company's decision shall be followed by the Trade Contractor.
  - b. This Contract and all Contract Documents are intended to supplement and complement each other and shall, where possible, be thus interpreted. If, however, any provision of this Contract irreconcilably conflicts with a provision of any other Contract Document, the provision imposing the greater duty on the Trade Contractor shall govern.
- 4) Trade Contractor has carefully examined the site and all Contract Documents and, from its own investigation, has satisfied itself as to the nature and location of the Work, the general and local conditions, and all matters which may, in any way, affect its Work or its performance and that as a result of its examination and investigation, Trade Contractor understands the requirements of its Work.
- 5) Trade Contractor shall submit Shop Drawings, Product Data, and Samples for all items of its Work and obtain approval of same prior to installation in accordance with procedure required by the Contract Documents.
- 6) Trade Contractor agrees within ten (10) days after written demand, to cause the effect of any lawsuit, lien and/or claim on bond, filed by a party employed and/or contracted by Trade Contractor, to be removed. In the event Trade Contractor shall fail to do so, Clark Construction Company is authorized to use whatever means in its discretion it may deem appropriate to cause said lawsuit, lien and/or bond claim to be removed or dismissed and Trade Contractor shall be responsible for the costs including all attorney fees incurred by Clark Construction Company. Trade Contractor may litigate any lawsuit, lien and/or bond claim provided Trade Contractor causes the effect thereof to be removed, promptly in advance, and shall do such things as are necessary to cause the Owner not to withhold any monies due to Clark Construction Company from Owner by reason of such lawsuit, lien and/or bond claim.
- 7) a. No extra or additional Work performed or material provided will be performed or provided by the Trade Contractor unless he obtains prior written approval and in the event Trade Contractor performs such extra or additional Work or provides material without prior written approval, then Trade Contractor thereby expressly waives any and all claims for additional payment for the extra or additional Work and/or the material provided.
  - b. Notwithstanding any other provision, if the Work for which the Trade Contractor claims extra compensation is determined by the Owner or the Architect/Engineer not to entitle Clark Construction Company to a change order or extra compensation, then Clark Construction Company shall not be liable to the Trade Contractor for any extra compensation for such Work, unless Clark Construction Company agreed, in writing prior to execution of the Work, to such extra compensation.
  - c. Trade Contractor shall make all claims for extras, additional compensation and for extensions of time to Clark Construction Company in accordance with the procedures set forth in the Prime Contract.

- d. If the Trade Contractor shall be materially delayed in the progress of the Work for reasons beyond the Trade Contractor's control, and under such circumstances as entitle Clark Construction Company to an extension of time under the Prime Contract, then Trade Contractor shall be entitled to a corresponding extension of time for completion of the Work. The extension of time herein provided shall be Trade Contractor's sole and exclusive remedy for any delay, and Trade Contractor hereby waives any claims for damages against Clark Construction Company or Owner by reason of delay.
- e. Trade Contractor agrees that Clark Construction Company shall not be liable to Trade Contractor for delay, acceleration, interference of any kind or changed conditions to Trade Contractor's Work by the act, neglect or fault of the Owner, Owner's agent, another Trade Contractor, or the Architect/Engineer, or by reason of fire or other casualty, or on account of any act of God, or any other cause beyond Clark Construction Company's control, or on account of any circumstances caused or contributed to by the Trade Contractor; provided, however, notwithstanding anything else contained herein, Clark Construction Company will be liable to the Trade Contractor for damages it incurs as a result of any acts or failures to act, by the Owner which delays, accelerates, interferes, interferes or changes the Trade Contractor's Work only to the extent the Owner is liable for such damages and actually pays Clark Construction Company for such damages; it being expressly understood that the only obligation Clark Construction Company has to the Trade Contractor under this provision is to pass on to the Owner any claim the Trade Contractor has for damages for delays caused by the Owner and to pay to the Trade Contractor any amounts which the Owner pays to Clark Construction Company as a result of such claim for Owner caused delays.
- 8) Trade Contractor shall be responsible for all clean-up in connection with this Contract. Trade Contractor shall be charged for clean-up services performed by Clark Construction Company for the Trade Contractor's operations.
- 9) As a condition precedent to payment, Trade Contractor shall furnish all waivers, releases and sworn statements required by any applicable construction lien law, bond law, or other applicable statute and such other evidence as may be required by Clark Construction Company, Owner, or Owner's lender to substantiate payment.
- 10) Payments hereunder will be on monthly installments commencing within 30 days after commencement of Work, equal to 90% of the value of completed Work, as estimated by Trade Contractor, approved by Clark Construction Company, and paid by Owner; so long as the Trade Contractor is not in default hereunder, installment payments shall be made within 10 days after Clark Construction Company has been paid by the Owner on account of the Trade Contractor's Work and to the extent therein; final payment within 30 days after completion and acceptance by the Architect and/or Engineer and approval and payment by the Owner. Acceptance of final payment by Trade Contractor shall constitute a full and final waiver of all of Trade Contractor's claims of whatever nature against Clark Construction Company and Owner on this project. No payments will be required unless Trade Contractor provides a sworn statement, waiver of lien and/or other evidence required by Clark Construction Company's billing instructions, any applicable construction lien or similar law and all documents necessary to assure Trade Contractor is meeting its Contractual obligations to the Trade Contractors and suppliers. All payments are to be made to the Trade Contractor only out of such equivalent payments actually received by Clark Construction Company for Work done by the Trade Contractor and Trade Contractor agrees actual payment by the Owner and receipt by Clark Construction Company shall be a condition precedent to any payment obligation of Clark Construction Company to the Trade Contractor. This paragraph shall be applicable even in the event Clark Construction Company posts a labor and material payment bond and performance bond. The Trade Contractor agrees that Clark Construction Company shall be under no obligation to pay the Trade Contractor for any Work performed on this construction project, unless Clark Construction Company has been paid therefore by the Owner. The provisions hereof, stating when progress and final payments are due are subject to the condition that Clark Construction Company receive payment from the Owner on account of Work performed by the Trade Contractor on this construction project. The Trade Contractor expressly contemplates that payments to it are contingent upon Clark Construction Company's receiving payment from the Owner, the Trade Contractor expressly agreeing to accept the risk that it will not be paid for Work performed on the credit and ability of the Owner, and not of Clark Construction Company, and thus the Trade Contractor agrees that payment by the Owner shall be a condition precedent to any payment obligation of Clark Construction Company to the Trade Contractor. The Trade Contractor agrees that the liability of the surety on Clark Construction Company's payment bond, if any, for payment to the Trade Contractor, is subject to the same conditions precedent as are applicable to Clark Construction Company's liability to the Trade Contractor.
- 11) To submit to Clark Construction Company applications for payment in such reasonable time as to enable Clark Construction Company to apply for payment. Trade Contractor shall make application for payment and invoices for monthly drawings in accordance with the billing instruction delivered with this Contract.
- 12) a. Clark Construction Company may deduct from any amount due or to become due to the Trade Contractor any sum or sums owing by the Trade Contractor to Clark Construction Company.
  - b. In the event of any breach by the Trade Contractor of any obligation of this Contract, or in the event Clark Construction Company deems it necessary to protect against claims by other Trade Contractors or any claim against the Owner, Clark Construction Company, the surety for Clark Construction Company or the premises upon which the Work is being performed, which claim arises out of the Trade Contractor's performance of this Contract, Clark Construction Company has the right, but is not obligated, to retain out of any amount due or to become

due to the Trade Contractor, an amount sufficient to completely protect Clark Construction Company from any and all loss, damage or expense until the claim has been resolved to the satisfaction of Clark Construction Company.

- c. Clark Construction Company shall have the right, but is not required, to issue joint checks comprising funds due or to become due to the Trade Contractor and any of its lower-tiered Contractors or material suppliers for which Trade Contractor's sworn statement or any sworn statement identifies a sub-tier party as having sums due and owing.
- d. This paragraph (12a-d) shall be applicable even though the Trade Contractor has posted a full payment and performance bond.
- 13) Trade Contractor shall be required to complete the Work and its several parts on or before the date or dates established by Clark Construction Company's progress schedule. Should the Trade Contractor, in the opinion of Clark Construction Company, at any time during the performance of the Contract, neglect or refuse to supply a sufficiency of properly skilled workman or of materials of proper quality, fail in any respect to prosecute the Work with promptness and diligence, fail to perform any terms of the conditions of the Contract, then Clark Construction Company may, after giving reasonable notice to the Trade Contractor, provide any necessary labor, materials or equipment required to assist Clark Construction Company in the completion of the Work and deduct the cost thereof from any monies due or thereafter due the Trade Contractor under this Contract, or any other Contract with Clark Construction Company. In the event the Trade Contractor shall fail to timely perform all or a portion of the Work required by the Contract, Clark Construction Company, upon three (3) working days' notice to the Trade Contractor, shall have the right to take over the Work required hereunder or any identifiable portion thereof with all materials, tools and appliances of the Trade Contractor. Clark Construction Company shall have the right to withhold all payments to the Trade Contractor until such costs have been paid to Clark Construction Company.
- 14) a. Trade Contractor agrees to employ sufficient manpower and methods to complete the Work in a timely fashion and will only employ persons who will Work at all times in harmony with other persons employed on the project and Trade Contractor agrees that labor disputes of any kind involving any other Contractors, Trade Contractors or their employees will not be permitted to occur on the project and such labor disputes as a result of those performing the Work hereunder shall not be cause for delay of the progress of Trade Contractor's Work and Trade Contractor will not be granted an extension of time in the event of such delay.
  - b. Trade Contractor agrees to perform the Work and all parts thereof at such time and in such order and sequence as Clark Construction Company may direct and as required for the best possible progress of the Work and to conduct its Work and perform all activities incidental thereto so as to facilitate and so as not to interfere with or delay the Work or the best possible progress of the Work in several or all parts simultaneously, if directed to do so by Clark Construction Company.
  - c. In the event the project's progress schedule shall fall behind or be delayed due to any cause of Trade Contractor, whether or not due to its own fault or neglect, the Trade Contractor may be required at the direction of Clark Construction Company, to Work such overtime or provide additional resources, for no additional compensation, as may be necessary to avoid delay and completion of the Work under this Contract.
  - d. Clark Construction Company reserves the right to modify any approved progress schedule with respect to the required sequence or duration of the Work or any portion thereof and Clark Construction Company makes no representations that Trade Contractor will be able to commence, prosecute or complete the Work in accordance with any progress schedule.
- 15) Trade Contractor shall defend, protect, indemnify and save harmless, the Owner, Clark Construction Company, their agents and assigns against any and all suits for claimed infringements of letters patent claimed to cover Trade Contractor's products, or any part thereof.
- 16) Trade Contractor shall, to the fullest extent permitted by law, secure, defend, protect, hold harmless and indemnify the Owner, Clark Construction Company, including any party Clark Construction Company is required to indemnify and the Architect and any of their respective agents, servants and employees against any liability, loss, claims, demands, suits, costs, fines, attorney fees, and any expenses whatsoever arising from bodily injury, sickness, disease, (including death resulting therefrom), of any persons, or the damage or destruction of any property or Work required by this Contract, including loss of use, arising out of or in connection with the performance of any Work relating to this Contract including extra Work assigned to the Trade Contractor, based upon any act or omission, negligent or otherwise, (including active or passive negligence) of (a) the Trade Contractor or any of its agents, employees or servants, (b) any Tiered Trade Contractor, supplier, or material man of the Trade Contractor, or any agents, employees or servants thereof, (c) any other person or persons. The Trade Contractor's obligations of indemnification contained herein shall exclude only those matters in which the claim is caused by the sole negligence of the Owner, the Architect, Clark Construction Company, or any of their respective agents, servants and employees. The obligations herein shall apply regardless of whether such suits, damages, judgments, liabilities, interest, attorney fees, costs, etc., are caused in part by a party indemnified hereunder or by anyone acting under their direction, control, or on their behalf. Also the obligation herein shall not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for the Trade Contractor under Workers Compensation Act, Disability Benefits Act or other employee benefit act. The obligation of indemnification shall survive the completion of the Contract. In the event an indemnified party (Clark Construction Company, the Owner, the Architect) takes

any action, to enforce this indemnification provision the indemnified party shall be entitled to payment of actual attorney fees and costs incurred, from the Trade Contractor.

- 17) Trade Contractor shall comply with all Federal, State and Local laws, codes, executive orders, and regulations and all local ordinances and regulations effective where the Work under this Contract is to be performed, and to pay all costs and expenses connected with such compliance, to pay for itself and its trade contractors, all wages, fees, and taxes of whatever nature required by the foregoing, and also to pay for itself and its trade contractors all taxes imposed by any State or Federal law for any employment insurance, pension, unemployment compensation, old age retirement/disability funds or any similar purpose and to furnish all necessary reports and information to the appropriate Federal, State and Local agencies, with respect to all of the foregoing the same as if Trade Contractor was in fact the Contractor, and to hold Clark Construction Company and the Owner harmless from any and all losses or damage occasioned by the failure of the Trade Contractor to comply with the terms of this paragraph.
- 18) a. In the event Clark Construction Company's Contract with the Owner contains a termination for convenience clause which is exercised by the Owner and/or Clark Construction Company, the rights, duties, and obligations of the parties herein shall be governed by the terms of the provisions of the convenience clause exercised by the Owner and/or Clark Construction Company.
  - b. Clark Construction Company shall, at its sole option and discretion, have the right to terminate the Agreement for any reason whatsoever by providing the Trade Contractor with a written Notice of Termination to be effective immediately upon receipt by Trade Contractor of the written Notice of Termination. A termination by default of the Agreement shall, if wrongfully made, be treated as a termination for convenience under this clause. Whenever the Trade Contractor shall only be entitled to the actual direct costs of all labor and material expended on the job prior to the effective date of termination plus eight (8) percent of the actual direct costs or Trade Contractor shall be entitled to be paid a pro rata percentage of the total Agreement price which is equal to its percent of completion, whichever of the two methods provides the lowest sum to be paid to the Trade Contractor. In no event shall the Trade Contractor be entitled to assert a claim in quantum or any other measure of damage other than that stated herein.
- 19) Not to discriminate against any employee or applicant for employment, to be employed in the performance of this Contract, with respect to his hire, tenure, terms, conditions or privileges of employment, or any matter directly or indirectly related to employment because of sex, race, color, religion, national origin or ancestry. Any nondiscrimination agreement required by the Prime Contract shall be executed by the Trade Contractor and all of its tiered Contractors.
- 20) a. To provide general liability insurance, including comprehensive, products and completed operations, hazards, contractual and independent Contractors, as well as automobile liability, Worker's Compensation and employee liability coverage on all its employees and his Trade Contractors' employees engaged in executing this Contract and to provide all other insurance coverage required by the Contract Documents to the Prime Contract.
  - b. Trade Contractor shall carry, and require its Trade Contractors to carry, identical insurance to that which Clark Construction Company is required to carry under the Prime Contract. Trade Contractor's insurance shall include contractual liability coverage applicable to its obligations pursuant to this Contract.
  - c. Trade Contractor shall cause its insurance carrier to add Clark Construction Company as an additional insured on all such policies excluding Worker's Compensation and automobile insurance prior to commencement of Work. Trade Contractor's Insurance carriers shall be directed to notify Clark Construction Company 15 days in advance of cancelation of any insurance coverage required herein. The additional insured endorsement included on the Trade Contractor's insurance policies shall state that coverage is afforded the additional insured with respect to claims arising out of operations performed by or on behalf of the Trade Contractor. If the additional insureds have other insurance which is applicable to the project, such other insurance shall be on an excess or contingent basis. The amount of the insurer's liability under this insurance policy shall not be reduced by the existence of such other insurance.
  - d. The Trade Contractor agrees that the insurance coverage required by contract shall be continued throughout the duration of the warranty period specified by the contract. For Contracts that require Pollution Insurance coverage, such coverage shall be continued for three years beyond the substantial completion date of the project, or as specified by the Prime Contract, whichever duration is greater. Upon request by Clark Construction and/or Owner, Trade Contractor shall provide evidence/proof of insurance that the insurance coverage is being continued throughout the period specified by contract.
- 21) a. Clark Construction Company and Trade Contractor waive all rights against each other for damages caused by fire or other perils to the extent covered by property insurance provided under the Prime Contract, except such rights as they may have to the proceeds of such insurance held by the Owner as trustee. Trade Contractor shall require similar waivers by its Trade Contractors.

- b. Clark Construction Company and Trade Contractor waive all rights against each other and the Architect for loss or damage to any equipment used in connection with the project which loss is covered by any property insurance. The Trade Contractor shall require similar waivers from all of its Trade Contractors. This provision shall not operate to relieve Clark Construction Company or Trade Contractor of responsibility for any loss or damage to their own or rented property of their employees which is not to be included in and remain a part of the permanent construction.
- 22) Trade Contractor agrees to provide Clark Construction Company with all warranties required by Clark Construction Company in the Prime Contract which relate to Trade Contractor's Work. Such warranties shall commence and continue in effect for the full period of Clark Construction Company's obligation to the Owner as provided in the Prime Contract.
- 23) Trade Contractor shall, at its own cost and expense, protect its employees and other persons from the risk of death, injury or bodily harm arising out of or in any way connected with the Trade Contractor's Work and Trade Contractor shall protect all persons on or near the premises from all unreasonable risk of injury that arises during or as a result of the Work. Trade Contractor shall provide a safety coordinator for the Work. In addition to the above, the Trade Contractor shall strictly comply with Clark Construction Company's safety program for the project with all safety policies and procedures of Clark Construction Company.
- 24) This Contract is subject to modifications by Clark Construction Company in the event of fire, accidents, strikes, government acts, or other conditions beyond its control.
- 25) Time is of the essence for both parties, and they mutually agree to see to the performance of their Work and the Work of their Trade Contractors in accordance with applicable standards of care so that the entire project may be completed in accordance with the Contract Documents.
- 26) a. No waiver by Clark Construction Company of any breach of Trade Contractor or of a provision of this Contract shall be deemed a waiver of any other provision hereof or of any subsequent breach by Trade Contractor of such provision.
  - b. If any provision of this Contract is invalid or unenforceable to any extent, the remainder of this Contract shall be valid and enforceable to the fullest extent permitted by law.
- 27) That this Contract is the entire agreement of the parties and there is no verbal understanding or agreement between Trade Contractor and Clark Construction Company other than the conditions stated in this Contract.
- 28) The Trade Contractor agrees that Clark Construction Company shall have the sole discretion to elect to demand arbitration with the Trade Contractor for any dispute or claim arising out of or relating to this Contract or breach thereof or performance thereof, and if arbitration is elected by Clark Construction Company, then the arbitration shall be decided in accordance with the Construction Industry Rules of the American Arbitration Association. Arbitration proceedings shall be held in Southfield, Michigan, at the offices of the American Arbitration Association or such other place as designated by Clark Construction Company. In the event that Clark Construction Company shall elect not to demand arbitration, then the matter shall proceed in a court of competent jurisdiction for the dispute.
- 29) Mediation. Except as stated in this Agreement or otherwise agreed in writing by the parties, claims arising out of or related to the Contract shall be subject to mediation as a condition precedent to the parties' agreed-upon dispute resolution process. The Project Owner or Construction Manager reserves the right in its discretion to require consolidation or joinder of any mediation or arbitration arising out of or relating to this Agreement with another mediation or arbitration involving a person or entity not a party to this Agreement but in any event the Owner or Construction Manager believes such consolidation or joinder is necessary in order to resolve a dispute or avoid duplication of time, expense or effort. Any mediation or arbitration hearing shall be held in the general location where the Project is located, unless another location is mutually agreed upon.
- 30) That Clark Construction Company may cancel this Contract if not executed as specified and no alteration of any provision hereunder and no substituted form will be accepted by Clark Construction Company.
- 31) Clark Construction Company may deduct or set-off against any amount due Trade Contractor or any amount to become due Trade Contractor, any claim that Clark Construction Company may have against Trade Contractor arising out of this Contract, or any other Contract/transaction between Trade Contractor and Clark Construction Company.
- 32) Clark Construction Company and Trade Contractor agree that signatures on documents required by this contract, including but not limited to Change orders, Sworn Statements etc., may be delivered by electronic mail in lieu of an original signature. The signature on a document delivered by electronic mail is agreed to be an original signature and shall have the same force and effect as an original signature. The Trade Contractor agrees that Clark Construction Company may rely on signatures affixed to a document delivered by electronic mail and is immediately binding upon the Trade Contractor upon receipt of the document by Clark Construction Company by electronic mail.

- 33) Asbestos-free product installation. It is hereby understood and agreed that no product and/or material containing asbestos including chrysolite, amosite, crocidolite, tremolite asbestos, anthophyllite asbestos, actinolite asbestos and any combination of these materials that have been chemically treated and/or altered shall be installed or introduced into the Work by the Trade Contractor or his employees, agents, subcontractors, or other individuals or entities over whom the Trade Contractor has control. The Trade Contractor shall be required to provide a signed certification statement ensuring that all products or materials installed or introduced into the Work all be asbestos-free. The Trade Contractor shall also be required to furnish certified statements from the manufacturers of supplied materials used during construction verifying their products to be asbestos-free. The Trade Contractor shall complete and submit to the Construction Manager and Owner a certification evidencing asbestos-free product installation prior to issuance of the final Certificate for Payment, in a form acceptable to the Owner.
- 34) Third Party Beneficiary. Trade Contractor agrees the Owner of the Project is a third party beneficiary of the Trade Contractor contract with the Construction Manager.

PLEASE SIGN AND RETURN THESE CONDITIONS

NAME: \_\_\_\_\_\_

## Change Order

Agency Change Order 2 Block

Date: To Subcontractor/Supplier:

Contract/PO Date:

FURNISH LABOR, MATERIAL, TAXES, SUPERVISION, MAN POWER AND ALL OTHER SERVICES REQUIRED PER YOUR CONTRACT TO COMPLETE THE FOLLOWING ADDITIONAL WORK:

PCO Item Budget Code Description

Add/Deduct

Amount

Trade Contractor agrees that this change order shall fully compensate Trade Contractor for all costs directly or indirectly related to this work and for any direct or indirect effect upon other Work or Schedule under the Trade Contract Documents . Trade Contractor agrees that any claim for direct costs, indirect costs, and cumulative impact costs are included in the change order amount. No claim for impact costs resulting from the performance of this Change Order will be permited except to the extent included in the change order payment above, and any such cost impact shall be considered waived and released. No extension of time for the performance of this Change Order work or any other work under the Contract Documents shall be allowed as a result of this Change Order, except as otherwise specifically provide within this Change Order. All other terms and conditions of the Agreement between Contractor and Trade Contractor remain unchanged .

The original Contract/PO Value was
Sum of changes by prior Change Orders
The Contract/PO Value prior to this Change Order was
The Contract/PO Value will be changed by this Change Order in the amount of
The new Contract/PO Value including this Change Order will be
The Contract/PO duration will be changed by
The revised Substantial Completion date as of this Change Order is

#### SUBCONTRACTOR/SUPPLIER

Address	Address
BY	BY
SIGNATURE	SIGNATURE
DATE	DATE



## **CERTIFICATE OF LIABILITY INSURANCE**

DATE (MM/DD/YYYY) 02/01/2012

THIS CERTIFICATE IS ISSUED AS A MATTER OF INFORMATION ONLY AND CONFERS NO RIGHTS UPON THE CERTIFICATE HOLDER. THIS CERTIFICATE DOES NOT AFFIRMATIVELY OR NEGATIVELY AMEND, EXTEND OR ALTER THE COVERAGE AFFORDED BY THE POLICIES BELOW. THIS CERTIFICATE OF INSURANCE DOES NOT CONSTITUTE A CONTRACT BETWEEN THE ISSUING INSURER(S), AUTHORIZED REPRESENTATIVE OR PRODUCER, AND THE CERTIFICATE HOLDER.					
IMPORTANT: If the certificate holder is an ADDITIONAL INSURED, the policy(les) must be endorsed. If SUBROGATION IS WAIVED, subject to the terms and conditions of the policy, certain policies may require an endorsement. A statement on this certificate does not confer rights to the certificate holder in lieu of such endorsement(s).					
PRODUCER	<u> </u>	CONTACT NAME:			_
Agent Name and Address		PHONE (A/C. No. Ext):		FAX (A/C, No):	
<b>•</b>		E-MAIL ADDRESS:			
			URER(S) AFFOR	DING COVERAGE	NAIC #
INSURED		INSURER A: A.M. BEST Rated "A" or Better INSURER B: A.M. BEST Rated "A" or Better			
Subcontractor Name and Address		INSURER C: A.M. BEST Rated A of Better			
		INSURER C: A.M. BEST Rated A of Better			
			EST Rated "A		
	·		EST Rated "A		
COVERAGES CERTIFICA	TE NUMBER:			REVISION NUMBER:	·
THIS IS TO CERTIFY THAT THE POLICIES OF INS	SURANCE LISTED BELOW HAY	VE BEEN ISSUED TO	THE INSUR	ED NAMED ABOVE FOR THE PO	LICY PERIOD
INDICATED. NOTWITHSTANDING ANY REQUIREN CERTIFICATE MAY BE ISSUED OR MAY PERTAIL EXCLUSIONS AND CONDITIONS OF SUCH POLICIE	MENT, TERM OR CONDITION N, THE INSURANCE AFFORD	OF ANY CONTRACT ED BY THE POLICIE	or other	DOCUMENT WITH RESPECT TO D HEREIN IS SUBJECT TO ALL	WHICH THIS
INSR TYPE OF INSURANCE ADDLISU		POLICY EFF (MM/DD/YYYY)	POLICY EXP (M.M/DD/YYYY)	LIMITS	
GENERAL LIABILITY			Townships of a set	EACH OCCURRENCE \$ 1,00	0.000
	_			DAMAGE TO RENTED	0.000
CLAIMS-MADE X OCCUR					0,000
	Policy Number	01/01/2012	01/01/2013		0,000
DEFECT PROD/COMP OPS	•			GENERAL AGGREGATE \$ 2,00	
GEN'L AGGREGATE LIMIT APPLIES PER:					0.000
				\$	*****
	-			COMBINED SINGLE LIMIT (Ea accident) \$ 1.00	0.000
				BODILY INJURY (Per person) \$	
ALL OWNED SCHEDULED			0410410010	BODILY INJURY (Per accident) \$	
AUTOS AUTOS X HIRED AUTOS X AUTOS	Policy Number	01/01/2012	01/01/2013	PROPERTY DAMAGE \$	
				(r dr dicuddini) \$	
				EACH OCCURRENCE \$ 1,00	00,000
EXCESS LIAB CLAIMS-MADE	Policy Number	01/01/2012	01/01/2013		00,000
DED RETENTION \$				\$	
WORKERS COMPENSATION AND EMPLOYERS' LIABILITY				X WC STATU- OTH- TORY LIMITS ER	
ANY PROPRIETOR/PARTNER/EXECUTIVE	Policy Number	01/01/2012	01/01/2013		0.000
OFFICE/MEMBER EXCLUDED?		0.0012012	5		0,000
If yos, describe under DESCRIPTION OF OPERATIONS below				E.L. DISEASE - POLICY LIMIT \$ 50	00,000
Errors and Omissions	-			Limit: \$1,000,000	
	Policy Number	01/01/2012	01/01/2013		
Contractors Pollution Liability				Limit: \$1,000,000	
DESCRIPTION OF OPERATIONS / LOCATIONS / VEHICLES (Atta	ch ACORD 101, Additional Remarks	Schedulo, if more space in	(boniupon d		
Clark Construction Company is an Additional Insu	ured for project (name) and pr	oject (number)			
	· · · · · · · · · · · · · · · · · · ·				
	രിവിപ്പ				
	22411112	112			
	Un HUU	السياليك			
CERTIFICATE HOLDER CANCELLATION					
CERTIFICATE HOLDER CANCELLATION					
Clark Construction Company SHOULD ANY OF THE ABOVE DESCRIBED POLICIES BE CANCELLED BEFORE					
THE EXPIRATION DATE THEREOF, NOTICE WILL BE DELIVERED IN					
3535 Moores River Drive		AUGURDANCE WI		TTAUNGIUNG.	
Lansing, MI 48911		AUTHORIZED REPRESE	NTATIVE	·	
		© 19	88-2010 AC	ORD CORPORATION. All rig	hts reserved.

The ACORD name and logo are registered marks of ACORD



Project:

#### **MEP Cost Breakdown**

PLUMBING	Value
1 Underground Sanitary Piping	\$ -
2 Above Ground Waste & Vent Piping	\$ -
3 Storm Drainage Piping	\$ -
4 Water Supply Piping	\$ -
5 Pipe Insulation	\$ -
6 Plumbing Fixtures	\$ -
7 Gas Piping	\$ -
8 Other / Special Systems:	\$ -
HVAC	\$ -
1 Hydronic Piping & Pumps	\$ -
2 Steam Piping & Pumps	\$ -
3 Refrigerant Piping	\$ -
4 Pipe Insulation	\$ -
5 Ductwork	\$ -
6 Duct Insulation	\$ -
7 VAV Boxes	\$ -
8 Grilles, Registers, Diffusers	\$ -
9 Heating Boilers	\$ -
10 Furnaces	\$ -
11 Packaged Water Chillers	\$ -
12 Cooling Towers	\$ -
13 Packaged Compressors and Condenser Units	\$-
14 Packaged Air Handler Units	\$-
15 Exhaust Fans	\$ -
16 Humidity Control Equipment	\$-
17 Temperature Controls & Energy Management	\$ -
18 Commissioning 19 Demolition & Excavation for Your Work	\$ - \$ -
20 Other / Special Systems:	ş - Ş -
	- ب
ELECTRICAL	\$ -
1 Incoming Primary	\$ -
2 Service & Distribution (Transformers, Switchgear, Panel Boards & Met	
3 Interior Lighting	\$ -
4 Exterior Lighting	\$ -
5 Outlets, Receptacles & Switches	\$ -
6 Equipment Connections	\$ -
7 Conduits, Pull Boxes, Junction Boxes, Bus Ways	\$ -
8 Wiring & Cabling	\$ -
9 Fire Alarm & Detection	\$ -
10 Lightning Protection	\$ -
11 Intercom & Sound	\$ -
12 Security & CCTV	\$-
13 Emergency Generator	\$ -
14 Low Voltage Cabling	\$ - \$ -
15 Demolition & Excavation for Your Work	\$ - \$ -
16 Other / Special Systems:	Ş -
General Items	\$ -
1 Bond Fee	\$ -
2 Submittals	\$ -
3 Close-Out	\$ -
4 As-Builts	\$ -
5 Warranty Call Back	
6 Mobilization	\$-
7 Testing & Balancing	\$-
8 Profit / Overhead	\$ -

## SWORN STATEMENT

STATE OF:	}	
COUNTY OF:	} SS }	
		(deponent), being sworn, states the following:
		is the (contractor) (subcontractor) for an
improvement to the fol	lowing described real property in	County, Michigan, described

as follows: (Insert legal description from Notice of Commencement or name and address of the Project)

That the following is a statement of each subcontractor and supplier and laborer, for whom payment of wages or fringe benefits and withholdings is due but unpaid, with whom the (contractor) (subcontractor) has (contracted) (subcontracted) for performance under the contract with the owner or lessee of the property and the amounts due to the persons as of the date of this statement are correctly and fully set forth opposite their names:

Name of subcontractor, supplier or laborer	Type of improvement furnished	Total contract price	Amount already paid	Amount currently owing	Balance to complete (optional)	Amount of laborer wages due but unpaid	Amount of laborer fringe benefits and withholdings due but unpaid
						-	
						_	
							<b>~</b>
•	-						
		•				 	<u>.</u>
· · ·							
Totals	0	0	0	0		<u> </u>	

(Some columns are not applicable to all persons listed)

The contractor has not procured material from, or subcontracted with any person other than those set forth, and owes no money for the improvement other than the sums set forth.

I make this statement as the (contractor) (subcontractor) or as \_\_\_\_\_\_\_\_ of the (contractor) (subcontractor) to represent to the owner or lessee of the property and his or her agents that the property is free from claims of construction liens, or the possibility of construction liens, except as specifically set forth in this statement for claims of construction liens by laborers that may be provided under section 109 of the construction lien act, 1980, P.A. 497, MCL 570.1109.

į

WARNING TO OWNER: AN OWNER OR LESSEE OF THE PROPERTY MAY NOT RELY ON THIS SWORN STATEMENT TO AVOID THE CLAIM OF A SUBCONTRACTOR, SUPPLIER, OR LABORER WHO HAS PROVIDED A NOTICE OF FURNISHING OR A LABORER WHO MAY PROVIDE A NOTICE OF FURNISHING UNDER SECTION 109 OF THE CONSTRUCTION LIEN ACT 1980 P.A. 497, MCL 570.1109 TO THE DESIGNEE OR THE OWNER OR LESSEE IF THE DESIGNEE IS NOT NAMED OR HAS DIED.

Deponent

WARNING TO THE DEPONENT: A PERSON, WHO GIVES A FALSE SWORN STATEMENT WITH THE INTENT TO DEFRAUD IS SUBJECT TO CRIMINAL PENALTIES AS PROVIDED IN SECTION 110 OF THE CONSTRUCTION LIEN ACT, 1980 PA 497, MCL 570.1110.

Subscribed and sworn to before me on (date) \_\_\_\_\_

Notary Public

,

. •

County, Michigan.

My commission expires: \_\_\_\_\_



## PARTIAL UNCONDITIONAL WAIVER

I / we have a contract with Clark Const	ruction Company to	provide work							
per contract and change orders for the improvement to the property described as:									
Clark Job #, Contract #, Project Description									
and hereby waive my/our construction lien t	o the amount of	\$0.00							
( <b>\$0.00</b> cumulative total paid to date) for labor and / or materials provided									
through This Partial Unconditional Waiver is a waiver and release of any and									
all claims arising out of work performed on the	project through the date of	<u> </u>							
	Subscribed and sworn to	before me							
(Company Name)									
By:	this day of	2002.							
(Authorized Signature)									
Title:									
Address:	Notary Public:								
	My commission expires:								
Telephone:									

#### WARNING DO NOT SIGN BLANK OR INCOMPLETE FORMS, RETAIN A COPY

NOTE: The following is supplied for informational purposes only and is not in limitation or modification of the above waiver.

Amount of Contract:	\$0.00
Previously Paid:	\$0.00
This Payment :	\$0.00
Balance of Contract:	\$0.00

Please sign and return this waiver to: Clark Construction Company 3535 Moores River Drive Lansing, MI 48911



## FULL UNCONDITIONAL WAIVER

/ we have a contract with Clark Construction Company provide work per										
contract and change orders for the improvement to the property described as:										
Clark job #, Contract #, Project description										
and having been fully paid and satisfied, all my/our construction lien rights against such										
property are hereby waived, released, and we release Clark Construction Company and the										
Owner in full, from any and all claims arising out of the Project.										
	Subscribed and sworn to before me									
(Company Name)	Subscribed and sworn to before the									
By:	thisday of20									
(Authorized signer)										
(Print or Type Name and Title of Signer)										
Address:	(Notary Public)									
	My commission expires:									
	· · · ·									

Telephone:

### WARNING DO NOT SIGN BLANK OR INCOMPLETE FORMS, RETAIN A COPY

Please sign and return this waiver to:

Clark Construction Company 3535 Moores River Drive Lansing, MI 48911

> Phone: (517) 372-0940 Fax: (517) 372-0668

### MATERIALS STORED PAYMENT FORM

Progress Payment Number\_\_\_\_\_Bid Package No./Category\_\_\_\_\_Description\_\_\_\_\_

1. The following items, if approved by the Owner, have been included in the above noted request for payment even though not yet incorporated into the work (attach additional sheets, if required):

ITEM	STORED AT	VALUE PER <u>ATTACHED INVOICE</u>
		<u>\$</u>

#### 2. TOTAL VALUE OF THE ABOVE ITEMS PER ATTACHED INVOICE(S):

- \$\_\_\_\_\_
- In addition to any insurance(s) the Owner may carry, the undersigned states that the 3. above items are covered by Trade Contractor's property insurance against all risk and that said insurance will remain in effect until final acceptance of the completed project by Trade Contractor must provide proof of insurance stating description, the Owner. location and value of items stored.
- 4. The undersigned hereby represents and guarantees that full payment has been made for the above items, and provides proof of payment, and by signature below, contingent upon receipt of payment on account of the above items, the Trade Contractor hereby transfers title to the above items to the Owner of the project.
- 5. Stored Materials are to be roped off or separated from all others. Boxes or pieces must be clearly marked.
- 6. Owner and/or Construction Manager reserves the right to inspect stored material at any time.
- 7. Receipt of payment on account for the above items shall in no manner or in any degree relieve the Trade Contractor of any obligation under its contract, particularly, but not limited to, its obligations with regard to the warranties and guarantees.

Trade Contractor:	Signature is by a member of firm or
Signade	Corporation entitled to sign Contract
Signed:	documents for the Trade Contract
Title:	
Date:	

## LABOR RATE CALCULATIONS

\_\_\_\_\_

.

#### TRADE:

Base Rate		STRAIGH	ΓTIME	TIME & ON	IE HALF	DOUBLE TIME		
WAGES		JOURNEYMAN	FOREMAN	JOURNEYMAN	FOREMAN	JOURNEYMAN	FOREMAN	
Base Rate	WAGES							
Vacation/Holiday Dues Taxable Wages Fringes Fr	Base Rate							
Dues	Vacation/Holiday							
Taxable Wages	Dues							
Fringes								
Fringes								
Fringes	Taxable Wages							
Health & Welfare	Taxable Hugee				<u>I                                    </u>		ļĮ	
Pension	Fringes							
Annuity	Health & Welfare							
S.U.B.       Other:       Image: Control of the second sec	Pension							
S.U.B.       Other:       Image: Control of the second sec	Annuity							
Image: Second	S.U.B.							
Payroll Tax & Ins.	Other:							
Payroll Tax & Ins.								
Payroll Tax & Ins.								
Payroll Tax & Ins.								
Payroll Tax & Ins.								
Payroll Tax & Ins.								
Payroll Tax & Ins.								
Payroll Tax & Ins.	Total Fringes							
F.I.C.A.       Image: Constraint of the second	-				·			
F.I.C.A.       Image: Constraint of the second	Payroll Tax & Ins.							
Medicare     Image: Constraint of the second s	F.I.C.A.							
F.U.T.     Image: Constraint of the second sec	Medicare							
F.U.T.     Image: Constraint of the second sec	S.U.I.T.							
Worker's Compensation       Image: Compensation       Image: Compensation         PL & PD Insurance       Image: Compensation       Image: Compensation         Total Payroll Tax & Ins.       Image: Compensation       Image: Compensation         Total Premium Cost       Image: Compensation       Image: Compensation         Total Hourly Cost       Image: Compensation       Image: Compensation         Overhead & Profit (15%)       Image: Compensation       Image: Compensation	F.U.I.T.							
PL & PD Insurance   Total Payroll Tax & Ins.     Total Premium Cost   Total Hourly Cost     Overhead & Profit (15%)	S.B.T.							
PL & PD Insurance   Total Payroll Tax & Ins.     Total Premium Cost   Total Hourly Cost     Overhead & Profit (15%)	Worker's Compensation							
Total Premium Cost	PL & PD Insurance							
Total Premium Cost								
Total Premium Cost	Total Pavroll Tax & Ins.							
Total Hourly Cost					<u> </u>			
Total Hourly Cost	Total Premium Cost							
Overhead & Profit (15%)				L				
Overhead & Profit (15%)	Total Hourly Cost							
			J					
	Overhead & Profit (15%)		]					
Total Hourly Rate	· · · ·							
	Total Hourly Rate							

#### FORM OF GUARANTEE

CLARK PROJECT NUMBER:
PROJECT NAME
LOCATION
NAME OF CONTRACTOR
CONTRACT FOR

We hereby agree that:

- a. We have completed our contract in full conformity with the intent of the contract documents and have made no substitutions in materials except as authorized in writing by the Architect.
- b. We will return to the project and commence work within three (3) working days of receipt of written notification from the Owner or the Architect and will provide at our expense all necessary labor, equipment and material to make proper repairs or corrections made necessary by defective materials or interior workmanship furnished or performed under contract, all to the satisfaction of the Owner and the Architect, and without cost to the Owner.
- c. This guarantee of our work and the work of all our subcontractors shall commence on \_\_\_\_\_\_ and remain in full force and effect \_\_\_\_\_\_ until \_\_\_\_\_. (Date of Substantial Completion) (One Year)
- d. We hereby certify that all payrolls, material bills, and other indebtedness connected with the work on the subject project have been paid in full.

SIGNED	
	(AUTHORIZED OFFICER
WITNESS	
DATE	
ADDRESS	

#### **State of Michigan**

WHPWRequest@michigan.gov

Official Request25Requestor:Saline Area SchoolsProjectTo Be DeterminedProject Number:To Be Determined

# Official 2017 Prevailing Wage Rate Schedule for Parking Lot, Road, Highway, Bridge and Airport Construction

Contr	Issue Date act must be award		-	/2017	
	Page	1 of 23			
Construction Mechanic Classifica		Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
 CARPENTERS CARPENTERS ZONE 1	RBCZ1 WAGE	\$29.47	======== \$44.21	×	 хнхххнн`
	6/6/2016 FRINGE	\$26.83		Λ	
Apprentice Rates:					
1ST 6 MONTHS	WAGE FRINGE	\$12.97 \$16.50			
2ND 6 MONTHS	WAGE FRINGE	\$16.21 \$18.53			
YEAR 2	WAGE FRINGE	\$19.16 \$20.38			
YEAR 3	WAGE FRINGE	\$22.10 \$22.22			
YEAR 4	WAGE FRINGE	\$25.05 \$24.06			
ARPENTERS ZONE 1					

Wayne, Oakland, Macomb, Sanilac, St. Clair, Monroe and the following townships of Livingston County Brighton, Deerfield, Genoa, Hartland, Osceola and Tyrone

comment

For all hours worked on Memorial Day, July 4, Labor Day, Thanksgiving Day, Christmas Day and New Years Day, double time is due

Official Request #: 25 Requestor: Saline Area Schools

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 2 of 23

	Page	2 of 23				
Construction Mechanic Classific	ation	Straight Time Rate	Time and One Half Rate	Double Time Rate	Overti	ime Code
CARPENTERS ZONE 2	RBCZ2 WAGE 6/14/2016 FRINGE	+	+	X	хнх	ххннү
Apprentice Rates:						
1ST YEAR	WAGE		+			
2ND YEAR	WAGE FRINGE	\$18.43	\$27.64			
3RD YEAR	WAGE FRINGE		\$31.59			
4TH YEAR	WAGE FRINGE					
CARPENTERS ZONE 2 The entire state except those counties and townships listed in Zone 1						
comment	SUNDAYS PAII HOLIDAYS PAI			F		
CEMENT MASONS						
CEMENT MASONS ZONE 1	RBCMZ1 WAGE 6/6/2016 FRINGE	+	+	Х	хнх	хнннү
Apprentice Rates:						
1ST YEAR	WAGE FRINGE	+ • • • • •				
2ND YEAR	WAGE FRINGE	+				
3RD YEAR	WAGE FRINGE		+			
CEMENT MASONS ZONE 1 Genesee, Oakland, Macomb, Monroe, Washtenaw, Wayne, Livingston and Saginaw Counties.						

comment

If only Sat & Sun worked, time and one-half due

Official Request #: 25 Requestor: Saline Area Schools

Project Description: To Be Determined

### **Official Rate Schedule**

Co	Issue Date		/2017 4/12	/2017	
	Page	3 of 23			
Construction Mechanic Classi	fication	Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
CEMENT MASONS ZONE 2	RBCMZ2 WAGE 6/6/2016 FRINGE	•	\$43.41		ххнххнннү
Apprentice Rates:					
1ST YEAR	WAGE FRINGE	-			
2ND YEAR	WAGE FRINGE				
3RD YEAR	WAGE FRINGE	+			
CEMENT MASONS ZONE 2 Il counties not listed in Zone 1					
comme	nt If only Sat & Su	n worked, t	ime and one	-half due	
Metal Fence & Guard Rail	IR-55-MF WAGE 10/9/2015 FRINGE	+		\$40.00 \$39.74	ННОННННОҮ
enawee	Monroe				
Make up day allowed comme	nt Four Tens allow to weather cond			e-up day allo	owed if M-Th schedule not worked d

Official Request #: 25 Requestor: Saline Area Schools

## **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Project Description: To Be Determined

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

		•	4 of 23 Straight Time	Time and One Half	Double Time Rate	
Construction Mechanic Classific	ation		Rate	Rate		Overtime Code
IRONWOKERS ZONE 1: Fence, sound barrier and guardrail	RBIRZ1 6/14/2016	WAGE	\$24.00	\$36.00	\$48.00	ХХНХХХНОҮ
erection/installation work, and	6/14/2016	FRINGE	\$11.55	\$11.55	\$11.55	
exterior signage work.						
Apprentice Rates:						
60%		WAGE	\$14.40	\$21.60	\$28.80	
		FRINGE	\$10.75	\$10.75	\$10.75	
65%		WAGE	\$15.60			
		FRINGE	\$10.85			
70%		WAGE	\$16.80			
		FRINGE	\$10.96			
75%		WAGE	\$18.00			
		FRINGE	\$11.05		\$11.05	
80%		WAGE	\$19.20			
		FRINGE	\$11.15		\$11.15	
85%		WAGE	\$20.40			
		FRINGE	\$11.25	\$11.25	\$11.25	
RONWORKERS ZONE 1	20					

Genesee, Oakland, Macomb, Washtenaw and Wayne Counties

Make up day allowed comment

Four 10 hour work days may be worked Monday-Saturday.

Official Request #: 25 Requestor: Saline Area Schools **Official Rate Schedule** 

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Project Description: To Be Determined

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 5 of 23

Construction Mechanic Classificat	ion		Time Rate	Time and One Half Rate	Double Time Rate						ode		
IRONWORKERS ZONE 2: Fence, sound barrier and guardrail erection/installation work, and	EEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEEE	WAGE	\$20.00 \$11.55		\$40.00 \$11.55	× ×				==== X	_	D	Y
exterior signage work.	0/14/2010	TRINGE	ψ11.00	φ11.00	φ11.00								
Apprentice Rates:			¢10.00	¢10.00	¢04.00								
60%		WAGE	\$12.00 \$10.75		\$24.00 \$10.75								
65%		WAGE	\$13.00		\$26.00								
0070		FRINGE	\$10.85		\$10.85								
70%		WAGE	\$14.00		\$28.00								
		FRINGE	\$10.96	\$10.96	\$10.96								
75%		WAGE	\$15.00	\$22.50	\$30.00								
		FRINGE	\$11.05	\$11.05	\$11.05								
80%		WAGE	\$16.00		\$32.00								
		FRINGE	\$11.15	•	\$11.15								
85%		WAGE FRINGE	\$17.00 \$11.25		\$34.00 \$11.25								
RONWORKERS ZONE 2 he entire state except those counties listed in Zone :Genesee, Oakland, Macomb, Monroe, Washtenaw nd Wayne and Lenawee													
Make up day allowed comment	Four 10	) hour wo	rk days ma	ay be worked	d Monday-Sa	aturda	ay.						
LABORERS													
LABORERS CLASS 1 ZONE 1	RBLABC1Z1 6/9/2016	WAGE FRINGE	\$22.32 \$16.14			ХХ	( Х	X	X	X	Х	Η	Y
Apprentice Rates:													
0-1000 WORK HOURS		WAGE	\$16.74	\$25.11									
		FRINGE	\$16.14	\$17.66									
1001-2000 WORK HOURS	;	WAGE	\$17.86										
		FRINGE	\$16.14										
2001-3000 WORK HOURS	i	WAGE	\$18.97										
		FRINGE	\$16.14										
3001-4000 WORK HOURS	i	WAGE	\$21.20 \$16.14										
ABORERS ZONE 1 ienesee, Macomb, Monroe, Oakland, Washtenaw nd Wayne			÷	÷									

Official Request #: 25 Requestor: Saline Area Schools

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

, Jago 6 of 22

		Page	6 of 23										
Construction Mechanic Classificat	ion		Straight Time Rate	Time and One Half Rate	Double Time Rate			-			e Co	de	
LABORERS CLASS 1 ZONE 2	RBLABC1Z2 6/9/2016	WAGE FRINGE									X	Х	НΥ
Apprentice Rates:													
0-1000 WORK HOURS		WAGE FRINGE	•										
1001-2000 WORK HOURS	i	WAGE FRINGE		+									
2001-3000 WORK HOURS	i	WAGE FRINGE	\$16.00	\$17.45									
3001-4000 WORK HOURS		WAGE FRINGE	+										
LABORERS ZONE 2 Allegan, Barry, Bay, Berrien, Branch, Calhoun, Cass, Clinton, Eaton, Gratiot, Hillsdale, Huron, Ingham, Jackson, Kalamazoo, Lapeer, Lenawee, Livingston, Midland, Muskegon, Saginaw, Sanilac, Shiawassee, St. Clair, St. Joseph, Tuscola, and Van Buren													
LABORERS CLASS 1 ZONE 3 & 4	RBLABC1Z3 6/9/2016	WAGE FRINGE		+		х	Х	Х	Х	Х	х	Х	ΗY
Apprentice Rates:													
0-1000 WORK HOURS		WAGE FRINGE											
1001-2000 WORK HOURS	i	WAGE FRINGE	<b>•</b>										
2001-3000 WORK HOURS	i	WAGE FRINGE		\$17.45									
3001-4000 WORK HOURS		WAGE FRINGE											
LABORERS ZONE 3 Alcona, Alpena, Antrim, Arenac, Benzie, Charlevoix, Cheboygan, Clare, Crawford, Emmet, Gladwin, Grand Traverse, Ionia, Iosco, Isabella, Kalkaska, Kent, Lake, Leelanau, Manistee, Mason, Mecosta, Missaukee, Montcalm, Montmorency, Newaygo, Oceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford	Houghton,	iga, Chippe Iron, Kewe	wa, Delta, Dic enaw, Luce, M	kinson, Gogebio lackinac, and Schoolcraft									

Official Request #: 25 Requestor: Saline Area Schools

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 7 of 23

		Page	7 of 23				
Construction Mechanic Classification	ı		Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Cod	de
LABORERS CLASS 2 ZONE 1 RB	BLABC2Z1 6/9/2016	WAGE FRINGE		\$33.68	X	<b>X X X X X</b>	хн
Apprentice Rates:							
0-1000 WORK HOURS		WAGE FRINGE					
1001-2000 WORK HOURS		WAGE FRINGE		+			
2001-3000 WORK HOURS		WAGE FRINGE	•				
3001-4000 WORK HOURS		WAGE FRINGE	+				
ABORERS ZONE 1 Genesee, Macomb, Monroe, Oakland, Washtenaw and Wayne							
LABORERS CLASS 2 ZONE 2 RB	BLABC2Z2	WAGE	+		х	<b>x x x x x</b> x	хн
	6/9/2016	FRINGE	\$16.00	\$17.45			
Apprentice Rates:							
0-1000 WORK HOURS		WAGE FRINGE	+				
1001-2000 WORK HOURS		WAGE FRINGE	+	+			
2001-3000 WORK HOURS		WAGE FRINGE	+ -	+			
3001-4000 WORK HOURS		WAGE FRINGE	+				
ABORERS ZONE 2 Allegan, Barry, Bay, Berrien, Branch, Calhoun, Cass, Clinton, Eaton, Gratiot, Hillsdale, Huron, Ingham, Isalacan, Kalanastan, Lanastan, Lanastan							

Official Request #: 25 Requestor: Saline Area Schools

Jackson, Kalamazoo, Lapeer, Lenawee, Livingston, Midland, Muskegon, Saginaw, Sanilac, Shiawassee, St. Clair, St. Joseph, Tuscola, and Van Buren

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 8 of 23

		Page	8 OT 23										
Construction Mechanic Classificat	ion		Straight Time Rate	Time and One Half Rate	Double Time Rate			0,	ver	time	e C	ode	ŗ
LABORERS CLASS 2 ZONES 3 & 4	RBLABC2Z4	WAGE FRINGE	====== \$19.87 \$16.00	\$29.81									н`
Apprentice Rates:													
0-1000 WORK HOURS		WAGE FRINGE	\$14.90 \$16.00										
1001-2000 WORK HOURS	5	WAGE FRINGE	\$15.90 \$16.00	\$23.85									
2001-3000 WORK HOURS	5	WAGE FRINGE	\$16.89 \$16.00	+									
3001-4000 WORK HOURS	5	WAGE FRINGE	\$18.88 \$16.00	+									
ABORERS ZONE 3 Vicona, Alpena, Antrim, Arenac, Benzie, Charlevoix, Cheboygan, Clare, Crawford, Emmet, Gladwin, Grand raverse, Ionia, Iosco, Isabella, Kalkaska, Kent, .ake, Leelanau, Manistee, Mason, Mecosta, Missaukee, Montcalm, Montmorency, Newaygo, Ceeana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford	Alger, Bara Houghton,	Iron, Kewee	wa, Delta, Dic enaw, Luce, M	kinson, Gogebio lackinac, and Schoolcrafi									
LABORERS CLASS 3 ZONE 1	RBLABC3Z1 6/9/2016	WAGE FRINGE	\$22.63 \$16.14			х	х	х	Х	Х	Х	х	н`
Apprentice Rates:													
0-1000 WORK HOURS		WAGE FRINGE	\$16.97 \$16.14	+									
1001-2000 WORK HOURS	5	WAGE FRINGE	\$18.10 \$16.14	+ -									
2001-3000 WORK HOURS	5	WAGE FRINGE	\$19.24 \$16.14										
3001-4000 WORK HOURS	5	WAGE FRINGE	\$21.50 \$16.14										
ABORERS ZONE 1													

LABORERS ZONE 1 Genesee, Macomb, Monroe, Oakland, Washtenaw

and Wayne

Official Request #: 25 Requestor: Saline Area Schools

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page Q of 23

		Page	9 of 23										
Construction Mechanic Classificati	on		Straight Time Rate	Time and One Half Rate	Double Time Rate			Ov	ert	ime	Co	de	
LABORERS CLASS 3 ZONE 2	RBLABC3Z2 6/9/2016	WAGE FRINGE	\$20.85 \$16.00	+		<b>X</b> 2					X .	ΧI	ΗY
Apprentice Rates:													
0-1000 WORK HOURS		WAGE FRINGE	\$15.64 \$16.00	+									
1001-2000 WORK HOURS		WAGE FRINGE	\$16.68 \$16.00	+									
2001-3000 WORK HOURS		WAGE FRINGE	\$16.00	\$17.45									
3001-4000 WORK HOURS		WAGE FRINGE	\$19.81 \$16.00										
ABORERS ZONE 2 Nllegan, Barry, Bay, Berrien, Branch, Calhoun, Cass, Clinton, Eaton, Gratiot, Hillsdale, Huron, Ingham, Iackson, Kalamazoo, Lapeer, Lenawee, Livingston, Midland, Muskegon, Saginaw, Sanilac, Shiawassee, St. Clair, St. Joseph, Tuscola, and Van Buren													
LABORERS CLASS 3 ZONES 3 & 4	RBLABC3Z3 6/9/2016	WAGE FRINGE	+			X	х	Х	х	Х	X	ΧI	ΗY
Apprentice Rates:													
0-1000 WORK HOURS		WAGE FRINGE	\$15.12 \$16.00										
1001-2000 WORK HOURS		WAGE FRINGE	\$16.13 \$16.00										
2001-3000 WORK HOURS		WAGE FRINGE	1	+ -									
3001-4000 WORK HOURS		WAGE FRINGE	\$19.15 \$16.00	+ -									
LABORERS ZONE 3 Alcona, Alpena, Antrim, Arenac, Benzie, Charlevoix, Cheboygan, Clare, Crawford, Emmet, Gladwin, Grand Traverse, Ionia, Iosco, Isabella, Kalkaska, Kent, Lake, Leelanau, Manistee, Mason, Mecosta, Missaukee, Montcalm, Montmorency, Newaygo, Oceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle Roscommon and Wexford	Alger, Bara Houghton,	Iron, Kewee	wa, Delta, Dic enaw, Luce, M	kinson, Gogebio ackinac, and Schoolcraft									

Official Request #: 25 Requestor: Saline Area Schools

Presque Isle, Roscommon and Wexford

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 10 of 23

	Page	10 of 23			
Construction Mechanic Classification		Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
	BC4Z1 <b>WAGE</b>	*	+	× >	 (
	6/9/2016 FRINGE	<b>\$</b> 16.14	\$17.66		
Apprentice Rates:					
0-1000 WORK HOURS	WAGE FRINGE		+		
1001-2000 WORK HOURS	WAGE FRINGE		+ -		
2001-3000 WORK HOURS	WAGE FRINGE		\$28.95		
3001-4000 WORK HOURS	WAGE FRINGE				
ABORERS ZONE 1 Senesee, Macomb, Monroe, Oakland, Washtenaw nd Wayne					
LABORERS CLASS 4 ZONE 2 RBLA	BC4Z2 WAGE	E \$21.20	\$31.80	х >	кххххнү
	6/9/2016 FRINGE	\$16.00	\$17.45		
Apprentice Rates:					
0-1000 WORK HOURS	WAGE FRINGE	+	+		
1001-2000 WORK HOURS	WAGE FRINGE	+	+ -		
2001-3000 WORK HOURS	WAGE FRINGE	•	+		
3001-4000 WORK HOURS	WAGE FRINGE	+ -			
ABORERS ZONE 2 Allegan, Barry, Bay, Berrien, Branch, Calhoun, Cass, Clinton, Eaton, Gratiot, Hillsdale, Huron, Ingham, Isakaan, Kalamazaa, Lanawaa, Livingstan					

Official Request #: 25 Requestor: Saline Area Schools

Jackson, Kalamazoo, Lapeer, Lenawee, Livingston, Midland, Muskegon, Saginaw, Sanilac, Shiawassee, St. Clair, St. Joseph, Tuscola, and Van Buren

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

		Page	11 of 23									
Construction Mechanic Classificat	ion		Straight Time Rate	Time and One Half Rate	Double Time Rate		0	vertii	me	Cod	e	
LABORERS CLASS 4 ZONES 3 & 4	RBLABC4Z3 6/9/2016	WAGE FRINGE				Х Х	==== Х	X .	==== X	=== X X	СН	Y
Apprentice Rates:												
0-1000 WORK HOURS		WAGE FRINGE										
1001-2000 WORK HOURS	3	WAGE FRINGE										
2001-3000 WORK HOURS	3	WAGE FRINGE										
3001-4000 WORK HOURS	3	WAGE FRINGE										
LABORERS ZONE 3 Alcona, Alpena, Antrim, Arenac, Benzie, Charlevoix,	LABOREI Alger, Bara			kinson, Gogebi	с,							

LABORERS ZONE 3 Alcona, Alpena, Antrim, Arenac, Benzie, Charlevoix, Cheboygan, Clare, Crawford, Emmet, Gladwin, Grand Traverse, Ionia, Iosco, Isabella, Kalkaska, Kent, Lake, Leelanau, Manistee, Mason, Mecosta, Missaukee, Montcalm, Montmorency, Newaygo, Oceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford

LABORERS CLASS 5 ZONE 1 R	BLABC5Z1 WAGE 6/9/2016 FRINGE	\$22.92 \$16.14	\$34.38 \$17.66	ххххххнү
Apprentice Rates:				
0-1000 WORK HOURS	WAGE FRINGE	\$17.19 \$16.14	\$25.78 \$17.66	
1001-2000 WORK HOURS	WAGE FRINGE	\$18.34 \$16.14	\$27.51 \$17.66	
2001-3000 WORK HOURS	WAGE FRINGE	\$19.48 \$16.14	\$29.22 \$17.66	
3001-4000 WORK HOURS	WAGE FRINGE	\$21.77 \$16.14	\$32.66 \$17.66	

Houghton, Iron, Keweenaw, Luce, Mackinac,

Marquette, Menominee, Ontonagon and Schoolcraft

LABORERS ZONE 1

Genesee, Macomb, Monroe, Oakland, Washtenaw and Wayne

Official Request #: 25 Requestor: Saline Area Schools

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 12 of 23

		Page	12 of 23										
Construction Mechanic Classification	on		Straight Time Rate	Time and One Half Rate	Double Time Rate			-			e Co		
LABORERS CLASS 5 ZONE 2	RBLABC5Z2 6/9/2016	WAGE FRINGE	\$21.07 \$16.00			×	X				X	-	ΗΥ
Apprentice Rates:													
0-1000 WORK HOURS		WAGE FRINGE											
1001-2000 WORK HOURS		WAGE FRINGE		\$17.45									
2001-3000 WORK HOURS		WAGE FRINGE											
3001-4000 WORK HOURS		WAGE FRINGE	\$20.02 \$16.00										
LABORERS ZONE 2 Allegan, Barry, Bay, Berrien, Branch, Calhoun, Cass, Clinton, Eaton, Gratiot, Hillsdale, Huron, Ingham, Jackson, Kalamazoo, Lapeer, Lenawee, Livingston, Midland, Muskegon, Saginaw, Sanilac, Shiawassee, St. Clair, St. Joseph, Tuscola, and Van Buren													
LABORERS CLASS 5 ZONES 3 & 4	RBLABC5Z3 6/9/2016	WAGE FRINGE				Х	х	Х	Х	Х	х	Х	ΗΥ
Apprentice Rates:													
0-1000 WORK HOURS		WAGE FRINGE	\$15.16 \$16.00										
1001-2000 WORK HOURS		WAGE FRINGE	\$16.18 \$16.00	*									
2001-3000 WORK HOURS		WAGE FRINGE	-										
3001-4000 WORK HOURS		WAGE FRINGE											
LABORERS ZONE 3 Alcona, Alpena, Antrim, Arenac, Benzie, Charlevoix, Cheboygan, Clare, Crawford, Emmet, Gladwin, Grand Traverse, Ionia, Iosco, Isabella, Kalkaska, Kent, Lake, Leelanau, Manistee, Mason, Mecosta, Missaukee, Montcalm, Montmorency, Newaygo, Oceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford	Houghton,	iga, Chippe Iron, Kewee	wa, Delta, Dic enaw, Luce, M	kinson, Gogebio lackinac, and Schoolcraft									

Official Request #: 25 Requestor: Saline Area Schools

Presque Isle, Roscommon and Wexford

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 13 of 23

		Page	13 of 23					
Construction Mechanic Classification	ı		Straight Time Rate	Time and One Half Rate	Double Time Rate		Overtime	e Code
LABORERS CLASS 6 ZONE 1 RB	BLABC6Z1 6/9/2016	WAGE	+ -			х х	ххх	ххн
Apprentice Rates:								
0-1000 WORK HOURS		WAGE FRINGE	\$17.42 \$16.14	+				
1001-2000 WORK HOURS		WAGE FRINGE	\$18.58 \$16.14					
2001-3000 WORK HOURS		WAGE FRINGE	\$19.74 \$16.14	+				
3001-4000 WORK HOURS		WAGE FRINGE	\$22.06 \$16.14					
ABORERS ZONE 1 Senesee, Macomb, Monroe, Oakland, Washtenaw nd Wayne								
LABORERS CLASS 6 ZONE 2 RB	BLABC6Z2	WAGE	\$21.41			хх	ххх	ххн
	6/9/2016	FRINGE	\$16.00	\$17.45				
Apprentice Rates:			• · · · · ·					
0-1000 WORK HOURS		WAGE	\$16.06 \$16.00	+				
1001-2000 WORK HOURS		WAGE	\$17.13 \$16.00	\$25.70				
2001-3000 WORK HOURS		WAGE	\$18.20 \$16.00	\$27.30				
3001-4000 WORK HOURS		WAGE FRINGE	\$20.34 \$16.00					
ABORERS ZONE 2 Nlegan, Barry, Bay, Berrien, Branch, Calhoun, Cass, Dinton, Eaton, Gratiot, Hillsdale, Huron, Ingham, Jackson, Kalamazon, Lanava, Lanavao, Liviagton								

Official Request #: 25 Requestor: Saline Area Schools

Jackson, Kalamazoo, Lapeer, Lenawee, Livingston, Midland, Muskegon, Saginaw, Sanilac, Shiawassee, St. Clair, St. Joseph, Tuscola, and Van Buren

Project Description: To Be Determined

## **Official Rate Schedule**

### Issue Date: 1/12/2017

#### Contract must be awarded by: 4/12/2017

Page 14 of 23

Construction Mechanic Classificati	ion	:	Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
LABORERS CLASS 6 ZONES 3 & 4	======================================	WAGE FRINGE	\$20.65 \$16.00			х х х х х х х н ү
Apprentice Rates:						
0-1000 WORK HOURS		WAGE FRINGE	\$15.49 \$16.00			
1001-2000 WORK HOURS	i	WAGE FRINGE	\$16.52 \$16.00			
2001-3000 WORK HOURS	i	WAGE FRINGE	\$17.55 \$16.00	+		
3001-4000 WORK HOURS	i	WAGE FRINGE	\$19.62 \$16.00	+		
ABORERS ZONE 3 Icona, Alpena, Antrim, Arenac, Benzie, Charlevoix, heboygan, Clare, Crawford, Emmet, Gladwin, Grand raverse, Ionia, Iosco, Isabella, Kalkaska, Kent,	Alger, Bara Houghton,	Iron, Kewee	va, Delta, Dicl naw, Luce, M	kinson, Gogebic ackinac, and Schoolcraft		
ake, Leelanau, Manistee, Mason, Mecosta, lissaukee, Montcalm, Montmorency, Newaygo, oceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford						
lissaukee, Montcalm, Montmorency, Newaygo, oceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa,	RBLABC72 6/9/2016	WAGE FRINGE	\$23.98 \$16.00			ххххххнү
lissaukee, Montcalm, Montmorency, Newaygo, Iceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford		-				ххххххнү
lissaukee, Montcalm, Montmorency, Newaygo, ceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford LABORERS CLASS 7 ZONES 2, 3, 4		-		\$17.45 \$26.97		ххххххнү
issaukee, Montcalm, Montmorency, Newaygo, ceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford LABORERS CLASS 7 ZONES 2, 3, 4 Apprentice Rates:	6/9/2016	FRINGE	\$16.00 \$17.98	\$17.45 \$26.97 \$17.45 \$28.77		ххххххнү
lissaukee, Montcalm, Montmorency, Newaygo, ceana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford LABORERS CLASS 7 ZONES 2, 3, 4 Apprentice Rates: 0-1000 WORK HOURS	6/9/2016	FRINGE WAGE FRINGE WAGE	\$16.00 \$17.98 \$16.00 \$19.18	\$17.45 \$26.97 \$17.45 \$28.77 \$17.45 \$30.57		ххххххнү
tissaukee, Montcalm, Montmorency, Newaygo, Joeana, Ogemaw, Osceola, Oscoda, Otsego, Ottawa, Presque Isle, Roscommon and Wexford LABORERS CLASS 7 ZONES 2, 3, 4 Apprentice Rates: 0-1000 WORK HOURS 1001-2000 WORK HOURS	6/9/2016	FRINGE WAGE FRINGE WAGE FRINGE WAGE	\$16.00 \$17.98 \$16.00 \$19.18 \$16.00 \$20.38	\$17.45 \$26.97 \$17.45 \$28.77 \$17.45 \$30.57 \$17.45 \$34.17		ххххххн

Official Request #: 25 Requestor: Saline Area Schools

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 15 of 23

		Page 15 of 23													
Construction Mechanic Classificat	tion		Straight Time Rate	Time and One Half Rate	Double Time Rate			c	)ve	rti	me	ə C	od	е	
LABORERS CLASS 7 ZONE 1	RBLABC7Z1 6/9/2016	WAGE FRINGE	-			X :	== X	== X		<	X	×	== ( X	( }	ΗY
Apprentice Rates:															
0-1000 WORK HOURS		WAGE FRINGE													
10001-2000 WORK HOURS		WAGE FRINGE		+ -											
2001-3000 WORK HOUR	S	WAGE FRINGE		+											
3001-4000 WORK HOURS		WAGE FRINGE		+											
LABORERS ZONE 1 Genesee, Macomb, Monroe, Oakland, Washtenaw and Wayne															
OPERATING ENGINEERS															
OPERATING ENGINEERS CLASS I ZONE 1& 2	RBOEC1Z1 6/6/2016 6/6/2016	WAGE				Η	Η	F	1	4	Η	H	1  -		ΗY
Apprentice Rates:	0/0/2010		<b>~</b> =····	φ <b>_</b> 0. <u>−</u> .											
1ST 6 MONTHS		WAGE FRINGE	· · · · · ·	+											
2ND 6 MONTHS		WAGE FRINGE		+											
3RD 6 MONTHS		WAGE FRINGE		+											
4TH 6 MONTHS		WAGE FRINGE	\$26.14	\$28.21											
5TH 6 MONTHS		WAGE FRINGE	\$26.14	\$28.21											
6TH 6 MONTHS		WAGE FRINGE	\$26.14	\$28.21											
OPERATING ENGINEERS ZONE 1 Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne counties	The entire	state excep , Oakland,		NE 2 es listed in Zone roe, Washtenav											

Make up day allowed comment

4 tens allowed M-Th or T-F. If work cannot be performed due to weather during the M-Th schedule, Friday may be worked as a make-up day.

Official Request #: 25 Requestor: Saline Area Schools

Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

		Page	16 of 23			
Construction Mechanic Classificat	ion		Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
OPERATING ENGINEERS CLASS II ZONE 1	RBOEC2Z1 6/6/2016	WAGE	\$21.68			 Н Н Н Н Н Н Н Н Ү
		FRINGE	\$26.26	\$27.89		
OPERATING ENGINEERS ZONE 1 Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne counties						
Make up day allowed comment					not be perfo nake-up day.	rmed due to weather during the M-Th
OPERATING ENGINEERS GREASE TRUCK CLASS II ZONE 1	RBOEC2Z1G T 6/6/2016	WAGE	\$22.81	\$34.22		ннннннү
		FRINGE	\$26.43	\$28.14		
OPERATING ENGINEERS ZONE 1 Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne counties						
Make up day allowed comment					not be perfo nake-up day.	rmed due to weather during the M-Th
OPERATING ENGINEERS CLASS 2 ZONE 2	RBOEC2Z2	WAGE	\$21.55	\$32.33		ннннннү
ZONE Z	6/6/2016 6/6/2016	FRINGE	\$26.24	\$27.86		
OPERATING ENGINEERS ZONE 2 The entire state except those counties listed in Zone 1:Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne						
Make up day allowed comment					not be perfo nake-up day	rmed due to weather during the M-Th
OPERATING ENGINEERS GREASE TRUCK CLASS 2 ZONE 2	RBOEC2Z2G T 6/6/2016	WAGE	\$22.68	\$34.02		ннннннү
		FRINGE	\$26.41	\$28.11		
OPERATING ENGINEERS ZONE 2 The entire state except those counties listed in Zone 1:Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne						
Make up day allowed comment					not be perfo nake-up day	rmed due to weather during the M-Th

Official Request #: 25 Requestor: Saline Area Schools

#### Project Description: To Be Determined

## **Official Rate Schedule**

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

		Page	17 of 23					
Construction Mechanic Classificat	tion		Straight Time Rate	Time and One Half Rate	Double Time Rate		Overtime Code	
OPERATING ENGINEERS CLASS III	RBOEC3Z1	WAGE	\$21.19	\$31.79		н н	=====================================	
ZONE 1	6/6/2016 6/6/2016	FRINGE	\$26.19	\$27.78				
OPERATING ENGINEERS ZONE 1 Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne counties								
Make up day allowed comment					not be perfo nake-up day		l due to weather during the	M-Th
OPERATING ENGINEERS CLASS III	RBOEC3Z2	WAGE	\$21.06	\$31.59		нн	ІНННННҮ	
ZONE 2	6/6/2016 6/6/2016	FRINGE	\$26.17	\$27.75				
OPERATING ENGINEERS ZONE 2 The entire state except those counties listed in Zone 1:Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne								
Make up day allowed comment					not be perfo nake-up day		I due to weather during the	M-Th
OPERATING ENGINEERS CLASS IV ZONE 1	RBOEC4Z1 6/6/2016	WAGE	• -			нн	ІННННННҮ	
	6/6/2016	FRINGE	\$26.17	\$27.75				
OPERATING ENGINEERS ZONE 1 Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne counties								
Make up day allowed comment					not be perfo nake-up day		l due to weather during the	M-Th
OPERATING ENGINEERS CLASS IV ZONE 2	RBOEC4Z2 6/6/2016	WAGE	\$20.78	\$31.17		нн	інннннү	
		FRINGE	\$26.13	\$27.69				
OPERATING ENGINEERS ZONE 2 The entire state except those counties listed in Zone 1:Genesee, Oakland, Macomb, Monroe, Washtenaw and Wayne								
Make up day allowed comment					not be perfo		due to weather during the	M-Th

schedule, Friday may be worked as a make-up day.

Official Request #: 25 Requestor: Saline Area Schools

#### Project Description: To Be Determined

## **Official Rate Schedule**
#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

act must de	eaward	ed by:	4/12	/2017			
	Page '	18 of 23					
		Time Rate	Time and One Half Rate	Double Time Rate		Overtime Code	le
							ни
4/17/2015	-						
TM247-2 4/17/2015 4/17/2015	WAGE				нн	ннннн	н
		<i>Ф</i> о. <u> </u>	<i>Ф</i> о. <u>-</u> .				
TM247-3 4/17/2015	WAGE				нн	ннннн	нι
4/17/2015	FRINGE	\$8.21	\$8.21				
	tion TM247 4/17/2015 4/17/2015 TM247-2 4/17/2015 4/17/2015	tion       WAGE         TM247       WAGE         4/17/2015       FRINGE         TM247-2       WAGE         4/17/2015       FRINGE         TM247-2:       WAGE         4/17/2015       FRINGE	Time Rate         TM247       WAGE       \$19.99         4/17/2015       4/17/2015       FRINGE       \$8.21         TM247-2       WAGE       \$24.49         4/17/2015       FRINGE       \$8.21         TM247-3       WAGE       \$23.24         4/17/2015       WAGE       \$23.24	Page 18 of 23         Straight Time Rate       Time and One Half Rate         TM247       WAGE       \$19.99       \$29.99         4/17/2015       FRINGE       \$8.21       \$8.21         TM247-2       WAGE       \$8.21       \$8.21         TM247-2       WAGE       \$8.21       \$8.21         TM247-2       WAGE       \$8.21       \$8.21         TM247-3       WAGE       \$8.21       \$8.21	Page 18 of 23Straight Time RateTime and One Half Time Rate RateDouble Time Rate RateTM247WAGE\$19.99\$29.994/17/2015 4/17/2015FRINGE\$8.21\$8.21TM247-2WAGE\$24.49\$36.744/17/2015 4/17/2015FRINGE\$8.21\$8.21TM247-3 4/17/2015WAGE\$23.24\$34.86	Page 18 of 23         Straight Time       Time and One Half       Double Time Rate         tion       Rate       Rate         TM247       WAGE       \$19.99       \$29.99       H         4/17/2015       FRINGE       \$8.21       \$8.21         TM247-2       WAGE       \$24.49       \$36.74       H         4/17/2015       FRINGE       \$8.21       \$8.21         4/17/2015       FRINGE       \$8.21       \$8.21         4/17/2015       FRINGE       \$8.21       \$8.21         TM247-3       WAGE       \$23.24       \$34.86       H	Page 18 of 23         Straight Time and Ouble Time Rate         Double Time Rate         Overtime Code           tion         Rate         Rate         Overtime Code           TM247         WAGE         \$19.99         \$29.99         H H H H H H H H           4/17/2015         FRINGE         \$8.21         \$8.21           TM247-2         WAGE         \$24.49         \$36.74         H H H H H H H H           4/17/2015         FRINGE         \$8.21         \$8.21           4/17/2015         FRINGE         \$8.21         \$8.21           170247-3         WAGE         \$23.24         \$34.86         H H H H H H H H

Statewide

Official Request #: 25 Requestor: Saline Area Schools

cleaning and televising work

### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Cont	ract must be		•	4/12	/2017		
Construction Mechanic Classifica	ation	Page	19 of 23 Straight Time Rate	Time and One Half Rate	Double Time Rate		Overtime Code
Boiler Operator: unit driver and operator of steam/water heater units and all ancillary equipment associated	TM247-4 4/17/2015 4/17/2015	WAGE	•	·		н н	 ННННННN
Statewide							
Combo Unit driver & Jetter-Vac Operator	TM247-5 4/17/2015 4/17/2015	WAGE FRINGE	·	·		нн	ннннни
Statewide							
Pipe Bursting & Slip-lining Equipment Operator	TM247-6 9/16/2016 9/16/2016	WAGE FRINGE	•			нн	ннннни

Statewide

Official Request #: 25 Requestor: Saline Area Schools

### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Page 20 of 23

	Page	20 of 23			
Construction Mechanic Classific	ation	Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
Roadway Electrical					
Signal Tech, Communication Tech, Tower Tech, Fiber Optic	EC-17 WAGE 6/14/2016	\$37.60		\$75.20	ННННННОҮ
Splicer	6/14/2016 FRINGE	\$16.28	\$21.54	\$26.81	
Apprentice Rates:					
1st 6 months	WAGE	\$22.56	\$33.84	\$45.12	
	FRINGE	\$12.07	\$15.23	\$18.39	
2nd 6 months	WAGE FRINGE			\$48.88 \$19.43	
3rd 6 months	WAGE FRINGE			\$52.64 \$20.49	
4th 6 months	WAGE FRINGE			\$56.40 \$21.55	
5th 6 months	WAGE FRINGE			\$60.16 \$22.59	
6th 6 months	WAGE FRINGE			\$67.68 \$24.71	
Statewide					

Make up day allowed comment4 10s allowed M-Th; F make up dayOperator A - operates at least 3 of<br/>the following: backhoe,<br/>excavator, directional bore orEC-17AWAGE\$31.81\$47.72\$63.62H H H H H H H H D Y6/14/20166/14/20166/14/20166/14/2016\$23.57

boom/digger truck.

Road Builder Statewide

Make up day allowed comment

4 10s allowed M-Th; F make-up day

Official Request #: 25 Requestor: Saline Area Schools

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Conne			21 of 23	-1/12	2011	
Construction Mechanic Classificat		90	Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
Operator B - operates any 2 of the	EC-17B	WAGE	\$29.72		\$59.44	ннннннрү
following: backhoe, excavator, directional bore or boom/digger	6/14/2016 6/14/2016 <b>F</b>	RINGE	\$14.07	\$18.23	\$22.39	
truck						
Apprentice Rates:						
	F	WAGE RINGE	+	) \$0.00 \$0.00		
Road Builder Statewide						
Make up day allowed comment	4 10s allo	owed N	1-Th; F mal	ke up day		
Groundman	EC-17G 6/14/2016 F	WAGE RINGE	+		\$55.26 \$21.23	ННННННОҮ
Make up day allowed comment TRUCK DRIVERS TRUCK DRIVERS ZONE 1 EUCLID TYPE EQUIPMENT	4 10s allo TD1 6/6/2016 6/6/2016 F	WAGE		\$ \$37.73		НННННННҮ
RUCK DRIVERS ZONE 1 Genesee, Oakland, Macomb, Monroe, Livingston, Vashtenaw and Wayne						
comment	4 tens all For hours			k, the OT rat	te is \$38.23	per hour wages & fringes combine
TRUCK DRIVERS ZONE 2 EUCLID	TD2	WAGE				ннннннн
TYPE EQUIPMENT	6/6/2016 6/6/2016 <b>F</b>	RINGE	\$19.20	\$11.46		
TRUCK DRIVERS ZONE 2 The entire state except those counties listed in Zone : Genesee, Oakland, Macomb, Monroe, Livingston, Vashtenaw and Wayne						
comment	4 tens all For hours			k, the OT rat	te is \$38.08	per hour wages & fringes combine

Official Request #: 25 Requestor: Saline Area Schools

### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

	F	Page 2	2 of 23			
Construction Mechanic Classifica	tion	ę	Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
TRUCK DRIVERS ZONE 1 8 YARD CAPACITY OR GREATER	TD81 6/6/2016	WAGE	\$25.00	\$37.50		ннннннү
	6/6/2016 <b>FI</b>	RINGE	\$19.20	\$11.46		
TRUCK DRIVERS ZONE 1 Genesee, Oakland, Macomb, Monroe, Livingston, Washtenaw and Wayne						
comment	4 tens allo For hours			<, the OT rat	e is \$38 per	hour wages & fringes combined.
TRUCK DRIVERS ZONE 2 8 YARD CAPACITY OR GREATER	TD82 6/6/2016 6/6/2016 <b>FI</b>	WAGE	\$24.90 \$19.20			ннннннү
TRUCK DRIVERS ZONE 2 The entire state except those counties listed in Zone 1: Genesee, Oakland, Macomb, Monroe, Livingston, Washtenaw and Wayne						
comment	4 tens allo For hours			<, the OT rat	e is \$37.85	per hour wages & fringes combined.
TRUCK DRIVERS ZONE 1 ALL TRUCKS (except dump trucks of	6/6/2016	WAGE	\$24.90			ннннннн
8 cubic yard capacity or over, tandem axle trucks, transit mix and semis, euclid type equipment, double bottoms and low boys)	6/6/2016 <b>FI</b>	RINGE	\$19.20	\$11.46		
TRUCK DRIVERS ZONE 1 Genesee, Oakland, Macomb, Monroe, Livingston, Washtenaw and Wayne						
comment	4 tens allo	owed M	-F			

comment

4 tens allowed M-F For hours after 40 per week, the OT rate is \$37.85 per hour wages & fringes combined.

Official Request #: 25 Requestor: Saline Area Schools

#### Project Description: To Be Determined

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

#### Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Contra	act must be	e award	ed by:	4/12	/2017	
		Page 2	23 of 23			
Construction Mechanic Classifica	tion		Straight Time Rate	Time and One Half Rate	Double Time Rate	Overtime Code
TRUCK DRIVERS ZONE 2 All trucks (except dump trucks of 8 cubic yard capacity or over,	TD92 6/6/2016 6/6/2016	WAGE	\$24.80 \$19.20			ннннннн
tandem axle trucks, transit mix and semis, euclid type equipment, double bottoms and low boys)						
TRUCK DRIVERS ZONE 2 The entire state except those counties listed in Zone 1: Genesee, Oakland, Macomb, Monroe, Livingston, Washtenaw and Wayne						

comment

4 tens allowed M-F

For hours after 40 per week, the OT rate is \$37.70 per hour wages & fringes combined.

Official Request #: 25 Requestor: Saline Area Schools

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

#### State of Michigan

WHPWRequest@michigan.gov

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined Project Number: To Be Determined

#### Washtenaw County

#### Official 2017 Prevailing Wage Rates for State Funded Projects

Issue Date: 1/12/2017

Contract must be awarded by: 4/12/2017 Page 1 of 33 Last Straight Time and a Double Overtime **Classification** Name Description Updated Hourly Half Time Provision Asbestos & Lead Abatement Laborer Asbestos & Lead Abatement Laborer MLDC \$41.25 \$55.00 \$68.75 H H H X X X X D Y 4 ten hour days @ straight time allowed Monday-Saturday, must be consecutive 9/16/2016 Asbestos & Lead Abatement, Hazardous Material Handler Asbestos and Lead Abatement, Hazardous AS207 **\$67.75** H H H X X X X D Y \$40.75 \$54.25 Material Handler 4 ten hour days @ straight time allowed Monday-Saturday, must be consecutive 10/30/2015 Boilermaker Boilermaker BO169 \$54.70 \$81.08 \$107.45 Н Н Н Н Н Н Н D Y 2/17/2015 Apprentice Rates: 1st 6 months \$40.31 \$59.49 \$78.67 2nd 6 months \$41.45 \$61.21 \$80.95 \$83.19 3rd 6 months \$42.57 \$62.88 4th 6 months \$43.69 \$64.57 \$85.43 5th 6 months \$44.81 \$66.24 \$87.67 \$72.50 6th 6 months \$48.63 \$96.36 7th 6 months \$73.01 \$96.69 \$49.32 8th 6 months \$51.58 \$76.40 \$101.21

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### Official Rate Schedule

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 1 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 2 of 33

	Page 2	of 33				
Classification	-	Last	Straight T			Overtime
Name Description		Updated	Hourly	Half ======	Time ========	Provision
Bricklayer						
Bricklayer	BR9-14-BL		\$54.12	\$69.62	<b>\$85.12</b> H	НННННРҮ
Four 10s allowed M-F						
<i>Make up day allowed comment</i> Saturday		11/4/2016				
	Apprentice Rates:					
	0-749 hours		\$40.64	\$49.72	\$58.81	
	750-1499 hours		\$42.32	\$52.24	\$62.17	
	1500-2249 hours		\$44.01	\$54.77	\$65.55	
	2250-2999 hours		\$45.70	\$57.31	\$68.93	
	3000-3749 hours		\$47.38	\$59.83	\$72.29	
	3750-4499 hours		\$49.06	\$62.35	\$75.65	
	4500-5249 hours		\$50.75	\$64.89	\$79.03	
	5250-6000 hours		\$52.44	\$67.42	\$82.41	
Pointer, Cleaner, Caulker & Stone 4 10s allowed M-F comment	Mason BR9-14-S	11/4/2016	\$54.12	\$69.62	\$85.12 H	НННННОҮ
Saturday makeup day						
	Apprentice Rates:					
	0-749 hours		\$42.32	\$52.24	\$62.17	
	750-1499 hours		\$44.01	\$54.77	\$65.55	
	1500-2249 hours		\$45.70	\$57.31	\$68.93	
	2250-2999 hours		\$47.38	\$59.83	\$72.29	
	3000-3749 hours		\$49.06	\$62.35	\$75.65	
	3750-4499		\$50.75	\$64.89	\$79.03	
Carpenter						
Diver Four 10s allowed M-Sat; double t over 12 hours worked per day	CA 687 D ime due when		\$67.75	\$87.50	\$107.25 X	ХНХХННОҮ
<i>Make up day allowed comment</i> Saturday		8/12/2016				

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 2 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 3 of 33

			3 of 33				
<u>Clas</u> ame	<u>sification</u> Description		Last Updated	Straight T Hourly	ime and a Half	a Double Time	Overtime Provision
includ	t and Resilient Floor Layer, (does no e installation of prefabricated formic et flooring which is to be paid carper	a &		\$51.11	\$72.33	\$93.54 X	X H X X X X D <sup>·</sup>
	A	nontine Determ	7/26/2016				
		rentice Rates:		<b>\$05.44</b>	<b>*• • • •</b>	<b><i>Ф</i></b> 40 4 4	
		6 months		\$25.41	\$34.28	\$43.14	
		6 months		\$29.40	\$40.26	\$51.12	
		6 months		\$31.57	\$43.52	\$55.46	
		6 months		\$33.74	\$46.77	\$59.80	
		6 months		\$35.91	\$50.03	\$64.14	
		6 months 6 months		\$38.08	\$53.28	\$68.48	
		6 months		\$40.25 \$42.42	\$56.54 \$59.79	\$72.82 \$77.16	
				•	•	•	
Saturo 12 ho Make	nter-four 10s allowed Monday thru day; double time due anytime when urs worked per day e up day allowed comment	CA687Z2 over	8/12/2016	\$57.69	\$73.27	\$88.85 X	ХНХХННД
Saturo 12 ho Make	day; double time due anytime when urs worked per day e up day allowed comment rdays	over		\$57.69	\$73.27	\$88.85 X	ХНХХННО
Saturo 12 ho Make	day; double time due anytime when a urs worked per day e up day allowed comment rdays App						ХНХХННД
Saturo 12 ho Make	day; double time due anytime when a urs worked per day e up day allowed comment rdays App 1st 8	over		\$35.37	\$43.94	\$52.51	ХНХХННД
Saturo 12 ho Make	day; double time due anytime when a urs worked per day e up day allowed comment rdays App 1st a 3rd a	over rentice Rates: & 2nd 6 months		\$35.37 \$37.86	\$43.94 \$47.21	\$52.51 \$56.56	ХНХХННД
Saturo 12 ho Make	day; double time due anytime when a urs worked per day e up day allowed comment rdays App 1st & 3rd 4 4th (	over rentice Rates: & 2nd 6 months 6 months		\$35.37	\$43.94 \$47.21 \$50.45	\$52.51	ХНХХННД
Saturo 12 ho Make	day; double time due anytime when a urs worked per day e up day allowed comment rdays App 1st a 3rd 4th a 5th a	over rentice Rates: & 2nd 6 months 6 months 6 months		\$35.37 \$37.86 \$40.32	\$43.94 \$47.21	\$52.51 \$56.56 \$60.57	X H X X H H D
Saturo 12 ho Make	day; double time due anytime when a urs worked per day e up day allowed comment rdays App 1st & 3rd 4 4th 6 5th 6	over rentice Rates: & 2nd 6 months 6 months 6 months 6 months 6 months		\$35.37 \$37.86 \$40.32 \$42.81	\$43.94 \$47.21 \$50.45 \$53.72	\$52.51 \$56.56 \$60.57 \$64.62	X H X X H H D
Saturo 12 ho <i>Make</i>	day; double time due anytime when a urs worked per day e up day allowed comment rdays App 1st a 3rd 5th 6th 7th	over rentice Rates: & 2nd 6 months 6 months 6 months 6 months 6 months 6 months		\$35.37 \$37.86 \$40.32 \$42.81 \$45.29	\$43.94 \$47.21 \$50.45 \$53.72 \$56.98	\$52.51 \$56.56 \$60.57 \$64.62 \$68.66	X H X X H H D
Saturo 12 ho <i>Make</i> Satur	day; double time due anytime when a urs worked per day e up day allowed comment rdays App 1st a 3rd 5th 6th 7th	over rentice Rates: & 2nd 6 months 6 months 6 months 6 months 6 months 6 months 6 months 6 months 6 months	8/12/2016	\$35.37 \$37.86 \$40.32 \$42.81 \$45.29 \$47.77	\$43.94 \$47.21 \$50.45 \$53.72 \$56.98 \$60.24	\$52.51 \$56.56 \$60.57 \$64.62 \$68.66 \$72.70 \$76.75	X H X X H H D

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 3 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Contrac	t must be awarded by		4/12/2017			
<u>Classification</u> Name Description	Page 4	Of 33 Last Updated	Straight Ti Hourly	me and a Half	a Double Time	Overtime Provision
Cement Mason						
Cement Mason - Four 10s allowed M-	F BR9-14-CM	11/4/2016	\$50.52	\$65.28	\$80.04 X	хнхххн
	Apprentice Rates:					
	0-749 hours		\$39.30	\$48.75	\$58.21	
	750-1499 hours		\$40.91	\$51.17	\$61.43	
	1500-2249 hours		\$42.51	\$53.57	\$64.63	
	2250-2999 hours		\$44.11	\$55.97	\$67.83	
	3000-3749 hours		\$45.71	\$58.37	\$71.03	
	3750-4499 hours		\$47.32	\$60.79	\$74.25	
Cement Mason	CE514	8/17/2016	\$48.55	\$68.40	<b>\$88.24</b> H	Н
	Apprentice Rates:					
	1st 6 months		\$28.17	\$38.03	\$47.88	
	2nd 6 months		\$30.14	\$40.98	\$51.82	
	3rd 6 months		\$34.09	\$46.91	\$59.72	
	4th 6 months		\$38.03	\$52.82	\$67.60	
	5th 6 months		\$39.99	\$55.75	\$71.52	
	6th 6 months		\$43.94	\$61.68	\$79.42	
Drywall						
Drywall Taper Four 10s allowed Monday-Thursday	PT-22-D		\$45.91	\$59.74	<b>\$73.56</b> H	HDHDDD
Make up day allowed comment		8/25/2016				
Friday make-up day for bad weather o	r holidays Apprentice Rates:					
	First 3 months		\$32.08	\$38.99	\$45.90	
	Second 3 months		\$34.85	\$43.14	\$51.44	
	Second 6 months		\$37.62	\$47.30	\$56.98	
	Third 6 months		\$40.38	\$51.44	\$62.50	
	4th 6 months		\$41.76	\$53.51	\$65.26	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

DΥ

DΝ

DΥ

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 4 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 5 of 33

			Page 5	01 33				
Class Name	<u>sification</u> Description		_	Last Updated	Straight T Hourly	ime and Half =======	a Double Time	Overtime Provision
Electric	ian							
Inside	e wireman		EC-252-IW		\$63.41	\$85.59	\$107.77 H	HDHDDDN
				10/14/2016				
		Apprentice	Rates:					
		1st Period			\$36.22	\$44.81	\$53.39	
		2nd Period			\$41.23	\$52.32	\$63.41	
		3rd Period			\$45.66	\$58.97	\$72.28	
		4th Period			\$50.11	\$65.64	\$81.17	
		5th Period			\$54.54	\$72.28	\$90.03	
		6th Period			\$58.98	\$78.95	\$98.91	
Sound	and Communications Insta	aller/Technician	EC-252-SC	10/14/2016	\$42.26	\$56.72	\$71.17 H	Н
		Apprentice	Rates:					
		Period 1			\$23.94	\$31.18	\$38.40	
		Period 2			\$25.38	\$33.33	\$41.28	
		Period 3			\$26.83	\$35.51	\$44.18	
		Period 4			\$28.27	\$37.66	\$47.06	
		Period 5			\$33.58	\$43.70	\$53.81	
		Period 6			\$36.50	\$48.08	\$59.65	
		Period 7			\$39.38	\$52.40	\$65.41	
		Period 8			\$40.82	\$54.56	\$68.29	
Elevato	or Constructor							
Elevate	or Constructor Mechanic		EL-85		\$70.77		<b>\$116.32</b> D	DDDDDDY
	comment			4/8/2013				
4 tens	s allowed M-TH	Apprentice	Rates:					
		1st year			\$50.27		\$75.32	
		2nd year			\$54.83		\$84.44	
		3rd year			\$57.10		\$88.98	
		ora your			<i>\\\</i> 01.10		Ψ00.00	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 5 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

	Contro		Page 6	•	12/2011			
<u>Clas</u> Name	<u>sification</u> Description 			Last Updated	Straight T Hourly	ime and a Half	Double Time	Overtime Provision
Blazier								
Glazier			GL-357		\$49.50	\$65.23	<b>\$80.95</b> H	нннннр
	ur 10 hour day workweek is s							
TOUL IN	Os must be consecutive, M-F.			8/24/2016				
		Apprentice R	ates:					
		1st 6 months			\$33.77	\$41.63	\$49.49	
		2nd 6 months			\$35.35	\$44.00	\$52.65	
		3rd 6 months			\$38.49	\$48.71	\$58.93	
		4th 6 months			\$40.06	\$51.07	\$62.07	
		5th 6 months			\$41.64	\$53.43	\$65.23	
		6th 6 months			\$43.21	\$55.79	\$68.37	
		7th 6 months			\$44.78	\$58.15	\$71.51	
		8th 6 months			\$47.93	\$62.87	\$77.81	
Spray	Insulation		AS25S	6/2/2016	\$25.29	\$36.51	Х	ххннннн
loat an	d Frost Insulator and As	shastas Wark	or					
	nd Frost Insulators and Asbe		AS25		\$62.65	\$78.41	<b>\$94.16</b> H	нннннр
weeks All hou double Four differe	0s must be worked for a min consecutively, Monday thru irs worked in excess of 10 wi time. All hours worked on <i>comment</i> 10s must be worked for a minime ent on a four 10 week. OT is 2x re time and one half. Sat first 8	Thursday. Il be paid at the fifth day, um of 2 consecution for hours beyond	10. All ho	urs on fifth day				
		Apprentice R	ates:					
		1st Year			\$46.90	\$54.78	\$62.66	
		2nd Year			\$50.05	\$59.50	\$68.96	
		3rd Year			\$53.20	\$64.23	\$75.26	
		4th Year			\$56.35	\$68.96	\$81.56	
I	Request #: 51 Requestor: Saline Area Schools lescription: To Be Determined			on t	-	r and subo tion site, i	contractor and a conspic	ate Schedule shall keep postec cuous place, a co

Project Number: To Be Determined County: Washtenaw on the construction site, in a conspicuous place, a co of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 6 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

			Page 7	of 33				
<u>Cla</u> Name	<u>ssification</u> Description			Last Updated	Straight Ti Hourly	ime and a Half	a Double Time	Overtime Provision
<u>Sub</u>	division of county	Twps of Ann Arbor, Augu Scio, Superior, Webster,			l, Salem, Sa	line,		
Mał	and Frost Insulator <i>(e up day allowed cor</i> ay for cancelled work i		AS47	12/11/2015	\$45.25	\$59.86	<b>\$74.47</b> H	Н Н Н Н Н Н Д Ү
1110		Apprentice	Rates:					
		1st year			\$26.38	\$33.69	\$40.99	
		2nd year			\$30.15	\$38.92	\$47.68	
		3rd year			\$33.92	\$44.15	\$54.37	
		4th year			\$37.70	\$49.39	\$61.08	
		5th year			\$41.48	\$54.63	\$67.78	
<u>Sub</u>	division of county	Bridgewater, Dexter, Fre Sylvan	edom, Lima, L	indon, Manches	ster, Sharon	and		
erect Four	e, Sound Barrier & G ion/installation and	Guardrail Exterior Signage work may be worked during	IR-25-F1		\$35.55	\$47.55	\$59.55 X	ХНХХХНДҮ
WO	ady outdrudy.			3/11/2016				
		Apprentice	Rates:					
		60% Level			\$25.15	\$32.35	\$39.55	
		65% Level			\$26.45	\$34.25	\$42.05	
		70% Level			\$27.76	\$36.16	\$44.56	
		75% Level			\$29.05	\$38.05	\$47.05	
		80% Level			\$30.35	\$39.95	\$49.55	
		85% Level			\$31.65	\$41.85	\$52.05	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw Official Rate Schedule Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 7 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 8 of 33

		Page 8 d	of 33				
Classification ame Description		-	Last Updated	Straight T Hourly	ime and a Half	a Double Time	Overtime Provision
	==============			=======			======
Siding, Glazing, Curtain <sup>1</sup> 4 tens may be worked M		IR-25-GZ2		\$47.16	\$58.82	\$70.48 X	хннннор
@ straight time.	ionuay thru m	ursuay					
Make up day allowed co Friday	mment		6/5/2015				
		Apprentice Rates:					
		Level 1		\$30.23	\$36.84	\$43.43	
		Level 2		\$32.34	\$39.58	\$46.80	
		Level 3		\$34.46	\$42.33	\$50.19	
		Level 4		\$36.58	\$45.08	\$53.57	
		Level 5		\$38.69	\$47.82	\$56.95	
	l	Level 6		\$40.81	\$50.57	\$60.33	
Pre-engineered Metal W	ork	IR-25-PE-Z1		\$47.49	\$58.03	\$68.56 X	хнхххх
Make up day allowed comment			4/6/2016				
4 tens allowed M-Th with		up day Apprentice Rates:					
		1st Year		\$28.36	\$34.09	\$39.81	
	:	3rd 6 month period		\$30.48	\$36.95	\$43.43	
		4th 6 month period		\$32.61	\$39.84	\$47.08	
	:	5th 6 month period		\$34.73	\$42.72	\$50.70	
		6th 6 month period		\$36.86	\$45.60	\$54.34	
Subdivision of county	East of US-2	3 and the University of Micl	higan.				
Pre-engineered Metal W Make up day allowed co	mment	IR-25-PE-Z2	4/6/2016	\$45.27	\$55.20	\$65.12 X	ХНХХХХ
4 tens allowed M-Th with		up day Apprentice Rates:					
		1st Year		\$28.36	\$34.09	\$39.81	
	:	3rd 6 month period		\$30.48	\$36.95	\$43.43	
		4th 6 month period		\$32.61	\$39.84	\$47.08	
	:	5th 6 month period		\$34.73	\$42.72	\$50.70	
		6th 6 month period		\$36.86	\$45.60	\$54.34	
Subdivision of county	West of US 2	3 excluding the Univ of Mic	ch				

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 8 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 9 of 33

			Page 9	of 33				
<u>Clas</u> Name ======	<u>sification</u> Description 			Last Updated	Straight Ti Hourly	ime and Half	a Double Time	Overtime Provision
Reinfo	prced Iron Work a up day allowed		IR-25-RF	8/19/2016	\$56.98	\$85.28		————— H H D H D D D D N
		Apprentice	Rates:					
		Level 1			\$37.63	\$54.83	\$72.77	
		Level 2			\$40.00	\$58.38	\$77.51	
		Level 3			\$42.36	\$63.05	\$83.73	
		Level 4			\$44.90	\$66.86	\$88.81	
		Level 5			\$47.43	\$69.53	\$92.37	
		Level 6			\$49.97	\$74.46	\$98.95	
Riggin	g Work		IR-25-RIG	8/19/2016	\$62.83	\$93.95	\$125.06	Н Н Н Н Н Н Н D N
		Apprentice	Rates:					
		Level 1& 2			\$38.13	\$55.74	\$74.10	
		Level 3			\$40.96	\$61.12	\$81.26	
		Level 4			\$43.78	\$65.34	\$86.90	
		Level 5			\$46.61	\$69.59	\$92.56	
		Level 6			\$49.44	\$73.83	\$98.22	
@ stra a ma Monda Friday must l	ng may be worked Monday f aight time. If bad weathe ke up day. If holiday cele ay, 4 10s may be worked . Work in excess of 12 h be paid @ double time. a up day allowed comment	r, Friday may be ebrated on a Fuesday thru	IR-25-SD	8/19/2016	\$54.79	\$81.86	\$108.92	ХХННННООҮ
	y for 4 tens M-Th day for 5 eights M-F							

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 10 of 33

Classification Name Description			Last Updated	Straight Ti Hourly	me and Half	a Double Time	Overtime Provision
Structural, ornamental, welder and 4 tens may be worked Monday thru @ straight time. If bad weather, F a make up day. If holiday celebra Monday, 4 10s may be worked Tues Friday. Work in excess of 12 hours must be paid @ double time.	Thursday riday may be ted on a sday thru	IR-25-STR		\$62.96	\$94.11	\$125.26 H	і Н Н Н Н Н D D Y
Make up day allowed			8/19/2016				
	Apprentice	Rates:					
	Levels 1 & 2	2		\$38.13	\$56.87	\$75.60	
	Level 3			\$40.96	\$61.12	\$81.26	
	Level 4			\$43.78	\$65.34	\$86.90	
	Level 5			\$46.61	\$69.59	\$92.56	
	Level 6			\$49.44	\$73.83	\$98.22	
	Level 7			\$52.26	\$78.06	\$103.86	
	Level 8			\$55.09	\$82.31	\$109.52	
Industrial Door erection & construct Make up day allowed comment Friday for bad weather when 4 tens so tens may be worked T-F. Work in exe time.	heduled for M-1				\$64.25	\$85.45 ⊦	I Н Н Н Н Н D D Y
Laborer							
Laborer, demolition, drywall handle jobsite cleanup	rs, general	L499-A-A		\$42.30	\$56.46	\$70.62 ×	х х н н н н н р ү
<i>Make up day allowed comment</i> Saturday			10/24/2014				
	Apprentice	Rates:					
	0-1,000 hou	rs		\$36.27	\$47.42	\$58.56	
	1,001-2,000	hours		\$37.48	\$49.23	\$60.98	
	2,001-3,000	hours		\$38.68	\$51.03	\$63.38	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 10 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 11 of 33

		Page 11	of 33				
<u>Classification</u> ame Description			Last Updated	Straight T Hourly	ime and a Half	Time	Overtime Provision
Mortar mixer, material mixer, air, gas tool operator, power buggy operator, setter, tender, scaffold builder or disr winlass operator, tar & kettle operator	stone nantler,	L499-A-B		\$42.50	\$56.76	\$71.02 X	ХННННО
Make up day allowed comment Saturday			10/24/2014				
	Apprentice I	Rates:					
	0-1,000 hours	S		\$36.42	\$47.64	\$58.86	
	1,001-2,000 l	nours		\$37.64	\$49.47	\$61.30	
	2,001-3,000	nours		\$38.85	\$51.28	\$63.72	
	3,001-4,000	nours		\$41.28	\$54.93	\$68.58	
Jack hammering and chipping on con Make up day allowed comment Saturday	crete	L499-A-B2	10/24/2014	\$42.80	\$57.21	\$71.62 X	ХНННННО
	Apprentice I	Rates:					
	0-1,000 hours	S		\$36.64	\$47.97	\$59.30	
	1,001-2,000 l	nours		\$37.88	\$49.83	\$61.78	
	2,001-3,000	nours		\$39.11	\$51.68	\$64.24	
	3,001-4,000	nours		\$41.57	\$55.36	\$69.16	
Crock or pipe laborer, caisson worker Make up day allowed comment Saturday		L499-A-C	10/24/2014	\$42.62	\$56.94	\$71.26 X	ХНННННО
	Apprentice I	Rates:					
	0-1,000 hours	S		\$36.51	\$47.78	\$59.04	
	1,001-2,000 l	nours		\$37.73	\$49.60	\$61.48	
	2,001-3,000	nours		\$38.95	\$51.44	\$63.92	
	3,001-4,000	nours		\$41.40	\$55.11	\$68.82	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 11 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 12 of 33

	Page 12	2 of 33				
<u>Classification</u> ame Description		Last Updated	Straight T Hourly	ime and a Half	a Double Time	Overtime Provision
Watchmen, Civil Engineer Helpers, or Rodmen <i>Make up day allowed comment</i> Saturday	L499-A-D	10/24/2014	\$41.51	\$55.28	\$69.04 X	ХННННО
Apprentic	e Rates:					
0-1,000 ho	urs		\$35.68	\$46.53	\$57.38	
1,001-2,00	0 hours		\$36.84	\$48.27	\$59.70	
2,001-3,00	0 hours		\$38.01	\$50.02	\$62.04	
3,001-4,00	0 hours		\$40.34	\$53.52	\$66.70	
Final cleaning: washing or cleaning of walls, partitions, ceilings, windows, bathrooms, kitchens, laboratories and all fixtures and facilitie therein. Clean-up mopping, washing, waxing and polishing or dusting of all floors or areas.	L499-A-E		\$39.30	\$51.96	\$64.62 X	ХНННННО
Make up day allowed comment Saturday		10/24/2014				
Plasterer Tender, Plastering Machine Operator	LPT-2		\$45.18	\$64.31	<b>\$83.44</b> H	НННННО
If conditions beyond the employer/employee's control prevent one or more hours of working during Mon-Fri, the employer may choose to work up to 10 hour straight time weekdays. Work may be scheduled up to 10 hours per Mon Fri for the purpose of reaching 40 hours @ straight time. Make up days may also include 8 hours of work on Saturdays @ straight time. <i>Make up day allowed comment</i>		6/17/2016				
Saturday						
Apprentic						
0 - 1,000 h			\$38.77	\$54.70	\$70.62	
1,001 - 2,0	00 hours		\$40.30	\$56.99	\$73.68	
2,001 - 3,0	00 hours		\$41.52	\$58.82	\$76.12	
3,001 - 4,0	00 hours		\$43.96	\$62.48	\$81.00	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 12 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

	Page 13	of 33				
<u>Classification</u> Name Description	_	Last Updated	Straight Ti Hourly	me and a Half	a Double Time	Overtime Provision
Laborer - Hazardous						
Class A performing work in conjunction with s preparation and other preliminary work prior actual removal, handling, or containment of hazardous waste substances not requiring use personal protective equipment required by sta or federal regulations; or a laborer performing work in conjunction with the removal, handlin or containment of hazardous waste substance when use of personal protective equipment le "D" is required.	to e of hte g, es		\$42.30	\$60.51	\$78.72 H	НННННОҮ
Make up day allowed comment		11/7/2014				
4 10s allowed M-Th or T-F; inclement weather ma Appren	ikeup day Friday tice Rates:					
0-1,000	work hours		\$36.27	\$51.47	\$66.66	
1,001-2,	000 work hours		\$37.48	\$53.28	\$69.08	
2,001-3,	000 work hours		\$38.68	\$55.08	\$71.48	
3,001-4,	000 work hours		\$41.09	\$58.69	\$76.30	
Class B performing work in conjunction with removal, handling, or containment of hazard waste substances when the use of personal protective equipment levels "A", "B" or "C" is required.			\$43.30	\$62.01	\$80.72 H	Н Н Н Н Н Н Д Ү
Make up day allowed comment 4 10s allowed M-Th or T-F; inclement weather ma	keup day Friday <b>tice Rates:</b>	11/7/2014				
0-1,000	work hours		\$37.02	\$52.59	\$68.16	
1,001-2,	,000 work hours		\$38.28	\$54.48	\$70.68	
2,001-3,	,000 work hours		\$39.53	\$56.35	\$73.18	
3,001-4,	000 work hours		\$42.04	\$60.12	\$78.20	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 13 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

		Page 14	of 33				
<u>Cla</u> Name	<u>issification</u> Description ====================================	_	Last Updated	Straight Ti Hourly	me and a Half	a Double Time	Overtime Provision
Labor	er Underground - Tunnel, Shaft	& Caisson					
dum	I - Tunnel, shaft and caisson laborer, o man, shanty man, hog house tender ng man (on gas), and watchman.			\$35.67	\$47.07	\$58.47 X	X X X X X X D Y
			10/30/2014				
	Арр	rentice Rates:					
	0-1,0	000 work hours		\$30.52	\$39.35	\$48.17	
	1,00	1-2,000 work hours		\$31.55	\$40.90	\$50.23	
	2,00	1-3,000 work hours		\$32.58	\$42.44	\$52.29	
	3,00	1-4,000 work hours		\$34.64	\$45.53	\$56.41	
build	5 II - Manhole, headwall, catch basin er, bricklayer tender, mortar man, ma r, fence erector, and guard rail builder			\$35.76	\$47.21	\$58.65 X	X X X X X X D Y
			10/30/2014				
	Арр	rentice Rates:					
	0-1,0	000 work hours		\$30.58	\$39.44	\$48.29	
	1,00	1-2,000 work hours		\$31.62	\$41.00	\$50.37	
	2,00	1-3,000 work hours		\$32.66	\$42.56	\$52.45	
	3,00	1-4,000 work hours		\$34.72	\$45.65	\$56.57	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 14 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 15 of 33

		Faye 15	55				
<u>Clas</u> Name	ssification Description		Last Updated	Straight Ti Hourly	me and a Half	Time	Overtime Provision
bush botto car pu form laboro tool c welde pea g tende man, vibrat wag	III - Air tool operator (jack hammer man, hammer man and grinding man), first m man, second bottom man, cage tender, usher, carrier man, concrete man, concrete man, concrete repair man, cement invert er, cement finisher, concrete shoveler, eyor man, floor man, gasoline and electric operator, gunnite man, grout operator, er, heading dinky man, inside lock tender, gravel operator, pump man, outside lock er, scaffold man, top signal man, switch track man, tugger man, utility man, tor man, winch operator, pipe jacking man, on drill and air track operator and concrete operator (under 40 h.p.).	LAUCT-Z2-3		\$35.86	\$47.36	\$58.85 X X	XXXXXDY
			10/30/2014				
	Apprentice R	lates:					
	0-1,000 work	hours		\$30.66	\$39.56	\$48.45	
	1,001-2,000 v	vork hours		\$31.70	\$41.12	\$50.53	
	2,001-3,000 v	vork hours		\$32.74	\$42.68	\$52.61	
	3,001-4,000 v	vork hours		\$34.82	\$45.80	\$56.77	
brace	IV - Tunnel, shaft and caisson mucker, r man, liner plate man, long haul dinky r and well point man.	LAUCT-Z2-4		\$36.02	\$47.60	\$59.17 X X	X
			10/30/2014				
	Apprentice R	lates:					
	0-1,000 work	hours		\$30.78	\$39.74	\$48.69	
	1,001-2,000 v	vork hours		\$31.83	\$41.32	\$50.79	
	2,001-3,000 v	vork hours		\$32.88	\$42.89	\$52.89	

3,001-4,000 work hours

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

\$46.02 \$57.07

\$34.97

Page 15 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

	Contract must	Page 16		#/ 1 <i>2/2</i> 017			
<u>Clas</u> Name	sification Description		Last Updated	Straight Ti Hourly	me and a Half	a Double Time	Overtime Provision
runne opera	V - Tunnel, shaft and caisson miner, dr r, keyboard operator, power knife tor, reinforced steel or mesh man (e.g. nesh, steel mats, dowel bars)			\$36.28	\$47.99	\$59.69 X	 X X X X X X D Y
WIICT			10/30/2014				
	Appre	ntice Rates:					
	0-1,000	) work hours		\$30.98	\$40.04	\$49.09	
	1,001-2	2,000 work hours		\$32.04	\$41.63	\$51.21	
	2,001-3	3,000 work hours		\$33.10	\$43.22	\$53.33	
	3,001-4	4,000 work hours		\$35.22	\$46.40	\$57.57	
Class	VI - Dynamite man and powder man.	LAUCT-Z2-6	10/30/2014	\$36.59	\$48.45	\$60.31 X	ххххххрү
	Apprei	ntice Rates:					
	0-1,000	) work hours		\$31.21	\$40.38	\$49.55	
	1,001-2	2,000 work hours		\$32.28	\$41.99	\$51.69	
	2,001-3	3,000 work hours		\$33.36	\$43.61	\$53.85	
	3,001-4	4,000 work hours		\$35.51	\$46.84	\$58.15	
soddir gradir replac	VII - Restoration laborer, seeding, ng, planting, cutting, mulching and tops ng and the restoration of property such ing mail boxes, wood chips, planter bo agstones.	as		\$28.86	\$36.86	\$44.85 X	X X X X X X D Y
and n			10/30/2014				
	Appre	ntice Rates:					
	0-1,000	) work hours		\$25.41	\$31.68	\$37.95	
	1,001-2	2,000 work hours		\$26.10	\$32.72	\$39.33	
	2,001-3	3,000 work hours		\$26.79	\$33.76	\$40.71	

3,001-4,000 work hours

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

\$35.82 \$43.47

\$28.17

Page 16 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

	Contract mus	i be awalueu by		+/12/2017			
<u>Classific</u> Name De	<u>ation</u> escription	Page 17	of 33 Last Updated	Straight Ti Hourly	me and a Half	a Double Time	Overtime Provision
Landscap	e Laborer						
Landscape equipmen lawn sprir where see trimming,	e Specialist includes air, gas, and out operator, skidsteer (or equivaler and skiller installer on landscaping work eding, sodding, planting, cutting, backfilling, rough grading or ance of landscape projects occurs.	nt),		\$28.98	\$40.04	\$51.09 X	ХНХХХНДҮ
Sundays p at double	paid at time & one half. Holidays time.	paid					
			10/13/2015				
operator, material r sodding, r rough gra projects o Sundays r	baid at time & one half. Holidays	ling,		\$24.76	\$33.71	\$42.65 X	ХНХХХНДҮ
at double	time.		10/13/2015				
Marble Fir	nisher						
	nisher orkweek may be worked Monday sday or Tuesday thru Friday.	BR1-MF		\$43.48	\$54.29	\$65.10 H	H D H D D D D Y
			10/20/2014				
		entice Rates:					
	Level			\$19.04	\$25.12	\$31.20	
	Level			\$20.24	\$26.92	\$33.60	
	Level			\$27.01	\$33.96	\$40.90	
	Level			\$28.47	\$36.14	\$43.82	
	Level	-		\$29.99	\$37.84	\$45.70	
	Level			\$31.61	\$39.86	\$48.10	
	Level			\$33.30	\$41.59	\$49.87	
	Level	0		\$34.79	\$43.48	\$52.17	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 17 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

		Page 18	of 33				
<u>Classification</u> Name Description			Last Updated	Straight T Hourly	Half	a Double Time	Overtime Provision
Marble Mason							
Marble Mason A 4 ten workweek may be worked thru Thursday or Tuesday thru Fri		BR1-MM		\$50.29	\$64.51	\$78.72 H	Н
			10/17/2014				
	Apprentice R	ates:					
	Level 1			\$25.14	\$32.65	\$40.15	
	Level 2			\$28.20	\$36.49	\$44.78	
	Level 3			\$33.41	\$41.97	\$50.53	
	Level 4			\$36.15	\$45.66	\$55.17	
	Level 5			\$38.42	\$48.17	\$57.92	
	Level 6			\$42.07	\$53.56	\$65.05	
	Level 7			\$42.74	\$54.38	\$66.02	
	Level 8			\$43.67	\$55.78	\$67.88	
Millwright							
Millwright <i>Make up day allowed comment</i> Friday		M1102	7/31/2014	\$61.90	\$92.85	\$123.80 H	Н
	Apprentice R	ates:					
	1st 6 months			\$40.96	\$61.45	\$81.92	
	2nd 6 months			\$43.57	\$65.36	\$87.14	
	3rd 6 months			\$46.19	\$69.28	\$92.38	
	4th 6 months			\$48.81	\$73.22	\$97.62	
	5th 6 months			\$51.42	\$77.13	\$102.84	
	6th 6 months			\$54.03	\$81.05	\$108.06	
Operating Engineer							
Crane with boom & jib or leads 12 comment Double time after 12 hours M-F	0' or longer	EN-324-A120	7/28/2016	\$58.61	\$76.37	\$94.13 X	ХННООООҮ

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 18 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 19 of 33

	Page 19	of 33				
<u>Classification</u> Name Description 		Last Updated	Straight Ti Hourly	me and Half	a Double Time	Overtime Provision
Crane with boom & jib or leads 140' or longer	EN-324-A140		\$59.43	\$77.60	\$95.77 X	ХННООООҮ
Work in excess of 12 per day M-F shall be paid at double time.		7/28/2016				
Crane with boom & jib or leads 220' or longer Work in excess of 12 per day M-F shall be paid at double time.	EN-324-A220		\$59.73	\$78.05	\$96.37 X	ХННДДДДҮ
		7/28/2016				
Crane with boom & jib or leads 300' or longer Work in excess of 12 per day M-F shall be paid at double time.	EN-324-A300		\$61.23	\$80.30	\$99.37 X	ХННООООҮ
		7/28/2016				
Crane with boom & jib or leads 400' or longer Work in excess of 12 per day M-F shall be paid at double time.	EN-324-A400		\$62.73	\$82.55	\$102.37 X	ХННДДДДҮ
		7/28/2016				
Compressor or welding machine Work in excess of 12 per day M-F shall be paid at double time.	EN-324-CW		\$47.76	\$60.10	\$72.43 X	ХННДДДДҮ
		7/28/2016				
Forklift, lull, extend-a-boom forklift Work in excess of 12 per day M-F shall be paid at double time.	EN-324-FL		\$55.07	\$71.06	\$87.05 X	ХННООООҮ
		7/28/2016				
Fireman or oiler Work in excess of 12 per day M-F shall be paid at double time.	EN-324-FO		\$46.73	\$58.55	\$70.37 X	ХННДДДДҮ
		7/28/2016				

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 19 of 33

## Official 2017 Prevailing Wage Rates for State Funded Projects Issue Date: 1/12/2017

Contract must be awarded by: 4/12/2017 Page 20 of 33

	Page 20	of 33				
<u>Classification</u> Name Description	_	Last Update	Straight Ti d Hourly	me and Half	a Double Time	Overtime Provision
Regular crane, job mechanic, concrete pump with boom	EN-324-RC		\$57.75	\$75.08	\$92.41 >	К Х Н Н D D D Y
Work in excess of 12 per day M-F shall be paid at double time.						
		7/27/2010	5			
Regular engineer, hydro-excavator, remote controlled concrete breaker Work in excess of 12 per day M-F shall be paid at double time.	EN-324-RE		\$56.78	\$73.63	\$90.47    )	К Х Н Н D D D Y
		7/27/2010	5			
Apprentice	Rates:					
0-999 hours	i		\$46.35	\$58.48	\$70.61	
1,000-1,999	hours		\$48.09	\$61.10	\$74.09	
2,000-2,999	hours		\$49.82	\$63.68	\$77.55	
3,000-3,999	hours		\$51.55	\$66.28	\$81.01	
4,000-4,999	hours		\$53.29	\$68.90	\$84.49	
5,000-5,999	hours		\$55.01	\$71.47	\$87.93	
Operating Engineer - DIVER						
Diver/Wet Tender/Tender/Rov Pilot/Rov Tender	GLF D	4/2/2014	\$52.80	\$79.20	\$105.60 H	ННННННО N
<b>Operating Engineer - Marine Construction</b>						
Diver/Wet Tender, Engineer (hydraulic dredge)	GLF-1		\$65.00	\$84.85	\$104.70 >	К Х Н Н Н Н Н D Y
Make up day allowed		2/12/2014	1			
Subdivision of county all Great Lakes, islands t	herein & conn	ectina & t	ributary waters			
Crane/Backhoe Operator, 70 ton or over Tug Operator, Mechanic/Welder, Assistant Engineer (hydraulic dredge), Leverman (hydraulic dredge), Diver Tender	GLF-2		\$63.50	\$82.60	\$101.70 >	<pre>&lt; X H H H H H D Y</pre>
Holiday pay = \$120.80 per hour, wages & <i>Make up day allowed</i>		2/12/2014	1			
Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined Project Number: To Be Determined County: Statewide				r and sub tion site, wage an	contractor in a consp	Rate Schedule shall keep posted icuous place, a copy enefit rates
County. Clatowide					Pad	ge 20 of 33

Page 20 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

	Contract must be a	Page 21 (		4/12/2017			
<u>Classification</u> Name Description		1 age 21 (	Last Updated	Straight Tii Hourly	me and Half	a Double Time	Overtime Provision
<u>Subdivision of county</u>	All Great Lakes, islands th	nerein, & conne	cting & tri	butary waters			
Friction, Lattice Boom or ( Certification	Crane License	GLF-2B		\$64.50	\$84.10	\$103.70 X	ХНННННОҮ
Holiday pay = \$123.30 <i>Make up day allowed</i>			2/12/2014				
Subdivision of county	All Great Lakes, islands, t	herein, & conne	ecting & tr	ibutary waters			
Deck Equipment Operator Maintenance of Crane (ov Backhoe (115,000 lbs or r Operator, Loader, Dozer c Machinery	, Machineryman, er 50 ton capacity) or nore), Tug/Launch	GLF-3	U	\$59.30	\$76.30	\$93.30 X	ХНННННОҮ
Holiday pay = \$110.30 pe	r hour, wages &						
Make up day allowed			2/12/2014				
Subdivision of county	All Great Lakes, islands th	nerein, & conne	cting & tri	butary waters			
Deck Equipment Operator (Machineryman/Fireman), more), Off Road Trucks, E Engineer, & Crane Mainter and under or Backhoe 115 Assistant Tug Operator	(4 equipment units or Deck Hand, Tug nance 50 ton capacity	GLF-4		\$53.60	\$67.75	\$81.90 X	ХНННННОҮ
Holiday pay = \$96.05 pe <i>Make up day allowed</i>	r hour, wages & fringes		2/12/2014				
Subdivision of county	All Great Lakes, islands th	nerein. & conne	ctina & tri	butarv waters			
Operating Engineer Ste		- ,	5	,,			
Forklift, 1 Drum Hoist Make up day allowed com 4 10s allowed M-Th with Fr	ment	EN-324-ef	8/4/2016	\$59.66	\$78.12	\$96.58 H	Н
Crane w/ 120' boom or lo Make up day allowed com 4 10s allowed M-Th with Fr	ment	EN-324-SW120	) 8/5/2016	\$62.36	\$82.17	\$101.98 H	Н
Crane w/ 120' boom or lo Make up day allowed com 4 10s allowed M-Th with Fr	ment	EN-324-SW120	)-O 8/5/2016	\$63.36	\$83.67	\$103.98 H	Н
Official Request #: 51 Requestor: Saline Are Project Description: To Be Det			o		and sub	ocontractor in a conspi	ate Schedule shall keep posted cuous place, a copy nefit rates

Project Number: To Be Determined County: Washtenaw

of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 21 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 22 of 33

	Page 22 of 33				
<u>Classification</u> Name Description	Last Updated	Straight T Hourly	ime and Half	a Double Time	Overtime Provision
Crane w/ 140' boom or longer <i>Make up day allowed comment</i> 4 10s allowed M-Th with Friday makeup day beca	EN-324-SW140 8/5/2016 ause of bad weather	\$63.54	\$83.94	\$104.34 H	Н
Crane w/ 140' boom or longer W/ Oiler Make up day allowed comment 4 10s allowed M-Th with Friday makeup day beca	EN-324-SW140-O 8/5/2016 ause of bad weather	\$64.54	\$85.44	\$106.34 H	Н
Boom & Jib 220' or longer <i>Make up day allowed comment</i> 4 10s allowed M-Th with Friday makeup day beca	EN-324-SW220 8/5/2016 ause of bad weather	\$63.81	\$84.35	\$104.88 H	Н
Crane w/ 220' boom or longer w/ Oiler Make up day allowed comment 4 10s allowed M-Th with Friday makeup day beca	EN-324-SW220-O 8/5/2016 ause of bad weather	\$64.81	\$85.85	\$106.88 H	Н
Boom & Jib 300' or longer <i>Make up day allowed comment</i> 4 10s allowed M-Th with Friday makeup day beca	EN-324-SW300 8/5/2016 ause of bad weather	\$65.31	\$86.60	\$107.88 H	Н
Crane w/ 300' boom or longer w/ Oiler Make up day allowed comment 4 10s allowed M-Th with Friday makeup day beca	EN-324-SW300-O 8/5/2016 ause of bad weather	\$66.31	\$88.10	\$109.88 H	Н
Boom & Jib 400' or longer <i>Make up day allowed comment</i> 4 10s allowed M-Th with Friday makeup day beca	EN-324-SW400 8/5/2016 ause of bad weather	\$66.81	\$88.80	\$110.82 H	Н
Crane w/ 400' boom or longer w/ Oiler <i>Make up day allowed comment</i> 4 10s allowed M-Th with Friday makeup day beca	EN-324-SW400-O 8/5/2016 ause of bad weather	\$67.81	\$90.35	\$112.88 H	Н

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 22 of 33

## Official 2017 Prevailing Wage Rates for State Funded Projects Issue Date: 1/12/2017

Contract must be awarded by: 4/12/2017

	Straight Tir Hourly \$62.00 \$49.22	Half ======	Time 	Overtime Provision ====== H H D H H H D D
2016	\$49.22			ННОНННОО
2016		\$62.96		
		\$62.96		
		\$62.96		
	<b><b><b>(</b>)</b></b>		\$76.70	
	\$51.18	\$65.90	\$80.62	
	\$53.15	\$68.85	\$84.56	
	\$55.11	\$71.80	\$88.48	
	\$57.07	\$74.74	\$92.40	
	\$59.04	\$77.69	\$96.34	
2016	\$63.00	\$83.13	\$103.26	ННОНННОО
2016	\$54.55	\$70.46	\$86.36	Н Н Д Н Н Н Д Д
	\$61.36	\$80.67	\$99.98	ННДНННДД
2016				
	\$53.14	\$68.34	\$83.54	ННДНННДД
2016				
	\$63.09	\$83.27	\$103.44	нноннноо
2016				
2016	\$64.09	\$84.77	\$105.44	ННДНННДД
2	016 016 016	\$59.04 \$63.00 016 \$54.55 016 \$61.36 016 \$53.14 016 \$63.09 016 \$64.09	\$59.04 \$77.69 \$63.00 \$83.13 016 \$54.55 \$70.46 \$61.36 \$80.67 016 \$53.14 \$68.34 016 \$63.09 \$83.27 016 \$64.09 \$84.77	\$59.04 \$77.69 \$96.34 \$63.00 \$83.13 \$103.26 1 \$63.00 \$83.13 \$103.26 1 \$54.55 \$70.46 \$86.36 1 \$61.36 \$80.67 \$99.98 1 \$61.36 \$80.67 \$99.98 1 \$63.09 \$83.27 \$103.44 1 \$63.09 \$83.27 \$103.44 1 \$64.09 \$84.77 \$105.44 1

Project Number: To Be Determined County: Washtenaw

prescribed in a contract.

Page 23 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

	Page 24 of 33				
<u>Classification</u> Name Description	Last Updated	Straight T Hourly	ime and a Half	a Double Time	Overtime Provision
Operating Engineer Unde	erground				
Class I Equipment	EN-324A1-UC1 12/6/2016	\$54.54	\$70.33	<b>\$86.12</b> H	Н Н Н Н Н Н D Y
	Apprentice Rates:				
	0-999 hours	\$43.26	\$54.31	\$65.36	
	1,000-1,999 hours	\$44.84	\$56.68	\$68.52	
	2,000-2,999 hours	\$46.43	\$59.07	\$71.70	
	3,000-3,999 hours	\$48.00	\$61.42	\$74.84	
	4,000-4,999 hours	\$49.58	\$63.79	\$78.00	
	5,000-5,999 hours	\$51.16	\$66.17	\$81.16	
Class II Equipment	EN-324A1-UC2 12/6/2016	\$49.81	\$63.24	\$76.66 H	НННННРҮ
Class III Equipment	EN-324A1-UC3 12/6/2016	\$49.08	\$62.14	\$75.20 H	НННННОҮ
Class IV Equipment	EN-324A1-UC4 12/6/2016	\$48.51	\$61.29	\$74.06 H	НННННРҮ
Master Mechanic	EN-324A1-UMM 12/6/2016	\$54.79	\$70.71	\$86.62 H	НННННОҮ

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 24 of 33

## Official 2017 Prevailing Wage Rates for State Funded Projects Issue Date: 1/12/2017

4/12/2017 Contract must be awarded by:

Page 25 of 33

	Page 2	o of 33			
Classification Name Description		Last Updated	Straight T Hourly	ime and a Half	Double Overtime Time Provision
Painter					
Painter (8 hours of repaint work performed on Sunday shall be paid time & one half rate)	PT-22-P		\$44.32	\$57.60	\$70.88 H H D H D D D Y
Four 10s allowed Monday-Thursday with Friday makeup day if job down due to weather, holiday or other conditions beyond the control o the employer.					
Make up day allowed comment Fridays for bad weather or holidays		8/25/2016			
Apprentic	ce Rates:				
Year 1			\$31.04	\$37.68	\$44.32
Year 2			\$33.70	\$41.67	\$49.64
Year 3			\$36.35	\$45.64	\$54.94
Year 4			\$40.34	\$51.63	\$62.92
Fifth 6 mo	nths		\$38.36	\$48.79	\$59.21
Final 6 mo	onths		\$39.66	\$50.73	\$61.81
Pipe and Manhole Rehab					
General Laborer for rehab work or normal cleaning and cctv work-top man, scaffold man, CCTV assistant, jetter-vac assistant	TM247		\$28.20	\$38.20	нннннни
		4/17/2015			
Tap cutter/CCTV Tech/Grout Equipment Operator: unit driver and operator of CCTV; grouting equipment and tap cutting equipment	TM247-2		\$32.70	\$44.95	нннннни
		4/17/2015			
CCTV Technician/Combo Unit Operator: unit driver and operator of cctv unit or combo unit in connection with normal cleaning and televising work	TM247-3 n		\$31.45	\$43.07	нннннни
WORK		4/17/2015			
Boiler Operator: unit driver and operator of steam/water heater units and all ancillary equipment associated	TM247-4		\$33.20	\$45.70	нннннни
		4/17/2015			
Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined Project Number: To Be Determined		on of	the construc	r and sub tion site, i g wage and	fficial Rate Schedule contractor shall keep posted n a conspicuous place, a cop d fringe benefit rates
County: Statewide		•			Page 25 of 33

Page 25 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 26 of 33

		Page 26	of 33				
<u>Classification</u> Name Description		-	Last Updated	Straight Ti Hourly	ime and a Half	Double Time	Overtime Provision
Combo Unit driver & Jetter-Vac Operato	or	TM247-5	4/17/2015	\$33.20	\$45.70	Н	ННННННN
Pipe Bursting & Slip-lining Equipment C	)perator	TM247-6	4/17/2015	\$34.20	\$47.20	Н	нннннн
Plasterer							
Plasterer - Four 10s allowed M-F		BR9-14-PL	11/4/2016	\$48.65	\$62.73	<b>\$76.81</b> H	ннннн рү
	Apprentice R	ates:					
(	0-749 hours			\$37.94	\$46.95	\$55.97	
7	750-1499 hou	rs		\$39.47	\$49.25	\$59.03	
1	1500-2249 ho	urs		\$41.00	\$51.54	\$62.09	
2	2250-2999 ho	urs		\$42.53	\$53.83	\$65.15	
3	3000-3749 ho	urs		\$44.06	\$56.13	\$68.21	
3	3750-4499 ho	urs		\$45.59	\$58.43	\$71.27	
Plasterer		PL67	9/8/2010	\$44.72	\$60.11	<b>\$75.50</b> H	ННХDDDDN
	Apprentice R	ates:					
1	1st 6 months			\$29.33	\$37.02	\$44.72	
2	2nd 6 months			\$30.87	\$39.34	\$47.80	
3	3rd 6 months			\$32.41	\$41.64	\$50.88	
4	4th 6 months			\$35.49	\$46.26	\$57.04	
Ę	5th 6 months			\$38.56	\$51.16	\$63.76	
6	6th 6 months			\$41.64	\$55.49	\$69.34	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 26 of 33

# Official 2017 Prevailing Wage Rates for State Funded Projects Issue Date: 1/12/2017 Contract must be awarded by: 4/12/2017

Contrac	t must be av	Page 27		4/12/2017			
<u>Classification</u> Name Description			Last Updated	Straight T Hourly	ime and Half	a Double Time	Overtime Provision
Plumber, Pipefitter, Welder & HV	AC						
Plumber, Pipefitter, Welder & HVAC Make up day allowed comment 4 tens can be worked between Monda	y and Friday	PL-190	7/29/2016	\$60.50	\$84.12	\$107.74 ⊦	інннннр
	Apprentice R	ates:					
	1st Year			\$40.16	\$53.61	\$67.06	
	2nd Year			\$43.68	\$58.89	\$74.10	
	3rd Year			\$47.19	\$64.16	\$81.12	
	4th Year			\$50.71	\$69.44	\$88.16	
	5th Year-1st 6	6 months		\$52.47	\$72.08	\$91.68	
	5th Year-2nd	6 months		\$54.23	\$74.72	\$95.20	
Roofer							
Commercial Roofer 4 consecutive tens allowed M-TH 5 consecutive nine hour days M-F als time over forty hours per/week shall Sat makeup day		RO-70-Z6		\$46.35	\$59.91	\$73.47 ⊦	іннхнннD
Make up day allowed comment Friday or Saturday			10/9/2015				
,	Apprentice R	lates:					
	1st Class			\$28.49	\$35.27	\$42.05	
	2nd Class			\$30.26	\$37.72	\$45.18	
	3rd Class			\$32.00	\$40.13	\$48.27	
	4th Class			\$34.31	\$43.40	\$52.48	
	5th Class			\$36.33	\$46.23	\$56.13	
	6th Class			\$37.82	\$48.26	\$58.70	
	7th Class			\$39.30	\$50.29	\$61.27	
Sewer Relining							
Class I-Operator of audio visual CCT including remote in-ground cutter an equipment used in conjunction with	d other	SR-I		\$43.66	\$59.01	\$74.36 ⊦	I Н Н Н Н Н Н D
			11/24/2015				
Official Request #: 51			_				Rate Schedule
Requestor: Saline Area Schools Project Description: To Be Determined Project Number: To Be Determined			on of		tion site, g wage an	in a conspi	shall keep posted cuous place, a cop enefit rates
County: Statewide			•			Par	ne 27 of 33

Page 27 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 28 of 33

		Page 28	3 of 33				
<u>Cla</u> Name	<u>ssification</u> Description		Last Updated	Straight Ti Hourly	me and Half	a Double Time	Overtime Provision
circul and r	<ul> <li>II-Operator of hot water heaters and lation system; water jetters; and vacuum mechanical debris removal systems and e assisting.</li> </ul>	SR-II		\$42.13	\$56.72	\$71.30 H	 Н Н Н Н Н Н D N
	g.		11/24/2015				
Sheet	Metal Worker						
A 4 1	t Metal Worker 0 schedule may be worked, 4 consecutive Monday thru Friday.	SHM-80		\$64.50	\$82.20	\$99.89 H	Н
5	5		8/30/2016				
	Apprentice	Rates:					
	1st & 2nd F	eriods		\$40.99	\$48.95	\$56.92	
	3rd & 4th P	eriods		\$42.75	\$51.60	\$60.44	
	5th & 6th P	eriods		\$44.52	\$54.26	\$63.98	
	7th & 8th P	eriods		\$46.29	\$56.91	\$67.52	
Sprink	kler Fitter						
4 ten	kler Fitter hour days allowed Monday-Friday le time pay due after 12 hours worked M-F	SP 704		\$66.92	\$85.66	\$104.39 H	Н
			1/3/2017				
	Apprentice	Rates:					
	1st Period			\$29.27	\$37.92	\$46.57	
	2nd Period			\$43.13	\$52.86	\$62.59	
	3rd Period			\$45.29	\$56.10	\$66.91	
	4th Period			\$47.45	\$59.34	\$71.23	
	5th Period			\$49.62	\$62.59	\$75.57	
	6th Period			\$51.78	\$65.83	\$79.89	
				\$53.94	\$69.07	\$84.21	
	7th Period			φ <b>3</b> 3.94	ψ05.07	ψ04.21	
				\$56.10	\$72.31	\$88.53	
	7th Period						

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 28 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

		Page 29	of 33				
Classification Name Description			Last Updated	Straight Ti Hourly	ime and a Half	Double Time	Overtime Provision
Terrazzo							
Terrazzo Finisher A 4 ten workweek may be worked Mo thru Thursday or Tuesday thru Friday		BR1-TRF		\$43.97	\$55.03	\$66.08 H	Н
			10/17/2014				
	Apprentice	Rates:					
	Level 1			\$19.04	\$25.12	\$31.20	
	Level 2			\$20.24	\$26.92	\$33.60	
	Level 3			\$27.01	\$33.96	\$40.90	
	Level 4			\$28.47	\$36.14	\$43.82	
	Level 5			\$29.99	\$37.84	\$45.70	
	Level 6			\$31.61	\$39.86	\$48.10	
	Level 7			\$33.30	\$41.59	\$49.87	
	Level 8			\$34.79	\$43.48	\$52.17	
Terrazzo Worker A 4 ten workweek may be worked Mo thru Thursday or Tuesday thru Friday		BR1-TRW		\$49.73	\$63.67	\$77.60 H	HDHDDDV
			10/17/2014				
	Apprentice	Rates:					
	Level 1			\$25.14	\$32.65	\$40.15	
	Level 2			\$28.20	\$36.49	\$44.78	
	Level 3			\$33.41	\$41.97	\$50.53	
	Level 4			\$36.15	\$45.66	\$55.17	
	Level 5			\$38.42	\$48.17	\$57.92	
	Level 6			\$42.07	\$53.56	\$65.05	
	Level 7			\$42.74	\$54.38	\$66.02	
	Level 8			\$43.67	\$55.78	\$67.88	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw Official Rate Schedule Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 29 of 33

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

		Page 30	0 of 33				
Classification Name Descrip	· .		Last Updated	Straight Ti Hourly	ime and a Half	a Double Time	Overtime Provision
Tile							
Tile Finisher A 4 ten workw	veek may be worked Monday v or Tuesday thru Friday.	BR1-TF		\$43.50	\$54.32	\$65.14 H	H
			10/17/2014				
	Apprentic	e Rates:					
	Level 1			\$19.04	\$25.12	\$31.20	
	Level 2			\$20.24	\$26.92	\$33.60	
	Level 3			\$27.01	\$33.96	\$40.90	
	Level 4			\$28.47	\$36.14	\$43.82	
	Level 5			\$29.99	\$37.84	\$45.70	
	Level 6			\$31.61	\$39.86	\$48.10	
	Level 7			\$33.30	\$41.59	\$49.87	
	Level 8			\$34.79	\$43.48	\$52.17	
	veek may be worked Monday v or Tuesday thru Friday.	BR1-TL		\$49.68	\$63.59	\$77.50 H	Н
			10/17/2014				
	Apprentic	e Rates:					
	Level 1			\$25.14	\$32.65	\$40.15	
	Level 2			\$28.20	\$36.49	\$44.78	
	Level 3			\$33.41	\$41.97	\$50.53	
	Level 4			\$36.15	\$45.66	\$55.17	
	Level 5			\$38.42	\$48.17	\$57.92	
	Level 6			\$42.07	\$53.56	\$65.05	
	Level 7			\$42.74	\$54.38	\$66.02	
	Level 8			\$43.67	\$55.78	\$67.88	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw Official Rate Schedule Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 30 of 33
# Official 2017 Prevailing Wage Rates for State Funded Projects

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

Page 31 of 33

Classification Name Description	1 age 31	Last Updated	Straight T Hourly	ime and a Half	a Double Time	Overtime Provision
Truck Driver						
on all trucks of 8 cubic yard capacity or less (except dump trucks of 8 cubic yard capacity or over, tandem axle trucks, transit mix and semis, euclid type equipment, double bottoms and low boys)	TM-RB1		\$44.10	\$48.81	нн	ннннн
		6/7/2016				
of all trucks of 8 cubic yard capacity or over	TM-RB1A	6/7/2016	\$44.20	\$48.96	нн	ннннн
on euclid type equipment <i>Make up day allowed</i>	TM-RB1B	6/7/2016	\$44.35	\$49.19	нн	ннннн
Underground Laborer Open Cut, Class I						
Construction Laborer	LAUC-Z2-1	10/29/2014	\$35.42	\$46.65	\$57.87 X X	XXXXXD
Apprentic	e Rates:					
0-1,000 wo	ork hours		\$30.44	\$39.18	\$47.91	
1,001-2,00	0 work hours		\$31.43	\$40.66	\$49.89	
2,001-3,00	0 work hours		\$32.43	\$42.16	\$51.89	
3,001-4,00	0 work hours		\$34.42	\$45.15	\$55.87	
Underground Laborer Open Cut, Class II						
Mortar and material mixer, concrete form man, signal man, well point man, manhole, headwall and catch basin builder, guard rail builders, headwall, seawall, breakwall, dock builder and fence erector.	LAUC-Z2-2		\$35.53	\$46.81	\$58.09 X X	X X X X X D Y
		10/29/2014				
Apprentic	e Rates:					
0-1,000 wo	ork hours		\$30.52	\$39.30	\$48.07	
1,001-2,00	0 work hours		\$31.52	\$40.80	\$50.07	
2,001-3,00	0 work hours		\$32.52	\$42.30	\$52.07	
	0 work hours		\$34.53	\$45.32	\$56.09	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 31 of 33

# Official 2017 Prevailing Wage Rates for State Funded Projects

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

	Page 32 of 33							
<u>Clas</u> Name	<u>sification</u> Description			Last Updated	Straight Ti Hourly	me and a Half	Double Time	Overtime Provision
Underg	ground Laborer Open Cu	ut, Class III						
Air, ga operat brace (e.g. v cemen man, v concre	asoline and electric tool oper tor, drillers, pump man, tar H ers, rodder, reinforced steel wire mesh, steel mats, dowe nt finisher, welder, pipe jacki wagon drill and air track ope ete saw operator (under 40 H ugger man, and directional b	ator, vibrator Li kettle operator, or mesh man I bars, etc.), ing and boring grator and n.p.), windlass	AUC-Z2-3	10/20/2014	\$35.65	\$46.99	\$58.33 X X	X X X X X D Y
		Appropriate Date		10/29/2014				
		Apprentice Rate			\$30.61	\$39.44	\$48.25	
		0-1,000 work hou				\$39.44 \$40.95		
		1,001-2,000 worl			\$31.62 \$32.62		\$50.27 \$52.27	
		2,001-3,000 worl			\$32.62	\$42.45 \$45.49		
		3,001-4,000 worl	Knours		\$34.64	\$45.48	\$56.31	
Underc	ground Laborer Open Cu	ut. Class IV						
-	h or excavating grade man.		AUC-Z2-4	10/29/2014	\$35.72	\$47.10	\$58.47 X X	хххххлү
		Apprentice Rate	es:					
		0-1,000 work hou	urs		\$30.66	\$39.51	\$48.35	
		1,001-2,000 worl	k hours		\$31.67	\$41.02	\$50.37	
		2,001-3,000 worl	k hours		\$32.68	\$42.54	\$52.39	
		3,001-4,000 worl	k hours		\$34.71	\$45.58	\$56.45	
Underg	ground Laborer Open Cu	ut, Class V						
Pipe L	ayer	L	AUC-Z2-5	10/29/2014	\$35.87	\$47.32	\$58.77 X X	X X X X X X D Y
		Apprentice Rate	es:					
		0-1,000 work hou	urs		\$30.77	\$39.68	\$48.57	
		1,001-2,000 worl	k hours		\$31.79	\$41.20	\$50.61	
		2,001-3,000 wor	k hours		\$32.81	\$42.74	\$52.65	
		3,001-4,000 worl			\$34.85	\$45.80	\$56.73	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

### **Official Rate Schedule**

Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 32 of 33

# Official 2017 Prevailing Wage Rates for State Funded Projects

**Issue Date:** 1/12/2017

Contract must be awarded by: 4/12/2017

		Page 33	of 33				
<u>Classification</u> Name Description			Last Updated	Straight Ti Hourly	me and a Half	Double Time	Overtime Provision ======
Underground Laborer Open Cut,	Class VI						
Grouting man, top man assistant, aud television operations and all other op connection with closed circuit televisi- inspection, pipe cleaning and pipe rel and the installation and repair of wat pipe and appurtenances.	erations in on ining work	AUC-Z2-6		\$33.17	\$43.27	\$53.37 X	X X X X X X D Y
			10/29/2014				
	Apprentice Rat	es:					
	0-1,000 work ho	ours		\$28.75	\$36.64	\$44.53	
	1,001-2,000 wo	rk hours		\$29.63	\$37.96	\$46.29	
	2,001-3,000 wo	rk hours		\$30.52	\$39.30	\$48.07	
	3,001-4,000 wor	rk hours		\$32.29	\$41.96	\$51.61	
Underground Laborer Open Cut,	Class VII						
Restoration laborer, seeding, sodding cutting, mulching and topsoil grading restoration of property such as replac boxes, wood chips, planter boxes, fla	and the ing mail	AUC-Z2-7		\$29.81	\$38.23	\$46.65 X	X X X X X X D Y
			10/29/2014				
	Apprentice Rat	es:					
	0-1,000 work ho	ours		\$26.23	\$32.86	\$39.49	
	1,001-2,000 wo	rk hours		\$26.94	\$33.93	\$40.91	
	2,001-3,000 wo	rk hours		\$27.66	\$35.01	\$42.35	
	3,001-4,000 wo	rk hours		\$29.09	\$37.16	\$45.21	

Official Request #: 51 Requestor: Saline Area Schools Project Description: To Be Determined

Project Number: To Be Determined County: Washtenaw

#### Official Rate Schedule Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing wage and fringe benefit rates prescribed in a contract.

Page 33 of 33

# **Complaint Investigation Non-Compliance Summary Report**

### **Closed with Violations**

Date Produced 01/01/2017

### **ER/CA Failure to provide records**

Employer	Address	Claim	Date
GREAT LAKES ELEVATOR	530 E GRAND RIVER RD WILLIAMSTON, MI 48895	182507	10/06/2015
MOSS TELECOMMUNICATIONS	561 CENTURY SW GRAND RAPIDS, MI	182798	06/25/2015
SERVICES	49503	182796	06/25/2015
		182795	09/17/2015
NEW CREATIONS GROUP PAINTING	1061 TREVOR PLACE DETROIT, MI 48207	172288	04/30/2014
LLC		172285	04/30/2014
	GREAT LAKES ELEVATOR MOSS TELECOMMUNICATIONS SERVICES NEW CREATIONS GROUP PAINTING	GREAT LAKES ELEVATOR530 E GRAND RIVER RD WILLIAMSTON, MI 48895MOSS TELECOMMUNICATIONS SERVICES561 CENTURY SW GRAND RAPIDS, MI 49503NEW CREATIONS GROUP PAINTING1061 TREVOR PLACE DETROIT, MI 48207	GREAT LAKES ELEVATOR530 E GRAND RIVER RD WILLIAMSTON, MI182507MOSS TELECOMMUNICATIONS SERVICES561 CENTURY SW GRAND RAPIDS, MI182798MOSS TELECOMMUNICATIONS SERVICES182796182795NEW CREATIONS GROUP PAINTING1061 TREVOR PLACE DETROIT, MI 48207172288

# **Complaint Investigation Non-Compliance Summary Report**

### **Closed with Violations**

Date Produced 01/01/2017

### **Contractor Violation Unpaid**

Violations	Employer	Address	Claim	Date
2	AAA ELECTRICAL SERVICES INC	5462 ORO DR SHELBY TWP, MI 48315	177031	09/04/2014
			177030	09/04/2014
1	ANDY J EGAN COMPANY	2001 WALDORF NW GRAND RAPIDS, MI 49544	167938	09/25/2014
1	BEAL CONSTRUCTION SERVICES INC	221 FELCH ST STE 11 ANN ARBOR, MI 48103	183806	09/09/2015
1	GREAT LAKES ELEVATOR	530 E GRAND RIVER RD WILLIAMSTON, MI 48895	182508	07/08/2015
1	MOSS AUDIO CORPORATION	561 CENTURY AVE SW GRAND RAPIDS, MI 49503	178867	02/10/2015
2	PENINSULA INSULATION	833 N AL MOSES RD LAKE CITY, MI 49651	173524	01/15/2014
			172748	12/11/2013
1	SOUTHWEST TRANSPORT	137 HILLSBOROUGH ST HARTFORD, MI 49057	186028	08/10/2016



### **STATE OF MICHIGAN**

### Wage and Hour Division PO Box 30476 Lansing, MI 48909 517-284-7800 Informational Sheet: Prevailing Wages on State Projects

### REQUIREMENTS OF THE PREVAILING WAGES ON STATE PROJECTS ACT, PUBLIC ACT 166 OF 1965

The State of Michigan determines prevailing rates pursuant to the Prevailing Wages on State Projects Act, Public Act 166 of 1965, as amended. The purpose of establishing prevailing rates is to provide minimum rates of pay that must be paid to workers on construction projects for which the state or a school district is the contracting agent and which is financed or financially supported by the state. By law, prevailing rates are compiled from the rates contained in collectively bargained agreements which cover the locations of the state projects. The official prevailing rate schedule provides an hourly rate which includes wage and fringe benefit totals for designated construction mechanic classifications. The overtime rates also include wage and fringe benefit totals. Please pay special attention to the overtime and premium pay requirements. Prevailing wage is satisfied when wages plus fringe benefits paid to a worker are equal to or greater than the required rate.

### State of Michigan responsibilities under the law:

The department establishes the prevailing rate for each classification of construction mechanic requested by a contracting agent prior to contracts being let out for bid on a state project.

### Contracting agent responsibilities under the law:

- If a contract is not awarded or construction does not start within 90 days of the date of the issuance of rates, a redetermination of rates must be requested by the contracting agent.
- Rates for classifications needed but not provided on the Prevailing Rate Schedule, *must* be obtained *prior* to contracts being let out for bid on a state project.
- The contracting agent, by written notice to the contractor and the sureties of the contractor known to the contracting
  agent, may terminate the contractor's right to proceed with that part of the contract, for which less than the prevailing
  rates have been or will be paid, and may proceed to complete the contract by separate agreement with another
  contractor or otherwise, and the original contractor and his sureties shall be liable to the contracting agent for any
  excess costs occasioned thereby.

### Contractor responsibilities under the law:

- Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all prevailing rates prescribed in a contract.
- Every contractor and subcontractor shall keep an accurate record showing the name and occupation of and the actual
  wages and benefits paid to each construction mechanic employed by him in connection including certified payroll, as
  used in the industry, with said contract. This record shall be available for reasonable inspection by the contracting
  agent or the department.
- Each contractor or subcontractor is separately liable for the payment of the prevailing rate to its employees.
- The prime contractor is responsible for advising all subcontractors of the requirement to pay the prevailing rate prior to commencement of work.
- The prime contractor is secondarily liable for payment of prevailing rates that are not paid by a subcontractor.
- A construction mechanic *shall only* be paid the apprentice rate if registered with the United States Department of Labor, Bureau of Apprenticeship and Training and the rate is included in the contract.

### **Enforcement:**

A person who has information of an alleged prevailing wage violation on a state project may file a complaint with the State of Michigan. The department will investigate and attempt to resolve the complaint informally. During the course of an investigation, if the requested records and posting certification are not made available in compliance with Section 5 of Act 166, the investigation will be concluded and a referral to the Office of Attorney General for civil action will be made. The Office of Attorney General will pursue costs and fees associated with a lawsuit if filing is necessary to obtain records.



### **STATE OF MICHIGAN**

### Wage and Hour Division PO Box 30476 Lansing, MI 48909 517-284-7800 Informational Sheet: Prevailing Wages on State Projects

### **General Information Regarding Fringe Benefits**

Certain fringe benefits may be credited toward the payment of the Prevailing Wage Rate:

- o If a fringe benefit is paid directly to a construction mechanic
- o If a fringe benefit contribution or payment is made on behalf of a construction mechanic
- If a fringe benefit, which may be provided to a construction mechanic, is pursuant to a written contract or policy
- o If a fringe benefit is paid into a fund, for a construction mechanic

When a fringe benefit is not paid by an hourly rate, the hourly credit will be calculated based on the annual value of the fringe benefit divided by 2080 hours per year (52 weeks @ 40 hours per week).

The following is an example of the types of fringe benefits allowed and how an hourly credit is calculated:

Vacation	40 hours X \$14.00 per hour = \$560/2080 =	\$.27
Dental insurance	\$31.07 monthly premium X 12 mos. = \$372.84 /2080 =	\$.18
Vision insurance	\$5.38 monthly premium X 12 mos. = \$64.56/2080 =	\$.03
Health insurance	\$230.00 monthly premium X 12 mos. = \$2,760.00/2080 =	\$1.33
Life insurance	\$27.04 monthly premium X 12 mos. = \$324.48/2080 =	\$.16
Tuition	\$500.00 annual cost/2080 =	\$.24
Bonus	4 quarterly bonus/year x \$250 = \$1000.00/2080 =	\$.48
401k Employer Contribution	\$2000.00 total annual contribution/2080 =	\$.96
Total Hourly Credit		

Other examples of the types of fringe benefits allowed:

- Sick pay
- Holiday pay
- Accidental Death & Dismemberment insurance premiums

The following are examples of items that will not be credited toward the payment of the Prevailing Wage Rate

- Legally required payments, such as:
  - Unemployment Insurance payments
  - Workers' Compensation Insurance payments
  - FICA (Social Security contributions, Medicare contributions)
  - Reimbursable expenses, such as:
    - Clothing allowance or reimbursement
    - Uniform allowance or reimbursement
    - Gas allowance or reimbursement
    - Travel time or payment
    - Meals or lodging allowance or reimbursement
    - Per diem allowance or payment
  - Other payments to or on behalf of a construction mechanic that are not wages or fringe benefits, such as:
    - Industry advancement funds
    - Financial or material loans

\$3.65



### **State of Michigan**

### OVERTIME PROVISIONS for MICHIGAN PREVAILING WAGE RATE COMMERCIAL SCHEDULE

1. Overtime is represented as a nine character code. Each character represents a certain period of time after the first 8 hours Monday thru Friday.

	Monday thru Friday	Saturday	Sunday & Holidays	Four 10s
First 8 Hours		4		
9th Hour	1	5	8	-
10th Hour	2	6		9
Over 10 hours	3	7		

Overtime for Monday thru Friday after 8 hours:

the 1st character is for time worked in the 9th hour (8.1 - 9 hours)

the 2nd character is for time worked in the 10th hour (9.1 - 10 hours)

the 3rd character is for time worked beyond the 10th hour (10.1 and beyond)

### Overtime on Saturday:

the 4th character is for time worked in the first 8 hours on Saturday (0 - 8 hours) the 5th character is for time worked in the 9th hour on Saturday (8.1 - 9 hours) the 6th character is for time worked in the 10th hour (9.1 - 10 hours) the 7th character is for time worked beyond the 10th hour (10.01 and beyond)

### Overtime on Sundays & Holidays

The 8th character is for time worked on Sunday or on a holiday

### Four Ten Hour Days

The 9th character indicates if an optional 4-day 10-hour per day workweek can be worked **between Monday and** *Friday* without paying overtime after 8 hours worked, unless otherwise noted in the rate schedule. To utilize a 4 ten workweek, notice is required from the employer to employee prior to the start of work on the project.

- 2. Overtime Indicators Used in the Overtime Provision:
  - H means TIME AND ONE-HALF due
  - X means TIME AND ONE-HALF due after 40 HOURS worked
  - D means DOUBLE PAY due
  - Y means YES an optional 4-day 10-hour per day workweek can be worked without paying overtime after 8 hours worked
  - N means NO an optional 4-day 10-hour per day workweek *can not* be worked without paying overtime after 8 hours worked
- 3. EXAMPLES:

HHHHHHDN - This example shows that the 1½ rate must be used for time worked after 8 hours Monday thru Friday (*characters 1 - 3*); for all hours worked on Saturday, 1½ rate is due (*characters 4 - 7*). Work done on Sundays or holidays must be paid double time (*character 8*). The N (*character 9*) indicates that 4 ten-hour days is not an acceptable workweek at regular pay.

XXXHHHHDY - This example shows that the  $1\frac{1}{2}$  rate must be used for time worked after 40 hours are worked Monday thru Friday (*characters 1-3*); for hours worked on Saturday,  $1\frac{1}{2}$  rate is due (*characters 4 - 7*). Work done on Sundays or holidays must be paid double time (*character 8*). The Y (*character 9*) indicates that 4 ten-hour days is an acceptable alternative workweek.

LARA is an equal opportunity employer. Auxiliary aids, services and other reasonable accommodations are available, upon request, to individuals with disabilities.

> Wage and Hour Division P.O. BOX 30476 • LANSING, MICHIGAN 48909 www.michigan.gov/wagehour • Phone : (517) 284-7800

### **UNDERGROUND ENGINEERS**

### CLASS I

Backfiller Tamper, Backhoe, Batch Plant Operator, Clam-Shell, Concrete Paver (2 drums or larger), Conveyor Loader (Euclid type), Crane (crawler, truck type or pile driving), Dozer, Dragline, Elevating Grader, End Loader, Gradall (and similar type machine), Grader, Power Shovel, Roller (asphalt), Scraper (self propelled or tractor drawn), Side Broom Tractor (type D-4 or larger), Slope Paver, Trencher (over 8' digging capacity), Well Drilling Rig, Mechanic, Slip Form Paver, Hydro Excavator.

### CLASS II

Boom Truck (power swing type boom), Crusher, Hoist, Pump (1 or more 6" discharge or larger gas or diesel powered by generator of 300 amps or more, inclusive of generator), Side Boom Tractor (smaller than type D-4 or equivalent), Tractor (pneu-tired, other than backhoe or front end loader), Trencher (8' digging capacity and smaller), Vac Truck.

### CLASS III

Air Compressors (600 cfm or larger), Air Compressors (2 or more less than 600 cfm), Boom Truck (non-swinging, non-powered type boom), Concrete Breaker (self-propelled or truck mounted, includes compressor), Concrete Paver (1 drum, ½ yard or larger), Elevator (other than passenger), Maintenance Man, Mechanic Helper, Pump (2 or more 4" up to 6" discharge, gas or diesel powered, excluding submersible pump), Pumpcrete Machine (and similar equipment), Wagon Drill Machine, Welding Machine or Generator (2 or more 300 amp or larger, gas or diesel powered).

### CLASS IV

Boiler, Concrete Saw (40HP or over), Curing Machine (self-propelled), Farm Tractor (w/attachment), Finishing Machine (concrete), Firemen, Hydraulic Pipe Pushing Machine, Mulching Equipment, Oiler (2 or more up to 4", exclude submersible), Pumps (2 or more up to 4" discharge if used 3 hrs or more a day-gas or diesel powered, excluding submersible pumps), Roller (other than asphalt), Stump Remover, Vibrating Compaction Equipment (6' wide or over), Trencher (service) Sweeper (Wayne type and similar equipment), Water Wagon, Extend-a-Boom Forklift.

### HAZARDOUS WASTE ABATEMENT ENGINEERS

### CLASS I

Backhoe, Batch Plant Operator, Clamshell, Concrete Breaker when attached to hoe, Concrete Cleaning Decontamination Machine Operator, Concrete Pump, Concrete Paver, Crusher, Dozer, Elevating Grader, Endloader, Farm Tractor (90 h.p. and higher), Gradall, Grader, Heavy Equipment Robotics Operator, Hydro Excavator, Loader, Pug Mill, Pumpcrete Machines, Pump Trucks, Roller, Scraper (self-propelled or tractor drawn), Side Boom Tractor, Slip Form Paver, Slope Paver, Trencher, Ultra High Pressure Waterjet Cutting Tool System Operator, Vactors, Vacuum Blasting Machine Operator, Vertical Lifting Hoist, Vibrating Compaction Equipment (self-propelled), and Well Drilling Rig.

### CLASS II

Air Compressor, Concrete Breaker when not attached to hoe, Elevator, End Dumps, Equipment Decontamination Operator, Farm Tractor (less than 90 h.p.), Forklift, Generator, Heater, Mulcher, Pigs (Portable Reagent Storage Tanks), Power Screens, Pumps (water), Stationary Compressed Air Plant, Sweeper, Water Wagon and Welding Machine.



### STATE OF MICHIGAN

### Wage and Hour Division PO Box 30476 Lansing, MI 48909 517-284-7800

### Informational Sheet: Prevailing Wages on State Projects

### **REQUIREMENTS OF**

### THE PREVAILING WAGES ON STATE PROJECTS ACT, PUBLIC ACT 166 OF 1965

The State of Michigan determines prevailing rates pursuant to the Prevailing Wages on State Projects Act, Public Act 166 of 1965, as amended. The purpose of establishing prevailing rates is to provide minimum rates of pay that must be paid to workers on construction projects for which the state or a school district is the contracting agent and which is financed or financially supported by the state. By law, prevailing rates are compiled from the rates contained in collectively bargained agreements which cover the locations of the state projects. The official prevailing rate schedule provides an hourly rate which includes wage and fringe benefit totals for designated construction mechanic classifications. The overtime rates also include wage and fringe benefit totals. Please pay special attention to the overtime and premium pay requirements. Prevailing wage is satisfied when wages plus fringe benefits paid to a worker are equal to or greater than the required rate.

#### State of Michigan responsibilities under the law:

 The department establishes the prevailing rate for each classification of construction mechanic requested by a contracting agent prior to contracts being let out for bid on a state project.

#### Contracting agent responsibilities under the law:

- If a contract is not awarded or construction does not start within 90 days of the date of the issuance of rates, a redetermination of rates must be requested by the contracting agent.
- Rates for classifications needed but not provided on the Prevailing Rate Schedule, *must* be obtained *prior* to contracts being let out for bid on a state project.
- The contracting agent, by written notice to the contractor and the sureties of the contractor known to the contracting
  agent, may terminate the contractor's right to proceed with that part of the contract, for which less than the prevailing
  rates have been or will be paid, and may proceed to complete the contract by separate agreement with another
  contractor or otherwise, and the original contractor and his sureties shall be liable to the contracting agent for any
  excess costs occasioned thereby.

### Contractor responsibilities under the law:

- Every contractor and subcontractor shall keep posted on the construction site, in a conspicuous place, a copy of all
  prevailing rates prescribed in a contract.
- Every contractor and subcontractor shall keep an accurate record showing the name and occupation of and the actual
  wages and benefits paid to each construction mechanic employed by him in connection including certified payroll, as
  used in the industry, with said contract. This record shall be available for reasonable inspection by the contracting
  agent or the department.
- Each contractor or subcontractor is separately liable for the payment of the prevailing rate to its employees.
- The prime contractor is responsible for advising all subcontractors of the requirement to pay the prevailing rate prior to commencement of work.
- The prime contractor is secondarily liable for payment of prevailing rates that are not paid by a subcontractor.
- A construction mechanic *shall only* be paid the apprentice rate if registered with the United States Department of Labor, Bureau of Apprenticeship and Training and the rate is included in the contract.

#### Enforcement:

A person who has information of an alleged prevailing wage violation on a state project may file a complaint with the State of Michigan. The department will investigate and attempt to resolve the complaint informally. During the course of an investigation, if the requested records and posting certification are not made available in compliance with Section 5 of Act 166, the investigation will be concluded and a referral to the Office of Attorney General for civil action will be made. The Office of Attorney General will pursue costs and fees associated with a lawsuit if filing is necessary to obtain records.



### **STATE OF MICHIGAN**

### Wage and Hour Division PO Box 30476 Lansing, MI 48909 517-284-7800 Informational Sheet: Prevailing Wages on State Projects

### **General Information Regarding Fringe Benefits**

Certain fringe benefits may be credited toward the payment of the Prevailing Wage Rate:

- o If a fringe benefit is paid directly to a construction mechanic
- o If a fringe benefit contribution or payment is made on behalf of a construction mechanic
- If a fringe benefit, which may be provided to a construction mechanic, is pursuant to a written contract or policy
- o If a fringe benefit is paid into a fund, for a construction mechanic

When a fringe benefit is not paid by an hourly rate, the hourly credit will be calculated based on the annual value of the fringe benefit divided by 2080 hours per year (52 weeks @ 40 hours per week).

The following is an example of the types of fringe benefits allowed and how an hourly credit is calculated:

Vacation	40 hours X \$14.00 per hour = \$560/2080 =	\$.27
Dental insurance	\$31.07 monthly premium X 12 mos. = \$372.84 /2080 =	\$.18
Vision insurance	\$5.38 monthly premium X 12 mos. = \$64.56/2080 =	\$.03
Health insurance	\$230.00 monthly premium X 12 mos. = \$2,760.00/2080 =	\$1.33
Life insurance	\$27.04 monthly premium X 12 mos. = \$324.48/2080 =	\$.16
Tuition	\$500.00 annual cost/2080 =	\$.24
Bonus	4 quarterly bonus/year x \$250 = \$1000.00/2080 =	\$.48
401k Employer Contribution	\$2000.00 total annual contribution/2080 =	\$.96

Total Hourly Credit

\$3.65

Other examples of the types of fringe benefits allowed:

- Sick pay
- Holiday pay
- Accidental Death & Dismemberment insurance premiums

The following are examples of items that will not be credited toward the payment of the Prevailing Wage Rate

- o Legally required payments, such as:
  - Unemployment Insurance payments
  - Workers' Compensation Insurance payments
  - FICA (Social Security contributions, Medicare contributions)
- Reimbursable expenses, such as:
  - Clothing allowance or reimbursement
  - Uniform allowance or reimbursement
  - Gas allowance or reimbursement
  - Travel time or payment
  - Meals or lodging allowance or reimbursement
  - Per diem allowance or payment
- Other payments to or on behalf of a construction mechanic that are not wages or fringe benefits, such as:
  - Industry advancement funds
  - Financial or material loans



### STATE OF MICHIGAN MICHIGAN PREVAILING WAGE RATE SCHEDULE For Parking Lot, Road, Highway, Bridge & Airport Construction

### **OPERATING ENGINEERS CLASSIFICATION DESCRIPTIONS**

Class I Asphalt Paver (self-propelled) Asphalt Planer (self-propelled) Asphalt Plant Operator Auto-Grader Blade Grader Operator Batch Plant (concrete-central mix) Backhoe (with over 3/8 yard bucket) **Bulldozer Operator** Concrete Pump 3" and over Conveyor Loader Operator (euclid type) **Crane Operator** Dragline Operator **Elevating Grader Operator** End-loader Operator (1 yard capacity or over) Slip Form Paver Finishing Machine Operator (asphalt) Gradall Operator (and similar type machines) Hoisting Engineer Hydro demolisher (water blaster) Locomotive Operator Mechanic Class II Sweeper (wayne type & similar equipment) Screening Plant Operator Washing Plant Operator **Crusher Operator** Vacuum Truck Operator Class II Grease Truck Class III Air Compressor Operator (600 cfm or more) Air Compressor (2 or more, less than 600 cfm) **Concrete Breaker** Tractor Operator (farm type with attachments) Wagon Drill Operator **Boiler Fireman** Class IV Oiler End-loader Operator (under 1 yard capacity) Roller Operator (other than asphalt) Curing Equipment Operator (self-propelled) Concrete Saw Operator (Over 40 HP) Power Bin Operator Plant Drier Operator (asphalt) Vibratory Compaction Equipment (6' wide or over) **Guard Post Driver Operator** All Mulching Equipment Boom or Winch Hoist Truck Operator

Paver Operator (5 bags or more) Pump Operator (6" discharge or over, gas, diesel powered, or generator of 300 amp or larger) Pile Driving Operator Roto Mill Roller Operator (Asphalt) Side Boom Tractor (type D-4, equivalent or larger) Self-Propelled or Tractor Drawn Scraper

Slurry Machine (asphalt) Swinging Boom Truck (over I2 ton capacity) Shouldering or Gravel Distributing Machine Operator (self-propelled) Shovel Operator Side Boom Tractor (type D-4 or equivalent or larger) Tractor Operator Trenching Machine Operator (ladder or wheel type) Tube Finisher (slip form paving) Farm type tractor with attached pan

Backhoe (with 3/8 yard bucket or less) Side Boom Tractor (smaller than D-4 type or equivalent) Batch Plant (concrete-dry mix)

Stump Remover Skid Steer Fireman Mechanic's Helper Trencher (service) Flexplane Operator Cleftplane Operator Grader Operator Self-propelled Fine-Grade or Form (concrete) Finishing Machine Operator (concrete) Concrete Pump (under 3") Farm Type Tractor Operator Mesh Installer (self-propelled)

End Dumps



### STATE OF MICHIGAN MICHIGAN PREVAILING WAGE RATE SCHEDULE For Parking Lot, Road, Highway, Bridge & Airport Construction

### LABORERS CLASSIFICATION DESCRIPTIONS

- **Class 1** Asphalt Shoveler or Loader, Asphalt Raker Tender, Asphalt Plant Misc., Railroad Track and Trestle Laborer, Burlap Man, Carpenter's Tender, Top Man, Yard Man, Guard Rail Builder's Tender, Earth Retention Barrier and Wall and Mechanically Stabilized Earthen Wall Installers Tender, Highway and Median Barrier Installer's Tender (including Sound, Retaining and Crash Barrier), Fence Erector's Tender, Dumper (wagon, truck, etc.) Joint Filling Labor, Misc., Unskilled Labor, Sprinkler Labor, Form Setting Labor, Form Stripper, Pavement Reinforcing, Handling and Placing (e.g. wire mesh, steel mats, dowel bars, etc.) Mason's or Bricklayer's Tender on Manholes, Manhole Builder, Headwalls, etc., Waterproofing (other than buildings), Seal Coating and Slurry Mix, Shoring, Underpinning, Bridge Painting, etc. (spray, roller and brush) Sandblasting, Pressure Grouting, and Bridge Pin and Hanger Removal, Material Recycling Laborer, Horizontal Paver (brick, concrete, clay, stone and asphalt) Ground Stabilization and Modification Laborer, Grouting, Waterblasting, Sign Installer and remote control operated equipment.
- Class 2 Mix Operator (less than 5 sacks), Air or Electric Tool Operator (jack hammer, etc.), Spreader, Boxman (asphalt, stone, gravel, etc.), Concrete Paddler, Power Chain Saw Operator, Paving Batch Truck Dumper, Tunnel Mucker (highway work only), Concrete Saw Operator (under 40 H.P.), Dry Pack Machine and Roto-Mill Grounds Person.
- Class 3 Tunnel Miner (highway work only), Finishers Tender, Guard Rail Builder, Highway and Median Barrier Installer, Fence Erector, Bottom Man, Powder Man, Wagon Drill and Air Track Operators, Curb and Side Rail Setters' Tender, Diamond & Core Drills, Earth Retention Barriers, Walls and Mechanically Stabilized Earthen Wall Installer (including sound, retaining and crash barrier), grade checker and certified welder.
- Class 4 Asphalt Raker
- Class 5 Pipe Layers, Oxy-gun
- Class 6 Line-Form Setter for Curb or Pavement and asphalt screed checker/screw man on asphalt paving machines.
- **Class 7** Concrete Specialist, finishing and troweling, of cast in place or precast concrete by any and all methods.



### STATE OF MICHIGAN

MICHIGAN PREVAILING WAGE RATE SCHEDULE

For Parking Lot, Road, Highway, Bridge & Airport Construction

### **OVERTIME PROVISIONS FOR Road Builder PREVAILING WAGE RATE SCHEDULE**

1. Overtime is represented as a nine character code. Each character represents a certain period of time after the first 8 hours Monday thru Friday.

	Monday thru Friday	Saturday	Sunday & Holidays	Four 10s
First 8 Hours		4		
9 <sup>th</sup> Hour	1	5	0	
10 <sup>th</sup> Hour	2	6	8 ·	9
Over 10 hours	3	7		

Overtime for Monday thru Friday after 8 hours:

the 1<sup>st</sup> character is for time worked in the 9<sup>th</sup> hour (8.1 - 9 hours)

the  $2^{nd}$  character is for time worked in the  $10^{th}$  hour (9.1 - 10 hours)

the 3<sup>rd</sup> character is for time worked beyond the 10<sup>th</sup> hour (10.1 and beyond)

### Overtime on Saturday:

the 4<sup>th</sup> character is for time worked in the first 8 hours on Saturday (0 - 8 hours)

the 5<sup>th</sup> character is for time worked in the 9<sup>th</sup> hour on Saturday (8.1 - 9 hours) the 6<sup>th</sup> character is for time worked in the 10<sup>th</sup> hour (9.1 - 10 hours)

the 7<sup>th</sup> character is for time worked beyond the 10<sup>th</sup> hour (10.01 and beyond)

### Overtime on Sunday & Holidays

the 8<sup>th</sup> character is for time worked on Sunday or on a holiday

### 4 Ten hour days @ Straight Time

The 9<sup>th</sup> character indicates if an optional 4-day 10-hour per day workweek can be worked between Monday and Friday without paying overtime after 8 hours worked. To utilize a 4 ten workweek, notice is required from the employer to employee prior to the start of work on the project.

#### 2. Overtime Indicators Used in the Overtime Provision:

H -means TIME AND ONE-HALF due

D -means DOUBLE PAY due

X means TIME AND ONE HALF due after 40 hours worked

Y means YES an optional 4-day 10-hour per day workweek can be worked without paying overtime after 8 hours worked

N -means NO optional 4-day 10-hour per day workweek can be worked without paying overtime after 8 hours worked

#### 3. EXAMPLES:

HHHHHDDY - This example shows that the 11/2 rate must be used for time worked after 8 hours Monday thru Friday (characters 1 - 3) and for all hours worked on Saturday, (characters 4 - 6), except hours worked after 10 hours on Saturday (7<sup>th</sup> character). Work done after 10 hours must be paid at the double time rate. Work done on Sunday or holidays must be paid double time (character 8). The Y (character 9) indicates that 4 ten-hour days is an acceptable alternative workweek at regular pay.

HHHHHHHY means that the 11/2 rate must be used for time worked after 8 hours worked Monday thru Friday (characters 1-3); and for any hours worked on Saturdays, Sundays or holidays (characters 4-8). The Y (character 9) indicates that 4 ten-hour days is an acceptable alternative workweek at regular pay.

XXHXXXHDY this example allows 4 ten hour days Monday thru Saturday to be worked. Hours worked beyond ten Monday thru Saturday OR hours worked after 40 hours in one week must be paid at time and one half. Sunday or holiday hours must be paid at double.

# DRAFT AIA Document A201<sup>™</sup> - 2007

### General Conditions of the Contract for Construction

### for the following PROJECT:

*(Name and location or address)* Saline Area Schools construction improvements in accordance with the relevant application for preliminary qualification of bonds, the approved plans and specifications, the Owner's fixed budget and as otherwise approved by the Owner. **THE OWNER:** 

*(Name and address)* Saline Area Schools 7265 Saline Ann Arbor Road Saline, Michigan 48176

#### THE ARCHITECT:

*(Name and address)* Kingscott Associates, Inc. 259 E. Michigan Avenue, Suite 308 Kalamazoo, Michigan 49007-6403

#### THE CONSTRUCTION MANAGER:

Clark Construction Company 29110 Inkster Road, Suite 150 Southfield, Michigan 48075

References to the "Contractor" herein with respect to the Construction Phase shall be deemed to apply to the Construction Manager, and references to the Owner/Contractor Agreement shall refer to the Owner/Construction Manager Agreement, unless context suggests otherwise. (It is acknowledged that references to the "Contractor" during the Pre-Construction Phase may refer to individual trade contractors.)

The inclusion of any reference to or use of the term "Construction Manager" in this document shall in no way be read to limit in any way the applicability of this document as provided in Owner-Construction Manager Agreement Section 1.3.

### TABLE OF ARTICLES

- 1 GENERAL PROVISIONS
- 2 OWNER
- 3 CONTRACTOR
- 4 ARCHITECT
- 5 SUBCONTRACTORS
- 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS
- 7 CHANGES IN THE WORK

#### ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.





**ELECTRONIC COPYING** of any portion of this AIA<sup>®</sup> Document to another electronic file is prohibited and constitutes a violation of copyright laws as set forth in the footer of this document.

- 8 TIME
- 9 PAYMENTS AND COMPLETION
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 INSURANCE AND BONDS
- 12 UNCOVERING AND CORRECTION OF WORK
- 13 MISCELLANEOUS PROVISIONS
- 14 TERMINATION OR SUSPENSION OF THE CONTRACT
- 15 CLAIMS AND DISPUTES







AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. (989098163)

#### INDEX

(Numbers and Topics in Bold are Section Headings)

Acceptance of Nonconforming Work 9.6.6, 9.9.3, 12.3 Acceptance of Work 9.6.6, 9.8.2, 9.9.3, 9.10.1, 9.10.3, 12.3 Access to Work 3.16, 6.2.1, 12.1 Accident Prevention 10 Acts and Omissions 3.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 8.3.1, 9.5.1, 10.2.5, 10.2.8, 13.4.2, 13.7.1, 14.1, 15.2 Addenda 1.1.1, 3.11.1 Additional Costs, Claims for 3.7.4, 3.7.5, 6.1.1, 7.3.7.5, 10.3, 15.1.4 **Additional Inspections and Testing** 9.4.2, 9.8.3, 12.2.1, 13.5 Additional Insured 1114 Additional Time, Claims for 3.2.4, 3.7.4, 3.7.5, 3.10.2, 8.3.2, 15.1.5 **Administration of the Contract** 3.1.3, 4.2, 9.4, 9.5 Advertisement or Invitation to Bid 1.1.1 Aesthetic Effect 4.2.13 Allowances 3.8, 7.3.8 All-risk Insurance 11.3.1, 11.3.1.1 **Applications for Payment** 4.2.5, 7.3.9, 9.2, 9.3, 9.4, 9.5.1, 9.6.3, 9.7.1, 9.10, 11.1.3 Approvals 2.1.1, 2.2.2, 2.4, 3.1.3, 3.10.2, 3.12.8, 3.12.9, 3.12.10, 4.2.7, 9.3.2, 13.5.1 Arbitration 8.3.1, 11.3.10, 13.1.1, 15.3.2, 15.4 ARCHITECT 4 Architect, Definition of 4.1.1 Architect, Extent of Authority 2.4.1, 3.12.7, 4.1, 4.2, 5.2, 6.3.1, 7.1.2, 7.3.7, 7.4, 9.2.1, 9.3.1, 9.4, 9.5, 9.6.3, 9.8, 9.10.1, 9.10.3, 12.1, 12.2.1, 13.5.1, 13.5.2, 14.2.2, 14.2.4, 15.1.3, 15.2.1 Architect, Limitations of Authority and Responsibility 2.1.1, 3.12.4, 3.12.8, 3.12.10, 4.1.2, 4.2.1, 4.2.2, 4.2.3, 4.2.6, 4.2.7, 4.2.10, 4.2.12, 4.2.13, 5.2.1, 7.4.1, 9.4.2, 9.5.3, 9.6.4, 15.1.3, 15.2

Architect's Additional Services and Expenses 2.4.1, 11.3.1.1, 12.2.1, 13.5.2, 13.5.3, 14.2.4 Architect's Administration of the Contract 3.1.3, 4.2, 3.7.4, 15.2, 9.4.1, 9.5 Architect's Approvals 2.4.1, 3.1.3, 3.5.1, 3.10.2, 4.2.7 Architect's Authority to Reject Work 3.5.1, 4.2.6, 12.1.2, 12.2.1 Architect's Copyright 1.1.7, 1.5 Architect's Decisions 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 4.2.14, 6.3.1, 7.3.7, 7.3.9, 8.1.3, 8.3, 1, 9, 2.1, 9, 4.1, 9.5, 9, 8.4, 9.9.1, 13.5.2, 15.2, 15.3 Architect's Inspections 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 13.5 Architect's Instructions 3.2.4, 3.3.1, 4.2.6, 4.2.7, 13.5.2 Architect's Interpretations 4.2.11, 4.2.12 Architect's Project Representative 4.2.10 Architect's Relationship with Contractor 1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5.1, 3.7.4, 3.7.5, 3.9.2, 3.9.3, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.2, 4.1.3, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3.7, 12, 13, 4.2, 13.5, 15.2 Architect's Relationship with Subcontractors 1.1.2, 4.2.3, 4.2.4, 4.2.6, 9.6.3, 9.6.4, 11.3.7 Architect's Representations 9.4.2, 9.5.1, 9.10.1 Architect's Site Visits 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5 Asbestos 10.3.1 Attorneys' Fees 3.18.1, 9.10.2, 10.3.3 Award of Separate Contracts 6.1.1, 6.1.2 Award of Subcontracts and Other Contracts for **Portions of the Work** 5.2 **Basic Definitions** 1.1 **Bidding Requirements** 1.1.1, 5.2.1, 11.4.1 **Binding Dispute Resolution** 9.7.1, 11.3.9, 11.3.10, 13.1.1, 15.2.5, 15.2.6.1, 15.3.1, 15.3.2, 15.4.1 **Boiler and Machinery Insurance** 11.3.2 Bonds, Lien 7.3.7.4, 9.10.2, 9.10.3 Bonds, Performance, and Payment 7.3.7.4, 9.6.7, 9.10.3, 11.3.9, 11.4

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

**Building Permit** 3.7.1 Capitalization 1.3 Certificate of Substantial Completion 9.8.3, 9.8.4, 9.8.5 **Certificates for Payment** 4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10.1, 9.10.3, 14.1.1.3, 14.2.4, 15.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Insurance 9.10.2, 11.1.3 **Change Orders** 1.1.1, 2.4.1, 3.4.2, 3.7.4, 3.8.2.3, 3.11.1, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, **7.2**, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3 Change Orders, Definition of 7.2.1 **CHANGES IN THE WORK** 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 7.4.1, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 **CLAIMS AND DISPUTES** 3.2.4, 6.1.1, 6.3.1, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 154 Claims and Timely Assertion of Claims 15.4.1 **Claims for Additional Cost** 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 **Claims for Additional Time** 3.2.4, 3.7.46.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 **Cleaning Up** 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 **Communications Facilitating Contract** Administration 3.9.1, 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 **COMPLETION, PAYMENTS AND** 9

Completion, Substantial 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, 9.8, 9.9.1, 9.10.3, 12.2, 13.7 Compliance with Laws 1.6.1, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 10.2.2, 11.1, 11.3, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14.1.1, 14.2.1.3, 15.2.8, 15.4.2, 15.4.3 Concealed or Unknown Conditions 3.7.4, 4.2.8, 8.3.1, 10.3 Conditions of the Contract 1.1.1, 6.1.1, 6.1.4 Consent, Written 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3, 11.3.1, 13.2, 13.4.2, 15.4.4.2 **Consolidation or Joinder** 15.4.4 **CONSTRUCTION BY OWNER OR BY** SEPARATE CONTRACTORS 1.1.4.6 Construction Change Directive, Definition of 7.3.1 **Construction Change Directives** 1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1 Construction Schedules, Contractor's 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 **Contingent Assignment of Subcontracts** 5.4, 14.2.2.2 **Continuing Contract Performance** 15.1.3 Contract, Definition of 1.1.2 **CONTRACT, TERMINATION OR** SUSPENSION OF THE 5.4.1.1, 11.3.9, 14 Contract Administration 3.1.3, 4, 9.4, 9.5 Contract Award and Execution, Conditions Relating to 3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1 **Contract Documents, The** 1.1.1 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.1.1 **Contract Sum** 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, 9.1, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7.1, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5 Contract Time, Definition of 8.1.1

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

### CONTRACTOR

3 Contractor. Definition of 3.1, 6.1.2 **Contractor's Construction Schedules** 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 Contractor's Employees 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 10.3, 11.1.1, 11.3.7, 14.1, 14.2.1.1, **Contractor's Liability Insurance** 11.1 Contractor's Relationship with Separate Contractors and Owner's Forces 3.12.5, 3.14.2, 4.2.4, 6, 11.3.7, 12.1.2, 12.2.4 Contractor's Relationship with Subcontractors 1.2.2, 3.3.2, 3.18.1, 3.18.2, 5, 9.6.2, 9.6.7, 9.10.2, 11.3.1.2, 11.3.7, 11.3.8 Contractor's Relationship with the Architect 1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5.1, 3.7.4, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.3, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3.7, 12, 13.5, 15.1.2, 15.2.1 Contractor's Representations 3.2.1, 3.2.2, 3.5.1, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.8.2 Contractor's Responsibility for Those Performing the Work 3.3.2, 3.18, 5.3.1, 6.1.3, 6.2, 9.5.1, 10.2.8 Contractor's Review of Contract Documents 3.2 Contractor's Right to Stop the Work 9.7 Contractor's Right to Terminate the Contract 14.1, 15.1.6 Contractor's Submittals 3.10, 3.11, 3.12.4, 4.2.7, 5.2.1, 5.2.3, 9.2, 9.3, 9.8.2, 9.8.3, 9.9.1, 9.10.2, 9.10.3, 11.1.3, 11.4.2 Contractor's Superintendent 3.9, 10.2.6 Contractor's Supervision and Construction Procedures 1.2.2, 3.3, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4, 7.1.3, 7.3.5, 7.3.7, 8.2, 10, 12, 14, 15.1.3 Contractual Liability Insurance 11.1.1.8, 11.2 Coordination and Correlation 1.2, 3.2.1, 3.3.1, 3.10, 3.12.6, 6.1.3, 6.2.1 Copies Furnished of Drawings and Specifications 1.5, 2.2.5, 3.11 Copyrights 1.5, 3.17 Correction of Work 2.3, 2.4, 3.7.3, 9.4.2, 9.8.2, 9.8.3, 9.9.1, 12.1.2, 12.2 **Correlation and Intent of the Contract Documents** 1.2 Cost, Definition of 7.3.7

Costs 2.4.1, 3.2.4, 3.7.3, 3.8.2, 3.15.2, 5.4.2, 6.1.1, 6.2.3, 7.3.3.3, 7.3.7, 7.3.8, 7.3.9, 9.10.2, 10.3.2, 10.3.6, 11.3, 12.1.2, 12.2.1, 12.2.4, 13.5, 14 **Cutting and Patching** 3.14, 6.2.5 Damage to Construction of Owner or Separate Contractors 3.14.2, 6.2.4, 10.2.1.2, 10.2.5, 10.4, 11.1.1, 11.3, 12.2.4 Damage to the Work 3.14.2, 9.9.1, 10.2.1.2, 10.2.5, 10.4.1, 11.3.1, 12.2.4 Damages, Claims for 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Damages for Delay 6.1.1, 8.3.3, 9.5.1.6, 9.7, 10.3.2 Date of Commencement of the Work, Definition of 8.1.2 Date of Substantial Completion, Definition of 8.1.3 Day, Definition of 8.1.4 Decisions of the Architect 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 15.2, 6.3, 7.3.7, 7.3.9, 8.1.3, 8.3.1, 9.2.1, 9.4, 9.5.1, 9.8.4, 9.9.1, 13.5.2, 14.2.2, 14.2.4, 15.1, 15.2 **Decisions to Withhold Certification** 9.4.1, 9.5, 9.7, 14.1.1.3 Defective or Nonconforming Work, Acceptance, Rejection and Correction of 2.3.1, 2.4.1, 3.5.1, 4.2.6, 6.2.5, 9.5.1, 9.5.2, 9.6.6, 9.8.2, 9.9.3, 9.10.4, 12.2.1 Defective Work. Definition of 3.5.1 Definitions 1.1, 2.1.1, 3.1.1, 3.5.1, 3.12.1, 3.12.2, 3.12.3, 4.1.1, 15.1.1, 5.1, 6.1.2, 7.2.1, 7.3.1, 8.1, 9.1, 9.8.1 Delays and Extensions of Time 3.2., 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4.1, 8.3, 9.5.1, 9.7.1, 10.3.2, 10.4.1, 14.3.2, 15.1.5, 15.2.5 Disputes 6.3.1, 7.3.9, 15.1, 15.2 **Documents and Samples at the Site** 3.11 Drawings, Definition of 1.1.5 Drawings and Specifications, Use and Ownership of 3.11 Effective Date of Insurance 8.2.2, 11.1.2 Emergencies 10.4, 14.1.1.2, 15.1.4 Employees, Contractor's 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 10.3.3, 11.1.1, 11.3.7, 14.1, 14.2.1.1

5

AIA Document A201<sup>mx</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163) Equipment, Labor, Materials or 1.1.3, 1.1.6, 3.4, 3.5.1, 3.8.2, 3.8.3, 3.12, 3.13.1, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2 Execution and Progress of the Work 1.1.3, 1.2.1, 1.2.2, 2.2.3, 2.2.5, 3.1, 3.3.1, 3.4.1, 3.5.1, 3.7.1, 3.10.1, 3.12, 3.14, 4.2, 6.2.2, 7.1.3, 7.3.5, 8.2, 9.5.1, 9.9.1, 10.2, 10.3, 12.2, 14.2, 14.3.1, 15.1.3 Extensions of Time 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3, 7.4.1, 9.5.1, 9.7.1, 10.3.2, 10.4.1, 14.3, 15.1.5, 15.2.5 **Failure of Payment** 9.5.1.3, 9.7, 9.10.2, 13.6, 14.1.1.3, 14.2.1.2 Faulty Work (See Defective or Nonconforming Work) **Final Completion and Final Payment** 4.2.1, 4.2.9, 9.8.2, 9.10, 11.1.2, 11.1.3, 11.3.1, 11.3.5, 12.3.1, 14.2.4, 14.4.3 Financial Arrangements, Owner's 2.2.1, 13.2.2, 14.1.1.4 Fire and Extended Coverage Insurance 11.3.1.1 **GENERAL PROVISIONS** 1 **Governing Law** 13.1 Guarantees (See Warranty) **Hazardous Materials** 10.2.4, 10.3 Identification of Subcontractors and Suppliers 5.2.1 Indemnification 3.17.1, **3.18**, 9.10.2, 10.3.3, 10.3.5, 10.3.6, 11.3.1.2, 11.3.7 Information and Services Required of the Owner 2.1.2, 2.2, 3.2.2, 3.12.4, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 11.4, 13.5.1, 13.5.2, 14.1.1.4, 14.1.4, 15.1.3 **Initial Decision** 15.2 Initial Decision Maker, Definition of 1.1.8 Initial Decision Maker, Decisions 14.2.2, 14.2.4, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 Initial Decision Maker, Extent of Authority 14.2.2, 14.2.4, 15.1.3, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 Injury or Damage to Person or Property 10.2.8, 10.4.1 Inspections 3.1.3, 3.3.3, 3.7.1, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 12.2.1, 13.5 Instructions to Bidders 1.1.1 Instructions to the Contractor 3.2.4, 3.3.1, 3.8.1, 5.2.1, 7, 8.2.2, 12, 13.5.2 Instruments of Service, Definition of 1.1.7

Insurance 3.18.1, 6.1.1, 7.3.7, 9.3.2, 9.8.4, 9.9.1, 9.10.2, 11 **Insurance, Boiler and Machinery** 11.3.2 **Insurance, Contractor's Liability** 11.1 Insurance, Effective Date of 8.2.2, 11.1.2 Insurance, Loss of Use 11.3.3 **Insurance, Owner's Liability** 11.2 **Insurance**, **Property** 10.2.5. 11.3 Insurance, Stored Materials 9.3.2, 11.4.1.4 **INSURANCE AND BONDS** 11 Insurance Companies, Consent to Partial Occupancy 9.9.1, 11.4.1.5 Insurance Companies, Settlement with 11.4.10 Intent of the Contract Documents 1.2.1, 4.2.7, 4.2.12, 4.2.13, 7.4 Interest 13.6 Interpretation 1.2.3, 1.4, 4.1.1, 5.1, 6.1.2, 15.1.1 Interpretations, Written 4.2.11, 4.2.12, 15.1.4 Judgment on Final Award 15.4.2 Labor and Materials, Equipment 1.1.3, 1.1.6, **3.4**, 3.5.1, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2 Labor Disputes 8.3.1 Laws and Regulations 1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13.1, 4.1.1, 9.6.4, 9.9.1, 10.2.2, 11.1.1, 11.3, 13.1.1, 13.4, 13.5.1, 13.5.2, 13.6.1, 14, 15.2.8, 15.4 Liens 2.1.2, 9.3.3, 9.10.2, 9.10.4, 15.2 8 Limitations, Statutes of 12.2.5, 13.7, 15.4.1.1 Limitations of Liability 2.3.1, 3.2.2, 3.5.1, 3.12.10, 3.17.1, 3.18.1, 4.2.6, 4.2.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 10.2.5, 10.3.3, 11.1.2, 11.2, 11.3.7, 12.2.5, 13.4.2 Limitations of Time 2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3, 11, 3.12.5, 3.15.1, 4.2.7, 5.2, 5.3.1, 5.4.1, 6.2.4, 7.3, 7.4, 8.2, 9.2.1, 9.3.1, 9.3.3, 9.4.1, 9.5, 9.6, 9.7.1, 9.8, 9.9, 9.10, 11.1.3, 11.3.1.5, 11.3.6, 11.3.10, 12.2, 13.5, 13.7, 14, 15 Loss of Use Insurance

6

11.3.3

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. (989098163)

Material Suppliers 1.5, 3.12.1, 4.2.4, 4.2.6, 5.2.1, 9.3, 9.4.2, 9.6, 9.10.5 Materials, Hazardous 10.2.4, 10.3 Materials, Labor, Equipment and 1.1.3, 1.1.6, 1.5.1, 3.4.1, 3.5.1, 3.8.2, 3.8.3, 3.12, 3.13.1, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1.2, 10.2.4, 14.2.1.1, 14.2.1.2 Means, Methods, Techniques, Sequences and Procedures of Construction 3.3.1, 3.12.10, 4.2.2, 4.2.7, 9.4.2 Mechanic's Lien 2.1.2. 15.2.8 Mediation 8.3.1, 10.3.5, 10.3.6, 15.2.1, 15.2.5, 15.2.6, 15.3, 15.4.1 **Minor Changes in the Work** 1.1.1, 3.12.8, 4.2.8, 7.1, 7.4 **MISCELLANEOUS PROVISIONS** 13 Modifications, Definition of 1.1.1 Modifications to the Contract 1.1.1, 1.1.2, 3.11, 4.1.2, 4.2.1, 5.2.3, 7, 8.3.1, 9.7.1, 10.3.2, 11.3.1 **Mutual Responsibility** 6.2 Nonconforming Work, Acceptance of 9.6.6, 9.9.3, 12.3 Nonconforming Work, Rejection and Correction of 2.3.1, 2.4.1, 3.5.1, 4.2.6, 6.2.4, 9.5.1, 9.8.2, 9.9.3, 9.10.4, 12.2.1 Notice 2.2.1, 2.3.1, 2.4.1, 3.2.4, 3.3.1, 3.7.2, 3.12.9, 5.2.1, 9.7.1, 9.10, 10.2.2, 11.1.3, 11.4.6, 12.2.2.1, 13.3, 13.5.1, 13.5.2, 14.1, 14.2, 15.2.8, 15.4.1 Notice, Written 2.3.1, 2.4.1, 3.3.1, 3.9.2, 3.12.9, 3.12.10, 5.2.1, 9.7.1, 9.10, 10.2.2, 10.3, 11.1.3, 11.3.6, 12.2.2.1, 13.3, 14, 15.2.8, 15.4.1 **Notice of Claims** 3.7.4, 4.5, 10.2.8, 15.1.2, 15.4 Notice of Testing and Inspections 13.5.1, 13.5.2 Observations, Contractor's 3.2, 3.7.4 Occupancy 2.2.2, 9.6.6, 9.8, 11.3.1.5 Orders, Written 1.1.1, 2.3, 3.9.2, 7, 8.2.2, 11.3.9, 12.1, 12.2.2.1, 13.5.2, 14.3.1 **OWNER** 2 Owner, Definition of 2.1.1

**Owner, Information and Services Required of the** 2.1.2, 2.2, 3.2.2, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.3.2, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 11.3, 13.5.1, 13.5.2, 14.1.1.4, 14.1.4, 15.1.3 **Owner's** Authority 1.5, 2.1.1, 2.3.1, 2.4.1, 3.4.2, 3.8.1, 3.12.10, 3.14.2, 4.1.2, 4.1.3, 4.2.4, 4.2.9, 5.2, 1, 5, 2.4, 5, 4.1, 6, 1, 6, 3, 1, 7.2.1, 7.3.1, 8.2.2, 8.3.1, 9.3.1, 9.3.2, 9.5.1, 9.6.4, 9.9.1, 9.10.2, 10.3.2, 11.1.3, 11.3.3, 11.3.10, 12.2.2, 12.3.1, 13.2.2, 14.3, 14.4, 15.2.7 **Owner's Financial Capability** 2.2.1, 13.2.2, 14.1.1.4 **Owner's Liability Insurance** 11.2 **Owner's Loss of Use Insurance** 11.3.3 Owner's Relationship with Subcontractors 1.1.2, 5.2, 5.3, 5.4, 9.6.4, 9.10.2, 14.2.2 **Owner's Right to Carry Out the Work 2.4**, 14.2.2 **Owner's Right to Clean Up** 6.3 **Owner's Right to Perform Construction and to Award Separate Contracts** 6.1 **Owner's Right to Stop the Work** 2.3 Owner's Right to Suspend the Work 14.3 Owner's Right to Terminate the Contract 14.2 **Ownership and Use of Drawings, Specifications** and Other Instruments of Service 1.1.1, 1.1.6, 1.1.7, 1.5, 2.2.5, 3.2.2, 3.11.1, 3.17.1, 4.2.12, 5.3.1 **Partial Occupancy or Use** 9.6.6, 9.9, 11.3.1.5 Patching, Cutting and 3.14, 6.2.5 Patents 3.17 Payment, Applications for 4.2.5, 7.3.9, 9.2.1, **9.3**, 9.4, 9.5, 9.6.3, 9.7.1, 9.8.5, 9.10.1, 14.2.3, 14.2.4, 14.4.3 Payment, Certificates for 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7.1, 9.10.1, 9.10.3, 13.7, 14.1.1.3, 14.2.4 Payment, Failure of 9.5.1.3, 9.7, 9.10.2, 13.6, 14.1.1.3, 14.2.1.2 Payment, Final 4.2.1, 4.2.9, 9.8.2, 9.10, 11.1, 2, 11.1.3, 11.4.1, 11.4.5, 12.3.1, 13.7, 14.2.4, 14.4.3 Payment Bond, Performance Bond and 7.3.7.4, 9.6.7, 9.10.3, 11.4.9, 11.4 **Payments**, **Progress** 9.3, 9.6, 9.8.5, 9.10.3, 13.6, 14.2.3, 15.1.3 PAYMENTS AND COMPLETION 9

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

Payments to Subcontractors 5.4.2, 9.5.1.3, 9.6.2, 9.6.3, 9.6.4, 9.6.7, 11.4.8, 14.2.1.2 PCB 10.3.1 **Performance Bond and Payment Bond** 7.3.7.4, 9.6.7, 9.10.3, 11.4.9, 11.4 Permits, Fees, Notices and Compliance with Laws 2.2.2, 3.7, 3.13, 7.3.7.4, 10.2.2 PERSONS AND PROPERTY, PROTECTION OF 10 Polychlorinated Biphenyl 10.3.1 Product Data, Definition of 3.12.2 **Product Data and Samples, Shop Drawings** 3.11, 3.12, 4.2.7 **Progress and Completion** 4.2.2, 8.2, 9.8, 9.9.1, 14.1.4, 15.1.3 **Progress Payments** 9.3, 9.6, 9.8.5, 9.10.3, 13.6, 14.2.3, 15.1.3 **Project**, Definition of the 1.1.4 **Project Representatives** 4.2.10 **Property Insurance** 10.2.5, 11.3 PROTECTION OF PERSONS AND PROPERTY 10 Regulations and Laws 1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 9.9.1, 10.2.2, 11.1, 11.4, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14, 15.2.8, 15.4 Rejection of Work 3.5.1, 4.2.6, 12.2.1 Releases and Waivers of Liens 9.10.2 Representations 3.2.1, 3.5.1, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.4.2, 9.5.1, 9.8.2, 9.10.1 Representatives 2.1.1, 3.1.1, 3.9, 4.1.1, 4.2.1, 4.2.2, 4.2.10, 5.1.1, 5.1.2, 13.2.1 Responsibility for Those Performing the Work 3.3.2, 3.18, 4.2.3, 5.3.1, 6.1.3, 6.2, 6.3, 9.5.1, 10 Retainage 9.3.1, 9.6.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3 **Review of Contract Documents and Field Conditions by Contractor** 3.2, 3.12.7, 6.1.3 Review of Contractor's Submittals by Owner and Architect 3.10.1, 3.10.2, 3.11, 3.12, 4.2, 5.2, 6.1.3, 9.2, 9.8.2 Review of Shop Drawings, Product Data and Samples by Contractor 3.12

**Rights and Remedies** 1.1.2, 2.3, 2.4, 3.5.1, 3.7.4, 3.15.2, 4.2.6, 4.5, 5.3, 5.4, 6.1, 6.3, 7.3.1, 8.3, 9.5.1, 9.7, 10.2.5, 10.3, 12.2.2, 12.2.4, 13.4, 14, 15.4 **Royalties, Patents and Copyrights** 3.17 Rules and Notices for Arbitration 15.4.1 Safety of Persons and Property **10.2**, 10.4 Safety Precautions and Programs 3.3.1, 4.2.2, 4.2.7, 5.3.1, 10.1, 10.2, 10.4 Samples, Definition of 3.12.3 Samples, Shop Drawings, Product Data and 3.11, 3.12, 4.2.7 Samples at the Site, Documents and 3.11 **Schedule of Values** 9.2, 9.3.1 Schedules, Construction 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 Separate Contracts and Contractors 1.1.4, 3.12.5, 3.14.2, 4.2.4, 4.2.7, 6, 8.3.1, 11.4.7, 12.1.2 Shop Drawings, Definition of 3.12.1 Shop Drawings, Product Data and Samples 3.11, 3.12, 4.2.7 Site, Use of 3.13, 6.1.1, 6.2.1 Site Inspections 3.2.2, 3.3.3, 3.7.1, 3.7.4, 4.2, 9.4.2, 9.10.1, 13.5 Site Visits, Architect's 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5 Special Inspections and Testing 4.2.6, 12.2.1, 13.5 Specifications, Definition of the 1.1.6 **Specifications**, The 1.1.1, 1.1.6, 1.2.2, 1.5, 3.11, 3.12.10, 3.17, 4.2.14 Statute of Limitations 13.7, 15.4.1.1 Stopping the Work 2.3, 9.7, 10.3, 14.1 Stored Materials 6.2.1, 9.3.2, 10.2.1.2, 10.2.4, 11.4.1.4 Subcontractor, Definition of 5.1.1 **SUBCONTRACTORS** 5 Subcontractors, Work by 1.2.2, 3.3.2, 3.12.1, 4.2.3, 5.2.3, 5.3, 5.4, 9.3.1.2, 9.6.7 **Subcontractual Relations** 5.3, 5.4, 9.3.1.2, 9.6, 9.10, 10.2.1, 11.4.7, 11.4.8, 14.1, 14.2.1

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

Submittals 3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3, 7.3.7, 9.2, 9.3, 9.8, 9.9.1, 9.10.2, 9.10.3, 11.1.3 Submittal Schedule 3.10.2, 3.12.5, 4.2.7 Subrogation, Waivers of 6.1.1, 11.4.5, 11.3.7 **Substantial Completion** 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, **9.8**, 9.9.1, 9.10.3, 12.2, 13.7 Substantial Completion, Definition of 9.8.1 Substitution of Subcontractors 5.2.3. 5.2.4 Substitution of Architect 4.1.3Substitutions of Materials 3.4.2, 3.5.1, 7.3.8 Sub-subcontractor, Definition of 5.1.2 Subsurface Conditions 3.7.4 Successors and Assigns 13.2 Superintendent **3.9**, 10.2.6 **Supervision and Construction Procedures** 1.2.2, 3.3, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4, 7.1.3, 7.3.7, 8.2, 8.3.1, 9.4.2, 10, 12, 14, 15.1.3 Surety 5.4.1.2, 9.8.5, 9.10.2, 9.10.3, 14.2.2, 15.2.7 Surety, Consent of 9.10.2, 9.10.3 Surveys 2.2.3 Suspension by the Owner for Convenience 14.3 Suspension of the Work 5.4.2, 14.3 Suspension or Termination of the Contract 5.4.1.1, 11.4.9, 14 Taxes 3.6, 3.8.2.1, 7.3.7.4 **Termination by the Contractor** 14.1, 15.1.6 Termination by the Owner for Cause 5.4.1.1, 14.2, 15.1.6 Termination by the Owner for Convenience 14.4 Termination of the Architect 4.1.3 Termination of the Contractor 14.2.2 **TERMINATION OR SUSPENSION OF THE** CONTRACT 14

**Tests and Inspections** 3.1.3, 3.3.3, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 10.3.2, 11.4.1.1, 12.2.1, 13.5 TIME 8 Time, Delays and Extensions of 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3, 1, 7.4.1, **8.3**, 9.5.1, 9.7.1, 10.3.2, 10.4.1, 14.3.2, 15.1.5, 15.2.5 Time Limits 2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3 11, 3.12.5, 3.15.1, 4.2, 4.4, 4.5, 5.2, 5.3, 5.4, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.4.1, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 11.1.3, 11.4.1.5, 11.4.6, 11.4.10, 12 2, 13.5, 13.7, 14, 15.1.2, 15.4 **Time Limits on Claims** 3.7.4, 10.2.8, 13.7, 15.1.2 Title to Work 9.3.2, 9.3.3 **Transmission of Data in Digital Form** 1.6 UNCOVERING AND CORRECTION OF WORK 12 **Uncovering of Work** 12.1 Unforeseen Conditions, Concealed or Unknown 3.7.4, 8.3.1, 10.3 Unit Prices 7.3.3.2, 7.3.4 Use of Documents 1.1.1, 1.5, 2.2.5, 3.12.6, 5.3 **Use of Site** 3.13, 6.1.1, 6.2.1 Values. Schedule of 9.2. 9.3.1 Waiver of Claims by the Architect 13.4.2 Waiver of Claims by the Contractor 9.10.5, 11.4.7, 13.4.2, 15.1.6 Waiver of Claims by the Owner 9.9.3, 9.10.3, 9.10.4, 11.4.3, 11.4.5, 11.4.7, 12.2.2.1, 13.4.2, 14.2.4, 15.1.6 Waiver of Consequential Damages 14.2.4, 15.1.6 Waiver of Liens 9.10.2, 9.10.4 Waivers of Subrogation 6.1.1, 11.4.5, **11.3.7** Warranty 3.5, 4.2.9, 9.3.3, 9.8.4, 9.9.1, 9.10.4, 12.2.2, 13.7.1 Weather Delays 15.1.5.2 Work, Definition of 1.1.3 Written Consent 1.5.2, 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3, 11.4.1, 13.2, 13.4.2, 15.4.4.2

9

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163) Written Interpretations 4.2.11, 4.2.12 Written Notice 2.3, 2.4, 3.3.1, 3.9, 3.12.9, 3.12.10, 5.2.1, 8.2.2, 9.7, 9.10, 10.2.2, 10.3, 11.1.3, 11.4.6, 12.2.2, 12.2.4, **13.3**, 14, 15.4.1 Written Orders 1.1.1, 2.3, 3.9, 7, 8.2.2, 11.4.9, 12.1, 12.2, 13.5.2, 14.3.1, 15.1.2





AIA Document A201<sup>TM</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

#### ARTICLE 1 GENERAL PROVISIONS § 1.1 BASIC DEFINITIONS § 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) which shall be awarded by the Owner and then assigned over by the Owner to the Construction Manager after award and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive or (4) a written order for a minor change in the Work issued by the Architect. Unless specifically enumerated in the Agreement in writing, the Contract Documents also include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, accepted portions of the Contractor's bid or proposal, and portions of Addenda relating to bidding requirements. The execution of the Owner/Construction Manager and subcontractor agreements shall constitute acceptance of all provisions of the Drawings, Addenda and all Contract Documents.

### § 1.1.2 THE CONTRACT

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, (3) between the Owner and the Architect or the Architect or the Architect's consultants or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate the Contractor's performance of its duties.

### § 1.1.3 THE WORK

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project. The Contractor acknowledges and agrees that the Contract Documents are sufficient to provide for the completion of the Work and that the Contract Documents include work (whether or not shown or described) which reasonably may be inferred to be required or useful for the completion of the Work in accordance with applicable laws, codes, and customary standards of the construction industry.

### § 1.1.4 THE PROJECT

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by separate contractors.

### § 1.1.5 THE DRAWINGS

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams.

### § 1.1.6 THE SPECIFICATIONS

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services.

### § 1.1.7 INSTRUMENTS OF SERVICE

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

### § 1.1.8 INITIAL DECISION MAKER

The Initial Decision Maker is the person identified in the Agreement to render initial decisions on Claims in accordance with Section 15.2 and certify termination of the Agreement under Section 14.2.2.

AIA Document A201<sup>ax</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

### § 1.2 CORRELATION AND INTENT OF THE CONTRACT DOCUMENTS

§ 1.2.1 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

**§ 1.2.2** Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade. Where responsibility for particular Work is required of the Contractor, the Contractor shall not be released from that responsibility by reason of the location of the specification or drawing information which establishes the responsibility. Thus, the Contractor shall be responsible for all Work required of him, even though that responsibility may be shown only in that portion of the documents typically pertaining to another contractor or trade.

**§ 1.2.3** Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings.

**§ 1.2.4** If there should be a conflict between two or more of the Contract Documents, the Architect reserves the right to determine which governs, consistent with the following order of interpretation:

- .1 Where requirements specifically set forth in the Agreement are in conflict with other Contract Documents, including but not limited to these General Conditions, the Agreement shall govern.
- .2 In all other instances, the conflict shall be resolved by complying with the provision that is most favorable to the Owner.
- .4 When a duplicate of material or equipment occurs in the Drawings, the Specifications or other Contract Documents, each Contractor shall be deemed to have bid on the basis of each furnishing such material or equipment. The Contractor will decide which Subcontractor(s) shall furnish the same.

**§ 1.2.5** The Contractor acknowledges that there may be items of the Work, which the Contractor is responsible to provide under the Agreement that are not drawn or specified in the Design Documents but are necessary for the proper execution and completion of the Work and are consistent with an reasonably inferable from the Drawings and Specifications. All such items shall be provided as part of the Work without delay in its progress and without any increase in the Contract Sum or Guaranteed Maximum Price.

### § 1.3 CAPITALIZATION

Terms capitalized in these General Conditions include those that are (1) specifically defined, (2) the titles of numbered articles or (3) the titles of other documents published by the American Institute of Architects.

### § 1.4 INTERPRETATION

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

### § 1.5 OWNERSHIP AND USE OF DRAWINGS, SPECIFICATIONS AND OTHER INSTRUMENTS OF SERVICE

§ 1.5.1 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and unless otherwise indicated in the Contract Documents or the Owner/Architect Agreement, the Architect and the respective consultants will retain all common law, statutory and other reserved rights, including copyrights. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Architect's or Architect's consultants' reserved rights.

**§ 1.5.2** The Contractor, Subcontractors, Sub-subcontractors and material or equipment suppliers are authorized to use and reproduce the Instruments of Service solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers may not use the Instruments of Service on

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner, Architect and the Architect's consultants.

### § 1.6 TRANSMISSION OF DATA IN DIGITAL FORM

If the parties intend to transmit Instruments of Service or any other information or documentation in digital form, they may endeavor to establish necessary protocols governing such transmissions, unless otherwise already provided in the Agreement or the Contract Documents.

### ARTICLE 2 OWNER

### § 2.1 GENERAL

**§ 2.1.1** The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to matters requiring the Owner's approval or authorization subject to parameters of authority established by the Owner's Board of Education. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.

### § 2.1.2

### § 2.2 INFORMATION AND SERVICES REQUIRED OF THE OWNER

**§ 2.2.1** Prior to commencement of the Work, the Contractor may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. Thereafter, the Contractor may only request such evidence if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) a change in the Work materially changes the Contract Sum; or (3) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish, as applicable, such evidence as a condition precedent to commencement or continuation of the Work or the portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.

§ 2.2.2 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including, but not limited to, those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities.

**§ 2.2.3** The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. Taking into account the Contractor's experience and expertise, the Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work. The Contractor shall not be entitled to additional compensation resulting from its failure to confirm the location of readily observable site utilities or existing structures prior to the opening of the Contractor's bid.

**§ 2.2.4** Upon specific written request of the Owner, the Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services. Contracts with other Contractors alone shall not constitute sufficient Owner control for purposes of this Section.

**§ 2.2.5** Unless otherwise provided in the Contract Documents, the Owner shall furnish to the Contractor one copy of the Contract Documents for purposes of making reproductions pursuant to Section 1.5.2.

### § 2.3 OWNER'S RIGHT TO STOP THE WORK

If the Contractor fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 or fails to carry out Work in accordance with the Contract Documents, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3. This right shall be in addition to and not in limitation of the Owner's rights under any provision of the Contract Documents.

AIA Document A201<sup>III</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

### § 2.4 OWNER'S RIGHT TO CARRY OUT THE WORK

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a seven (7)-day period after receipt of written notice from the Owner or the Owner's designee to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, including any claim against the Contractor's performance bond, correct such deficiencies. In the event the Contractor's default or neglect results in a threat to the safety of persons or property, the Contractor shall immediately commence and continue correction; otherwise, the Owner may undertake the same actions as permitted in the prior sentence. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the reasonable cost of correcting such deficiencies, including Owner's expenses, including any and all legal expenses incurred to effectuate and enforce this provision, and compensation for the Architect's, Construction Manager's and/or other Contractor's additional services made necessary by such default, neglect or failure. Exercise of such rights shall in no way limit or jeopardize the Owner's right to any claim against the Performance Bond or Contractor. If payments then or thereafter due the Contractor shall pay the difference to the Owner.

In the event the Owner directs another entity to perform Work pursuant to this section that otherwise is the obligation of the Contractor, including correction of safety violations, either at the Contractor's request or as a result of the Contractor's failure to perform such Work, that other entity shall charge the Contractor all costs for labor, material and equipment plus that other entity's administrative, profit and overhead costs. The Contractor shall pay that other entity within ten (10) days of the date of invoice. If not paid within ten (10) days, the Contractor authorizes the Owner to withhold that amount from the Contractor and to pay the same to that other entity from the next payment due the Contractor. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

# ARTICLE 3 CONTRACTOR § 3.1 GENERAL

§ 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor shall designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.

§ 3.1.2 The Contractor shall perform the Work in accordance with the Contract Documents.

§ 3.1.3 The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor.

### § 3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

**§ 3.2.1** Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed and correlated personal observations with requirements of the Contract Documents.

**§ 3.2.2** Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.2.3, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Contractor shall promptly report to the Architect any errors, inconsistencies or omissions discovered by or made known to the Contractor as a request for information in such form as the Architect may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.

§ 3.2.3 The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require, with a copy of same to be forwarded to the Owner.

AIA Document A201<sup>ax</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

§ 3.2.4 If the Contractor believes that additional cost or time is involved because of clarifications or instructions the Architect issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2 or 3.2.3, the Contractor shall make Claims as provided in Article 15. If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities.

§ 3.2.5 Prior to submitting its bid, the Contractor shall have studied and compared the Contract Documents and shall have reported to the Architect any error, inconsistency or omission in the Contract Documents. It will be presumed that the Contractor's bid and the Contract Sum include the cost of correcting any such error, inconsistency or omission, which could have been discovered by the exercise of reasonable diligence. Unless the Contractor establishes that such error, inconsistency or omission could not have been discovered by the exercise of reasonable diligence, the Contractor will make such corrections without additional compensation so that the Work is fully functional.

### § 3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

§ 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instruction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures shall give timely written notice to the Owner and Architect and shall not proceed with that portion of the Work without further written instructions from the Architect. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures. The Contractor shall immediately notify the Architect of delays of any other Contractors that could impact timely coordination and completion of the Work.

§ 3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of, the Contractor or any of its Subcontractors.

**§ 3.3.3** The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

### § 3.4 LABOR AND MATERIALS AND UTILITIES

**§ 3.4.1** Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work. Such provision of labor and materials shall occur in sufficient time to satisfy the existing Project schedule. The Contractor bears the risk of any failure to timely provide such labor and materials for any reason. The Contractor agrees to execute the appropriate UCC forms to effectuate the Owner's ownership of the material and equipment furnished pursuant to this Agreement.

**§ 3.4.2** Except in the case of minor changes in the Work authorized by the Architect in accordance with Sections 3.12.8 or 7.4, the Contractor may make substitutions only with the consent of the Owner, after evaluation by the Architect and in accordance with a Change Order or Construction Change Directive.

§ 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163) **§ 3.4.4** The Contractor agrees that neither it nor its Subcontractors will discriminate against any employee or applicant for employment, to be employed in the performance of this Contract, with respect to hire, tenure, conditions or privilege of employment, or any matter directly or indirectly related to employment, because of race, age, sex, color, religion, national origin, ancestry or physical disability. Brach of this covenant may be regarded as a material breach of this Contract.

### § 3.4.5 ASBESTOS-FREE PRODUCT INSTALLATION

- .1 It is hereby understood and agreed that no product and/or material containing asbestos including chrysolite, amosite, crocidolite, tremolite asbestos, anthophyllite asbestos, actinolite asbestos and any combination of these materials that have been chemically treated and/or altered shall be installed or introduced into the Work by the Contractor or its employees, agents, Subcontractors, or other individuals or entities over whom the Contractor has control. If applicable, the Contractor shall be required to provide a signed certification statement ensuring that all products or materials installed or introduced into the Work will be asbestos-free.
- .2 The Contractor also shall be required to furnish certified statements from the manufacturers of supplied materials used during construction verifying their products to be asbestos-free in accordance with the requirements of Section 3.4.5.1.
- .3 The Contractor shall complete and submit to the Owner a certification evidencing asbestos-free product installation prior to issuance of the final Certificate for Payment in a form acceptable to Owner.

### § 3.5 WARRANTY

The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment. In addition to any other warranties, guarantees or obligations set forth in the Contract Documents or applicable as a matter of a law, and not in limitation of the terms of the Contract Documents, the Contractor warrants and guarantees that:

- .1 The Owner will have good title to the Work and all materials and equipment incorporated into the Work and, unless otherwise expressly provided in the Contract Documents, will be new;
- .2 The Work and all materials and equipment incorporated into the Work will be free from all defects, including any defects in workmanship or materials;
- .3 The Work and all equipment incorporated into the Work will be fit for the purpose for which they are intended;
- .4 The Work and all materials and equipment incorporated into the Work will be merchantable; and
- .5 The Work and all materials and equipment incorporated into the Work will conform in all respects to the Contract Documents.

Upon notice of the breach of any of the foregoing warranties or guarantees or any other warranties or guarantees under the Contract Documents, the Contractor, in addition t any other requirements in the Contract Documents, will commence to correct such breach within seven (7) days after written notice thereof and thereafter will use its best efforts to correct such breach to the satisfaction of the Owner; provided that if such notice is given after final payment hereunder, such seven (7) day period shall be extended to fourteen (14) days. The foregoing warranties and obligations of the Contractor shall survive the final payment and/or termination of the Contract. The Contractor shall, at the time of final completion of the Work and as a condition precedent to final payment to the Contractor, assign to the Owner all manufacturers' warranties related to the materials and labor used in the Work. The Contractor further agrees to perform the Work in such manner as to preserve any and all such manufacturers' warranties and deliver to the Architect the warranties, project manuals, operating procedures, and other materials related to each of the building systems and materials included in the Contractor's Work and as required by the Specifications.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

### § 3.6 TAXES

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect. The Contractor shall pay all state and federal taxes levied on its business, income or property and shall make all contributions for social security and other wage or payroll taxes. The Contractor shall be solely responsible for such payments and shall indemnify the Owner and hold it harmless from same.

#### § 3.7 PERMITS, FEES, NOTICES, AND COMPLIANCE WITH LAWS

§ 3.7.1 The Contractor shall secure and pay for the building permit as well as for other permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations concluded.

§ 3.7.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work.

§ 3.7.3 If the Contractor performs Work contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.

§ 3.7.4 Concealed or Unknown Conditions. If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature, that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide written and dated notice to the Owner and the Architect before conditions are disturbed and in no event later than 21 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Owner and Architect determine that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, they will recommend an equitable adjustment in the Contract Sum or Contract Time, or both. If the Owner and Architect determine that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Contractor in writing, stating the reasons. If Contractor disputes the determination or recommendation, the Contractor shall proceed as provided in Article 15. The requirements of Section 2 of 1998 PA 57, as amended, are hereby incorporated into this document. The Contractor shall be alert to any indication or evidence of existing underground or concealed utilities or structures not shown on the Contract Documents and shall immediately notify the Owner of discovery of such evidence. It the Contractor encounters such utilities or structures, it shall cease operations immediately to minimize damage and shall notify the Owner and Architect. The Contractor shall bear the cost of damage resulting from its failure to exercise reasonable care in its construction activity or from continuing operations without notifying the Owner.

§ 3.7.4.1 The Contractor bidding on the Work is responsible for visiting the site and determining all local conditions that may in any way affect its Work.

§ 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall provide written and dated notification to the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features shall be made, as needed, as provided in Article 15.

### § 3.8 ALLOWANCES

§ 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection.

§ 3.8.2 Unless otherwise provided in the Contract Documents,

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847 1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

- .1 allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and all required taxes, less applicable trade discounts;
- .2 Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowances; and
- .3 whenever costs are less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2.

§ 3.8.3 Materials and equipment under an allowance shall be selected by the Owner with reasonable promptness.

### § 3.9 SUPERINTENDENT

§ 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor. The superintendent shall be satisfactory to the Owner in all respects, and the Owner shall have the right to require the Contractor to remove any superintendent from the Project whose performance is not satisfactory to the Owner and to replace such superintendent with a superintendent who is satisfactory to the Owner. The Contractor shall not replace its superintendent without the consent of the Owner, except with another superintendent who is satisfactory to the Owner.

**§ 3.9.2** The Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the name and qualifications of a proposed superintendent. The Owner and/or Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to the proposed superintendent or (2) that the Owner and/or Architect requires additional time to review.

**§ 3.9.3** The Contractor shall not employ a proposed superintendent to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, except with a superintendent who is satisfactory to the Owner.

### § 3.10 CONTRACTOR'S CONSTRUCTION SCHEDULES

§ 3.10.1 The Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's and Architect's information a Contractor's construction schedule for the Work. The schedule shall not exceed time limits current under the Contract Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work. In no event shall the Contractor's Construction Schedule be extended due to action or inaction of the Contractor, except with prior written approval of the Owner within the Owner's sole discretion.

**§ 3.10.2** The Contractor shall prepare a submittal schedule, promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, and shall submit the schedule(s) for the Owner's and Architect's approval, which shall not unreasonably be delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, (2) allow for a reasonable amount of time to review submittal, and (3) shall provide for expeditious and practical execution of the Work. If the Contractor fails to submit a submittal schedule, the Contractor shall not be entitled to any increase in Contract. Sum or extension of Contract Time based on the time required for review of submittals.

**§ 3.10.3** The Contractor shall perform the Work in general accordance with the most recent approved Project schedules and the most recent work schedule submitted to the Owner and Architect consistent therewith.

§ 3.10.4 In no event shall the Contractor's Construction Schedule be extended due to action or inaction of the Contractor, except with prior written approval of the Owner within the Owner's sole discretion.
§ 3.10.5 Progress Meetings: Meetings of representatives of the various Contractors and Trades will be held for the purpose of coordination and furthering the progress of the Work. Contractor and subcontractor attendance is mandatory. Meetings shall be held at regular intervals as provided in the General Requirements; special meetings may be held if deemed necessary by the Owner, Contractor and Architect/Engineer.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

**§ 3.10.6** The Contractor shall proceed strictly (not substantially) in accordance with the critical path set forth in the Construction Schedule. the Contractor shall monitor the progress of the Work for conformance with the requirements of the Construction Schedule and shall promptly advise the Owner of any delays or potential delays. If any progress report indicates any delays, the Contractor shall propose an affirmative plan to correct the delay, including overtime and/or additional labor, if necessary. In no event shall any progress report constitute an adjustment of the Contract Time or any Milestone Date or the Guarantee Maximum time unless any such adjustment is agreed to by the Owner and authorized pursuant to a Change Order.

### § 3.11 DOCUMENTS AND SAMPLES AT THE SITE

The Contractor shall maintain at the site for the Owner one copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and one copy of approved Shop Drawings, Product Data, Samples and similar required submittals. These shall be available to the Architect and shall be delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed.

### § 3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

§ 3.12.1 Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.

§ 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor for submittal to and review by the Architect to illustrate materials or equipment for some portion of the Work. All Work shall be furnished and installed in accordance with the Drawings, Specifications and as additionally required by the manufacturer's printed instructions. The Contractor shall review the manufacturer's instructions, and where conflict occurs between the Drawings or Specifications and the manufacturer's instructions, the Contractor shall request clarification from the Architect prior to commencing the Work.

**§ 3.12.3** Samples are physical examples that illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.

**§ 3.12.4** Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. Their purpose is to demonstrate the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review by the Architect is subject to the limitations of Section 4.2.7. Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Architect without action.

§ 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve and submit to the Architect Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of separate contractors.

§ 3.12.6 By submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.

**§ 3.12.7** The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples or similar submittals until the respective submittal has been approved by the Architect.

**§ 3.12.8** The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Architect's review and approval of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the Architect in detailed writing of such deviation at the time of submittal and (1) the Architect has given written

AIA Document A201<sup>™</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the Architect's approval thereof.

**§ 3.12.9** The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such written notice, the Architect's approval of a resubmission shall not apply to such revisions.

§ 3.12.10 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents for a portion of the Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. The Contractor shall not be required to provide professional services in violation of applicable law. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Architect will specify all performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by a properly licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Architect. The Owner and the Architect shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications and approvals performed or provided by such design professionals, provided the Owner and Architect have specified to the Contractor all performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.10, the Architect will review, approve or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents.

### § 3.13 USE OF SITE

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not unreasonably encumber the site with materials or equipment.

### § 3.14 CUTTING AND PATCHING

§ 3.14.1 The Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting and patching shall be restored to the condition existing prior to the cutting, fitting and patching, unless otherwise required by the Contract Documents.

§ 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or separate contractors by cutting, patching or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the Owner or a separate contractor except with written consent of the Owner and of such separate contractor; such consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold from the Owner or a separate contractor the Contractor's consent to cutting or otherwise altering the Work.

### § 3.15 CLEANING UP

§ 3.15.1 The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials from and about the Project.

§ 3.15.2 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and Owner shall be entitled to reimbursement from the Contractor.

**§ 3.15.3** Any areas and/or concurrently occupied space both occupied by the Owner and used in the progress of the Work, both within the limits of the construction site and the adjacent areas leading to it, shall be maintained, opened to travel and kept in a clean condition. Failure by the Contractor to maintain said areas will result in the Owner's cleaning of same, at the expense of the Contractor.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163) **§ 3.15.4** In addition to removal of rubbish, the Contractor and its subcontractors, under Contractor's direction, shall replace any broken glass, remove stains, spots, marks, and dirt from decorated work, clean hardware, and/or remove spots and smears from all surfaces which were affected by the Work.

### § 3.16 ACCESS TO WORK

The Contractor shall provide the Owner and Architect access to the Work in preparation and progress wherever located.

### § 3.17 ROYALTIES, PATENTS AND COPYRIGHTS

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and indemnify and hold harmless hold the Owner and Architect from any and all cost, damage, or loss on account thereof, including, but not limited to, actual attorneys fees. but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications or other documents prepared by the Owner or Architect. However, if the Contractor has reason to believe that the required design, process or product is an infringement of a copyright or a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Architect. The review by the Owner or Architect of any method of construction, invention, appliance, process, article, device or materials of any kind shall be for its adequacy in the Work and shall not be an approval for the use thereof by the Contractor in violation of any patent or other rights of any third person.

### § 3.18 INDEMNIFICATION

§ 3.18.1 To the fullest extent permitted by law the Contractor shall indemnify and hold harmless the Owner, Architect, Architect's consultants, and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work or the duties or obligations of the Owner/Construction Manager Agreement, these General Conditions or the failure of the Contractor or the Work to conform with the Contract Documents, caused in whole or in part by any negligent or wrongful acts or omissions of the Contractor, a Subcontractor, or anyone directly or indirectly employed by them or anyone for whose acts of any of them may be liable, to the fullest degree of Contractor's fault, on a comparative basis (or the fault of any others for whom the Contractor is responsible). The Contractor shall be responsible to the Owner, Architect, Architect's consultants, and agents and employees of any of them from and against all amounts such parties may be required to pay in attorney fees in order to pursue enforcement of this provision against the Contractor or otherwise obtain indemnification from the Contractor provided under the terms of this Article. Such obligation shall not be construed to negate, abridge or reduce any other rights or obligations of indemnity which would otherwise exist as to any party or person set forth in this section. To the fullest extent permitted by law, the Contractor shall indemnify the Owner and save the Owner harmless against all loss by fines, penalties or corrective measures resulting from acts of the Contractor or omissions by the Contractor, its Subcontractors, agents, employees or assigns, with respect to the violation of safety requirements of this Contract, including reasonable attorney fees.

**§ 3.18.2** In addition to and not in limitation of the Contractor's other indemnity obligations, the Contractor hereby accepts and assumes exclusive liability for and shall indemnify, protect and save harmless the Owner and Architect from and against the payment of the following:

All contributions, taxes or premiums (including interest and penalties thereon) which may be payable under the unemployment insurance law of any state, the federal Social Security Act, federal, state, county and/or municipal tax withholding laws, or any other law, measured upon the payroll of or required to be withheld from employees by whomsoever employed, engaged in the Work to be performed and furnished under this Contract;

All sales, use, personal property and other taxes (including interest and penalties thereon) required by any federal, state, county, municipal or other law to be paid or collected by the Contractor or any of its Subcontractors or vendors or any other person or persons acting for, through or under it or any of them, by reason of the performance of the Work or the acquisition, ownership, furnishing, or use of any materials, equipment, supplies, labor, services or other items for or in connection with the Work;

In accordance with the June 5, 1997, decision by the Sixth Circuit Court of Appeals, Michigan prevailing wage rates are valid and enforceable. All contractors must comply with the Michigan Prevailing Wage Act. Current prevailing

wage rates are published by the Michigan Department of Consumer and Industry Services. Bidders are responsible for obtaining and complying with the current applicable rates during the course of the Work. The Contractor shall pay the prevailing wage and fringe benefits in accordance with Michigan Act No. 166, Public Acts of 1965, as amended, for contracts performed on projects funded by or upon the credit of the State of Michigan. If the Contractor is in violation of the Prevailing Wage Act, that will constitute a material breach of the Contract, warranting termination. The Contractor shall be financially responsible for the payment of prevailing wages by all Subcontractors contracted by it for work on the Project;

All pension, welfare, vacation, annuity and other benefit contributions payable under or in connection with respect to all persons by whomsoever employed, engaged in the Work to be performed and furnished under this Contract.

The Contractor shall indemnify, defend and hold the Owner harmless from any claim, damage, loss or expense, including but not limited to actual attorney fees, incurred by the Owner related to any hazardous material or waste, toxic substance, pollution or contamination brought into the Project site or caused by the Contractor or used, handled, transported, stored, removed, remediated, disturbed or dispersed of by Contractor.

**§ 3.18.3** In the event that any claim is made or asserted, or lawsuit filed for damages or injury arising out of or resulting from the performance of the Work, whether or not the Owner is named as a party, the Contractor shall immediately advise the Owner , in writing, of such claim or lawsuit and shall provide a full and complete copy of any documents or pleadings thereto, as well as a full and accurate report of the facts involved.

### ARTICLE 4 ARCHITECT

### § 4.1 GENERAL

§ 4.1.1 The Architect is lawfully licensed to practice architecture or an entity lawfully practicing architecture in the jurisdiction where the Project is located. That person or entity is identified as the Architect in the Agreement and is referred to throughout the Contract Documents as if singular in number. The term "Architect," "Architect/Engineer," "Engineer," or "Design Professional" as used herein means the Architect or the Architect's authorized representative.

**§ 4.1.2** Duties, responsibilities and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified or extended without written consent of the Owner and Architect.

**§ 4.1.3** If the employment of the Architect is terminated, the Owner shall employ a successor architect whose status under the Contract Documents shall be that of the Architect.

### § 4.2 ADMINISTRATION OF THE CONTRACT

§ 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Architect issues the final Certificate For Payment and with the Owner's written concurrence during the correction period. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.

**§ 4.2.2** The Architect will visit the site at intervals appropriate to the stage of construction, or more frequently as agreed with the Owner or required by law, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine in general if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. Subject to the Owner/Architect Agreement for the Project, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. Except as otherwise set forth herein, the Architect will not have control over, charge of, or responsibility for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents, except as provided in Section 3.3.1.



AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)
§ 4.2.3 On the basis of the site visits, the Architect will keep the Owner informed about the progress and quality of the portion of the Work completed, will guard the Owner against defects and deficiencies in the Work, report to the Owner (1) known deviations from the Contract Documents and from the most recent construction schedule submitted by the Contractor, and (2) defects and deficiencies observed in the Work. Except as required by the Owner/Architect Agreement, the Architect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect will not have control over or charge of and will not be responsible for acts or omissions of the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.§ 4.2.4 COMMUNICATIONS FACILITATING CONTRACT ADMINISTRATION Except as otherwise provided in the Contract Documents or when direct communications have been specially authorized, the Owner and Contractor shall endeavor to communicate with each other through the Architect about matters arising out of or relating to the Contract. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and material suppliers shall be through the Contractor. Communications by and with separate contractors shall be through the Owner.

§ 4.2.5 Based on the Architect's evaluations of the Contractor's Applications for Payment, the Architect will review and certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.

§ 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect will have authority to require inspection or testing of the Work in accordance with Sections 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.

§ 4.2.7 The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Owner and Architect or, in the absence of an approved submittal schedule, with reasonable promptness as to cause no delay in the Work while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5 and 3.12. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component. However, should the Architect discover during the course of such review any inaccuracies, incompleteness, or other irregularities, the Architect shall immediately notify the Owner of the same to determine an appropriate corrective course of action or notify the Contractor of the same to correct the irregularities.

**§ 4.2.8** The Architect will prepare Change Orders and Construction Change Directives, and may authorize minor changes in the Work as provided in Section 7.4. The Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4.

**§ 4.2.9** The Architect will conduct inspections to determine, with the Owner's concurrence, the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9.8; receive and forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract and assembled by the Contractor pursuant to Section 9.10; and issue a final Certificate for Payment pursuant to Section 9.10.

**§ 4.2.10** If the Owner and Architect agree, the Architect will provide one or more project representatives to assist in carrying out the Architect's responsibilities at the site. The duties, responsibilities and limitations of authority of such project representatives shall be as set forth in an exhibit to be incorporated in the Contract Documents.

**§ 4.2.11** The Architect will interpret matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests will be

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

made in writing within any time limits agreed upon or otherwise with reasonable promptness given the particular circumstances.

§ 4.2.12 Interpretations of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations, the Architect will endeavor to secure faithful performance by the Contractor and will not be liable for results of interpretations or decisions rendered in good faith and without negligence.

§ 4.2.13 The Architect's interpretations on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.

§ 4.2.14 The Architect will review and respond to requests for information about the Contract Documents. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness given the particular circumstances. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information.

#### **ARTICLE 5 SUBCONTRACTORS** § 5.1 DEFINITIONS

§ 5.1.1 A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a separate contractor or subcontractors of a separate contractor. The term "Subcontractor" shall also include material and equipment suppliers. Each and every subcontract shall be understood to have the Owner as a third-party beneficiary, and the Owner shall enjoy all third-party beneficiary rights permitted by law.

§ 5.1.2 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Subsubcontractor.

#### § 5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

§ 5.2.1 Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the names of persons or entities (including those who are to furnish supplies, materials or equipment, including those fabricated to a special design) proposed for each principal portion of the Work. The Architect will reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to any such proposed person or entity or (2) that the Architect requires additional time for review. All contractual agreements with additional persons or entities serving as a subcontractor shall expressly identify the Owner as a third-party beneficiary, and the Owner shall enjoy all third-party beneficiary rights not prohibited by law.

§ 5.2.1.1 These General Conditions refer to the relationship between the Owner and Contractor. As to the contract between the Contractor and its Sub-Contractors, the General Conditions shall be read as the Contractor having the position of the Owner and the Sub-Contractors having the position of the Contractor. The Sub-Contractors are bound to the Contractor just as the Contractor is bound to the Owner. The Sub-Contractor shall have all the rights, duties and obligations to the Contractor as the Contractor has rights, duties and obligations to the Owner. The Sub-Contractors shall agree to and accept the same responsibility to the Owner as the Contractor. In the event any failure of a Sub-Contractor causes any type of injury or loss to the Owner, direct or indirect, the Contractor shall be jointly and severally liable to the Owner for such injury in addition to any responsibility or liability of the Sub-Contractor.

§ 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.

§ 5.2.3 If the Owner or Architect has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Architect has no reasonable objection. If the proposed but rejected Subcontractor was reasonably capable of performing the Work, despite the Architect's or Owner's reasonable objection, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any,

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute Subcontractor's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Contractor has acted promptly and responsively in submitting names as required.

**§ 5.2.4** The Contractor shall not substitute a Subcontractor, person or entity previously selected if the Owner or Architect makes reasonable objection to such substitution. The Contractor shall notify the Owner and Architect of any proposed substitution a minimum of ten days prior to such proposed change.

#### § 5.3 SUBCONTRACTUAL RELATIONS

By appropriate agreement, written where legally required for validity, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor, by these Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies and redress against the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents. Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

#### § 5.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS

§ 5.4.1 Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that

- .1 assignment is effective only after termination of the Contract by the Owner for cause pursuant to Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor in writing; and
- .2 assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract.

**§ 5.4.2** Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation may be adjusted as negotiated by the parties.

§ 5.4.3 Upon such assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity.

### ARTICLE 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS

#### § 6.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS

§ 6.1.1 The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other construction or operations on the site under Conditions of the Contract identical or substantially similar to these including those portions related to insurance. The Contractor shall be responsible for coordinating the Work with the work of other Contractors, including the Owner's own forces or separate contractors, so as to complete the Work in accordance with the Project time schedule.

**§ 6.1.2** When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.

§ 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces and of each separate contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

other separate contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to the construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, separate contractors and the Owner until subsequently revised.

#### § 6.1.4

#### § 6.2 MUTUAL RESPONSIBILITY

**§ 6.2.1** The Contractor shall afford the Owner and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.

**§ 6.2.2** If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Architect apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects not then reasonably discoverable.

**§ 6.2.3** The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a separate contractor because of the Contractor's delays, improperly timed activities or defective construction.

**§ 6.2.4** The Contractor shall promptly remedy damage the Contractor causes to completed or partially completed construction or to property of the Owner or separate contractors as provided in Section 10.2.5.

**§ 6.2.5** The Owner and each separate contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Section 3.14.

#### § 6.3 OWNER'S RIGHT TO CLEAN UP

If a dispute arises among the Contractor, subcontractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and allocate the cost among those responsible. The Owner's right to clean up shall in no event be deemed a duty, and should the Owner choose not to pursue this remedy, the Contractor necessitating such action shall remain fully responsible for the same.

## ARTICLE 7 CHANGES IN THE WORK § 7.1 GENERAL

§ 7.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, only by Change Order, Construction Change Directive, written contract amendment, or order for a minor

change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.§7.1.2 A Change Order shall be based upon agreement among the Owner, Contractor and Architect; a Construction

§ 7.1.2 A Change Order shall be based upon agreement among the Owner, Contractor and Architect; a Construction Change Directive may be issued by the Owner and Architect and may or may not be agreed to by the Contractor; an order for a minor change in the Work may be issued by the Architect alone, if so authorized.

§ 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise provided in the Change Order, Construction Change Directive or order for a minor change in the Work.

#### § 7.2 CHANGE ORDERS

§ 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Owner, Contractor and Architect stating their agreement upon all of the following:

- .1 The change in the Work;
- .2 The amount of the adjustment, if any, in the Contract Sum; and
- .3 The extent of the adjustment, if any, in the Contract Time.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163) § 7.2.2 The Contractor's agreement on any change order shall constitute a final statement from the Contractor on all matters relating to the direct and indirect costs associated with such change, as well as any and all related adjustments to the Contract Sum and Contract Time.

#### § 7.3 CONSTRUCTION CHANGE DIRECTIVES

§ 7.3.1 A Construction Change Directive is a written order prepared by the Architect and signed by the Owner and Architect, directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and Contract Time being adjusted accordingly.

§ 7.3.2 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order.

§ 7.3.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one or more of the following methods:

- .1 Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation;
- .2 Unit prices stated in the Contract Documents or subsequently agreed upon;
- .3 Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
- .4 As provided in Section 7.3.7.

However, the Contract time shall be adjusted only if the Contractor demonstrates to the Owner that the changes in the Work required by the Construction Change Directive adversely affect the critical path of the Work.

§ 7.3.4 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed in a proposed Change Order or Construction Change Directive so that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted, unless the Contractor provided such unit prices as a part of a competitive bid.

§ 7.3.5 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time. Contractor agreements to a Construction Change Directive shall require a follow-up writing or signature as contemplated in Section 7.3.6.

§ 7.3.6 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

§ 7.3.7 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect shall determine, with the Owner's approval, the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.7 shall be limited to a reasonable amount of the following that are actually incurred by the Contractor:

- Costs of labor, including social security, old age and unemployment insurance, fringe benefits .1 required by agreement or custom, and workers' compensation insurance;
- .2 Costs of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed;
- .3 Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor or others;
- .4 Costs of premiums for all bonds and insurance, permit fees, and sales, use or similar taxes related to the Work; and

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847 1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

.5 Additional costs of supervision and field office personnel directly attributable to the change.

§ 7.3.8 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

**§ 7.3.9** Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for undisputed Work completed under the Construction Change Directive in Applications for Payment. For those undisputed portions, the Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the Architect's professional judgment, to be reasonably justified. The Architect's interim determination of cost, if agreed to by the Owner in writing, shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of the Contractor to disagree and assert a Claim in accordance with Article 15.

§ 7.3.10 When the Owner and Contractor agree in writing with a determination made by the Architect concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments in writing, such agreement shall be effective immediately and the Architect will prepare a Change Order. Change Orders may be issued for all or any part of a Construction Change Directive.

#### § 7.4 MINOR CHANGES IN THE WORK

The Architect has authority to order minor changes in the Work not involving adjustment in the Contract Sum or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes will be effected by written order signed by the Architect and shall be binding on the Owner and Contractor.

## ARTICLE 8 TIME

#### § 8.1 DEFINITIONS

**§ 8.1.1** Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.

§ 8.1.2 The date of commencement of the Work is the date established in the Agreement.

§ 8.1.3 The date of Substantial Completion is the date certified by the Architect in accordance with Section 9.8.

**§ 8.1.4** The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

#### § 8.2 PROGRESS AND COMPLETION

§ 8.2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement the Contractor confirms that the Contract Time is a reasonable period for obtaining all supplies, materials, tools and equipment necessary to perform the Work and for properly performing the Work.

**§ 8.2.2** The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance.

**§ 8.2.3** The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time. All work shall be completed in sufficient time to allow for clean up and preparation for Owner move-in prior to the Date of Substantial Completion.

#### § 8.3 DELAYS AND EXTENSIONS OF TIME

**§ 8.3.1** Provided the Contractor submits a written request for an extension not more than fourteen days after the occurrence that gives rise to the delay, if the Contractor is delayed at any time in the commencement or progress of the Work by an act or neglect of the Owner or Architect, or of an employee of either, or of a separate contractor employed by the Owner; or by changes ordered in the Work; or by fire, unavoidable casualties or other causes beyond the Contractor's control; or by delay authorized by the Owner pending litigation, mediation or arbitration, as applicable; or by other causes that the Architect determines may justify delay, then the Contract Time shall be

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

extended by Change Order for such reasonable time as the Architect may determine and the Owner agrees. Failure of the Contractor to submit a timely request for an extension shall irrevocably waive the Contractor's right to such an extension of time. If the contract time is subject to extension pursuant to this subparagraph, such extension shall be the exclusive remedy of the Contractor and the Contractor shall not be entitled to recover damages from the Owner.

§ 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Article 15.

**§ 8.3.3** Section 8.4 precludes recovery of damages for delay by the Contractor under other provisions of the Contract Documents.

#### § 8.4 DELAY DAMAGE CLAIMS

**§ 8.4.1** In the event the Contractor is delayed or hindered in the commencement or progress of the Work, including but not limited to those delays caused by the Work or lack of Work of the Architect or the Owner's consultants, and the Contractor claims monetary damages as a direct and proximate consequence thereof (including, but not limited to, extended general conditions, overhead, profit, overtime, interest, supervision or other costs or profits whatsoever), then the Contractor shall not assert such claims against the Owner and, as to the Owner, the Contractor's claims of such delay damages are hereby waived. The Contractor's sole and exclusive remedy regarding claims for monetary delay damages shall be to pursue such claims directly against any individual or entity which may have caused the delay.

**§ 8.4.2** For any delay claims raised against the Owner, the Contractor's sole and exclusive remedy is an extension of time to perform the Work not to exceed the time frame of any proven delay. Under no circumstances is the Contractor entitled to monetary delay damages from the Owner.

## ARTICLE 9 PAYMENTS AND COMPLETION

#### § 9.1 CONTRACT SUM

The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

#### § 9.2 SCHEDULE OF VALUES

Before the first Application for Payment, the Contractor shall submit a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment.

#### § 9.3 APPLICATIONS FOR PAYMENT

**§ 9.3.1** At least fourteen (14) days before the date established for each progress payment, the Contractor shall submit to the Architect an itemized Application for Payment prepared in accordance with the schedule of values for completed portions of the Work. Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to payment as the Owner or Architect may require, such as copies of requisitions from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents. Applications for Payment are due to the office of the Construction Manager by the designated day of the month. Applications for Payment that are received after the specified date will not be processed until the following month.

**§ 9.3.1.1** As provided in Section 7.3.9, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Architect, but not yet included in Change Orders. A request for payment of sums related to work regarding Construction Change Directives shall, unless qualified in writing at the time of request, constitute full and complete consent to the Construction Change Directive(s) and to the issuance of a Change Order.

§ 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

§ 9.3.1.3 The Contractor shall make available to the Owner, upon request, along with the Application for Payment (1) an Affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the previous Application was submitted and the Owner might in any way be responsible have been paid or otherwise satisfied, and (2) a release or waiver of liens arising out of the Contract from each Contractor and/or Subcontractor, materialman, supplier and laborer of the Contractor addressing all previous Applications for Payment submitted for the Project.

§ 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site. Payment to Contractor for materials stored off site is discouraged. When circumstances indicate that the Owner's best interest is served by off-site storage, the Contractor shall make written request to the Owner for approval to include such material costs in his next progress payment. The Contractor's request shall include the following information:

- .1 A list of the fabricated materials consigned to the Project (which shall be clearly identified, giving the place of storage, together with copies of invoices and reasons why materials cannot be delivered to the site.
- .2 Certification that items have been tagged for delivery to the Project and that they will not be used for another purpose.
- .3 A letter from the Contractor's Surety indicating agreement to the arrangements and that payment to the Contractor shall not relieve either party of their responsibility to complete the Work.
- .4 Evidence of adequate insurance covering the material in storage, which shall name the Owner as additionally insured.
- .5 Costs incurred by the Architect to inspect material in off-site storage shall be paid by the Contractor.
- .6 Subsequent pay requests shall itemize the materials and their cost which were approved on previous pay requests and remain in off-site storage.
- When a partial payment is allowed on account of material delivered on the site of the Work or in the .7 vicinity thereof or under possession and control of the Contractor, but not yet incorporated therein, such material shall become the property of the Owner, but if such material is stolen, destroyed or damaged by casualty before being used, the Contractor will be required to replace it at its own expense.

§ 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims. security interests or encumbrances in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the Work.

#### § 9.4 CERTIFICATES FOR PAYMENT

§ 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in whole or in part as provided in Section 9.5.1.

§ 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's evaluation of the Work and the data comprising the Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion and to specific qualifications expressed by the Architect, in writing, together with the certification to which it pertains. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed, unless otherwise required by the contract or law, construction means, methods,

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

#### § 9.5 DECISIONS TO WITHHOLD CERTIFICATION

§ 9.5.1 The Architect may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of

- .1 defective Work not remedied, or the Contractor in default on the Agreement;
- .2 third party claims filed or reasonable evidence indicating probable filing of such claims unless security acceptable to the Owner is provided by the Contractor;
- .3 failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment;
- .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
- .5 damage to the Owner or a separate contractor;
- .6 reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay;
- .7 repeated failure to carry out the Work in accordance with the Contract Documents;
- .8 the Work not having progressed to the extent set forth in the Application for payment; or
- .9 representations of the Contractor are untrue.
- .10 failing to conform to project schedule
- .11 default in the performance of any obligation to Owner under another contract
- .12 failure to provide sufficiently skilled workers.

**§ 9.5.2** When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.

**§ 9.5.3** If the Architect withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or material or equipment suppliers to whom the Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Architect will reflect such payment on the next Certificate for Payment.

**§ 9.5.4** If the Contractor disputes any determination by the Owner with regard to any Certificate for Payment, the Contractor shall nevertheless continue to expeditiously perform the Work and such dispute shall provide no basis for any manner of suspension of the Contractor's performance of the Work.

#### § 9.6 PROGRESS PAYMENTS

**§ 9.6.1** After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Architect.

**§ 9.6.2** The Contractor shall pay each Subcontractor no later than seven days after receipt of payment from the Owner the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.

**§ 9.6.3** The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

**§ 9.6.4** The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors to ascertain whether they have been properly paid. Neither the Owner nor Architect shall have an obligation to pay or to see to the payment of money to a Subcontractor, except as may otherwise be required by law.

**§ 9.6.5** Contractor payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.

**§ 9.6.6** A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.

**§ 9.6.7** Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors and suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, shall create any fiduciary liability or tort liability on the part of the Contractor for breach of trust or shall entitle any person or entity to an award of punitive damages against the Contractor for breach of the requirements of this provision.

**§ 9.6.8** Subject to applicable law, if a petition in bankruptcy or any other arrangement or proceeding regarding insolvency, assignment for the benefit of creditors, trust, chattel mortgage, or similar state or federal proceeding, whether voluntary or involuntary, shall be filed with respect to the Contractor, the Owner may withhold the final balance, or any other payments, whether or not an application for progress payment has been properly filed, until expiration of the period of any guarantees or warranties required for the Contractor, and the Owner may pay out such funds the amount necessary to satisfy any claims or costs that otherwise would have been covered by such guarantees or warranties.

#### § 9.7 FAILURE OF PAYMENT

If without justifiable basis under the Contract Documents the Owner does not pay the Contractor within seven days after the date established in the Contract Documents the undisputed amount asserted by the Contractor in its Application for Payment or awarded by a court, then the Contractor may, upon 21 additional days' written notice to the Owner and Architect, stop the Work until payment of the undisputed amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shut-down, delay and start-up, plus interest as provided for in the Contract Documents.

#### § 9.8 SUBSTANTIAL COMPLETION

§ 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents and when all required occupancy permits, if any, have been issued, so that the Owner can occupy or utilize the Work for its intended use.

**§ 9.8.2** When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.

**§ 9.8.3** Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item immediately. In such case, the Contractor shall then submit a request for another inspection by the Architect to determine Substantial Completion.

**§ 9.8.4** When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163) Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.

**§ 9.8.5** The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate. Upon such acceptance and consent of surety, if any, the Owner shall make payment of retainage applying to such Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

**§ 9.8.6** Notwithstanding Sections 9.8.1 and 9.8.2, as a condition precedent to establishing the date of Substantial Completion, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or correct (a "punch list"). The Contractor shall respond immediately to correct Work deficiencies and/or punch list items. Should the Contractor fail to make corrections in a timely fashion, but not later than thirty (30) calendar days from the date of Substantial Completion or notification of the required corrections, whichever is earlier, such Work may be corrected by the Owner at the Contractor's sole expense, and the Contract Sum may be adjusted accordingly.

#### § 9.9 PARTIAL OCCUPANCY OR USE

**§ 9.9.1** The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer as required under Section 11.3.1.5 and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete. The Contractor shall proceed with the work in such a manner as reasonably directed and shall cooperate with the Owner to limit interruptions. In the event of such partial occupancy or use, the Owner shall determine responsibilities of the Owner and Contractor with respect to payments, retainage if any, security, maintenance, heat, utilities, damage to the Work, insurance, and the commencement of any warranties required by the Contract Documents.

**§ 9.9.2** Immediately prior to such partial occupancy or use, the Owner, Contractor and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

**§ 9.9.3** Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

**§ 9.9.4** Any agreement as to the acceptance of non-conforming Work not complying with the requirements of the Contract Documents, shall be in writing in the form of a Change Order, acceptable to the Owner's authorized representative and signed by all parties.

#### § 9.10 FINAL COMPLETION AND FINAL PAYMENT

§ 9.10.1 Upon receipt of the Contractor's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection and, when the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

**§ 9.10.2** Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner, (3) a written statement that the Contract Documents, (4) consent of surety, if any, to final payment, (5) an affidavit that states the Work is fully completed and performed in accordance with the Contract Documents, (6) in the event of Contractor bankruptcy, at the Owner's option, an order entered by the court having jurisdiction of the Contractor's insolvency proceeding authorizing such payment; (7) a general release executed by the Contractor on a form provided by the Architect ,

and (8) if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and actual attorneys' fees.

**§ 9.10.3** If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims.

§ 9.10.4 The making of final payment shall not constitute a waiver of Claims by the Owner.

**§ 9.10.5** Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver of all claims of the Contractor, except those previously made by the Contractor in writing and identified by the Contractor as unsettled at the time of final Application for Payment and specifically referenced as being an exception to the waiver contained in this section.

# ARTICLE 10 PROTECTION OF PERSONS AND PROPERTY § 10.1 SAFETY PRECAUTIONS AND PROGRAMS

The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract.

#### § 10.2 SAFETY OF PERSONS AND PROPERTY

§ 10.2.1 The Contractor shall take every reasonable precaution for safety of, and shall provide reasonable protection to prevent damage, injury or loss to

- .1 employees on the Work and other persons who may be affected thereby;
- .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors; and
- .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.

**§ 10.2.2** The Contractor shall take all reasonable safety precautions with respect to its Work and the work of others, shall comply with all standard industry safety measures and shall comply with all applicable laws, ordinances, rules, regulations and orders of any public authority and all other requirements of the Contract Documents, including those applicable to the safety of persons or property. The Contractor shall be responsible for the safety of all of the Contractor's employees and the safety of all of the Contractor's Subcontractors, suppliers, and their employees. The Contractor shall report in writing to the Architect any injury to any of to any of Contractor's or its Subcontractors' employees at the site within one (1) day after the occurrence of such injury.

**§ 10.2.3** The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable, necessary and appropriate safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities.

**§ 10.2.4** When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel. The Contractor shall be solely and fully responsible for any and all damage claims and for defense of all actions against the Owner relating to such explosives, hazardous materials and/or unusual methods.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

§ 10.2.5 The Contractor shall promptly remedy damage and loss to property referred to in Sections 10.2.1.2 and 10.2.1.3 caused in whole or in part by the Contractor, a Subcontractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.

§ 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect.

§ 10.2.7 The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

§ 10.2.7.1 The Contractor acknowledges that the safety of the Owner's students, employees and guests is of the utmost importance. The Contractor will take no action which would unnecessarily jeopardize the safety of the Owner's students, employees and guests and, without the Owner's written approval, shall take no action which would unreasonably interfere with the Owner's activities. Without limiting the foregoing provisions, the Contractor shall comply with all laws applicable to students and/or school safety.

#### § 10.2.8 INJURY OR DAMAGE TO PERSON OR PROPERTY

If Contractor suffers injury or damage to person or property because of an act or omission of the Owner, or of others for whose acts the Owner is legally responsible, written notice of such injury or damage, whether or not insured, shall be given to the Owner within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the Owner to investigate the matter. The Contractor's failure to do so shall be an irrevocable waiver of any claim arising out of such injury or damage. Injury or damage to persons or property suffered by the Owner because of an act or omission of the Contractor or others for whose acts the Contractor is legally responsible shall be subject to the limitations provisions established by Michigan law.

§ 10.2.8.1 The Contractor causing damage to the Work of another shall be responsible for the repair and replacement of such damaged Work. Back charges shall be made against the Contract sum of the damaging Contractor when corrections are not made promptly.

§ 10.2.8.2 If the Contractor or any Subcontractor chooses to use any systems, equipment, facilities, or services which have been incorporated in the Project as a permanent part thereof by any other, the Contractor shall assume full responsibility for damages caused to said systems, equipment, facilities or services, and have damages repaired as required, so that in no case will the performance of the used systems, equipment, facilities or services be diminished from the specified criteria as a result of such use.

#### § 10.3 HAZARDOUS MATERIALS

§ 10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner and Architect in writing.

§ 10.3.2 Upon receipt of the Contractor's written notice, the Owner, in its discretion, shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall, as a courtesy, furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriatelyto address shut-down, delay and start-up.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847 1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

**§ 10.3.4** The Owner shall not be responsible under this Section 10.3 for materials or substances the Contractor brings to the site. To the extent the Contract requires the removal, transport and disposal of hazardous materials, the Contractor agrees that it assumes responsibility for said tasks as a part of the Agreement.

§ 10.3.5 The Contractor shall indemnify the Owner for the cost and expense the Owner incurs (1) for remediation of a material or substance the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.

#### § 10.4 EMERGENCIES

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's reasonable discretion, to prevent threatened damage, injury or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7. Nothing in this paragraph will be construed as relieving Contractor from the cost and responsibilities for emergencies covered hereby.

#### § 10.5 NOTIFICATION OF UTILITY COMPANIES

**§ 10.5.1** At least five (5) working days prior to the start of work in areas which may involve existing utility lines, the Contractor shall notify by certified mail with return receipt requested, the Registered Utility Protection Service of the utility company possibly affected by the planned work.

**§ 10.5.2** The utility company should, upon receipt of notice, stake, mark or otherwise designate the location (and depth) of their lines, or temporarily move the line(s).

**§ 10.5.3** The Contractor shall immediately report to the respective utility company any break or leak in its lines, or any dent, gouge, groove or other damage to the utility line or to its coating or cathodic protection made or discovered in the course of the Work.

**§ 10.5.4** The Contractor shall immediately alert the Owner, Architect and occupants of nearby premises of any and all emergencies caused or discovered in the utility line(s) in the course of the Work.

#### ARTICLE 11 INSURANCE AND BONDS

#### § 11.1 CONTRACTOR'S LIABILITY INSURANCE

§ 11.1.1 The Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located such insurance as will protect the Contractor from claims set forth below which may arise out of or result from the Contractor's operations and completed operations under the Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or by a Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:

- .1 Claims under workers' compensation, disability benefit and other similar employee benefit acts that are applicable to the Work to be performed;
- .2 Claims for damages because of bodily injury, occupational sickness or disease, or death of the Contractor's employees;
- .3 Claims for damages because of bodily injury, sickness or disease, or death of any person other than the Contractor's employees;
- .4 Claims for damages insured by usual personal injury liability coverage;
- .5 Claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
- .6 Claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle;
- .7 Claims for bodily injury or property damage arising out of completed operations;
- .8 Claims involving contractual liability insurance applicable to the Contractor's obligations under Section 3.18
- .9 liability insurance shall include all major divisions of coverage and e on a comprehensive basis including:

- .1 Premises' Operations (deleting X, C, or U exclusions);
- .2 Owner's and Contractor's Protective;
- .3 Products and Completion Operations;
- .4 Contractual including specific for the Contractor's obligations under Paragraph 3.18;
- .5 Any auto; and
- .6 Broad Form Property Damage, including Completed Operations; and
- .10 All bonds required by law, including bid bond, performance bond and payment bond.

**§ 11.1.2** The insurance required by Section 11.1.1 shall be written for not less than limits of liability specified in the Contract Documents or required by law, whichever coverage is greater. Coverages, whether written on an occurrence or claims-made basis, shall be maintained without interruption from the date of commencement of the Work until the date of final payment and termination of any coverage required to be maintained after final payment, and, with respect to the Contractor's completed operations coverage, until the expiration of the period for correction of Work or for such other period for maintenance of completed operations coverage as specified in the Contract Documents.

**§ 11.1.3** Certificates of insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work and thereafter upon renewal or replacement of each required policy of insurance. These certificates and the insurance policies required by this Section 11.1 shall contain a provision that coverages afforded under the policies will not be canceled, reduced or allowed to expire until at least 30 days' prior written notice has been given to the Owner. An additional certificate evidencing continuation of liability coverage, including coverage for completed operations, shall be submitted with the final Application for Payment as required by Section 9.10.2 and thereafter upon renewal or replacement of such coverage on account of revised limits or claims paid under the General Aggregate, or both, shall be furnished by the Contractorto the Owner and Architect no less than thirty days prior to any reduction of coverage.

**§ 11.1.4** The Contractor shall cause the commercial liability coverage required by the Contract Documents to include (1) the Owner, the Architect and the Architect's Consultants as additional insureds for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's operations; and (2) the Owner as an additional insured for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's negligent

**§ 11.1.5** On all insurance contracts under which the Contractor is obligated to have its insurance company name the Owner as additional insured, the Contractor shall require such insurance company to add to the policy the following clause: "The insurance afforded to the Additional Insured is primary insurance. If the Additional Insureds have other insurance which is applicable to the loss on an excess or contingent basis, the amount of the insurance company's liability under this policy shall not be reduced by the existence of such other insurance."

#### § 11.2 OWNER'S LIABILITY INSURANCE

The Owner shall be responsible for purchasing and maintaining the Owner's usual liability insurance.

#### § 11.3 PROPERTY INSURANCE

**§ 11.3.1** Unless otherwise provided, the Owner shall purchase and maintain, in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located, property insurance written on a builder's risk "all-risk" or equivalent policy form in the amount of the initial Contract Sum, plus value of subsequent Contract Modifications and cost of materials supplied or installed by others, comprising total value for the entire Project at the site on a replacement cost basis without optional deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Section 9.10 or until no person or entity other than the Owner has an insurable interest in the property required by this Section 11.3 to be covered, whichever is later. This insurance shall include interests of the Owner, the Contractor, Subcontractors and Subsubcontractors in the Project. The term "temporary building", as used in this section, shall not include job trailers or any party.

This policy will exclude any tools, equipment, scaffolding, glass breakage, etc. owned or rented by the Contractor or Subcontractors and material stored on the site but not incorporated into the Project. The Contractor shall be responsible for protecting all product until the Date of Substantial Completion is established by the

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

Architect/Engineer. The Contractor shall replace any Work if damaged before Substantial Completion. The contractor may assume the risk itself or obtain insurance in amounts it deems sufficient.

**§ 11.3.1.1** Property insurance shall be on an "all-risk" or equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, earthquake, flood, windstorm, falsework, testing and startup, temporary buildings and debris removal including demolition occasioned by enforcement of any applicable legal requirements, and shall cover reasonable compensation for Architect's and Contractor's services and expenses required as a result of such insured loss.

**§ 11.3.1.2** If the Owner does not intend to purchase such property insurance required by the Contract and with all of the coverages in the amount described above, the Owner shall so inform the Contractor in writing prior to commencement of the Work. The Contractor may then effect insurance that will protect the interests of the Contractor, Subcontractors and Sub-subcontractors in the Work, and by appropriate Change Order the cost thereof shall be charged to the Owner. If the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain insurance as described above, without so notifying the Contractor in writing, then the Owner shall bear all reasonable costs properly attributable thereto.

§ 11.3.1.3 If the property insurance requires deductibles, the Contractor shall pay the deductibles.

**§ 11.3.1.4** This property insurance shall cover portions of the Work stored off the site, and also portions of the Work in transit.

**§ 11.3.1.5** Partial occupancy or use in accordance with Section 9.9 shall not commence until the insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Owner and the Contractor shall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse or reduction of insurance.

#### § 11.3.2 BOILER AND MACHINERY INSURANCE

The Owner shall purchase and maintain boiler and machinery insurance required by the Contract Documents or by law, which shall specifically cover such insured objects during installation and until final acceptance by the Owner; this insurance shall include interests of the Owner, Contractor, Subcontractors and Sub-subcontractors in the Work, and the Owner and Contractor shall be named insureds.

#### § 11.3.3 LOSS OF USE INSURANCE

The Owner, at the Owner's option, may purchase and maintain such insurance as will insure the Owner against loss of use of the Owner's property due to fire or other hazards, however caused.

§ 11.3.4 If the Contractor requests in writing that insurance for risks other than those described herein or other special causes of loss be included in the property insurance policy, the Owner shall, if possible, include such insurance, and the cost thereof shall be charged to the Contractor by appropriate Change Order.

**§ 11.3.6** Before an exposure to loss may occur, the Owner shall file with the Contractor a copy of each policy that includes insurance coverages required by this Section 11.3. Each policy shall contain all generally applicable conditions, definitions, exclusions and endorsements related to this Project. Each policy shall contain a provision that the policy will not be canceled or allowed to expire, and that its limits will not be reduced, until at least 30 days' prior written notice has been given to the Contractor.

#### § 11.3.7 WAIVERS OF SUBROGATION

Superior to any other term in this Agreement, any reference in the General conditions to "waiver of subrogation" or such similar term are hereby deleted and shall be declared to no effect.

**§ 11.3.8** A loss insured under the Owner's property insurance shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Section 11.3.10. The Contractor shall pay Subcontractors their just shares of insurance proceeds received by the Contractor, and by appropriate agreements, written where legally required for validity, shall require Subcontractors to make payments to their Sub-subcontractors in similar manner.

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

#### § 11.3.9

**§ 11.3.10** The Owner shall use its best efforts, with consultation of the Construction Manager, to reach a quick and fair settlement for all interested parties, with the insurance companies after a loss.

#### § 11.4 PERFORMANCE BOND AND PAYMENT BOND

**§ 11.4.1** The Owner hereby requires the Contractor to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder each in the penal sum of 100% of the Contract Sum and in accordance with applicable law on the date of execution of the Contract.

**§ 11.4.2** Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.

**§ 11.4.3** The Contractor shall deliver copies of the Performance Bond and Payment Bond required by the Agreement to the Owner prior to Contractor beginning performance pursuant to the Agreement. The Contractor's obligation to provide such bonds shall not be waived in any fashion, including any failure to secure such bonds prior to Contractor beginning performance pursuant to the Agreement.

# ARTICLE 12 UNCOVERING AND CORRECTION OF WORK § 12.1 UNCOVERING OF WORK

§ 12.1.1 If a portion of the Work is covered contrary to the Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Architect, be uncovered for the Architect's examination and be replaced at the Contractor's expense without change in the Contract Time or Contract Sum.

**§ 12.1.2** If a portion of the Work has been covered that the Architect has not specifically requested to examine prior to its being covered, the Architect may request with the Owner's consent to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. At the time Owner's consent is sought as described herein, the Architect shall notify the Owner that additional costs may apply if the Work is in accordance with the Contract Documents, such costs and the cost of correction shall be at the Contractor's expense unless the condition was caused by the Owner or a separate contractor in which event the Owner or the separate contractor shall be responsible for payment of such costs.

#### § 12.2 CORRECTION OF WORK

#### § 12.2.1 BEFORE OR AFTER SUBSTANTIAL COMPLETION

If any portion of the Work is determined by the Owner or Architect, either during performance of the Work or during any applicable warranty period, to be defective or not in compliance with the requirements therefor, the Owner shall notify the Contractor in writing that such Work is rejected. Thereupon, the Contractor shall immediately replace and/or correct such Work by making the same comply strictly with all the requirements therefor. The Contractor shall bear all costs of correcting such rejected Work, including work of other Subcontractors and including compensation for the Architect's and Construction Manager's additional services and any delay or related damages to the owner made necessary thereby. The Owner shall have the right to charge the Contractor for any compensation payable for the Architect's or Construction Manager's additional services required by the Contractor's rejected Work and deduct the payment from the next payment due the Contractor.

#### § 12.2.2 AFTER SUBSTANTIAL COMPLETION

**§ 12.2.2.1** In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.4.

AIA Document A201<sup>28</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

**§ 12.2.2** The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.

**§ 12.2.3** The one-year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Section 12.2.

**§ 12.2.3** The Contractor shall remove from the site portions of the Work that are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.

**§ 12.2.4** The Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner or separate contractors caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.

**§ 12.2.5** Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one-year period for correction of Work as described in Section 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

**§ 12.2.6** The Contractor shall respond immediately to correct Work deficiencies and/or punch list items. Failure to correct Work deficiencies and/or punch list items in a timely fashion shall be a material breach, and the Owner may terminate the Contract. Whether or not the Contract is terminated, if the Contractor fails to make corrections in a timely fashion, such Work may be corrected by the Owner, in its sole discretion, at the Contractor's expense and the Contract Sum may be adjusted by backcharge accordingly. The Contractor shall promptly notify the Architect in writing when Work deficiencies and/or punch list items are completed. If upon review of the Work by the Architect, after such notification by the Contractor, Work deficiencies and/or punch list items shall continue to exist, the Contractor shall reimburse the Owner for any costs incurred by the Owner. Failure to pay such costs within ten (10) days of receipt of a demand regarding the same shall permit the Owner to pay such costs out of retainage held by the Owner on the Contractor's contract.

#### § 12.3 ACCEPTANCE OF NONCONFORMING WORK

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made. The acceptance of nonconforming Work by the Owner shall be by written Change Order signed by the Owner's authorized representative. Acceptance of nonconforming Work may only occur pursuant to such written Change Order.

## ARTICLE 13 MISCELLANEOUS PROVISIONS § 13.1 GOVERNING LAW

The Contract shall be governed by the law of the State of Michigan in all respects, except that claims and causes of action brought by the Owner shall not be deemed untimely if filed within six (6) years of substantial completion of the Project.

#### § 13.2 SUCCESSORS AND ASSIGNS

§ 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns and legal representatives to covenants, agreements and obligations contained in the Contract Documents. Except as provided in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

**§ 13.2.2** The Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate such assignment.

AIA Document A201<sup>ax</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1967, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

#### § 13.3 WRITTEN NOTICE

Written notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended; or if delivered at, or sent by registered or certified mail or by national overnight courier service providing a tracking system and proof of delivery to, the last business address known to the party giving notice.

Wherever the Contract Documents require the Contractor to give "Notice" or "Timely Notice" to the Architect, Public Authority, and/or others, it shall be the Contractor's responsibility to furnish all such notices sufficiently in advance to allow the party receiving the notice reasonable time to react to such notice, including travel time on the job sit as necessary, when such notices require the on-site presence of the Architect, Public Authority, their authorized representatives, or others for field observation of inspections, testing or approvals. Reasonable time shall be defined as no less than 24 hours plus normal travel time from the home office of the party being notified to the job site and must also accommodate known, standard, or reasonable processing periods.

#### § 13.4 RIGHTS AND REMEDIES

§ 13.4.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.

§ 13.4.2 No action or failure to act by the Owner, Architect or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach there under, except as may be specifically agreed in writing.

#### § 13.5 TESTS AND INSPECTIONS

§ 13.5.1 Tests, inspections and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made so that the Architect may be present for such procedures. The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable laws or regulations prohibit the Owner from delegating their cost to the Contractor.

**§ 13.5.2** If the Architect, Owner or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Section 13.5.1, the Architect will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Architect of when and where tests and inspections are to be made so that the Architect may be present for such procedures. Such costs, except as provided in Section 13.5.3, shall be at the Owner's expense.

**§ 13.5.3** If such procedures for testing, inspection or approval under Sections 13.5.1 and 13.5.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents or applicable law, all costs made necessary by such failure including those of repeated procedures and compensation for the Architect's services and expenses shall be at the Contractor's expense.

**§ 13.5.4** Required certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Architect.

**§ 13.5.5** If the Architect is to observe tests, inspections or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.

§ 13.5.6 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid delay in the Work.

#### § 13.6 INTEREST

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

#### § 13.7 TIME LIMITS ON CLAIMS

§ 13.7.1 The Owner shall commence all claims and causes of action in accordance with Section 13.1, regardless of any other time frames identified in this Agreement. The Contractor shall commence all claims and causes of action in accordance with Section 15.1, other provisions of this Agreement and in accordance with Michigan law.

**§ 13.7.2** Regardless of any provisions to the contrary, the statute of limitations with respect to any claim by the Owner with respect to defective or nonconforming Work shall not commence until the discovery of such defective or nonconforming Work by the Owner. See also Section 13.1.

#### § 13.8 SURETY NOTICE AND PRIOR APPROVAL

Except where otherwise expressly required by the terms of the Agreement, the Contract Documents or the General Conditions, exercise by the Owner of any contractual or legal right or remedy without prior notice to or approval by the Contractor's surety shall in no way bar or prohibit the Owner's ability to pursue such right or remedy. Further, pursuit of such a right or remedy without prior notice to or approval of surety shall in no way compromise, limit or bar any claim by the Owner against a surety bond of the Contractor.

## ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT § 14.1 TERMINATION BY THE CONTRACTOR

§ 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 30 consecutive days through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, for any of the following reasons:

- .1 Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be stopped;
- .2 An act of government, such as a declaration of national emergency that requires all Work to be stopped;
- .3 Because the Owner has not made payment on an undisputed Certificate for Payment within the time stated in the Contract Documents, subject to justifiable withholding of payment as described herein or in the contract documents; or
- .4 The Owner has failed to furnish to the Contractor promptly, upon the Contractor's request, reasonable evidence as required by Section 2.2.1.

**§ 14.1.2** The Contractor may terminate the Contract if, through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, repeated suspensions, delays or interruptions of the entire Work by the Owner as described in Section 14.3 constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.

**§ 14.1.3** If one of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' written notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work executed, including reasonable overhead and profit, costs incurred by reason of such termination, and damages.

**§ 14.1.4** If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing portions of the Work under contract with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

#### § 14.2 TERMINATION BY THE OWNER FOR CAUSE

- § 14.2.1 The Owner may terminate the Contract if the Contractor
  - .1 refuses or fails to supply enough properly skilled workers or proper materials to the point of negatively impacting the Project and/or the related schedule;

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

- .2 fails to make payment to Subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors;
- .3 disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority;.4 otherwise is guilty of substantial breach of a provision of the Contract Documents; or
- .5 the Contractor fails to prosecute the Work or any part thereof with promptness and diligence or fails to perform any provisions of this Contract, or goes into bankruptcy, liquidation, makes an assignment for the benefit of creditors, enters into a composition with its creditors, or becomes insolvent.

**§ 14.2.2** When any of the above reasons exist, the Owner, upon certification by the Initial Decision Maker that sufficient cause exists to justify such action, may without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' written notice, terminate the Contractor's right to proceed with the Work, or such part of the Work as to which such defaults have occurred, and may take any one or more of the following actions:

- .1 Exclude the Contractor from the site and take possession of all materials, equipment, tools, and construction equipment and machinery thereon owned by the Contractor;
- .2 Accept assignment of subcontracts pursuant to Section 5.4; and
- .3 Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work.

In the event the Contractor's surety bond requires notice of intent to declare a default of the Contractor and if such bond notice is provided by the Owner, such notice shall be adequate to satisfy the seven (7) day written notice described above in this section.

**§ 14.2.3** When the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.

**§ 14.2.4** If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner in pursuing termination and completion of the Work, including actual attorney and legal fees and costs, and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case may be, shall be certified by the Initial Decision Maker, upon application, and this obligation for payment shall survive termination of the Contract.

### § 14.3 SUSPENSION BY THE OWNER FOR CONVENIENCE

**§ 14.3.1** The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine.

**§ 14.3.2** The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in Section 14.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent

- .1 that performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible; or
- .2 that an equitable adjustment is made or denied under another provision of the Contract.

#### § 14.4 TERMINATION BY THE OWNER FOR CONVENIENCE

§ 14.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.

**§ 14.4.2** Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor shall

- .1 cease operations as directed by the Owner in the notice;
- .2 take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; and
- .3 except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.

**§ 14.4.3** In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination.

#### ARTICLE 15 CLAIMS AND DISPUTES § 15.1 CLAIMS § 15.1.1 DEFINITION

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, or other relief with respect to the terms of the Contract, including but not limited to additional sums, additional time for performance, or damages for delay. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim. The Contractor shall not knowingly (as "knowingly" is defined in the Federal False Claims Act, 31 USC 3729, *et seq.*) present or cause to be presented a false or fraudulent Claim. As a condition precedent to making a Claim by the Contractor, the Claim shall be accompanied by an affidavit sworn to before a notary public or other person authorized to administer oaths in the State of Michigan and executed by an authorized representative of the Contractor, which states that: "The Claim which is submitted herewith complies with Section 15.1.1 of the General Conditions, as amended, which provides that the Contractor shall not knowingly present or cause to be presented a false or shall not knowingly present or cause to be presented a false or shall not knowingly authorized herewith complies with Section 15.1.1 of the General Conditions, as amended, which provides that the Contractor shall not knowingly present or cause to be presented a false or fraudulent claim."

#### § 15.1.2 NOTICE OF CLAIMS

Claims by Contractor must be initiated by written notice to the other party and to the Initial Decision Maker with a copy sent to the Architect, if the Architect is not serving as the Initial Decision Maker. Claims by the Contractor must be initiated within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the Contractor first recognizes the condition giving rise to the Claim, whichever is later. Failure to timely and properly initiate a claim shall be an irrevocable waiver of such claim. Claims by the Owner shall be governed by the applicable statute of limitations period, except as such time frame may be longer in accordance with Section 13.1 and Section 13.7.

#### § 15.1.3 CONTINUING CONTRACT PERFORMANCE

Pending final resolution of a Claim, including mediation, arbitration and/or litigation, as applicable, except as otherwise agreed in writing or as provided in Section 9.7 and Article 14, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make undisputed payments in accordance with the Contract Documents. The Architect will prepare Change Orders and issue Certificates for Payment in accordance with the decisions of the Initial Decision Maker.

#### § 15.1.4 CLAIMS FOR ADDITIONAL COST

If the Contractor wishes to make a Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the Work. Failure to provide such notice shall serve as an absolute bar against a claim for such an increase in the Contract Sum. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4. A Project delay shall not be a basis for a Claim for additional cost. Delays may be remedied only through an extension of time per Section 15.1.5.

#### § 15.1.5 CLAIMS FOR ADDITIONAL TIME

**§ 15.1.5.1** If the Contractor wishes to make a Claim for an increase in the Contract Time, additional written notice as provided herein shall be given in addition to the general requirements for filing a claim. The Contractor's Claim shall include an estimate of probable effect of delay on progress of the Work, due to the increase in Contract Time sought. In the case of a continuing delay, only one Claim is necessary.

**§ 15.1.5.2** If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated and had an adverse effect on the scheduled construction.

§ 15.1.6

#### § 15.2 INITIAL DECISION

**§ 15.2.1** Claims, excluding those arising under Sections 10.3, 10.4, 11.3.9, and 11.3.10, shall be referred to the Initial Decision Maker for initial interpretation. The Architect will serve as the Initial Decision Maker, Except for those Claims excluded by this Section 15.2.1, an initial interpretation shall be required as a condition precedent to mediation, arbitration and/or litigation, of any Claim brought by the Contractor against the Owner, arising prior to

AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

the date final payment is due, unless 30 days have passed after the Claim has been referred to the Initial Decision Maker with no interpretation having been rendered. Unless the Initial Decision Maker and all affected parties agree, the Initial Decision Maker will not decide disputes between the Contractor and persons or entities other than the Owner.

**§ 15.2.2** The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions: (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, (4) suggest a compromise, or (5) advise the parties that the Initial Decision Maker is unable to resolve the Claim if the Initial Decision Maker lacks sufficient information to evaluate the merits of the Claim or if the Initial Decision Maker concludes that, in the Initial Decision Maker's sole discretion, it would be inappropriate for the Initial Decision Maker to interpret the Claim. Within ten (10) days of a written request, the Contractor shall make available to the Owner or its representative all of its books, records, or other documents in its possession or to which it has access relating to a Claim and shall require its subcontractors, regardless of tier, and material men to do the same.

**§ 15.2.3** In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Decision Maker in rendering an interpretation. The Initial Decision Maker may request the Owner to authorize retention of such persons at the Owner's expense.

**§ 15.2.4** If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of such request, and shall either (1) provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished or (3) advise the Initial Decision Maker that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Initial Decision Maker will, based on its interpretation, either reject or approve the Claim in whole or in part.

**§ 15.2.5** The Initial Decision Maker will render an initial interpretation approving or rejecting the Claim, or indicating that the Initial Decision Maker is unable to resolve the Claim. This initial interpretation shall (1) be in writing; (2) state the reasons therefor; and (3) notify the parties and the Architect, if the Architect is not serving as the Initial Decision Maker, of any recommended change in the Contract Sum or Contract Time or both. If the Claim is timely and properly filed, the initial interpretation shall be subject to the parties' agreed-upon dispute resolution process.

**§ 15.2.6** Regardless of any other time frames identified herein, claims and causes of action brought by the Owner shall be governed in accordance with the statute of limitations periods under Michigan law, except for such longer periods of time as may be permitted in Section 13.1 and Section 13.7

**§ 15.2.7** In the event of a Claim against the Contractor, the Owner, Architect or Initial Decision Maker may, but are not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner, Architect or Initial Decision Maker may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.

#### § 15.3 MEDIATION

**§ 15.3.1** Except as stated in this Agreement or otherwise agreed in writing by the parties, claims arising out of or related to the Contract except those waived as provided for in Sections 9.10.4, 9.10.5, and 15.1.6 shall be subject to mediation as a condition precedent to the parties' agreed-upon dispute resolution process.

**§ 15.3.2** The parties shall endeavor to resolve their Claims by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the commencement of the parties' agreed-upon dispute resolution proceedings but, in such event, mediation shall proceed in advance of such proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order

AIA Document A201<sup>ax</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1967, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA<sup>®</sup> Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA<sup>®</sup> Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

**§ 15.3.3** The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

**§ 15.3.4** The parties shall share the mediator's fee and any filing fees equally. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof. During the mediation process, all applicable limitations period shall be tolled during the pendency of mediation.

#### § 15.4 ARBITRATION

#### § 15.4.4 CONSOLIDATION OR JOINDER

**§ 15.4.4.1** The Owner, at its sole discretion, may consolidate a mediation conducted under this Agreement with any other mediation to which it is a party provided that (1) the mediation agreement governing the other mediation permits consolidation, (2) the mediations to be consolidated substantially involve common questions of law or fact, and (3) the mediations employ materially similar procedural rules and methods for selecting mediator(s).

**§ 15.4.4.2** The Owner, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in mediation, provided that the party sought to be joined consents in writing to such joinder. Consent to mediation involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

Modified: 02/09/16; 11:25am





AIA Document A201<sup>24</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. WARNING: This AIA® Document is protected by U.S. Copyright Law and International Treaties. Unauthorized reproduction or distribution of this AIA® Document, or any portion of it, may result in severe civil and criminal penalties, and will be prosecuted to the maximum extent possible under the law. This draft was produced by AIA software at 09:39:39 on 06/02/2009 under Order No.1000386847\_1 which expires on 2/3/2010, and is not for resale. User Notes: (989098163)

## SECTION 007216 GENERAL CONDITIONS

## 1. <u>GENERAL</u>

- 1.1. The Trade Contractor shall be bound to Clark Construction Company by the terms of all Contract Documents of Clark Construction Company's Contract with the Owner including, but not limited to Drawings, Specifications, General Conditions, Special and Supplemental Conditions and Addenda, and to assume toward Clark Construction Company all obligations and responsibilities that Clark Construction Company by said documents, assume toward the Owner. The Trade Contractor binds himself to Clark Construction Company for the performance of Trade Contractor's Work in the same manner as Clark Construction Company is bound to the Owner for such performance under the prime Contract. The prime Contract or the pertinent parts of such Contract will be made available upon Trade Contractor's request.
  - A. Copies of the General Conditions are enclosed under section 006200.

## END OF SECTION

## 1. <u>GENERAL</u>

- 1.1. Trade Contractor shall provide general liability insurance, including comprehensive, products and completed operations, hazards, contractual and independent Contractors, as well as automobile liability, workman's compensation and employee liability coverage on all their employees and their Subcontractors' employees engaged in executing this Contract and to provide all other insurance coverage required by the Contract Documents to the Prime Contract.
- 1.2. The Trade Contractor agrees that the insurance coverage required by contract shall be continued throughout the duration of the warranty period specified by the contract. For Contracts that require Pollution Insurance coverage, such coverage shall be continued for three years beyond the substantial completion date of the project, or as specified by the Prime Contract, whichever duration is greater. Upon request by Clark Construction and/or Owner, Trade Contractor shall provide evidence/proof of insurance that the insurance coverage is being continued throughout the period specified by contact.
- 1.3. Trade Contractor shall carry, and require its Subcontractors to carry, identical insurance to that which the Construction Manager is required to carry under the Prime Contract. Trade Contractor's insurance shall include contractual liability coverage applicable to its obligations pursuant to this Contract.
- 1.4. Trade Contractor shall cause its insurance carriers to add the Construction Manager as an additional named insured on all such policies excluding Workers Compensation and automobile insurance prior to commencement of Work. Trade Contractor's insurance carriers shall be directed to notify the Construction Manager (thirty) 30 days in advance of cancellation of any insurance coverage required herein. The additional insured endorsement included on the Trade Contractor's operations performed by or on behalf of the Trade Contractor. If the additional insureds have other insurance which is applicable to the project, such other insurance shall be on an excess or contingent basis. The amount of the insurer's liability under this insurance policy shall not be reduced by the existence of such other insurance.
- 1.5. As a condition of being awarded a Contract, the Trade Contractor must provide Clark Construction Company with satisfactory evidence of insurance coverage as follows. (Refer to sample Certificate of Insurance included in Certificates and Other Forms Section).
  - A. Certificate of Insurance must be provided on the ACORD Certificate Form #25 or 25-S.
  - B. AM Best rating for each insurance carrier shall be at  $\underline{A, V}$  or better.
  - C. The issuing company shall provide a thirty (30) day written notice of cancellation to the certificate holder.
  - D. The Owner, Clark Construction Company and the Architect/Engineer shall be listed on the Certificate as <u>"Additional Insured's"</u> on the General Liability policy. The additional insured status shall include On-going Operations using <u>ISO General Liability Form #CG 20 10 01</u> and Products & Completed Operations using <u>ISO General Liability Form #CG 2037 10 01</u> or the insurance companies equivalent coverage endorsement.

- E. Clark Construction Company is the Certificate Holder.
- 1.6. The Trade Contractor shall purchase from and maintain a company or companies' lawfully licensed and authorized to do business in the jurisdiction in which the Project is located such insurance as will protect the Trade Contractor from claims set forth which may arise out of or result from the Trade Contractor's operations under the Contract and for which the Trade Contractor may be legally liable whether such operations be performed by the Trade Contractor or by a Subcontractor or by anyone for whose acts any of them may be liable:
  - A. Claims under workers' compensation disability benefit and other similar employee benefit acts which are applicable to the Work to be performed;
  - B. Claims for damages because of bodily injury, occupational sickness, disease or death of the Trade Contractor's employees;
  - C. Claims for damages because of bodily injury, sickness or disease or death of a person other than the Trade Contractor's employees;
  - D. Claims for damages insured by usual personal injury liability coverage which are sustained (1), by a person as a result of an offense directly or indirectly related to employment of such person by the Trade Contractor, or (2) by another person;
  - E. Claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
  - F. Claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of motor vehicle;
  - G. Claims involving contractual liability insurance applicable to the Trade Contractor's obligations under the Contract;
  - H. The Trade Contractor's liability insurance shall be written on a Commercial General Liability basis with coverage at least as broad as an <u>ISO General Liability</u> <u>Endorsement CG 0001 12 04</u> or later revised version and such coverage shall specifically include but shall not be limited to:
    - 1. Premises/Operations;
    - 2. Independent Trade Contractor's Protective;
    - 3. Products and Completed Operations:
    - 4. Personal Injury Liability with Employment Exclusion deleted;
    - 5. Contractual, including coverage for Trade Contractor's obligations under its Contract;
    - 6. Owned, non-owned and hired motor vehicles; and
    - 7. Broad Form Property Damage including Completed Operation.

1.7. The Trade Contractor shall, for the protection and benefit of the Owner, Clark Construction Company and the Architect/Engineer (hereinafter these parties are collectively referred to as "Trade Contractor Indemnities") and the Trade Contractor and as part of the Trade Contractor's efforts to satisfy the insurance obligation of Section 1. Procure, pay for and maintain in full force and effect at all times during the performance of the Trade Contractor's Work until final acceptance of the Trade Contractor's Work or for such duration as required policies of insurance issued by a responsible carrier or carriers acceptable to the Owner and Clark Construction Company which afford the coverages in the limits set forth below:

## 2. <u>LIMITS OF LIABILITY</u>

- 2.1. <u>Commercial General Liability</u> including Premises Operations: Independent Trade Contractor's Protective; Products and Completed Operations; Broad Form Property Damage; Personal and Advertising Injury.
  - A. General Aggregate shall not be less than <u>\$2,000,000/per occurrence</u> and it shall apply in total to this project only using the <u>ISO General Liability Endorsement</u> <u>CG 2503 03 97</u> or Trade Contractor's Insurance Carrier's equivalent coverage endorsement.
  - B. Products Completed/Operations Aggregate shall not be less than <u>\$2,000,000</u> aggregate limit.
  - C. Personal and Advertising Injury shall not be less than <u>\$1,000,000</u> aggregate.
  - D. Each Occurrence shall not be less than <u>\$1,000,000</u>.
  - E. Fire Damage shall not be less than \$100,000 on any one (1) fire.
  - F. Medical Expense shall not be less than \$10,000 on any one (1) person.
  - G. Property Damage Liability shall provide X, C and U coverages.
  - H. Broad Form Property Damage coverage shall include completed operations.
  - I. All such insurance shall be written on an occurrence basis.

## 2.2. <u>Automobile Liability</u>

- A. Automobile liability coverage shall be at least as broad as a CA 00 01 (3-10) or insurance companies equivalent endorsement.
- B. Coverage shall include: Any Auto, Hired Autos and Non-Owned Autos.
- C. Combined Single Limit shall not be less than <u>\$1,000,000.</u>
- D. All such insurance shall be written on an occurrence basis.

#### 2.3. Excess Liability

A. The Umbrella Form is to be provided.

- B. Each Occurrence shall not be less than \$1,000,000 over primary insurance.
- C. Self-Insured Retention (SIR) shall not be more than <u>\$10,000</u> for each occurrence.
- D. All such insurance shall be written on an occurrence basis.
- 2.4. <u>Workers Compensation and Employers Liability</u> (including the Proprietor/Partners/Executive Officers)
  - A. Workers Compensation coverage shall be at least as broad as an **ISO**
  - B. <u>Workers Compensation Endorsement WC 00 03 13</u> or insurance companies equivalent endorsement.
  - C. Each Accident shall not be less than \$1,000,000.
  - D. Disease Policy Limit shall not be less than \$1,000,000.
  - E. Disease Each Employee shall not be less than \$1,000,000.
  - F. All such insurance shall be written on an occurrence basis.

#### 2.5. **Pollution Liability**

- A. Trade Contractors responsible for remediation, including containerization, transportation, or disposal of any hazardous or toxic wastes, materials, or substances requiring permits or licenses by state or Federal Law or regulation shall maintain Pollution Legal Liability Insurance with limit no less than \$1,000,000.
- B. Coverage shall apply to sudden and gradual pollution legal liability including defense costs and completed operations.
- C. All such insurance shall be written on an occurrence basis.
- 2.6. The Trade Contractor hereby agrees to deliver to Clark Construction Company within seven (7) days of the date of the Contract and prior to any equivalent or personnel being brought onto the Project site, certified copies of all insurance policies procured by the Trade Contractor or with consent of the Owner and Clark Construction Company Certificates evidencing the required coverages with limits not less than those specified in Section 1.5 hereto. Trade Contractor's indemnities shall be included as additional insured on Trade Contractor's Commercial General Liability Policy and shall be primary coverage to any valid and collectible insurance carried separately by any of the Trade Contractor's Indemnities. Further, all policies and Certificates of Insurance shall expressly provide that not less than thirty (30) days prior written notice shall be given the Owner, Clark Construction Company, and the Trade Contractor's Indemnities in the event of material alteration, cancellation, non-renewal or expiration of the coverage contained in such policy or evidenced by such certified copy or Certificate of Insurance.
- 2.7. In no event shall any failure of Clark Construction Company to receive certified copies or certificates of policies required under Section 1.0 or to demand receipt of such certified copies or certificates prior to the Trade Contractor commencing the Work be construed as a waiver by the Owner, and Clark Construction Company, of the Trade Contractor's

obligations to obtain insurance pursuant to this Section 1.0. The obligation to procure and maintain any insurance required by this Section 1.0 is a separate responsibility of the Trade Contractor and independent of the duty to furnish a certified copy or certificate of such insurance policies.

- 2.8. If the Trade Contractor fails to purchase and maintain or require to be purchased and maintained, any insurance required under this Section, the Owner, or Clark Construction Company, may, but shall not be obligated to, upon five (5) days written notice to the Trade Contractor, purchase such insurance on behalf of the Trade Contractor and shall be entitled to be reimbursed by the Trade Contractor upon demand, or deduct the cost of insurance from the Contract amount.
- 2.9. When any required insurance, due to the attainment of a normal expiration date or renewal date shall expire, the Trade Contractor shall supply the Owner and Clark Construction Company Certificates of Insurance and amendatory riders or endorsements that clearly evidence the continuation of all coverage as is required by this Section. In the event any renewal or replacement policy, for whatever reason obtained or required is written by a carrier other than that with whom the coverage was previously placed or the subsequent policy differs in any way from the previous policy, the Trade Contractor shall also furnish the Owner and Clark Construction Company with a certified copy of the renewal or replacement policy unless, Clark Construction Company provides the Trade Contractor with prior written consent to submit only a Certificate of Insurance for any such policies. All renewal and replacement policies shall be in form and substance satisfactory to the Owner and Clark Construction Company.
- 2.10. Any aggregate limit under the Trade Contractor's liability insurance shall, by endorsement, apply to this Project separately.
- 2.11. The Trade Contractor shall cause each of its Subcontractors to (1) procure insurance reasonably satisfactory to the Owner and Clark Construction Company, and (2) name the Trade Contractor, the Owner, Clark Construction Company, and Architect as additional insured under the Subcontractor's comprehensive general liability policy. The additional insured endorsement included on the Subcontractor's comprehensive general liability policy shall state that coverage is afforded the additional insured with respect to claims arising out of operations performed by or on behalf of the Subcontractor. If additional insured have other insurance which is applicable to the Project, such other insurance shall be for the purpose hereof, on an excess or contingent basis. The amount of the insurer's liability under this insurance policy shall not be reduced by the existence of such other insurance.
- 2.12. The Trade Contractor shall provide for the Owner, Clark Construction Company, and Architect as additional named insured parties under the comprehensive general liability insurance and completed operations required herein, and the insurance shall be primary and non-contributory.

### 3. <u>INDEMNIFICATION</u>

3.1. The Trade Contractor shall secure, defend, protect, hold harmless and indemnify the Owner, Clark Construction Company, including any party that Clark Construction Company is required to indemnify, the Architect and any of their respective agents, servants and employees against any liability, loss, claims, demands, suits, costs, fines,

attorney fees, and any expenses whatsoever arising from bodily injury, sickness, disease (including death resulting therefrom), of any persons, or the damage or destruction of any property or Work required by this Contract including loss of use, arising out of, or in connection with, the performance of any Work relating to this Contract including additional Work assigned to the Trade Contractor, based upon any act or omission, negligent or otherwise, (including active or passive negligence) of (a) the Trade Contractor or any of its agents, employees or servants, (b) any Sub/Subcontractor, supplier, or materialman of the Trade Contractor, or any agents, employees or servants thereof, (c) any other person or persons. The obligations of indemnification contained herein shall exclude only those matters in which the claim is caused by the sole negligence of the Owner, the Architect, Clark Construction Company, or any of their respective agents, servants and employees. The obligations herein shall apply regardless of whether such suits, damages, judgments, liabilities, interest, attorney fees, costs, etc., are caused in part by a party indemnified hereunder or by anyone acting under their direction, control, or on their behalf. Also the obligations herein shall not be limited in any way by any limitation on the amount or type of damages, compensation, or benefits payable by or for the Trade Contractor under workmen's compensation act, disability benefits act or other employee benefits act. This obligation for Indemnification shall survive the completion of the Contract.

3.2. In the event an Indemnified Party (Clark Construction Company, the Owner, the Architect) takes any action, to enforce the Indemnification Provision (3.1) the Indemnified Party shall be entitled to payment of actual attorney fees and costs incurred including all actual attorney fees and costs incurred by the Indemnified Party to enforce the Indemnification Provision from the Trade Contractor.

## 4. WAIVER OF SUBROGATION

4.1. A waiver of subrogation shall be provided by the Trade Contractor's insurance carrier for general liability and workers compensation related claims.

END OF SECTION

## SECTION 012100 ALLOWANCES

## 1. <u>GENERAL</u>

1.1. Drawings and General Provisions of the Contract, including General and Supplementary Conditions, Division 00 - Bidding and Contracting Requirements and Division 01 - General Requirements Specification Sections, apply to the Work of this Section.

## 2. <u>LUMP SUM ALLOWANCES</u>

- 2.1. Bidder shall include, in its base bid amount, allowances itemized in the Bid Proposal Form for its Bid Category.
- 2.2. Work shall not be performed and funded from the allowance without written direction from the Construction Manager.
- 2.3. Work performed which is to be funded from the allowance shall be monitored and signed for on a daily basis by the Construction Manager. The Trade Contractor shall be solely responsible for pursuing daily signed verification.
- 2.4. Claims for Work performed that are not accompanied by a signed ticket in accordance with the Contract shall be rejected. Payment will not be approved.
- 2.5. A line item shall be included in the Payment Schedule of Values for each allowance.
- 2.6. Unused portions or entire amounts of funds under allowance shall be credited by deductive change order when directed by the Construction Manager.
- 2.7. Lump sum allowance shall include all costs of labor and material, equipment, incidentals, supervision, overhead and profit.

## 3. <u>MATERIAL ALLOWANCES</u>

3.1. Allowances for a specific material shall cover all material related cost including taxes, freight and delivery cost to the Project site. All other costs including labor, overhead and profit and incidentals shall be included in the base bid.

### END OF SECTION

## SECTION 012300 ALTERNATES

## 1. <u>GENERAL</u>

- 1.1. Drawings and General Provisions of the Contract, including General and Supplementary Conditions, Division 00 Procurement and Contracting Requirements and Division 01-General Requirements Specifications and Addenda, apply to the Work of this Section.
- 1.2. Alternate Work shall include miscellaneous devices, accessories and other items incidental to or required for a complete installation whether or not indicated as part of the alternate.
- 1.3. Base bid amounts shall be in accordance with the Bid Documents and shall not include additional or deducted costs for alternates.
- 1.4. Cost for Labor and Material Performance and Payment Bonds, if such bonds are required by the Contract Documents, shall be included in the cost of the alternate.

### 2. VOLUNTARY ALTERNATES

- 2.1. Bidders may propose alternative materials to those specified in the form of a Voluntary Alternate.
- 2.2. Proposals for voluntary alternates shall accompany the base bid proposal with additional cost to or cost to be deducted from the base bid amount. Voluntary Alternate costs shall not be included in the base bid amount.

#### 3. <u>SPECIFIED ALTERNATES</u>

- 3.1. Alternate material or Work to the base bid is itemized and described in this section and in the Contract Documents.
- 3.2. Bidders shall carefully review the proposed alternate to be fully informed as to the material, incidental material and Work to be performed. Bidder shall not later than five (5) days prior to the date for receipt of bids, make written request for interpretation or correction of any ambiguity, inconsistency or error discovered.
- 3.3. Each bidder shall submit with his proposal in the space provided on the Bid Proposal Form alternate proposals stating the additions to or deductions from the base bid lump sum amount for substituting, omitting, adding, changing, or altering materials, equipment or construction from that indicated on the Drawings and/or Specifications.
- 3.4. The Owner and Construction Manager reserve the right to waive any irregularities, reject, any or all alternates, which in its opinion serve the Owner's best interest.

## SECTION 012300 ALTERNATES

## 4. <u>SUMMARY OF REQUESTED ALTERNATES</u>

1.	4.1	Alternate #1 –	
	a.	Add/Deduct (Circle one)	\$
2.	Alternate #2 –		
	a.	Add/Deduct (Circle one)	\$
3.	Alternate #3 –		
	a.	Add/Deduct (Circle one)	\$

## END OF SECTION

## SECTION 012600 CONTRACT MODIFICATION PROCEDURES

## 1. <u>GENERAL</u>

- 1.1. Trade Contractors shall not proceed with any Change in the Work without written direction to proceed by the Construction Manager. Increases to the Contract will not be allowed for any extra Work performed by the Trade Contractor without this written authorization to perform the Work.
- 1.2. The Architect may issue documentation for Changes in the Scope of Work in the form of Bulletins or other form as provided in the Contract Documents. The Construction Manager may issue documentation for minor Changes in the Work.

## 2. <u>PROCESS</u>

- 2.1. Change documentation issued by the Architect to the Construction Manager for review will in turn be transmitted to the affected Trade Contractors for quotation of cost.
- 2.2. The Trade Contractor shall provide written quotations for Changes in the Work within seven (7) days unless otherwise noted in the request for quotation.
- 2.3. Written quotations shall be broken down in detail as required by Clark Construction Company. Quotation breakdown for Changes in the Work as a minimum shall include actual labor cost (hours x Clark Construction Company approved direct labor cost), itemized material cost, itemized equipment cost, overhead and profit.
- 2.4. The Trade Contractor shall submit a response to the Construction Manager regarding the proposed Change within the specified time period regardless of whether the Change involves a Change in the Contract price or not.
- 2.5. Should the Trade Contractor fail to respond within the specified time period, and after a second request by the Construction Manager and still no response received from the Trade Contractor within an additional seven (7) days, the Construction Manager will assume that the Trade Contractor has reviewed the proposed Change document and that no additional costs are associated with the Work. The Trade Contractor shall be held to complete any Work associated with the Change at no increase to the Contract.
- 2.6. In the event of credits to the Contract for deleted Work which is not quoted by the Trade Contractor as noted above, the Construction Manager shall calculate a credit cost based on local industry standards for the Work and submit a quotation on behalf of the Trade Contractor and subsequently issue a deduct Change Order.
- 2.7. Upon acceptance of the proposed cost quotations by the Construction Manager, the Construction Manager will prepare and issue a Change Order for approval by the Architect and Owner.
- 2.8. Following approval of the Trade Contractor quotation by the Architect and Owner, the Construction Manager will issue a Change Order to the Trade Contractor for the Change in Work. Only after receipt of the written Change Order from the Construction Manager and execution by the Trade Contractor may the Trade Contractor include related Work on its Application for Payment.

## SECTION 012600 CONTRACT MODIFICATION PROCEDURES

- 2.9. The Construction Manager may direct the Trade Contractor in writing to proceed with the Work at any point in time prior to formal approval of the Change Order. The Trade Contractor shall promptly proceed with the Work.
- 2.10. The Construction Manager reserves the option to instruct a Trade Contractor to proceed with Work on a time and material cost basis.

#### 3. <u>COMPENSATION FOR CHANGES IN THE WORK</u>

- 3.1. The price to be paid for Change in the Work shall be the actual and reasonable cost of:
  - A. Necessary materials (including transportation to the site); plus
  - B. Necessary direct labor cost (direct labor shall be limited to amounts paid to employees for hourly wages, fringe benefits, and payroll taxes). (Refer to "Labor Rate Calculation" form included in Certificates and Forms section); plus
  - C. All insurance required by reason of the performance of the extra Work; plus
  - D. Payments required to be made to labor organizations under existing labor agreements, plus;
  - E. Sales and personal property taxes, if any, required to be paid on materials incorporated in such extra Work, plus;
  - F. Maintenance, operation, and rental of, or reasonable rental value of, contractorowned, necessary plant and equipment other than small tools (including gas, oil, electric current, etc.); plus
  - G. Necessary installation and dismantling of such plant and equipment (including transportation to and from the site), if any items; plus
  - H. For Work performed by Trade Contractor's own forces fifteen percent (15%) of the total of Items A through G as compensation for all other items of cost or expense including administration, overhead, general superintendence, small tools and profit; plus
  - I. For Work performed by Trade Contractor's subcontractors Trade Contractor shall add five percent (5%) of the total subcontractor cost (unless otherwise required by Clark Construction Company's Contract with the Owner) as compensation for all other items of cost or expense including administration, overhead, general superintendence, small tools and profit for Trade Contractors.
  - J. For Work performed by Trade Contractor's subcontractors subcontractors for their work shall add fifteen percent (15%) (unless otherwise required by Clark Construction Company's Contract with the Owner) of the total of Items A through G as compensation for all other items of cost or expense including administration, overhead, general superintendence, small tools and profit for Trade Contractors.
- 3.2. Reference Certificates and Other forms 006200 for Sample Trade Contractor Change Order.
# SECTION 012600 CONTRACT MODIFICATION PROCEDURES

END OF SECTION

# SECTION 012900 PAYMENT PROCEDURES

# 1. <u>GENERAL</u>

1.1. Payment procedures shall be strictly adhered to as a condition of payment.

### 2. <u>PROJECT START-UP DOCUMENTATION</u>

- 2.1. Prior to beginning Work at the project site, the following documentation shall be submitted to and approved by Clark Construction Company:
  - A. Current Certificate of Insurance with appropriate limits and Clark Construction Company named as additional insured. Contract Number and project description shall be noted on the Certificate of Insurance. (Refer to "Insurance Requirements" section).
  - B. Contract must be returned executed and acceptable to Clark Construction Company.
  - C. Clark Construction Company "Safety Requirements" letter must be completed and fully executed.

### 3. <u>APPLICATION FOR PAYMENT</u>

- 3.1. All Applications for Payment shall be submitted through Construction Payment Management noted in article 4.
- 3.2. A Sworn Statement and Waivers of Lien as applicable shall be submitted with Applications for Payment through Construction Payment Management noted in article 4. Applications which are not accompanied by an appropriate Sworn Statement will be rejected.
- 3.3. Payment applications which include application for payment for stored materials shall be accompanied by a Materials Stored Payment Form (Refer to "Certificates and Other Forms" section).
- 3.4. Applications are on or before the 25<sup>th</sup> of each month. Late billings will not be processed for payment. Late billings shall be resubmitted for payment the following month by the Trade Contractor.
- 3.5. The following shall be clearly stated on all Applications for Payment:
  - A. Trade Contractor name
  - B. Address
  - C. Telephone number
  - D. Contract Number

#### 4. <u>CONSTRUCTION PAYMENT MANAGEMENT (CPM)</u>

4.1. Trade Contractor applications for payment shall be submitted electronically through the Textura Construction Payment Management program (CPM). The CPM will expedite the payment application process in addition to the following benefits.

# SECTION 012900 PAYMENT PROCEDURES

- A. Automatic generation of AIA G702/703 and lien release documents
- B. Email notification as the payment process progresses
- C. Timely response to payment application questions
- D. Real-time visibility into compliance status
- E. Expedited direct electronic payments
- F. Email notification of electronic direct deposit payments made
- 4.2. A nominal fee will be charged to each Contractor for use of the Textura CPM service payable to Textura as follows.

Contract Amount	Contractor Cost
\$0-27,777	\$50.00
\$27,778 - \$1,388,888	.18% of contract amount
\$1,388,889+	\$2,500

- 4.3. Additional information can be found on the Textura website at <u>www.texturacorp.com</u>.
- 4.4. Trade Contractors shall take part in a Textura\_training webinar. Textura Consultants facilitate webinars three times per week, Monday / Wednesday / Friday. Register for this training webinar at: <u>http://www.texturacorp.com/training/webinarschedule.htm</u>
- 5. After taking part in the training session Contractors may call Textura Customer Support team at 866-TEXTURA (866-839-8872) for additional assistance in using the Textura service. The call center is staffed from 8am– 9pm ET.

END OF SECTION

# SECTION 012973 SCHEDULE OF VALUES

# 1. <u>GENERAL</u>

- 1.1. Trade Contractor shall submit to the Construction Manager for approval, a Schedule of Values prior to award of Contract.
- 1.2. Upon request by the Construction Manager, the Trade Contractor shall submit supporting data to substantiate the accuracy of the values provided.
- 1.3. The Schedule of Values as approved by the Construction Manager shall be used as the basis of all applications for payment.

## 2. FORM OF SUBMITTAL

- 2.1. The Schedule of Values shall be broken down sufficiently and as approved by the Construction Manager for ease of assessment of work completed throughout performance of the Work.
- 2.2. Each line item value shall be broken down into labor and material components including overhead and profit. The aggregate value for all items shall be equal to the Contract sum.
- 2.3. Trade Contractor shall provide further breakdown as Work progresses if required\_by the Construction Manager.
- 2.4. The following items shall be included in the Schedule of Values as a minimum:
  - A. Performance and Labor and Material Payment Bonds (if applicable)
  - B. Mobilization
  - C. Field Supervision, Layout, Barricades and Safety Precautions, Temporary Office and Equipment
  - D. Shop Drawings, Submittals and Samples
  - E. Project Clean-up
  - F. Closeout (The following items shall be included in addition to the contract required retention amount)
    - 1. As-Built Documents 1.0% of Contract Amount
    - 2. Training and O&M Manuals 1.0% of Contract Amount
    - 3. Attic Stock Material 0.5% of Contract Amount
    - 4. Punch list Completion 2.0% of Contract Amount
    - 5. Warranty 0.25% of Contract Amount
    - 6. Consent of Surety (if applicable) 0.25% of Contract Amount

#### END OF SECTION

## 1. **GENERAL**

- 1.1. Trade Contractor, Project Manager and Superintendent and/or Foreman shall attend and participate in meetings indicated in this section and other meetings as required by the Construction Manager.
- 1.2. Trade Contractor representative attending meetings shall be fully informed and have authority to make decisions for the Trade Contractor.

### 2. <u>PRE-AWARD CONFERENCE</u>

2.1. Pre-Award Conference Meetings will be held prior to award of Contract to review conditions, requirements of the Contract and Scope of Work.

## 3. <u>PULL PLANNING MEETINGS</u>

- 3.1. Trade Contractor representatives shall attend One (1) Pull Planning Meeting (Project Master Scheduling) conducted by the Construction Manager. Trade Contractor representatives required to attend shall include at a minimum the Project Manager and the on-site Foreman. Representatives attending the meeting shall have detailed knowledge of the scope of work as well as trade hour requirements to accomplish the work.
- 3.2. Project Master Scheduling Meeting durations are typically 1 full day (8 hours).
  - A. Note:
    - 1. On more complex or phased projects more than 1 Pull Planning Meeting may be required. In that event revise the above accordingly.
    - In the event specific Trade Contractors will be required to attend more than one (1) Pull Planning Meeting, include a note as follows in the Scope of Work

       Bid Category Specific Notes for each bid category responsible to attend more than one (1) session.
      - a. Trade Contractor representatives shall attend \_\_\_\_\_ (#) Pull Planning Meetings conducted by the Construction Manager in accordance with Project Meetings specification section Pull Planning Meetings. This note supersedes the number of Pull Planning Meetings stated in the Project Meetings specification section.

## 4. **PRE-INSTALLATION MEETINGS**

- 4.1. Trade Contractor representative attending meetings shall be fully informed and have authority to make decisions for the Trade Contractor. In attendance should be the trade contractors' project manager and superintendent.
- 4.2. Pre-Installation Meeting Agenda
  - A. Drawing & Specification Review
  - B. Installation

- C. Construction Schedule Three Week Look Ahead Schedule
- D. Testing & Inspections
- E. Safety / Clean Up
- F. Environmental Impacts
- G. Warranty Requirements
- H. Misc. / Other Items

### 5. **PROGRESS MEETINGS**

- 5.1. Attend Project Progress and Coordination Meetings held weekly by the Construction Manager and attended by others.
- 5.2. Progress Meeting Agenda
  - A. <u>SAFETY</u>
    - 1. Clark Construction Company's Safety Policy
    - 2. Tool Box Talks
    - 3. Observed Violations/Safety Inspections
    - 4. Housekeeping and clean-up
    - 5. Temporary barricades, protection and lighting
    - 6. Accident Reports
    - 7. Material Safety Data Sheets (MSDS)
    - 8. Emergency contacts

#### B. <u>ADMINISTRATION</u>

- 1. Review current Document List
- 2. Submittal status
- 3. Applications for Payment
- 4. Insurance/Bonds
- 5. Manpower reporting
- 6. Permits and approvals
- 7. Governmental issues

8. Eventual closeout requirement

## C. <u>CHANGES IN WORK</u>

- 1. Potential Change Order Log/Bulletins/Field Orders
- 2. RFI Status Report
- 3. Change Orders
- 4. Cost related issues

### D. <u>FIELD OBSERVATIONS</u>

- 1. Problem resolutions
- 2. Conflict resolutions
- 3. Quality Control/workmanship Notice to Comply Log
- 4. Material storage and site staging issues
- 5. Installation procedures
- 6. Field condition issues
- 7. Design issues

#### E. <u>SCHEDULE</u>

- 1. Milestone dates
- 2. Short term schedule/three week look ahead schedule
- 3. Material/equipment deliveries
- 4. Long lead items
- 5. Manpower
- 6. Submittals
- 7. Sequencing/coordination issues
- 8. Occupancy issues work completion and punchlist
- 9. Pre Task Plan

#### F. <u>ENVIRONMENTAL ISSUES</u>

1. Review operational controls for significant aspects

2. Complaints/corrective actions required

END OF SECTION

## SECTION 013323 SHOP DRAWINGS, PROJECT DATA AND SAMPLES

## 1. <u>DEFINITIONS</u>

- 1.1. Shop drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Trade Contractor or any Subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.
- 1.2. Product data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Trade Contractor to illustrate a material, product or system for some portion of the Work.
- 1.3. Samples are physical examples which illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.

### 2. <u>SUBMISSION REQUIREMENTS</u>

- 2.1. Shop Drawings
  - A. Trade Contractor shall furnish all shop drawings through Connect in the Trade Contractor Transmittal Module.
  - B. Shop drawings shall follow or be keyed to the nomenclature, terminology and symbolism of the Contract Documents and shall be identified by the Bid Category designation.
  - C. A copy of the shop drawings will be returned to the Trade Contractor through Connect with review comments.
- 2.2. Product data
  - A. Submit manufacturer's product data printed literature, catalogs, pre-printed drawings and brochures for the Construction Manager's and Architect's review through Connect in the Trade Contractor Transmittal Module.
  - B. Submittals shall be identified by the CSI Specification section, location within the building and a descriptive explanation as to the materials intended use.
  - C. A copy of submitted product data will be returned to the Trade Contractor through Connect with review comments.
- 2.3. Samples
  - A. Submit four (4) samples to illustrate functional and aesthetic characteristics of the product, with integral parts and attachment devices. Coordinate sample submittals for interfacing Work.
  - B. Submit samples of finishes from the full range of manufacturer's standard colors, textures and patterns for Architect's selection.
  - C. Include identification on each sample, with full project information.
  - D. Submit four (4) samples unless otherwise specified in individual Specification sections, two (2) of which will be retained by the Architect.

## SECTION 013323 SHOP DRAWINGS, PROJECT DATA AND SAMPLES

E. Reviewed samples which may be used in the Work are indicated in individual Specification sections.

### 3. <u>MANUFACTURER'S INSTALLATION INSTRUCTIONS</u>

- 3.1. When specified in individual Specification sections, submit manufacturer's printed instructions for delivery, storage, assembly, installation, start-up, adjusting and finishing, in quantities specified for product data.
- 3.2. Indicate special procedures, perimeter conditions requiring special attention and special environmental criteria required for application or installation.

### 4. <u>MANUFACTURER'S CERTIFICATES</u>

- 4.1. When specified in individual Specification sections, submit manufacturer's certificate for review, in quantities specified for product data.
- 4.2. Indicate material or product conforms to or exceeds specified requirements. Submit supporting reference data, affidavits and certifications as appropriate.
- 4.3. Certificates may be recent or previous test results on material or product, but must be acceptable to the Architect.

### 5. TRADE CONTRACTOR RESPONSIBILITIES

- 5.1. The Trade Contractor shall review, check, approve and submit, with reasonable promptness and in such sequence as to cause no delay in the Work or in the Work of the Owner or any separate Trade Contractor, all shop drawings, product data and samples required by the Contract Documents.
- 5.2. Shop drawings shall be thoroughly reviewed and approved by the Trade Contractor before submitting to the Construction Manager.
- 5.3. The Construction Manager will not review any shop drawings which are not reviewed and approved by the Trade Contractor prior to submittal.
- 5.4. By approving and submitting shop drawings, product data and samples the Trade Contractor represents that he has determined and verified all materials, field measurements and field construction criteria related thereto, or will do so, and that he has checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.
- 5.5. Trade Contractors shall not be relieved of responsibility for any deviation from the requirements of the Contract Documents by the Construction Manager's or Architect/Engineer's review of shop drawings, product data or samples unless the Trade Contractor has specifically informed the Construction Manager in writing of such deviation at the time of submission and the Architect/Engineer has given written approval to the specific deviation.
- 5.6. The Trade Contractor shall not be relieved from responsibility for error or omissions in the shop drawings, product data or samples by the Construction Manager's review thereof.

## SECTION 013323 SHOP DRAWINGS, PROJECT DATA AND SAMPLES

- 5.7. The Trade Contractor shall direct specific attention, in writing or on resubmitted shop drawings, product data or samples, to revisions other than those requested by the Construction Manager on previous submittals.
- 5.8. No portion of the Work requiring submission of a shop drawing, product data or sample shall be commenced until the submittal has been approved by the Architect/Engineer. All such portions of the Work shall be in accordance with approved submittals.
- 5.9. After final review by the Architect/Engineer, the Trade Contractor shall furnish prints from the transparencies of all approved shop drawings, as required by construction operations in the field, to the Construction Manager for distribution to the Trade Contractor affected.
- 5.10. The requirements that corrections be made on shop drawings, or delay in resubmittal of shop drawings, shall not be justification for delay of the progress of the Work.
- 5.11. Extreme care is to be used in distribution of all drawings, shop drawings, equipment cuts and operating instructions to protect the Owner. These items shall be kept in strict confidence and all copies shall be accounted for.

# 6. ARCHITECT'S ACTION

- 6.1. Except for submittals for the record and similar purposes, where action and return on submittals is required or requested, the Architect will review each submittal, mark with appropriate "action" and return as soon as possible. Where the submittal must be held for coordination, the Construction Manager will so advise the Trade Contractor.
- 6.2. The architect will stamp each submittal to be returned with a uniform, self-explanatory action stamp, appropriately marked.

END OF SECTION

# 1. <u>GENERAL</u>

- 1.1. Clark Construction Company holds health and safety to be its highest priority at all times, under every circumstance.
  - A. Health and safety shall govern every course and method of action.
  - B. Potential risk to health and safety shall be evaluated prior to commencement of all work activities. Work practices shall be employed which create conditions free of potential injury.
  - C. An environment of concern for the health and safety of each person shall be promoted on the project site.
- 1.2. Trade Contractor shall be responsible to take all necessary precautions for the safety of employees, others working on the project site and the general public.
- 1.3. Trade Contractor shall comply with all applicable federal, state and local health, safety and environmental laws, codes and requirements.
- 1.4. Trade Contractor shall assure that each of its employees, and employees of its subcontractors and material suppliers on the project site are familiar with and abiding by all aspects of the Trade Contractor Health and Safety Plan.
- 1.5. Provisions contained herein shall not relieve the Trade Contractor of its obligations and liabilities under federal, state and local laws including all additions and revisions thereto. Nor shall any provision herein transfer obligations and/or liabilities of the Trade Contractor to Clark Construction Company.

## 2. TRADE CONTRACTOR RESPONSIBILITIES

- 2.1. Trade Contractors and subcontractors and material suppliers to the Trade Contractor at any tier shall comply with all health, safety and environmental requirements. Clark Construction Company shall have authority over such subcontractors and material suppliers at any tier to same extent it has over the Trade Contractor.
- 2.2. Trade Contractor Health and Safety Program and Hazard Communication Program Documentation:
  - A. Trade Contractor shall have in place and implement a comprehensive written Health and Safety and Hazard Communication Program. Trade Contractor shall ensure that all of its subcontractors and material suppliers at any tier adhere to the Health and Safety Program, Hazard Communication Program and all health, safety and environmental requirements of the Contract.
  - B. Trade Contractor, its subcontractors and material suppliers at any tier shall provide its current electronic Health and Safety Program and Hazard Communication Program to Clark Construction Company for review and acceptance as a condition of award of a Contract. Clark Construction Company will receive and keep such program on file to completion of the project.

- C. Trade Contractor shall ensure that subcontractors and material suppliers to the Trade Contractor at any tier have in place and implement a comprehensive Health and Safety and Hazard Communication Program.
- D. Trade Contractor and its subcontractors and material suppliers at any tier shall have on site and available at all times for its employees and Clark Construction Company, a current copy of its Health and Safety and Hazard Communication Program.
- E. Health and Safety and Hazard Communication Programs shall be supplemented by the Trade Contractor and any of its subcontractors and material suppliers at any tier with a project specific Health and Safety and Hazard Communication Plan ensuring compliance to all requirements of the Contract Documents and project conditions as required.

### 2.3. **Designated Safety Representative**

- A. Trade Contractor shall designate a safety representative from the site staff in addition to any craft safety representative when less than 40 employees of the Trade Contractor and/or subcontractor employees are working on the project site. Anytime 40 or more employees of the Trade Contractor and/or subcontractor employees are working on the project site the Trade Contractor shall have a full time safety representative on site responsible 100% of his/her time to health and safety.
- B. Trade Contractor shall provide Clark Construction Company with a contact telephone number of a representative of the Trade Contractor that can take action and may be contacted 24 hours per day, seven days per week in the event of an emergency.
- C. Safety Representative shall:
  - 1. have adequate training and experience as required by applicable regulations and be knowledgeable in health and safety aspects of the Work to be performed;
  - 2. have authority to make decisions and take action relative to health and safety on behalf of the Trade Contractor, its subcontractors and material suppliers;
  - 3. participate in Clark Construction Company Preconstruction Health and Safety Meetings, other project health and safety meetings and hazard identification walk-through meetings as required by Clark Construction Company; and
  - 4. assure immediate correction of any health and safety and environmental issues by the Trade Contractor at the Trade Contractor's expense and report corrective action taken to Clark Construction Company in writing.

#### 2.4. Regulations/Compliance

A. Trade Contractor shall comply with the current Federal Occupational Safety and Health Act (OSHA), current State Health and Safety requirements, other applicable

federal, state and local requirements, health and safety requirements of the (sub) Contract and environmental regulations.

B. Cost of correction of health and safety infractions shall be paid by the Trade Contractor.

### 2.5. **Disciplinary Action/Termination**

- A. In response to Trade Contractor's failure to comply with health and safety and environmental requirements or otherwise performing in an un-safe manner Clark Construction Company shall have authority to:
  - 1. order immediate Work stoppage and corrective action to be taken by the Trade Contractor;
  - 2. withhold any payments due to the Trade Contractor; and
  - 3. take action to correct any issue which is not corrected by the Trade Contractor in an acceptable time period.
- B. Trade Contractor shall be responsible to ensure compliance by all employees, subcontractors and material suppliers' employees at any tier. Trade Contractor supervision unable or unwilling to secure safe performance by these employees shall be deemed unqualified and the Trade Contractor shall, upon direction of the Construction Manager, replace the unqualified person with a qualified person.
- C. Trade Contractor, subcontractor and material supplier employees as a condition of working on the project site shall comply with all health, safety and environmental regulations and requirements.
- D. In response to Trade Contractor, subcontractor and material supplier employee(s) failure to comply with health, safety and environmental requirements or otherwise performing in an unsafe manner, Clark Construction Company shall have authority to take action up to and including barring an individual from the project site.
  - 1. Disciplinary actions may be implemented against employee(s) of the Trade Contractor, subcontractor and material supplier by Clark Construction in response to violations of health and safety and environmental requirements as follows:
    - a. first notice: verbal and/or written notification to the Trade Contractor;
    - b. second notice: written notification to the Trade Contractor;
    - c. third notice: termination of the employee from the project;
    - d. Clark Construction Company may at its discretion immediately bar the employee(s) in violation from the project site without a first or second notice if in the opinion of Clark Construction Company the violation warrants such action.

### 2.6. Health and Safety Implementation Planning

- A. Prior to the Trade Contractor beginning work on the project site the Trade Contractor shall develop and submit a detailed Project Specific Safety Plan as required by Clark Construction Company.
- B. The Project Specific Safety Plan shall be revised as project conditions and activities change.

#### 2.7. Employee Training

A. Trade Contractor shall provide health and safety training to each of its employees as required by governing health and safety authorities and shall provide training as required to perform specific duties in a safe manner.

### 2.8. Daily Health, Safety and Environmental Job Safety Analysis (JSA)

- A. Prior to commencement of work Trade Contractors will submit to Clark Construction the project specific JSA plans. JSA's will be reviewed by Clark personnel and accepted or sent back to the contractor for re-work. JSA's should identify all hazards and mediation of such hazards for each work related tasks.
- B. On a daily basis <u>prior to performing the Work</u> the Trade Contractor supervisor or foreman shall submit a Job Safety Analysis (409FR041), for work to be performed, to Clark Construction Company's Superintendent. If there is extra work involved not previously noted on the JSA, then the supervisors will fill out the back portion of the Sign-In Sheet (409FR045) to identify new job steps and job hazards.
- C. JSA's shall identify sequence of Work, required personal protective equipment, potential hazards, other persons, property, potentially affected environment, and action to be taken to eliminate risks.
- D. JSA's shall be reviewed daily with all personnel involved in the execution of the Work and other personnel on site that may be affected or have need to be made aware.
- <u>E.</u> Trade Contractors will be required to perform a daily stretching program prior to each shift with Clark field representatives and other Trade Contractors. <u>Locations of stretch and flex are as follows:</u>
  - 1. High School/ Harvest at Highschool
  - 2. Woodland Meadows/Heritage at Woodland Meadows
  - 3. Pleasant Ridge On Site
  - 4. Liberty On Site
  - 1.5. Middle School On Site

#### 2.9. Meeting Attendance

Saline Area Schoosl 2015 Bond Program – 2017 Projects Phase 1 Bid Release No.4 Section 013529 Trade Contractor Health and Safety Plan Version 8 - pending

- A. All Trade Contractor employees shall be required to attend;
  - 1. A project orientation session the first day the employee is on the project site prior to start of any Work activity. Orientation sessions will be approximately one hour long;
  - 2. Weekly "Tool Box Talk" meetings shall be conducted by the Trade Contractor with topics of discussion relevant to the Work to be performed and project site conditions. Attendees shall sign the weekly "Tool Box" meeting notes. "Tool Box" meeting attendance sheets shall be submitted to Construction Manager;
  - 3. Other health, safety and environmental meetings as may be required by Clark Construction Company.

## 2.10. Safety Guide Program

- A. Each tradesperson, prior to beginning work on the project site, will be provided with and shall read thoroughly and understand Clark Construction Company's Safety Guide information (Refer to CCC Form )
- B. Tradespersons after fully understanding the information shall sign and return the Safety Guide to Clark Construction Company.
- C. All tradespersons working on the project site shall have Clark Construction Company's safety sticker on their hard hat signifying participation in the Safety Guide Program.

## 2.11. BRAVO Program (Being Resistant Against Violations Openly)

- A. Trade Contractors shall actively participate in the BRAVO Program.
- B. The purpose of the BRAVO Program is to engage all persons on the project site in maintaining safety of self and others as the highest priority and to supportively encourage safe work practices by all.
- C. Trade Contractor onsite personnel are encouraged to actively identify safe work procedures and environments of other Trade Contractors and of others within their own firm.
- D. In the process of actively identifying safe work procedures and environments, each person is urged to detect any unsafe work procedures and environments.
  - 1. Trade Contractor personnel are encouraged to take action to prevent injury to others in the event an unsafe situation is observed. Action may include:
    - a. pointing out the potential threat to the individual(s) involved and suggesting the work be stopped and the situation corrected. This is approach may be appropriate under various circumstances; it should always be employed in the event of imminent danger;

- b. promptly report the situation to the person's supervisor who in turn shall promptly report to a designated Clark Construction Company representative; or
- c. promptly report the situation to a designated Clark Construction Company representative.
- E. Trade Contractors shall utilize Clark Construction Company BRAVO Observation cards to document both safe and if any, unsafe work procedures and environments.
  - 1. Each Trade Contractor shall submit a minimum of 2 completed BRAVO Observation cards per week to a designated Clark Construction Company representative.
  - 2. BRAVO Observation cards are a 2 sided card. One side for safe observations and the other for any unsafe observations.
  - 3. The intent of the information provided on a BRAVO Observation card is not to point fingers, rather to keep everyone mindful of safety being the project priority. As such BRAVO cards do not request the name of the firm for which the observation is being reported.
- F. Clark Construction Company will employ an incentive program in appreciation of the onsite personnel "looking out" for one another by Being Resistant Against Violations (BRAVO).

#### 2.12. First Aid/CPR Training

- A. Each Trade Contractor shall have a minimum of one First Aid/CPR trained person on the project site whenever work is being performed by its employees.
- B. Each Trade Contractor shall have a first aid supply kit on site available to its employees.

## 2.13. Communication/Foreign Languages

- A. In the event that one or more Trade Contractor employees working on site do not speak the English language fluently, the Trade Contractor shall assure that a minimum of one of its employees on site is fluent in the English language as well as the language of non-English speaking employee(s).
- B. The employee fluent in both languages shall be on site at all times when a non-English speaking employee is on site and shall have authority to make decisions and take action on behalf of the Trade Contractor.

### 2.14. Housekeeping

- A. Trade Contractors shall maintain the project free of debris.
- B. Thoroughly clean up on a daily basis and more frequently as required.

- C. Debris shall be transported to dumpsters on a daily basis.
- D. Material and equipment in storage and in use shall be located out of means of ingress, egress; stairways, walkways, etc.
- E. Location of stored material and equipment shall be coordinated with Clark Construction Company and other trades.

# 3. <u>HEALTH AND SAFETY ORIENTATION</u>

### 3.1. General

- A. Trade Contractor shall notify Clark Construction Company Superintendent of any employee arriving on the project site for the first time. Trade Contractor shall schedule a project-specific orientation sessions to be conducted prior to the employee performing work on-site. The orientation session will be conducted by a Clark Construction Company representative.
- B. Trade Contractor shall provide project-specific orientation training with each of its employees specific to the work to be performed to assure a safe\_project environment for all.
- C. Trade Contractor shall assure that employees visiting the site for meetings, project review, or other "non-trade Work" related activities and walking into or through construction areas, shall attend an abbreviated Health and Safety Orientation Session.

#### 3.2. **Orientation Sessions**

- A. The following topics may be reviewed at the Health and Safety Orientation Session.
  - 1. Trade Worker Orientation Sessions
    - a. Clark Construction Company site safety representative contact information
    - b. Emergency contact information
    - c. First Aid supply location
    - d. Personal protective equipment (PPE) and clothing requirements
    - e. Health and Safety General Rules
    - f. Clark Construction Company's Safety Guide Program (to be signed by each Trade Contractor employee).
    - g. Fire Protection and Prevention Plan Hazard Communication Program and SDS file location
    - h. Accident and incident reporting

- i. Environmental Program
- j. Cell Phone and Electronic Device Policy
- 2. Non-Trade Worker Orientation Sessions
  - a. Clark Construction Company's site safety representative\_contact information
  - b. Emergency contact information
  - c. First Aid supply location
  - d. Personal protective equipment (PPE) and clothing requirements
  - e. Cell Phone and Electronic Device Policy

### 4. <u>EMERGENCY RESPONSE PLAN</u>

#### 4.1. Medical Services

- A. Prior to commencement of work Trade Contractor shall;
  - 1. make provisions for prompt medical attention in the event of a serious injury;
  - 2. ensure that adequate first aid supplies are easily accessible when needed;
  - 3. provide proper equipment for prompt transportation of the injured person to proper medical care or system for contacting necessary ambulance service;
- B. In the event of any incident or accident the following procedure shall ensue.
  - 1. Attend to the injured party and/or stabilize the area as may be required.
  - 2. Contact 911 as required.
  - 3. Notify Clark Construction Company's project superintendent when first aid is being administered.
  - 4. Assist in ascertaining and reporting events surrounding the incident or Accident.
  - 5. Trade Contractor or injured person shall;
    - a. Complete an Incident Observation Report (409FR012) and provide a copy to Clark Construction Company site safety representative.
    - b. Provide status reports to Clark Construction Company's site safety representative as conditions require.

c. Participate in accident/incident investigation meetings for the purpose of establishing the "Root Cause" of the accident or incident.

#### 4.2. Emergency Evacuation Plan

- A. Have a sounding device and a sign to identify it as an evacuation alarm instructing all personnel to sound three short blasts in the event of emergency (a car horn or air horn can be used).
- B. Pre-determine a meeting place to account for all personnel.
- C. Each Trade Contractor's Designated Safety Representative shall account for all personnel to arrive at the pre-determined meeting place in the event of an emergency.
- D. All persons working on-site shall be familiar with emergency egress routes from their position on the project at all times.

#### 4.3. Blood–Borne Pathogens

A. In accordance with OSHA, Trade Contractors shall assure that each employee exposed or potentially exposed to blood and other infectious materials be advised of the potential Blood-borne pathogen hazards and how to guard against those hazard.

### 5. <u>CRISIS COMMUNICATION PLAN</u>

- 5.1. Clark Construction Company will designate a spokesperson to make all public comments during a crisis situation.
- 5.2. Trade Contractor personnel are encouraged to refrain from comments to the media. Trade Contractor employees are encouraged, questioned by the media, to state <u>"The incident is being investigated and Clark Construction Company's spokesperson will make a statement shortly".</u>

#### 6. PERSONAL PROTECTIVE EQUIPMENT AND CLOTHING

- 6.1. Personal protective equipment and clothing shall be worn on the project site with exception of construction office areas separated from construction work.
- 6.2. Personal protective equipment shall meet the most stringent standard established by federal, state and local authorities, requirements of the employer or Clark Construction Company.

#### 6.3. **Personal Protective Equipment**

- A. <u>Hard hats</u> shall be worn <u>at all times</u> on the project site with exception of construction office areas separated from construction work.
- B. <u>Safety glasses</u> shall be worn <u>at all times</u> on the project site with exception of construction office areas separated from construction work.

- C. Safety vest and/or shirt of high visibility fluorescent colored fabric shall be worn when ever performing project site construction activities or in the vicinity of moving site work equipment.
- D. Gloves shall be work <u>at all times</u> on the project site.

## 6.4. Clothing

- A. Shirt sleeve lengths shall be a minimum of four inches.
- B. Shirts cut off above the waist are not allowed.
- C. Pants shall extend to cover ankles.
- D. Shorts are not allowed.
- E. Work boots shall extend to protect the ankle.
- F. Clothing shall be worn in a neat fashion and be in good repair.
- G. Clothing imprinted with language or images as determined by Clark Construction Company to be offensive in any way or advertise and/or promote the use of alcohol or illegal substances shall not be worn on the project site.

## 7. <u>CELL PHONE AND ELECTRONIC USE POLICY</u>

- 7.1. Cell phone and electronic devise use
  - A. Cell phone use includes listening, talking, texting, emailing, net surfing or other cell phone device activity.
  - B. "Electronic device" includes any internet reception, video, electronic tablets, computers, Bluetooth, walkie-talkies, etc.
  - C. Cell phone and electronic device use is allowed only in hazard free areas.
  - D. Cell phone and electronic device users shall remain stationary, no walking during cell phone or electronic device use.
  - E. Persons not designated as Foreman, Superintendent, Manager, General Foreman, may use a cell phone or electronic device only during breaks in a hazard free area.
  - F. Cell phone and electronic device use is not allowed;
    - 1. While engaged in any job task which requires the use of the hands and/or other attention to work activity.
    - 2. While operating equipment, tools or motorized vehicles including cars, trucks and motorized carts, etc.

- G. Motorized vehicles must be parked in a location free of hazards and turned off while using a cell phone or electronic device.
- 7.2. Crane and other lifting device cell phone and walkie-talkie use
  - A. Cell phones and walkie-talkies shall not be used while operating a crane, hoist, powered industrial trucks or other lifting device (unless identified in a Pre-Task Plan requiring communication for ground person to Operator).
  - B. Cell phones must be turned off whenever in the control cab or at the control panel unless identified for use in a Pre-Task Plan.
  - C. No communications other than directly between the lift coordinators. Pre-Task planning shall include the communication method/plan.

## 8. <u>SUBSTANCE ABUSE POLICY</u>

- 8.1. Clark Construction Company is committed to an alcohol and drug free work environment.
  - A. Possession, distribution, or sale on the project premises, facility, or other work places of alcoholic beverages, intoxicants, drugs and related drug paraphernalia is strictly prohibited.
  - B. Person's shall not report for duty or perform work while under the influence of any drug, alcoholic beverage, or intoxicant.

## 9. <u>SAFE WORK REQUIREMENTS</u>

#### 9.1. Asbestos Abatement

- A. Trade Contractors shall assure that any employee that may be exposed to airborne asbestos fibers shall be trained in the recognition of the hazards and appropriate controls.
- B. Trade Contractors engaged in asbestos abatement shall be licenses by the Department of Labor and the State in which the Work is being performed.

## 9.2. Confined Space Entry

- A. Trade Contractor shall employ a confined space entry procedure when Contractor's employees are required to enter a confined area or space.
- B. Confined space entry procedures shall conform to OSHA and any State requirements.

#### 9.3. Excavation

A. A competent person trained in soil classification and the recognition of trenching and excavation hazards must be on the project site when excavating or trenching is to be done.

- B. Trenches and excavations 5-0 or more in depth and shall be shored or walls cut back to protect persons from cave-in. Trenches and excavations less than 5'-0 with unstable walls shall be shored or cut back to protect persons from cave-in.
- C. Trenches and excavations shall be properly barricaded to prevent persons from falling into them.

## 9.4. Fall Protection

- A. A fall protection system shall be utilized when an adjacent surface is six-feet (6'-0) or more below foot level.
  - 1. Ladders
    - a. Fall protection on ladders is not required providing that the persons body remains within the vertical side rails.
  - 2. Personal fall arrest systems
    - a. Shall be properly worn and actively used by all workers when an adjacent surface is six feet (6'-0) or more below the foot level of the person unless other adequate fall protection such as guardrails are in place.
    - b. All components (anchorage points, lanyards, clips etc.) of a fall arrest system shall be of a type recommended for the work being performed and the conditions to be encountered;
    - c. The entire system shall be as approved by the system manufacturer and all local and federal health and safety requirements.
  - 3. Wire rope guard rails;
    - a. Wire rope guard rail systems providing fall protection shall incorporate loop type connections with a minimum of two (2) wire rope clamps. Turnbuckles shall be installed at each side or at more frequent intervals to maintain required tautness of the wire rope.
    - b. Wire rope horizontal lifelines shall be designed by a registered professional engineer and installed and maintained by a competent person. Wire rope life lines shall be designed to meet, at a minimum, the requirements of OSHA.
  - 4. Flat roofs and decks
    - a. A warning barrier meeting, at a minimum, the requirements of OSHA may be used 15 feet from the fall hazard. Warning tape or other such means is not allowed, persons between the warning barrier and fall hazard shall utilize a positive means of fall protection.
  - 5. Scaffolding

Saline Area Schoosl 2015 Bond Program – 2017 Projects Phase 1 Bid Release No.4 Section 013529 Trade Contractor Health and Safety Plan Version 8 - pending

- a. Persons erecting, using and dismantling scaffolds shall be trained in the hazards and safe procedures to be followed to eliminate exposure to those hazards and shall utilize fall protection when six feet (6'-0) or more above an adjacent surface.
- b. Trade Contractor's designated Competent Person shall inspect all scaffolds prior to each work shift.
- c. All scaffolds shall bear a tag, signed and dated by the contractor's Competent Person, denoting that the scaffold has been inspected and is safe to use prior to inspected and is safe to use prior to an employee utilizing the scaffold on that work shift.

## 9.5. **Fire Protection and Prevention Plan**

- A. Purpose
  - 1. The following plan has been developed in accordance with MIOSHA Fire Protection and Prevention Standard Part 18.
    - a. This plan exists to:
      - i. Prevent inception of fire
      - ii. Prevent loss of life and personal injury
      - iii. Protect property from damage
      - iv. Provide uninterrupted operations
- B. Trade Contractor shall be responsible for fire prevention associated with its work and operational areas 24 hours per day 7 days per week for the duration of the contract.
- C. Major workplace fire hazard control procedures to them
  - 1. Prior to the start of the construction and as construction progresses, Trade Contractors shall identify hazards and make on-site personnel aware.
  - 2. Maintenance of equipment and systems to prevent or control fires
    - a. The Clark Construction Company Project Superintendent will inspect fire extinguishers at least monthly.
- D. Employee training
  - 1. Trade Contractors shall be responsible to train all of its personnel on-site on the general principles of fire extinguishers use and the hazards involved with the various stages of firefighting.

- 2. All personnel on site will be responsible to review the Fire Protection and Prevention Plan and be familiar with its contents.
- E. Hot work procedures
  - 1. Hot work procedures shall be followed when the possibility of a source of ignition exists. This includes open flames, spark-producing tools, hot surfaces (above 450 degrees F), and static discharging items.
  - 2. The following minimum hot work procedures shall be followed when directed by Clark Construction Company. In the event that the Trade Contractor or project specific requirements are more stringent the most stringent shall be employed after review with Clark Construction Company.
    - a. Ensure that a Hot Work Permit form is completed. A Hot Work Permit is good for one working shift only.
    - b. Hot work permit shall be posted at the jobsite with a copy in the Clark Construction Company Superintendent field office.
    - c. Hot work is not permitted where a flammable liquid is present in an open tank.
    - d. Obtain any other appropriate permits.
    - e. At the site of the work the following minimum action is required:
      - i. Clean up of combustibles such as paper, oil and grease.
      - ii. Cover immovable combustibles such as wooden steps with flame retardant tarpaulins.
      - iii. Place fire-extinguishing equipment at stand-by readiness.
      - iv. Have a fire watch person at stand-by readiness.
      - v. Perform explosive atmosphere testing.
      - vi. Provide necessary welding screens.
      - vii. Determine if wind may present a potential hazard.
      - viii. Cover sewers and drains.
      - ix. Protect lower levels.
      - x. Provide adequate ventilation.
      - xi. Assure equipment is purged, flushed, and/or filled with water as required.

- xii. Post and rope off areas.
- xiii. Wear required safety clothing and devices including respiratory equipment.
- xiv. Following completion of the hot work:
  - 1. Cleanup area
  - 2. Replace extinguishers
  - 3. Provide fire watch for one half hour
  - 4. Return copy of Hot Work Permit form to Clark Construction Company Superintendent.
- f. Construction related fire hazards:
  - i. Rubbish and trash accumulation
    - 1. Trash containers shall be provided for proper disposal of debris; and all trash containers shall be emptied or removed from the building at the end of each shift.
    - 2. Trash dumpsters shall be located a sufficient distance from the building.
    - 3. Care must be taken not to mix hot items (cigarettes, etc.) in the trash containers.
    - 4. Do not block or stack material against exits, fire extinguishers or electrical switches or panels.
  - ii. Cutting and welding
    - 1. Store acetylene and oxygen cylinders in safe locations, protected from high temperatures and separated from each other by a distance of 20 feet or a five foot, one hour fire-rated flame barrier.
    - 2. All cylinder valves shall be shut off and capped when not in use.
    - 3. Remove all combustibles from the area or be sure they are safely covered prior to starting hot work.
    - 4. Fire watchman shall be posted with a fire extinguisher during the work period and for at least 30 minutes after completion when the possibility of sparks or slag starting a fire exists.

013529-15 Project No. 2714

- 5. Fire resilient enclosures or fire blankets shall be utilized where possible.
- 6. Use "Hot Work" permit procedure if work needs to be done in an area where combustibles and flammables are present or otherwise required by Clark Construction Company.
- iii. Fireproofing
  - 1. When fireproofing is specified, it should be applied as soon as possible to afford fire protection to beams and columns in the event of a fire.
- iv. Flammable/combustible liquids
  - 1. Small quantities can be safely handled in approved, metal safety containers (not plastic), and should be kept in isolated locations.
  - 2. No Smoking in areas of flammable or combustible liquids in use or stored.
  - 3. Discharge devices requiring the container to be pressurized are prohibited.
  - 4. Pumping devices or faucets shall be well maintained.
  - 5. Only Underwriter's Lab (UL), Factory Mutual (FM) listed equipment shall be used.
  - 6. Bulk containers used to dispense flammable liquids shall be kept outside at least ten feet from any building or inside a building used exclusively for the storage of flammable liquids and located at least ten feet from any other building.
  - 7. Tanks and containers used for the storage of flammable liquids above ground shall be labeled (placarded), and "No Smoking" signs conspicuously placed.
  - 8. Tanks shall be grounded during dispensing operations.
  - 9. Area near storage tanks shall be graded away from building and diked.
  - 10. Do not use flammable materials near electrical panels, switches, fork trucks or other combustion engines
  - 11. Use Hot Work Permit procedures.

- 12. When combustible materials are present, a water supply shall be available.
- v. Temporary heaters
  - 1. Constant supervision is required.
  - 2. Only Underwriters Lab (UL) or Factory Mutual (FM) listed equipment shall be used.
  - 3. Area shall be cleared of all combustible material.
  - 4. Adequate ventilation shall be maintained.
  - 5. A fire extinguisher shall be readily available.
  - 6. Care shall be taken not to block means of egress.
  - 7. All spare cylinders shall be stored outside the building.
  - 8. Hoses supplying fuel to heaters shall be secured off of the floor and out of harm's way.
  - 9. Hoses shall be properly supported by materials which will not cause damage to the hose (a <u>wire</u> hanging wire is not allowed).
- vi. Internal Combustion Engines
  - 1. Care shall be taken to avoid exhaust discharge near or in contact with combustible material and flammable liquids.
  - 2. Engines shall be shut down and allowed to cool before refueling to prevent flash fires.
  - 3. A fire extinguisher shall be kept in close proximity to the unit.
- vii. Concrete Forms
  - 1. Non-combustible forms should be used whenever possible.
  - 2. Concrete shall be poured as quickly as possible after combustible forms have been constructed, and the forms removed as soon as possible after the concrete has set.
  - 3. During the time combustable forms are in place, storage and construction operations on that floor should be held to a minimum and ignition sources should be eliminated.

013529-17 Project No. 2714

- viii. Temporary Enclosures of Buildings
  - 1. Only flame-resistant tarpaulins or material of similar fire retardant characteristics shall be used.
  - 2. If plastic is used, it shall be the type that does not readily ignite and that exhibits slow burning characteristics.
  - 3. Temporary enclosure material shall be securely fastened or guarded to prevent contact with heaters or other sources of ignition.

#### ix. Tar Kettles

- 1. The use of tar kettles shall be continually supervised.
- 2. Tar kettles are permitted only outdoors away from combustibles or on a non-combustible floor or roof in the building.
- 3. Metal covers shall be provided to smother potential fires.
- 4. A fire extinguisher shall be kept in close proximity to the tar kettle. Should a fire occur, do not use water.
- 5. Roofing mops shall never be left indoors or near ignition sources or combustible materials. Note: roofing mops have been known to spontaneously ignite.

#### x. Electrical fires

- 1. Proper extinguishers must be used for electrical fires.
- 2. Extreme care must be used when dealing with energized circuits during fires.
- 3. Toxic fumes could be given off from certain electrical insulations.
- 4. Fire damage to wiring may extend beyond main involvement and proper investigation is required to isolate all damaged wiring.
- 5. Before re-energizing circuits, proper testing should be performed to assure that all damaged wiring has been eliminated.
- 6. Preventive maintenance of electrical systems will significantly reduce the amount of electrical fires.

#### xi. Housekeeping

- 1. Work and storage areas shall be kept clean, orderly and free of combustible materials.
- 2. Combustible liquids shall be stored away from arc producing devices.

### 9.6. Electrical Safety

- A. Temporary lighting
  - 1. Temporary lighting may not rest on any metal unless properly insulated.
  - 2. Proper illumination levels shall be maintained.
  - 3. Temporary lights shall be removed after permanent lighting is installed and the area is turned over.
- B. Temporary power
  - 1. GFCI circuits shall be supplied and regularly tested.
  - 2. Temporary power wiring shall be supported as per code requirements.
  - 3. Proper height restrictions shall be adhered to regarding temporary wiring installations.
- C. Tools
  - 1. All tools requiring grounds shall have grounded plugs. Double insulated tools do not need grounded plugs.
  - 2. Damaged tools and ladders are shall be repaired or replaced.
  - 3. All electrical power tools shall be tested for electrical safety prior to use.

#### 9.7. Hazard Communication

- A. Each Trade Contractor shall implement a Hazard Communication Program in accordance with federal, state and local and other applicable requirements.
- B. Each Trade Contractor utilizing hazardous chemicals on the project site shall:
  - 1. maintain project specific Safety Data Sheets (SDS) on the project site available to its employees for all hazardous chemicals and post the location of (SDS), as well as the contact person and telephone number, of the person responsible for managing this file;
  - 2. provide employees of other employer(s) on the project site access to (SDS) for each hazardous chemical their employees may be exposed to;

- 3. inform other employer(s) of any precautionary measures to be taken to protect employees during the workplace's normal operating conditions and in foreseeable emergencies;
- 4. inform the other employer(s) of the container labeling system used in the workplace; and
- 5. submit to Clark Construction Company a substance inventory list and (SDS) for hazardous chemicals prior to bringing such chemicals to the project site and cooperate in maintaining a current (SDS) record held by Clark Construction Company on the project site.
- 6. Trade Contractors shall train employees in chemical hazards and controls for hazardous chemicals used. Training shall include:
  - a. symptoms of exposure;
  - b. exposure prevention including control procedures, work practices and personal protective equipment;
  - c. exposure reaction procedures;
  - d. hazard label and (SDS) understanding; and
  - e. proper disposal of hazardous chemicals.
- 7. Label all hazardous chemical containers. Labels shall include the following minimum information:
  - a. container contents;
  - b. hazard warning(s); and
  - c. name and address of manufacturer.
- 8. Notify Clark Construction Company and other Trade Contractor in the work area when hazardous chemicals will be in use and potential hazards which may be encountered.

## 9.8. Lockout/Tagout

- A. No one shall be permitted to work on and energized circuit. Trade Contractor shall employ procedures to ensure that no one will be exposed to hazards as a result of an energized circuit.
- B. Trade Contractor shall adhere to the most stringent of the following lockout/tagout procedures as required:
  - 1. trade contractors own requirements;
  - 2. owner requirements;

Saline Area Schoosl 2015 Bond Program – 2017 Projects Phase 1 Bid Release No.4 Section 013529 Trade Contractor Health and Safety Plan Version 8 - pending 013529-20 Project No. 2714

3. Clark Construction Company requirements; or other project specific requirements.

### 9.9. Motor Vehicles and Equipment

- A. All equipment shall be inspected daily before each use by the operator.
- B. Defective equipment shall be removed form service immediately.
- C. All operators shall be properly licensed and certified. Copies of certifications shall be maintained on site and shall be made available to Clark Construction Company upon request.

### 9.10. **Rigging**

- A. Lifting and Rigging Planning
  - 1. All loads shall be rigged by a qualified rigger.
  - 2. Materials being hoisted shall be rigged to prevent unintentional displacement.
  - 3. All hoisting operations shall be pre-planned to ensure that all applicable local and federal health and safety agency requirements are met.
  - 4. Routes for suspended loads shall be preplanned to ensure that no employee is required to work directly below a suspended load, except for employees as allowed by local and federal health and safety agencies (i.e., structural steel connector making initial connection).
  - 5. Tag lines shall be used on all loads.
- B. Rigging Equipment
  - 1. Rigging equipment for material handling shall be inspected at the time of installation, before each job, and at the beginning of each shift if in use, by an employee qualified to perform this inspection.
  - 2. Defective rigging equipment shall be removed from service.
  - 3. Hooks with self-closing safety latches or their equivalent shall be used to prevent components from slipping out of the hook.

#### 10. NON-CONFORMANCE

- 10.1. Safety violations shall be promptly corrected by the Trade Contractor.
- 10.2. A "Notice to Comply" (NTC) may be issued by the Construction Manager for follow-up and response by the Trade Contractor (Refer to "Non-Conformance Procedure Section 014517 in front end documents)

# END OF SECTION

# SECTION 013543 ENVIRONMENTAL PROCEDURES

## 1. PROJECT SITE REQUIREMENTS

- 1.1. Clark Construction Company is committed to constructing and delivering to the client a project completed in an environmentally conscious manner from Design/Bid to Turnover.
- 1.2. Clark Construction Company's Environmental Policy is available for review upon request. Adherence to this policy shall be required.
- 1.3. Trade Contractors shall cooperate with the Construction Manager in evaluating Significant Environmental Aspects of the Work which it has control over and plan and implement such action as required to prevent significant impacts on the environment.
- 1.4. Trade Contractors shall take all necessary action and comply with applicable federal, state, county and municipal laws to prevent pollution of rivers, streams, waterways, lakes, soils and air.
- 1.5. Material, equipment, supplies or workmanship found not to comply with Clark Construction Company, and regulatory agency environmental requirements shall be promptly corrected by the Trade Contractor.
- 1.6. A "Notice to Comply" (NTC) may be issued by the Construction Manager for follow up action and response by the Trade Contractor. (Refer to Section 01410 for Non-Conformance Procedures).

## 2. <u>SOIL EROSION AND SEDIMENTATION CONTROL</u>

- 2.1. Trade Contractor shall be responsible to maintain all soil erosion and sedimentation control within the project limits in accordance with Public Act 347, 1972 as amended and as required by the Contract Documents. As a minimum, Trade Contractor shall keep a log of weekly inspections of the soil erosion and sedimentation control measures.
- 2.2. Trade Contractor shall conduct all excavation, embankment, grading and clean-up operations in a manner such that siltraceous soil materials and waterborne objects are not discharged into any storm sewer, drainage ditch, river, lake or underground utility system. In addition, soil shall be managed so that it does not become airborne.
- 2.3. Soil erosion and sedimentation extending beyond the project will not be permitted. This includes soil washing across both turf and paved areas and soil washing into underground utility systems.
- 2.4. All violations of this requirement shall be corrected within 12 hours of notification by the Project Representative to eliminate reoccurrence.
- 2.5. Maintain all erosion control measures in good working order at all times to provide for maximum efficiency of the control provision. Trade Contractor shall keep a log of weekly inspections of the soil erosion and sedimentation control measures. All control measures shall be re-inspected after any significant rain fall.

# SECTION 013543 ENVIRONMENTAL PROCEDURES

## 3. <u>DUST CONTROL</u>

3.1. Dust control shall be provided by each Trade Contractor as required to prevent nuisance and hazard.

### 4. <u>WATER CONTROL</u>

4.1. Maintain excavations free of water. Provide, operate and maintain pumping equipment.

### 5. <u>CATCH BASIN FILTERS</u>

- 5.1. Water from trenches and other excavations shall not be pumped directly into catch basins (or manholes being temporarily used as catch basins) without first being properly filtered through the approved sediment filter system.
- 5.2. All manholes and catch basins within the project site, and all such structures offsite where sediment is likely to flow from the construction site or be carried by construction traffic, shall be securely covered with the approved sediment filter system.
- 5.3. Filter system shall be maintained clean and shall be periodically replaced by the Trade Contractor as required to maintain adequate water flow.
- 5.4. Install 36" high filter fence around structures that have not been paved.

#### 6. DECONSTRUCTION AND SALVAGE OPPORTUNITIES

- 6.1. Separate dumpsters shall be required and set up through local recycling centers for the different types of recyclable materials, ie. metals, concrete, masonry, wood, gypsum etc.
- 6.2. Contractors shall be responsible for the disposal of recyclable items in the proper containers/ dumpsters. Compound materials are to be broken down for proper recycling ex. Wood doors with metal hardware, etc.
- 6.3. Separation of materials is required for inclusion in the proper dumpsters ie. removing door hardware off existing doors, wood door in wood dumpster and door hardware in the metals dumpster.
- 6.4. Salvage opportunities, including debris sorting to maximize recycling efforts, is required and will need to be reported to Clark Construction Company on a monthly basis, if dumpsters are not provided by Clark Construction Company. Percentage of waste diverted from landfills must also be reported.

## 7. <u>CONSTRUCTION INDOOR AIR QUALITY</u>

- 7.1. All HVAC duct work coming into the building for new installation shall have all vents and ends capped with a plastic covering to prevent dust building up and other debris from entering the duct prior to installation.
- 7.2. New HVAC units shall not be run for temporary heating unless deemed there are no other reasonable methods for temporary heat. In the event that new equipment will be required to

# SECTION 013543 ENVIRONMENTAL PROCEDURES

be used MERV 13 filters shall be required. The schedule of filter change out will depend on the intensity level of work on the project.

- 7.3. No smoking shall be allowed on the districts property. This is also in correlation with Clark Construction Company jobsite policies.
- 7.4. Reference specification section 015000 Temporary Facilities and Controls for the requirements of "temporary heating and cooling" during construction.

END OF SECTION
# SECTION 014516 FIELD QUALITY CONTROL PROCEDURES

## 1. <u>GENERAL</u>

1.1. Trade Contractor shall provide premium quality workmanship and service of all Work under its Contract.

## 2. <u>QUALITY ASSURANCE</u>

- 2.1. Trade Contractor shall be responsible to implement a quality control program relative to its suppliers, manufacturers, products, services and workmanship, to assure Work of specified quality.
- 2.2. Comply fully with manufacturer's requirements. Conflicts with manufacturer's requirements and the Contract Documents shall be brought to the attention of the Construction Manager for review and direction by the Architect prior to proceeding with the Work.
- 2.3. Standards of quality required in the Contract Documents shall be considered as minimum standard. Trade Contractor shall meet higher standards of quality as required by authorities having jurisdiction over the project or other entities as applicable.
- 2.4. Conflicts in the Contract Documents shall be brought to the attention of the Construction Manager for clarification and direction prior to proceeding with the Work.
- 2.5. All Work shall be performed by qualified crafts persons to produce workmanship of specified quality.

### 3. <u>REFERENCES</u>

3.1. Conform to reference standard current as of date of Contract Documents.

### 4. <u>MATERIAL MOCK-UP</u>

- 4.1. Construct mock-ups at the site for review and approval as required in the Contract Documents.
- 4.2. Quality of workmanship shall be equal to or better than the quality of workmanship of the approved mock-up.
- 4.3. Trade Contractor shall remove mock-ups from the project site upon completion of the Work.

### 5. <u>REQUESTS FOR INFORMATION</u>

5.1. Clarification of the Contract Documents shall be requested in the form of Request for Information (RFI) utilizing the Request for Information Form module in the Project specific Website (Connect) (Refer to "Certificates and Other Forms" section).

### 6. INSPECTION AND TESTING LABORATORY SERVICES

6.1. Cooperate with independent material inspection and testing firm employed by others to accomplish all inspection and testing required to assure the quality of materials and workmanship.

# SECTION 014516 FIELD QUALITY CONTROL PROCEDURES

6.2. Costs for any re-testing required as a result of non-conforming or apparent non-conforming Work will be charged to the Trade Contractor.

### 7. <u>MANUFACTURER'S FIELD SERVICES AND REPORTS</u>

- 7.1. When specified in individual Specification sections, Trade Contractor shall require material or product suppliers or manufacturers to provide qualified representatives to observe site conditions and installation procedures as specified or as required for warranties or other requirements.
- 7.2. Report observations and direction made by representatives that are supplemental or contrary to manufacturer's written instructions, Contract Documents or other requirements.

### 8. <u>REGULATORY REQUIREMENTS</u>

8.1. Standards and regulations referred to throughout the Contract Documents establish the minimum requirements of the Work. Where the Contract Document requirements exceed the applicable code, the Contract Documents drawings shall be met. Unless specifically noted, the current edition of all applicable regulatory codes shall be met.

### 9. <u>TOLERANCES</u>

9.1. Comply with manufacturer's quality tolerances. Manufacturer's tolerances which conflict with Contract Documents shall be brought to the attention of the Construction Manager for clarification and direction prior to proceeding with the Work.

#### 10. <u>NON-CONFORMANCE</u>

- 10.1. Material, equipment, supplies or workmanship found not to comply to the Contract requirements shall be promptly corrected by the Trade Contractor.
- 10.2. A "Notice to Comply" (NTC) may be issued by the Construction Manager for follow up action and response by the Trade Contractor. (Refer to "Non-Conformance Procedure" section).

# SECTION 014517 NON-CONFORMANCE PROCEDURE

# 1. <u>GENERAL</u>

- 1.1. Trade Contractor performance found not to comply with requirements relative to safety, quality and environmental issues, shall be promptly corrected by the Trade Contractor.
- 1.2. In the event of non-conforming performance, a Notice to Comply (NTC) may be issued by the Construction Manager for follow-up, action and response by the Trade Contractor. (Refer to "Certificates and Other Forms" section).

### 2. NOTICE TO COMPLY PROCEDURES

- 2.1. Construction Manager issues Notice to Comply (NTC).
- 2.2. Trade Contractor, jointly with the Construction Manager, determine the cause for nonconformity. Cause for non-conformity is recorded on the Notice to Comply (NTC).
- 2.3. Trade Contractor, jointly with the Construction Manager; propose corrective action to ensure non-conformity does not recur on project. Corrective action is recorded on the Notice to Comply (NTC).
- 2.4. Construction Manager reviews proposed corrective action and approves if acceptable. Approval of proposed corrective action may require review and acceptance by the Owner, Architect/Engineer or other entity as deemed appropriate by the Construction Manager. Approval is documented on the Notice to Comply (NTC).
- 2.5. Upon approval of proposed corrective action, the Trade Contractor implements approved corrective action. The Trade Contractor shall document corrective action taken and date completed on the Notice to Comply (NTC) and return form to the Construction Manager.

### 1. <u>TEMPORARY UTILITIES</u>

#### 1.1. **TEMPORARY ELECTRICAL SERVICE**

- A. The Electrical Trade Contractor shall provide power for construction operations including wiring, transformers, connections, and protection.
  - 1. Maintain building power throughout the entire construction phase.
- B. The Electrical Trade Contractor shall provide a complete lighting system throughout the building for construction operations.
  - 1. Maintain building lighting throughout the entire construction phase. If unable temporary lighting shall consist of a minimum of one (1) 150 Watt lamp, or equivalent, for each 500 square feet of floor area, or a minimum of 10 foot-candle over the entire floor area of the building (or as required to meet current MIOSHA Standards). Provide a minimum of one (1) light in each room (excluding closets and utility chases, corridors, stairwells, etc.). Provide additional lighting in corridors and stairwells as directed by the Construction Manager as conditions require.
  - 2. Provide spare lamps and maintain temporary lighting system until permanent lighting is installed and activated.
- C. Trade Contractor's maximum individual load connected to the system shall be as follows:

VOLTAGE-PHASE	MAXIMUM LOAD
120 Volt, 1 phase	1.5 kVA
240 Volt, 1 phase	2.5 Kva

- D. Trade Contractors shall not use temporary light and power system for electric heating, welders or for tools requiring larger than minimum loads specified above.
- E. If Work of separate Trade Contractors or Subcontractors requires three phase power or voltage exceeding 120 Volts for tools or equipment, that Trade Contractor or subcontractor with additional power requirements shall provide required temporary distribution system.
- F. Each separate Trade Contractor and Subcontractor shall make arrangements for electrical service, lighting and power for their field office, storage sheds and other temporary buildings.
- G. Electrical Trade Contractor shall remove temporary light and power system when no longer required. Salvaged materials shall become property of the Electrical Trade Contractor.

### 1.2. TEMPORARY HEATING AND COOLING

- A. The HVAC Trade Contractor shall be responsible for all costs associated with the material and Work described herein.
- B. The HVAC Trade Contractor shall place permanent heating and cooling equipment in operation as directed by the Construction Manager. The HVAC Trade Contractor shall operate and maintain permanent system while in operation prior to substantial completion of the project.
- C. The HVAC Trade Contractor agrees to provide testing, cleaning, adjusting and flushing, including precautions to be taken to prevent unusual wear or damage to the permanent heating system during its use for temporary heat and ventilation and shall submit to the Construction Manager its detailed plan.
- D. Permanent heating and cooling systems and equipment shall not be put into operation for temporary heating until the systems' components and equipment have been tested and cleaned in compliance with the approved plan. In addition:
  - 1. No permanent fan system shall be started, or ducts used, unless proper and adequate temporary air filters are installed and the Construction Manager has checked and approved the installation. Permanent air filters shall not be used for any temporary heating.
  - 2. Temporary air filters shall be replaced when the pressure drop across the filter bank exceeds the manufacturer's recommended maximum.
  - 3. Temporary filters shall be removed and replaced by permanent filters, at the termination of temporary heat, and prior to any testing or balancing of permanent heating system.
  - 4. All warranties for the permanent heating system, as used for temporary heat and ventilation, shall commence on the date of project substantial completion, prior use notwithstanding, unless accepted in advance of this date by the Owner.
  - 5. Use of permanent systems for temporary construction heating and cooling shall not effect the requirement that the warranty start date shall coincide with the substantial completion date as determined by the Architect and included in the Certificate of Substantial Completion. HVAC Trade Contractor shall include any additional cost to meet this provision in its base bid.
- E. Upon completion of use of temporary heating, permanent heating system shall be restored to operative conditions in manner acceptable to the Architect/Engineer. Entire duct system and all its components shall be thoroughly and properly cleaned to satisfaction of architect/engineer and Construction Manager. HVAC Trade Contractor shall guarantee material and equipment cleanliness. Cleaning shall be done so as not to damage other Work of the Contract or other contracts.
- F. Refer to Environmental Management specification section 015300 for specific details regarding the construction indoor air quality plan.

### 1.3. TEMPORARY TELEPHONES

A. Temporary telephones shall be arranged and paid for by each individual Trade Contractor per their requirements.

#### 1.4. TEMPORARY DATA CONNECTIONS

A. Data connections if required by Trade contractor shall be arranged for and paid for by each individual Trade Contractor per their requirements.

### 1.5. **<u>TEMPORARY WATER</u>**

A. Temporary water will be available from an on-site fire hydrant upon completion of underground utilities. Each Trade Contractor shall transport the water to their area of Work as required.

### 1.6. **<u>TEMPORARY TOILETS</u>**

A. The Construction Manager will provide temporary toilets.

### 1.7. FIRE SAFETY

A. The General Trades Contractor shall provide a minimum of 20 pound U.L. rated multi-purpose (Type A, B, C) fire extinguishers on stands spaced throughout the building at all times spaced such that the travel distance to a fire extinguisher shall not exceed 100 feet. Fire extinguisher stand shall be "Fire-Safety Mate" or equal as manufactured by the Goria Corporation, (telephone 888.464.6742).

#### 1.8. **TEMPORARY CONTROLS**

#### A. <u>CONSTRUCTION CLEANING</u>

#### 1. **On-going Clean-up**

- a. Project clean-up and disposal of debris is the responsibility of each Trade Contractor for their own material and shall be performed on a daily basis.
- b. Each contractor will be responsible to compensate Clark Construction 4 hour each week for every 5 workers on site at a cost of \$56.00/hour. This is for just on going clean and not contractor identifiable clean up.
- c. Trade Contractor shall furnish labor for clean-up of unidentifiable debris including broom clean as directed by the Construction Manager in proportion to its Workforce.
- d. Trade Contractor shall be responsible for disposal of debris and dumpster costs associated with its Work.
- e. In the event that the Trade Contractor fails to clean up and dispose of all debris promptly after written direction to comply, the Construction

Manager will clean-up the debris and Trade Contractor shall be responsible for all associated costs.

### 2. Work in Place Progress Cleaning

- a. Each Trade Contractor shall be responsible for cleaning of all material and equipment installed upon completion of installation of same.
- b. Work in place progress cleaning by each Trade Contractor shall include but not limited to the following:
  - i. Remove labels which are not required to be permanent.
  - ii. Clean transparent materials, including mirrors and window/door glass to a polished condition.
  - iii. All interior and exterior building material and equipment finished surfaces and components shall be cleaned to be free of dirt, dust, oils, film, and other foreign material.
  - iv. Clean debris and dust from limited-access spaces including roofs, plenums, ducts, shafts, trenches, equipment vaults, manholes, attics and similar spaces.
  - v. Each Trade Contractor, after completing its Work in any space, shall thoroughly clean any and all debris, dirt and dust caused by its Work or presence in a space. Such clean-up shall include but not be limited to:
    - 1. Vacuum or washing of floor, wall, ceiling and other surfaces.
    - 2. Restoring interior and exterior areas to condition that existed prior to occupying the space or commencement of the Work.

### 3. **Debris disposal**

- a. All debris shall be legally disposed of off-site. Salvage opportunities, including debris sorting, to maximize recycling efforts is required and will need to be reported to Clark Construction Company on a monthly basis, if dumpsters are not provided by Clark Construction Company. Percentage of waste diverted from landfills must also be reported.
- b. Debris shall not be buried, covered up or left in walls or above ceilings.
- c. Burning of debris will not be permitted on the project site.
- d. Trade Contractor shall be responsible for disposal of debris and dumpster costs associated with its Work.

#### 4. Street Cleaning

- a. Trade Contractors shall be responsible for keeping all paved surfaces and public streets in the vicinity of the project clean of dirt and other debris that is a direct result of his operations.
- b. Cleaning shall consist of cleaning the streets with a pick-up type sweeper in such a manner that the pavement and gutters are maintained broom clean for the duration of the Contract.
- c. In the event that the Trade Contractor fails after written direction to promptly comply with street cleaning requirements, the Construction Manager shall perform the Work and charge the Trade Contractor all associated costs.

### 1.9. MATERIAL AND EQUIPMENT HANDLING

- A. Material and equipment shall be handled and stored utilizing means and methods to avoid damage, soiling, exposure to adverse weather, etc.
- B. Store weather sensitive products in weather-tight enclosures; maintain within humidity ranges required by manufacturer.
- C. Products stored on-site shall be supported as not to be in contact with ground and to avoid soiling.
- D. In the event that products become soiled, exposed to adverse weather or damaged, the Trade Contractor shall clean, repair or replace if necessary the product to achieve new condition acceptable to the Owner and Architect.
- E. Coordinate material and equipment deliveries to meet requirements of the construction schedule.
- F. Coordinate all delivery dates and storage spaces with the Construction Manager.
- G. Trade Contractor shall be responsible for receipt, unloading and handling of products under its Contract. The Construction Manager or others will not receive deliveries for Trade Contractors unless noted otherwise in the Documents.

#### 1.10. PROTECTION OF INSTALLED WORK

- A. Protect installed Work and provide special protection where specified in Contract Documents.
- B. Trade Contractors required to Work on roof surfaces shall be responsible to take all precautions necessary to avoid puncture to roofing materials and shall place not less than the equivalent of <sup>1</sup>/<sub>2</sub> inch plywood to protect Work surface.

### 1.11. **SECURITY**

A. Trade Contractors shall provide all means of security and facilities as necessary to protect tools and equipment from theft or vandalism. Trade Contractor's shall be responsible for insurance and cost of replacement of all tools and equipment.

## 1.12. OFFICE/STORAGE FACILITIES

- A. Trade Contractor shall not place storage and/or office facilities on-site without prior approval of the Construction Manager.
- B. Storage and office facilities shall be in good condition and appearance as determined by Clark Construction Company. Clark Construction Company reserves the right to demand removal of facilities that do not meet this requirement.
- C. Field offices and storage facilities shall remain portable and are subject to relocation.
- D. Cost of set-up and any relocation of field offices and storage facilities shall be the responsibility of the Trade Contractor.
- E. Electric heat or air-conditioning will not be allowed for field offices or storage facilities unless the Trade Contractor arranges for its own metered electrical service.

### 1.13. REMOVAL OF UTILITIES, FACILITIES AND CONTROLS

- A. Remove all temporary above grade or buried utilities, equipment, facilities and materials when directed by the Construction Manager.
- B. Clean and repair damage caused by installation or use of temporary Work.

# SECTION 017123 FIELD ENGINEERING

## 1. <u>GENERAL</u>

1.1. The Owner will employ the services of a Civil Engineer or registered land surveyor to establish benchmarks, control points, and building corners. Trade Contractor shall be responsible for all other engineering and layout required for performance of its Work.

### 2. <u>REQUIREMENTS</u>

- 2.1. Confirm accuracy of information indicated on the drawings with property survey and existing benchmarks before commencing layout of the Work. Locate and protect existing benchmarks and control points. Preserve permanent reference points during construction.
- 2.2. Discrepancies between the Contract Drawings and existing conditions shall be brought to the attention of the Construction Manager prior to layout of the Work.
- 2.3. Confirm engineering and layout of previous Trade Work. Do not rely on Work or layout of other Trade Contractors. Proceeding with Work adjacent to or in concert with the Work of other trades shall indicate acceptance of responsibility for accuracy of layout of previous Trade Contractor Work.
- 2.4. Engage the services of a registered land surveyor for any layout associated with the performance of site related Work.
- 2.5. Establish and maintain a minimum of two permanent benchmarks on the site, referenced to data established by survey control points.
- 2.6. Prior to layout and construction, verify location and invert elevation at points of connection of all utilities including sanitary sewer, storm sewer, and water service piping.
- 2.7. Working from lines and levels established by the property survey, establish benchmarks and markers to set lines and levels at each story of construction and elsewhere as needed to properly locate each element of the Project. Calculate and measure required dimensions within indicated or recognized tolerances. Do not scale drawings to determine dimensions.
- 2.8. Confirm all major elements for line, level and plumb as Work progresses.
- 2.9. Record benchmark locations, with horizontal and vertical data, on Project Record Documents.
- 2.10. Do not change or relocate benchmarks or control points without prior written approval. Promptly report damaged, missing or disturbed control points.
- 2.11. Promptly replace disturbed or destroyed control points. Replace control points in exact original location.

### 3. <u>EXISTING CONDITIONS</u>

3.1. Location of underground utilities, such as sewers, electrical power, water piping, conduits, structures, etc., on the drawings is as accurate as can be determined from available information. Accuracy or completeness of this information is not guaranteed. Exact locations and elevations shall be verified by the Trade Contractor prior to starting Work.

# SECTION 017123 FIELD ENGINEERING

- 3.2. Exercise extreme care when excavating at or near the general location of underground utilities.
- 3.3. Notify Miss Dig or other utility location services as required by the Contract Documents, other authorities or quality construction practice, a minimum of three (3) working days prior to commencing site excavation or interruption of utility services.

## 1. <u>GENERAL</u>

- 1.1. Trade Contractor shall be required to perform all cutting and patching as required for its Work and to accomplish the following:
  - A. New construction into existing conditions.
  - B. Relocation of existing equipment or devices including patching areas where items were removed and any cutting and patching required at the new location.
  - C. Repair existing surfaces damaged as a result of incidental demolition Work. Repaired area shall match adjacent construction in all aspects.
  - D. Fit several parts properly.
  - E. Uncover Work to provide for installation of ill-timed Work.
  - F. Remove and replace defective Work.
  - G. Remove and replace Work not conforming to requirements of the Contract Documents.
  - H. Remove samples of installed Work as specified for testing.
  - I. Certain construction Work may be completed before the Work of this bid category begins. Cutting and patching of such Work shall be included in the Work of this Bid Category.

### 2. <u>DEFINITIONS</u>

- 2.1. Cutting and patching includes cutting into existing construction to provide for the installation or performance of other Work and subsequent fitting and patching as required to restore surfaces to their original condition.
- 2.2. Cutting and patching performed during the manufacture of products or during the initial fabrication, erection, or installation processes is not considered to be cutting and patching under this definition. Drilling of holes to install fasteners and similar operations are not considered to be cutting and patching.
- 2.3. Selective demolition is recognized as related but separate categories of Work, which may or may not require cutting and patching as defined in this section.

### 3. <u>PERFORMANCE</u>

- 3.1. Before cutting, examine surfaces to be cut and patched and conditions under which the Work is to be performed. Take corrective action prior to proceeding with cutting that would result in an unsafe condition.
- 3.2. Notify the Construction Manager immediately of any materials encountered which are suspected to contain asbestos. Asbestos-containing material will be removed and disposed of by the Owner.

- 3.3. Modification of the Work of other Trade Contractors or existing conditions shall not be performed without the consent of the Construction Manager.
- 3.4. Protect other Work during cutting and patching to prevent damage. Provide protection from adverse weather conditions for that area of the project that may be exposed during cutting and patching operations.
- 3.5. Take precautions to avoid cutting existing pipe, conduit, duct and other items to remain in service either temporarily or permanently.
- 3.6. Coordinate with other Trade Contractors installation of any sleeves and box outs for the performance of the Work. Include layout and installation of sleeves and box outs to be installed into the Work of other Trade Contractors.
- 3.7. Trade Contractor needing sleeves and box outs shall fill voids around sleeves and box outs with appropriate material to meet required ratings. Provide escutcheon plates where Work will be exposed.
- 3.8. Coordinate date and time of cutting and removal of Work to be uncovered for review and inspection by others.
- 3.9. Cutting and patching of existing materials shall be the responsibility of the Trade Contractor requiring the same.
- 3.10. Cutting and patching shall be performed by craftspersons skilled in working with the material being cut and patched.
- 3.11. Where cutting or patching is required of one Trade Contractor as a result of negligence of another Trade Contractor, cost shall be borne by the negligent Trade Contractor.
- 3.12. Comply with requirements of earth work specification sections where cutting and patching require excavation and backfilling.
- 3.13. All patching shall be done in a neat workmanlike manner with materials to match existing. Fit work tight to adjacent elements. Maintain integrity of wall, ceiling and floor construction and completely seal voids.
- 3.14. Use hand or small power tools designed for sawing or grinding to the greatest extent possible.
- 3.15. Cut using methods that are least likely to damage Work to be retained or adjoining Work. Where necessary, review proposed procedures with the original installer and Construction Manager. Comply with methods that will achieve highest quality results.
- 3.16. Joint and seam tolerances shall meet the most stringent quality standards in accordance with the Contract Documents or industry standard tolerances for the material being patched.
- 3.17. Coordinate with other Trade Contractors for fitting the Work.
- 3.18. Cutting of existing Work shall be straight, true and of proper size.

- 3.19. Cutting and patching of concrete floors and decks shall be performed in a neat and workmanlike manner, using a coring machine. Openings over six inch diameter must be formed by the Trade Contractor requiring the opening. Each Trade Contractor shall pack and grout openings around sleeves or Work penetrating the floor or deck associated with its Work.
- 3.20. Submit shop drawings and obtain approval of the Architect/Engineer prior to cutting or altering structural members. Shop drawings shall clearly identify each cut or alteration. Include detailed cutting plan.
- 3.21. Provide temporary support to prevent failure of Work to be cut.
- 3.22. Hot work permits shall be obtained from the Construction Manager for any cutting or patching work which requires the use of flame of any kind.

## 4. <u>SUBMITTALS</u>

- 4.1. Where prior approval of cutting and patching is required, submit proposed procedures for Work to be performed and request approval to proceed. Include the following information, as applicable:
  - A. Describe the nature of the Work and detailed performance plan indicating why cutting and patching cannot be avoided. Describe anticipated results of the Work in terms of changes to existing Work, including structural, operational and visual changes as well as other significant elements.
  - B. List of products to be used and firms that will perform Work.
  - C. Dates Work is expected to be performed.
  - D. List utilities that will be disturbed or otherwise affected by Work, including those that will be relocated and those that will be out of service temporarily. Include duration utility service will be disrupted.
- 4.2. Approval by the Architect/Engineer to proceed with cutting and patching Work does not waive the Architect/Engineer's right to later require complete removal and replacement of Work found to be cut and patched in an unsatisfactory manner.
- 4.3. Submit cost estimate prior to cutting and patching to be performed and considered to be a change in scope of Work.
- 4.4. In the event conditions of Work or schedule require a change of materials or methods, submit request and recommendation. Include the following information:
  - A. Describe condition requiring change.
  - B. Recommendations for alternative materials or methods.
  - C. Submittals as required for substitutions.
  - D. Hot work permit if flame is used in any form to perform the Work.

# 5. <u>MATERIALS</u>

5.1. Except as otherwise indicated, or as directed by the Architect/Engineer, use materials for cutting and patching that are identical to existing materials. In the event identical materials are not available, or cannot be used, materials that match existing adjacent surfaces to the fullest extent possible with regard to visual effect shall be used. Materials used for patching shall result in equal or better performance characteristics.

# 6. <u>CLEANING</u>

6.1. Thoroughly clean areas and spaces where Work is performed or spaces used as access to Work. Remove paint, mortar, oils, putty and items of similar nature completely. Thoroughly clean materials before painting or other finishing is applied.

# SECTION 017700 CLOSEOUT PROCEDURES

### 1. <u>GENERAL</u>

1.1. Specific project closeout requirements are the responsibility of the Trade Contractor. Coordinate all requirements with the Construction Manager.

### 2. <u>SUBMITTALS</u>

### 2.1. **Operating and Maintenance Date**

- A. Operating and Maintenance Manuals shall be submitted upon completion of 80% of the Work or earlier if required to meet other Contract requirements.
- B. Maintenance Data shall be provided in accordance with "Operation and Maintenance Data" section.
- C. Submit special tools and spare parts upon completion of 90% of the work in accordance with "Operation and Maintenance Data" section.

#### 2.2. Project Record Documents

- A. Project Record Documents shall be submitted upon completion of the Work.
- B. Record Documents shall be provided in accordance with "Project Record Documents" section.
- C. Project Record Documents shall include:
  - 1. "As-built" Drawings.
  - 2. "As-built" Material List.
  - 3. Certificates of Inspection and approval
  - 4. Warranties and Guarantees.
  - 5. Asbestos Free Document (Sample provided by Architect)
  - 6. Smoke and Fire Spread Documentation.
  - 7. Other documentation as required.
- D. "As-built" drawings shall be submitted in electronic format on the project specific website, as required by Clark Construction Company and one (1) hard copy.

### 3. FINAL APPLICATION FOR PAYMENT

- 3.1. Final Application for Payment shall be submitted in accordance with the Contract Documents.
- 3.2. Final Application for Payment shall be submitted only after all Project Closeout submittals, procedures and processes have been completed.

# SECTION 017700 CLOSEOUT PROCEDURES

### 4. <u>CERTIFICATE OF SUBSTANTIAL COMPLETION</u>

- 4.1. Submit Certificate of Substantial Completion on form included in "Certificates and Other Forms" section.
- 4.2. The date of Substantial Completion will be determined by the Architect.

#### 5. <u>INSTRUCTION OF OWNER PERSONNEL</u>

- 5.1. Provide instruction to Owner personnel in accordance with "Operation and Maintenance Data" section.
- 5.2. Coordinate instruction of Owner personnel to occur as early as possible following material and equipment being placed in complete operation.
- 5.3. Provide additional instruction sessions as required to accomplish instruction of material and equipment under differing conditions (i.e., HVAC equipment under heating and cooling conditions).
- 5.4. Video record Owner training sessions and turnover to the owner for their use.
- 5.5. Provide a signed transmittal if the owner declines any and/or all training.

#### 6. WORK COMPLETION/PUNCH LIST

#### 6.1. **PROCEDURE**

- A. Trade Contractor shall develop a work completion list by area of the building as areas of Work are completed. Corrective Work shall be undertaken immediately upon development of the list.
- B. The Construction Manager will review and add additional items to the completion list as needed for completion by the Trade Contractor.
- C. Immediately upon completion of the Work the Trade Contractor shall review the Work and develop a detailed list of punch list and incomplete items of Work and within seven (7) days of completion of the Work, submit the list to the Construction Manager.
- D. Trade Contractor list shall include location of item, description and scheduled date of completion.
- E. Trade Contractor shall immediately upon development of the list begin corrective Work.
- F. Construction Manager will review the list along with all material and equipment installed by the Trade Contractor and add additional items to the list found to be of unacceptable quality or incomplete.
- G. Construction Manager will coordinate with Owner's representatives, Architect and Engineers to review the Work and add additional items as required.

# SECTION 017700 CLOSEOUT PROCEDURES

- H. The complete list of defective or incomplete items which includes review by Owner, Architect and Engineers will be provided to the Trade Contractor for correction/completion of the Work.
- I. Trade Contractor shall be held responsible to correct/complete the Work within thirty (30) calendar days of transmittal of the list by the Construction Manager.
- J. Trade Contractor shall conduct a confirmation walk-through with Construction Manager after correction/completion of the list.
- K. Any punch list items not completed by the Trade Contractor after thirty (30) calendar days or shorter duration as required by the Contract following transmittal of the punch list by the Construction Manager will be completed by others. All costs including Construction Manager time, overhead and profit will be charged to the Trade Contractor.
- L. Final approval and acceptance of completed punch list and completed Work shall be by the Architect and Owner.

## 1. <u>GENERAL</u>

- 1.1. Provide complete product data and related information appropriate for Owner's maintenance and operation of products furnished under the Contract, and provide operating and maintenance data as specified in the section and as referenced in other pertinent Specification sections.
- 1.2. Instruct Owner's personnel in the maintenance of products and in the operation of equipment and systems. If owner training is required video all sessions and turnover to the owner for their use.

## 2. <u>QUALITY ASSURANCE</u>

2.1. Preparation of data shall be done by personnel trained and experienced in maintenance and operation of the described products.

### 3. <u>OPERATING AND MAINTENANCE MANUALS</u>

- 3.1. Furnish to the Construction Manager an electronic copy uploaded to the Project Specific Website and one (1) complete manuals as approved detailing the operation and maintenance of all operable systems and products furnished under the Contract.
- 3.2. Provide operation and maintenance manuals in the following format:
  - A. Binder type
    - 1. Commercial quality three-ring binders with durable and cleanable plastic covers
    - 2. Minimum ring size one inch
    - 3. When multiple binders are used, correlate the data into related consistent groupings
    - 4. Binder cover
      - a. Identify each volume with title "OPERATING AND MAINTENANCE INSTRUCTIONS", and the following information:
        - i. Title of project
        - ii. Identity of separate structure as applicable
        - iii. Identity of general subject matter covered in manual
      - b. Size 8 ½" x 11"
      - c. Paper 20 pound minimum, white, for typed pages
      - d. Text Manufacturer's printed data, or neatly typewritten
    - 5. Drawings (included in manuals)

- a. Provide reinforced, punched binder tab, bind in with text
- b. Fold larger drawings to the size of the text pages
- 6. Table of Contents
  - a. By system, for each volume
  - b. List each major component as subheading to system
- 7. Organization
  - a. Group all components of each system into complete sections
  - b. Provide indexed divider for each system
  - c. Provide index tabs at each major component of each system using same identifications as Table of Contents
- 8. Content to be included in manuals shall be as follows:
  - a. Manuals shall be suitable for use in training of Owner's personnel
  - b. Each manual section shall include the following, as applicable:
    - i. Operating data
    - ii. Narrative operating instructions for total system, organized sequentially and tailored to project installation
    - iii. For architectural products, applied materials and finishes
      - 1. Manufacturer's data, giving full information on products:
        - a. Catalog number, size, composition
        - b. Color and texture designations
        - c. Information required for re-ordering specialmanufactured products
      - 2. Maintenance data
        - a. Procedures
          - i. Recommended maintenance schedules
          - ii. Spare parts lists including manufacturer's catalog numbers
          - iii. Manufacturer's printed data

- iv. All applicable diagrams and system layouts
- v. Recommended cleaning products and methods. Include products or methods that are detrimental to the material.
- vi. Recommended schedule for cleaning and maintenance
- vii.Supplier and installer contact information

#### 4. POSTED OPERATING INSTRUCTIONS

- 4.1. Where applicable, provide operating instructions for posting for each major piece of equipment and systems of equipment, to aid in the operation of the equipment at or near local control panels or major piece of equipment.
- 4.2. Provide posted operating instructions in the following format:
  - A. Name plate data and manufacturers standard instructions, permanently affixed to product by manufacturer.
  - B. Operating instructions and diagrams furnished by Trade Contractor shall conform to the following:
    - 1. Size not larger than 24" x 36" or smaller than  $8\frac{1}{2}$ " x 11".
    - 2. Mounting provide clean rigid protective covering and fasteners to permanently mount instructions to substrates encountered.
    - 3. Groupings of instructions shall be of uniform size.
  - C. Content to be included shall be as follows:
    - 1. Layouts
      - a. Equipment
      - b. Electrical
      - c. Piping
      - d. Other as required
  - D. Instructions simplified narrative
    - 1. Operation
    - 2. Cautions
    - 3. <u>Supplementary information required to aid in the safe operation of the equipment</u>

### 5. <u>WARRANTIES</u>

- 5.1. Furnish to the Construction Manager an electronic copy uploaded to the Project Specific website and one (1) warranty manuals separate from other project documentation.
- 5.2. Provide warranty manuals in the following format:
  - A. Binder type
    - 1. Commercial quality three-ring binders with durable and cleanable plastic covers
    - 2. Minimum ring size one inch
    - 3. When multiple binders are used, correlate the data into related consistent groupings

### 6. INSTRUCTION OF OWNER'S PERSONNEL

- 6.1. Prior to final acceptance of the Work, and in accordance with specification sections, provide training encompassing all equipment and systems furnished under the Contract, for the Owner's designated operating and maintenance personnel. This training will be video recorded and turned over to the owner for their use. If Owner declines training provide a signed transmittal in lieu of the video recording.
- 6.2. Provide instruction documentation as follows:
  - A. Instruction schedule submittal shall include the following information
    - 1. Systems to be covered
    - 2. Total hours scheduled for each
    - 3. Percentage of classroom instruction versus field instruction
    - 4. Name of instructor and company represented
    - 5. Qualifications of instructor
    - 6. Operating and maintenance manuals, and posted operating instructions shall be the basis of instruction
- 6.3. The Trade Contractor shall maintain a training log with the following information
  - A. Date
  - B. System in which instruction was given
  - C. Name and signature of instructor
  - D. Hours of instruction

- E. Names of Owner's personnel in attendance
- F. Log to be given to Construction Manager as part of the close-out documents at the completion of the training
- 6.4. Instruction shall include the following:
  - A. Complete instruction in the safe operation, emergency procedures, and basic maintenance of equipment furnished under the Contract.
  - B. Instruction in the care and maintenance of finish materials and products.
  - C. Cautions on limitations of Owner's personnel service and conditions under which warranties may be voided.
- 6.5. Deferred Training
  - A. Instruction and training in seasonal systems may be deferred until the appropriate season, provide warranties of such systems will be in effect at such times.

### 7. <u>SPECIAL TOOLS</u>

7.1. Deliver to the Construction Manager four (4) sets of any special tools and keys required for operation, adjustment, resetting or maintenance. Identify each item and package each set in a separate container identified as to function.

### 8. <u>ATTIC STOCK</u>

- 8.1. Deliver Owner stock materials as required by each specification section.
- 8.2. Properly label stock material relative to product information and location on the project.
- 8.3. Deliver stock material when directed by the Construction Manager.
- 8.4. Stock material deliveries shall be accompanied by a transmittal listing all materials provided as well as quantities and specification section to which materials apply.
- 8.5. Stock material deliveries shall be signed for by the Construction Manager.

# SECTION 017839 PROJECT RECORD DOCUMENTS

## 1. <u>GENERAL</u>

1.1. Trade Contractor shall maintain at the jobsite, one (1) copy of Contract Drawings, Specifications, Addenda, approved shop drawings, Change Orders and other modifications to the Contract, and field test reports.

## 2. <u>RECORDING</u>

- 2.1. Label each document "PROJECT RECORD" in one inch high printed letters.
- 2.2. Keep record documents current; failure to maintain current documents may result in reduction of progress payments.
- 2.3. Provide updated set of "as-built" documents for review on a monthly basis as condition of payment.
- 2.4. Do not permanently conceal any Work until required information has been recorded.

### 3. <u>CONTRACT DRAWINGS</u>

3.1. Legibly mark to record "as-built" construction conditions which differ from Contract Documents. Include details as required and not noted on original Drawings.

### 4. <u>SUBMITTALS</u>

- 4.1. At completion of the Work, deliver record documents to the Construction Manager.
- 4.2. Accompany submittal with transmittal letter, in duplicate, containing:
  - A. Date
  - B. Project title and number, Architect's name and project number
  - C. Trade Contractor's name and address
  - D. Title and number of each record document
  - E. Certification that each document as submitted is complete and accurate
  - F. Signature of Trade Contractor or authorized representative

### 5. <u>AS-BUILT LIST</u>

5.1. Furnish one (1) completed copy of an "as-built" materials list. This list shall indicate manufacturer, vendor or supplier and installer for each applicable item listed and any other item specified by trade name herein.

### 6. <u>CERTIFICATES OF INSPECTION</u>

6.1. Provide a copy of all Certificates of Inspection as required by applicable codes and/or the Contract documents.

# SECTION 017839 PROJECT RECORD DOCUMENTS

# 7. <u>WARRANTY</u>

- 7.1. All Work and material on the project shall be guaranteed for a minimum period of twelve (12) months, or longer as may be required by the Specifications, Owner's Contract and local ordinances or applicable codes. The guarantee period shall commence at the date of substantial completion as determined by the Owner, Architect and Construction Manager.
- 7.2. Trade Contractor shall be responsible for any and all consequential damages as a result of failure of its Work under warranty.
- 7.3. Guarantee form shall be same as form included in "Certificates and Other Forms" section or other as required by the Architect, Owner or Construction Manager.
- 7.4. Trade Contractor shall return to the project approximately 11 months after the date of substantial completion for a warranty review meeting.

# 8. <u>OTHER</u>

8.1. Provide other close-out items as required by local ordinances and authorities having jurisdiction.